กำลังต้านทานการเกิดสภาวะลิควีแฟคชันของชั้นดินทรายในภาคเหนือของประเทศไทย

นายพิชัย ภัทรรัตนกุล

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

วิทยานิพนธ์นี้เป็นส่วนหนึ่งของการศึกษาตามหลักสูตรปริญญาวิศวกรรมศาสตรดุษฎีบัณฑิต สาขาวิชาวิศวกรรมโยธา ภาควิชาวิศวกรรมโยธา คณะวิศวกรรมศาสตร์ จุฬาลงกรณ์มหาวิทยาลัย ปีการศึกษา 2546 ISBN: 974-17-4074-3 ลิขสิทธิ์ของจุฬาลงกรณ์มหาวิทยาลัย

LIQUEFACTION RESISTANCE OF SANDS IN THE NORTHERN PART OF THAILAND

Mr. Pichai Pattararattanakul

A Dissertation Submitted in Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy in Civil Engineering Department of Civil Engineering Chulalongkorn University Academic Year 2003 ISBN 974-17-4074-3

Liquefaction Resistance of Sands in the Northern Part of
Thailand
Mr. Pichai Pattararattanakul
Civil Engineering
Assistant Professor Supot Teachavorasinskun

Accepted by the Faculty of Civil Engineering, Chulalongkorn University in Partial Fulfillment of the Requirements for the Doctor's Degree

...... Dean of Faculty of Engineering (Professor Somsak Panyakeow, D.Eng.)

THESIS COMMITTEE

...... Chairman (Assistant Professor Thavee Thanacharoengit, Dr.Ing.)

...... Thesis Advisor (Assistant Professor Supot Teachavorasinskun, D.Eng.)

(Assistant Professor Surachat Sambhandharaksa, Sc.D.)

...... Member (Assistant Professor Boonchai Ukritchon, Sc.D.)

...... Member

(Dr. Tirawat Boonyatee, D.Eng.)

พิชัย ภัทรรัตนกุล: กำลังต้านทานการเกิดสภาวะลิควีแฟคชันของชั้นดินทรายในภาคเหนือ ของประเทศไทย (LIQUEFACTION RESISTANCE OF SANDS IN THE NORTHERN PART OF THAILAND) อ. ที่ปรึกษา: ผู้ช่วยศาสตราจารย์ ดร. สุพจน์ เตชวรสินสกุล, 286 หน้า. ISBN 974-17-4074-3.

เป็นที่ยอมรับกันว่าพื้นที่ในแถบจังหวัดทางภาคเหนือของประเทศไทย เป็นบริเวณที่มี ความเสี่ยงภัยต่อแผ่นดินไหวในระดับปานกลาง และจากการสำรวจทางธรณีวิทยา พบว่ารอย เลื่อนหลาย ๆ รอยเลื่อน ในบริเวณภาคเหนือและภาคตะวันตกของประเทศ เป็นรอยเลื่อนที่มีพลัง ซึ่งเป็นไปได้ว่ารอยเลื่อนที่พบเหล่านี้ จะสามารถทำให้เกิดแผ่นดินไหวขนาด 5.5 ถึง 6.5 ริคเตอร์ แม้ว่าได้มีการประกาศให้อาคารหรือสิ่งก่อสร้าง ในพื้นที่ทางภาคเหนือและภาคตะวันตก ต้องทำ การออกแบบต้านแรงแผ่นดินไหวตั้งแต่ปี ค.ศ. 1980 แต่ไม่ค่อยได้รับความสนใจ เนื่องจากวิศวกร ส่วนใหญ่ยังขาดพื้นฐานความรู้ทางด้านพลศาสตร์ โดยเฉพาะพลศาสตร์ของดิน จากข้อมูลการ เจาะสำรวจชั้นดินที่รวบรวมได้ในจังหวัดเชียงใหม่และเชียงราย พบว่าชั้นดินส่วนใหญ่ ประกอบด้วยชั้นทรายหลวมถึงแน่นปานกลางกระจายอยู่ในระดับที่มีความลึกไม่มาก และมี SPT N-value ที่ปรับแก้แล้วประมาณ 5-20 ครั้งต่อฟุต ข้อมูลการเจาะสำรวจชั้นดินดังกล่าว และ แบบจำลองทางสถิติซึ่งได้จากการวิเคราะห์ข้อมูลการเกิดลิควีแฟคชั่นที่รวบรวมจากทั่วโลก จะถูก นำมาใช้ในการวิเคราะห์เพื่อหาแรงดันน้ำส่วนเกินที่เกิดจากแรงแผ่นดินไหวโดยวิธีหน่วยแรง ประสิทธิผล ซึ่งผลการวิเคราะห์ที่ได้ จะถูกเสนอในรูปของความสัมพันธ์ระหว่างความน่าจะเป็น ของการเกิดลิควีแฟคชั่น แรงดันน้ำส่วนเกิน และความเร่งสูงสุดที่ผิวดิน เพื่อใช้ประเมินความเสี่ยง และกำลังต้านทานการเกิดสภาวะลิควีแฟคชั่นของชั้นดินทรายในจังหวัดเซียงใหม่และเซียงราย

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

ภาควิชา	วิศวกรรมโยธา	ลายมือชื่อนิสิต
สาขาวิชา	วิศวกรรมโยธา	ลายมือชื่ออาจารย์ที่ปรึกษา
ปีการศึกษา	.2546	

##4271811621: MAJOR CIVIL ENGINEERING

KEYWORD: AMPLIFICATION / EARTHQUAKES / LIQUEFACTION / SEISMIC RISK PICHAI PATTARARATTANAKUL: LIQUEFACTION RESISTANCE OF SANDS IN THE NORTHERN PART OF THAILAND. THESIS ADVISOR: ASST. PROF. SUPOT TEACHAVORASINSKUN, D.Eng., 286 pp. ISBN 974-17-4074-3.

The northern part of Thailand is located in low to medium seismic risk zones. There are a few active faults recently found in the western and northern parts of the country. These could possibly induce earthquakes of magnitude (M_L) of 5.5-6.5. Although seismic design code has been enforced in the area since 1980, the fundamental knowledge on dynamic soil behavior has not been extensively attained. Literature reviews of the existing boreholes from the two largest provinces in the north, including Chiang Mai and Chiang Rai, revealed that the areas are underlain by loose to medium dense sand layers found at shallow depths. The corrected SPT N-value of those sand layers varies in the range of 5-20 blows/ft. These borehole information, together with the result obtained from the logistic regression based on worldwide liquefaction database are used to conduct the effective stress analysis. A simple tool correlating the liquefaction probability, estimated excess pore water pressure and peak ground acceleration is proposed. Preliminary risk zones in these two provinces were identified.

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

Department	Civil Engineering	Student's signature
Field of study	Civil Engineering	Advisor's signature
Academic year	2003	

ACKNOWLEDGEMENTS

First and foremost, I would like to express my sincere gratitude to my thesis advisor Dr. Supot Teachavorasinskun, who made it possible for me to study in the area of earthquake geotechnical engineering at Chulalongkorn University. He has played a major role in guiding the direction of this study. In addition, he has provided support and guidance in many other ways. Without his vision, effort and encouragement, this thesis could never reached completion.

Prof. Satoshi Morio of Maizuru National College of Technology, Japan for giving me an excellence computer program FLIP and his advice during the analytical phase of this study. At the same time, I also wish to express my appreciation to the members of the thesis committees, including Dr. Thavee Thanacharoengit, Dr. Surachat Sambhandharaksa, Dr. Boonchai Ukritchon, Dr. Tirawat Boonyatee, and especially Dr. Chitchai Anantasech of Chiang Mai University, for their constructive criticism and suggestions to this study.

Krungthep Engineering Consultants (KEC) Co., Ltd. and Worman and Associates Consultants (WAC) Co., Ltd. for providing borehole data in Chiang Mai and Chiang Rai.

I would like to thank my best younger friend Yuttana Kurojjanawong, who has made my life in geotechnical engineering division very enjoyable and has always provided me great suggestions for studying in the geotechnical engineering area.

Most of all, I would like to express my special appreciation to Piyawan Yimanan, my sweetest friend who always stood beside me and traveling with me in past few years. Her love, strength, and wisdom have inspired me to be the best I can be through many difficult times.

Finally, I owe my deepest gratitude to my parents, Arthorn and Pikun Pattararattanakul; their love is the greatest power in the world and has driven me to succeed in every destination I have dreamed of. I respectfully dedicate this thesis to my loving parents.

TABLE OF CONTENTS

PAGE

AB	STRA	ACT (TH	IAI)iv
AB	STRA	ACT (EN	VGLISH)v
AC	KNO	WLEDO	JEMENTS vi
TA	BLE	OF CON	VTENTS vii
LIS	ST OF	TABLE	ESxi
LIS	ST OF	FIGUR	ESxii
CI	HAP'	TER	
1.	IN	TROD	UCTION1
	1.1	OVER	VIEW
	1.2	OBJEC	TIVE AND SCOPE
2.	LĽ	FERA	ΓURE REVIEW6
	2.1	INTRO	DUCTION
	2.2	BASIC	SEISMOLOGY
		2.2.1	THE NATURE OF EARTHQUAKES
		2.2.2	EARTHQUAKE TERMINOLOGY
		2.2.3	SEISMIC WAVES
	2.3	EARTI	IQUAKE CHARACTERISTICS
		2.3.1	INTENSITY SCALE
		2.3.2	RICHTER MAGNITUDE SCALE
		2.3.3	RICHTER MAGNITUDE CALCULATION
		2.3.4	PEAK GROUND ACCELERATION 11
		2.3.5	SEISMOMETER AND ACCELEROMETER 12
		2.3.6	PREDOMINANT PERIOD AND DURATION
		2.3.7	SITE PERIOD
	2.4	DYNA	MIC SOIL PROPERTIES
		2.4.1	SHEAR WAVE VELOCITY OR SHEAR MODULUS
			2.4.1.1 CORRELATIONS BASED ON LABORATORY MEASUREMENT 15
			2.4.1.2 CORRELATIONS BASED ON FIELD MEASUREMENT
			(a) CORRELATIONS BASED ON DEPTH
		242	(D) COKKELATIONS BASED ON SPI N-VALUE
		2.4.2	MODULUS REDUCTION
		∠.4.3	DAWIFUNU KATIU

TABLE OF CONTENTS (CONT.)

CI	HAP	TER		PAGE
	2.5	GROU	ND RESPONSE ANALYSIS	
		2.5.1	ONE-DIMENSIONAL GROUND RESPONSE	
		2.5.2	EQUIVALENT LINEAR METHOD	
	2.6	AMPL	IFICATION OF EARTHQUAKE GROUND MOTIONS	
		2.6.1	AMPLIFICATION FACTOR	
		2.6.2	EFFECTS OF LOCAL SITE CONDITIONS ON GROUND MOTION	
	2.7	LIQUE	FACTION	
		2.7.1	GENERAL	
		2.7.2	CAUSES OF SOIL LIQUEFACTION	
		2.7.3	FAILURE MECHANISMS FOR SLOPE GROUND	
			2.7.3.1 LATERAL SPREADS	
			2.7.3.2 FLOW FAILURES	
		2.7.4	FAILURE MECHANISMS FOR HORIZONTAL GROUND	
			2.7.4.1 SAND BOILS	
			2.7.4.2 GROUND OSCILLATION	
			2.7.4.3 LOSS OF BEARING STRENGTH	
			2.7.4.4 SUBSIDENCE AND SETTLEMENT	
	2.8	EVAL	UATION OF LIQUEFACTION RESISTANCE OF SOILS	41
		2.8.1	DETERMINISTIC APPROACH OR CYCLIC STRESS APPROACH	41
			2.8.1.1 CYCLIC STRESS RATIO (CSR) AND CYCLIC RESISTANCE	
			RATIO (CRR)	
			2.8.1.2 EVALUATION OF CYCLIC STRESS RATIO (CSR)	
			2.8.1.3 EVALUATION OF LIQUEFACTION RESISTANCE (CRR)	45
			2.8.1.4 MAGNITUDE SCALING FACTORS (MSF)	
			2.8.1.5 EFFECTS OF INITIAL STATIC SHEAR STRESSES AND HIGI	Н
			OVERBURDEN STRESSES ON LIQUEFACTION RESISTANC	CE 48
			2.8.1.6 FACTOR OF SAFETY AGAINST LIQUEFACTION	
		2.8.2	PROBABILISTIC APPROACH	50
3.	PR	OFILI	E OF THE STUDY AREA	52
	3.1	LOCA	TION AND TOPOGRAPHY	52
	3.2	SEISM	ICITY	53
	3.3	SOIL F	PROFILE	54
	3.4	INPUT	MOTIONS FOR ANALYSIS	54
	3.5	ESTIM	IATION OF SHEAR WAVE VELOCITY	55

TABLE OF CONTENTS (CONT.)

CI	APTER	PAGE
	3.5.1 FORMULA FOR CLAY	55
	3.5.2 FORMULA FOR SAND	56
	3.6 SUMMARY	56
4.	AMPLIFICATION OF EARTHQUAKE GROUND MOTIONS	58
	4.1 INTRODUCTION	58
	4.2 COMPUTER PROGRAM SHAKE	58
	4.3 MODULUS REDUCTION AND DAMPING RATIO	60
	4.4 RESULTS OF ANALYSES	61
	4.5 SUMMARY	62
5.	EVALUATION OF LIQUEFACTION RESISTANCE BY	
	PROBABILISTIC APPROACH	64
	5.1 INTRODUCTION	64
	5.2 LIQUEFACTION DATA CATALOG	65
	5.3 REGRESSION METHOD FOR LIQUEFACTION DATA ANALYSIS	69
	5.4 LOGISTIC MODELS OF LIQUEFACTION BEHAVIOR	70
	5.5 SUMMARY	72
6.	FINITE ELEMENT ANALYSIS OF PORE WATER PRESSURE	
	IN SAND DEPOSITES DURING EARTHQUAKES	73
	6.1 INTRODUCTION	73
	6.2 CONSTITUTIVE MODEL OF SOILS	74
	6.2.1 BASIC CONSTITUTIVE EQUATIONS	74
	6.2.2 PHYSICAL BACKGROUND OF THE MODEL	77
	6.2.3 SHEAR MECHANISM	78
	6.2.4 VOLUMETRIC MECHANISM	80
	6.3 COMPUTER PROGRAM FLIP	81
	6.4 LIQUEFACTION TESTS ON SANDS USING A CYCLIC TRIAXIAL APPARATU	JS 82
	6.5 DETERMINATION OF MODEL PARAMETERS	82
	6.6 RESULTS OF ANALYSES	
	6.7 SUMMARY	85

TABLE OF CONTENTS (CONT.)

CHAPTER		PAGE
7. CON	NCLUSIONS AND RECOMMENDATIONS	
7.1 C	CONCLUSIONS	
7.2 R	RECOMMENDATIONS FOR FUTURE RESEARCH	
REFERI	ENCES	91
TABLES	S	
FIGURE	ES	
APPENI	DICES	
APPENDIX	X A. LIQUEFACTION DATA CATALOG	
APPENDIX	X B. BOREHOLE DATA USED IN THIS STUDY	
	B.1 CHIANG MAI	
	B.2 CHIANG RAI	
VITAE		

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

LIST OF TABLES

TABLI	TABLEPAGE	
1.1	Examples of recent earthquakes felt in Thailand105	
2.1	Modified Mercalli intensity scale (Kramer, 1996)106	
2.2	Richter magnitude scale (M _L)106	
2.3	Parameters affecting shear modulus and damping of soils subjected to	
	dynamic loading (Hardin and Drnevich, 1972a)107	
2.4	Values of G_{max}/S_u (Weiler, 1988)107	
2.5	Parameters for Ohsaki and Iwasaki (1973) relationship between	
	SPT N-value and V _s (Sykora, 1987)107	
2.6	Ohta and Goto (1976) relationships between	
	SPT N-value and V _s (Sykora, 1987)108	
2.7	Imai and Tonouchi (1982) relationships between	
	SPT N-value and V _s (Sykora, 1987)109	
2.8	Effect of environmental and loading conditions on modulus ratio	
	(at given strain level) of normally consolidated and moderately	
	overconsolidated soils (modified from Dobry and Vucetic, 1987)109	
2.9	Effect of environmental and loading conditions on damping ratio	
	of normally consolidated and moderately overconsolidated soils	
	(modified from Dobry and Vucetic, 1987)110	
2.10	Magnitude scaling factor (Seed and Idriss, 1982)	
3.1	Maximum estimated M_L for seismic source zones in Thailand region	
	(after Nutalaya et al., 1985 and modified by Warnitchai and	
	Lisantono, 1996)111	
3.2	Summary of input motions used in analyses111	
6.1	Physical properties of sand used in cyclic triaxial test (Gauchan, 1984) 112	
6.2	Model parameters (Iai et al., 1993)112	
6.3	Model parameters for dilatancy112	
6.4	Summary of the estimated values of factor of safety based on	
	the procedure proposed by Seed et al. (1983)113	

LIST OF FIGURES

FIGUE	RE PAG	GE
1.1	Evidence indicating the occurrence of liquefaction	
	in the northern area of Thailand	115
1.2	General study methodology adopted	116
1.3	Example of deterministic method of liquefaction evaluation derived	
	from empirical data for soils with $D_{50} > 0.25$ mm (Seed et al., 1983)	117
2.1	Earthquake terminology (Lindeburg, 1998)	18
2.2	Types of seismic waves: (a) P-wave; (b) S-wave; (c) R-wave; (d) L-wave	
	(Kramer, 1996)	18
2.3	Typical seismometer amplitude trace (Lindeburg, 1998)	119
2.4	Richter magnitude correction nomograph (Lindeburg, 1998)	19
2.5	Two hypothetical Fourier amplitude spectra with the same predominant	
	period but very different frequency contents. The upper curve describes	
	a wideband motion and the lower a narrowband motion	120
2.6	Variation of k-coefficient with plasiticity index (Dickenson, 1994)	120
2.7	Backbone curve showing typical variation of G _{sec} with shear strain	21
2.8	Variation of the modulus ratio with shear strain	121
2.9	Modulus reduction curves for fine-grained soils of different plasticity	
	(Vucetic and Dobry, 1991)1	122
2.10	Influence of mean effective confining pressure on modulus reduction	
	curves for (a) nonplastic ($PI = 0$) soil, and (b) plastic ($PI = 50$) soil	
	(Ishibashi, 1992)	122
2.11	Variation of damping ratio of fine-grained soil with cyclic shear strain	
	amplitude and plasticity index (Vucetic and Dobry, 1991)	123
2.12	Ground response nomenclature: (a) soil overlying bedrock;	
	(b) no soil overlying bedrock	123
2.13	Relationship between hysteresis loop and: (a) shear modulus;	
	(b) damping ratio	124
2.14	Iteration toward strain-compatible shear modulus and damping ratio in	
	equivalent linear analysis	124
2.15	Kanai's amplification factor for soft ground (Kanai, 1957)	125

FIGUR	RE PAGE
2.16	Approximate relationships between peak accelerations on rock and
	other local site conditions (Seed et al., 1976)
2.17	Approximate relationship between peak accelerations on rock and
	soft soil sites (Idriss, 1990)
2.18	Average normalized response spectra (5% damping) for different
	local site conditions (Seed et al., 1976)
2.19	Relationship between limiting epicentral distance of sites at which
	liquefaction has been observed and moment magnitude for shallow
	earthquakes. Deep earthquakes (focal depths > 50 km) have produced
	liquefaction at greater distances. (Ambraseys, 1988)127
2.20	Cyclic shear stresses on soil element during ground shaking: (a) idealized
	field loading conditions; (b) shear stress variation determined by response
	analysis
2.21	Lateral spreading adjacent to a river channel: (a) before earthquake;
	(b) after earthquake (Youd, 1984)
2.22	Examples of flow failure caused by liquefaction and loss of strength of
	soils lying on a steep slope
2.23	Formation of water interlayers in shaking table tests of
	Liu and Qiao, (1984)
2.24	Ground oscillation: (a) before earthquake; (b) after earthquake
	(Youd, 1984)
2.25	Example of structure tiled due to loss of bearing strength. Liquefaction
	weakens the soil reducing foundation support which allows heavy
	structures to settle and tip
2.26	Tilting of apartment buildings, Niigata (1964)131
2.27	Procedure for determining maximum shear stress, $(\tau_{\text{max}})_r$
	(Seed and Idriss, 1982)
2.28	r_d versus depth curves developed by Seed and Idriss (1971) with
	add mean-value lines plotted from Eq. (2.34) (Youd and Idriss, 2001) 132
2.29	Time history of shear stresses during earthquake (Seed and Idriss, 1982)133

FIGUE	FIGURE PAGE	
2.30	Plot used to determine the cyclic resistance ratio for clean and	
	silty sands for $M = 7.5$ earthquakes (Seed et al., 1985)	
2.31	Approximate relationships between the moment magnitude	
	scale M_w and other magnitude scales. Shown are the short-period	
	body wave magnitude scale m_b , the local magnitude scale M_L , the	
	long-period body wave magnitude scale m_B , the Japan Meteorological	
	Agency magnitude scale M_{JMA} , and the surface-wave magnitude scale	
	M _s (reproduced from Day, 2002)	
2.32	Variation of correction factor, K_{α} , with initial shear/normal stress ratio	
	(Seed and Harder, 1990)	
2.33	Variation of correction factor, K_{σ} , with effective overburden pressure	
	(Seed and Harder, 1990)	
3.1	Location of the study area	
3.2	Seismic source zones in Thailand and vicinity (Nutalaya et al., 1985)	
3.3	Map showing contours of peak ground acceleration (in units of	
	acceleration of gravity) with 10% chance of being exceeded in a 50-year	
	exposure time, and seismic zones for earthquake-resistant design	
	(Wanitchai and Lisantono, 1996)138	
3.4	Typical subsoil section in Chiang Mai Province (modified from	
	Anantasech and Thanadpipat, 1985)139	
3.5	Typical subsoil profile in Chiang Rai province140	
3.6	Examples of the soil profiles and soil properties collected from	
	Chiang Mai141	
3.7	Examples of the soil profiles and soil properties collected from	
	Chiang Rai	
3.8	Grain size distribution of sands: (a) Chiang Mai; (b) Chiang Rai143	
3.9	Acceleration time history of input motions	
3.10	Acceleration response spectra of input motions used in this study145	
4.1	One-dimensional idealization of a horizontally layered soil deposit	
	over a uniform half-space (Idriss and Sun, 1992)146	

FIGU	URE PAGE
4.2	Shear modulus and damping variation with shear strain for clays of
	varying plasticity (Vucetic and Dobry, 1991)147
4.3	Shear modulus and damping variation with shear strain for sands
	(Seed et al., 1984)148
4.4	Variations of amplification factors with base rock acceleration in
	Chiang Mai149
4.5	Variations of amplification factors with base rock acceleration in
	Chiang Rai
4.6	Comparison between peak ground accelerations of Chiang Mai and
	observed results from 1985 Mexico and 1989 Loma Preita earthquakes
	(modified from Idriss, 1991)151
4.7	Comparison between peak ground accelerations of Chiang Rai and
	observed results from 1985 Mexico and 1989 Loma Preita earthquakes
	(modified from Idriss, 1991)152
5.1	Example of deterministic method of liquefaction evaluation derived from
	empirical data for soils with $D_{50} > 0.25$ mm (Seed et al., 1983)
5.2	Example of liquefaction data catalog (Liao and Whitman, 1986a)154
5.3	Contours of equal probability of liquefaction (P_L) for magnitude, M = 7.5
	with deterministic line by Seed et al. (1985)
5.4	Contours of equal probability of liquefaction (P_L) for magnitude, M = 5.5
	with deterministic line modified from Seed et al. (1985)
6.1	Flow diagram for evaluation of pore water pressure generation
6.2	Schematic figure for multiple simple shear mechanisms (pairs of circles
	indicate mobilized virtual shear strain in positive and negative modes of
	compression shear and simple shear) (Iai et al., 1992)
6.3	Schematic figure of contact normal n_k , tangential direction t_k and contact
	force increment dP _k (Iai et al., 1993)157
6.4	Schematic figure of liquefaction front, state variable <i>S</i> and shear stress
	ratio r (Iai et al., 1992)157

FIGURE PAG	
6.5	Relationship between normalized plastic shear work w and liquefaction
	front parameter S_0 (Iai et al., 1992)158
6.6	Grain size distribution of sands: (a) Chiang Mai; (b) Chiang Rai159
6.7	Liquefaction resistance curves from undrained cyclic triaxial test
	(Gauchan, 1984)
6.8	Test and computed liquefaction resistance curves
6.9	Relationship between amplification factor and maximum pore water
	pressure ratio for Chiang Mai162
6.10	Relationship between amplification factor and maximum pore water
	pressure ratio for Chiang Rai164
6.11	Relationship between cyclic stress ratio for earthquake magnitude of 5.5
	and maximum pore water pressure ratio for Chiang Mai
6.12	Relationship between cyclic stress ratio for earthquake magnitude of 5.5
	and maximum pore water pressure ratio for Chiang Rai
6.13	Relationship between maximum ground acceleration and maximum pore
	water pressure ratio for Chiang Mai
6.14	Relationship between maximum ground acceleration and maximum pore
	water pressure ratio for Chiang Rai
6.15	Relationship between maximum ground acceleration and maximum pore
	water pressure ratio at GL -2.5 m for Chiang Mai
6.16	Relationship between maximum ground acceleration and maximum pore
	water pressure ratio at GL -2.5 m for Chiang Rai

ุฬาลงกรณมหาวทยาลย

CHAPTER 1

INTRODUCTION

1.1 OVERVIEW

Thai people were not concerned about earthquake until recent occurrence of several moderate earthquakes. On April 22, 1983, an earthquake of magnitude 5.9 on the Richter scale occurred near a dam site in Kanchanaburi, about 200 kilometers from Bangkok, the capital city of Thailand. The main tremor of this earthquake was felt all over the western part and most of the central part of the country. Five years later (November 6, 1988), an earthquake of magnitude 7.3 hit the southern part of China near the Burmese border. This earthquake was felt in Bangkok even though the epicenter was at a distance of more than 1,000 kilometers; this is a consequence of Bangkok's deep, soft-alluvial soil which tends to amplify the motion of incoming seismic waves. On September 29 and October 1 of the following year, several moderate earthquakes (5.3-5.4 on the Richter scale) hit the northern part of Thailand along the Burmese border. In the city of Chiang Mai, about 180 kilometers from the epicenter, the intensity of ground shaking was rated as VI on the Modified Mercalli Intensity (MMI) scale. Fortunately, the strong ground shaking in the vicinity of these earthquake epicenters have never coincided with any town or city, hence neither buildings have been destroyed nor people have been killed so far. However, damage from the 11 September 1994 Phan earthquake in northern Thailand being quickly followed by the 17 January 1995 Kobe earthquake served as a wake up call to the people of Thailand. Examples of recent earthquakes felt in Thailand are summarized in Table 1.1.

Among the northern provinces of Thailand, Chiang Mai and Chiang Rai were selected as the studied area because of the following reasons:

- 1. They are located close to some of the recently found active faults.
- 2. They are the most densely populated areas in the north.
- 3. They are underlain by layers of soft to medium clay and/or loose to medium dense sand at shallow depths as illustrated by Anantasech and Thanadpipat (1985). The existence of the loose to medium dense sand layer at shallow depths (2 8 m from ground surface) implies a certain liquefaction risk of those two provinces.
- 4. From the metropolitan records of both provinces, more than 80% of housings (1 2 stories building) in the center of the cities were built on shallow foundation. These are the structures most prone to damages due to liquefaction and/or partially increase in excess pore water pressure.
- 5. Figure 1.1 shows evidences indicating the occurrence of liquefaction in a suburban area in Chiang Mai. Trace of sand extruding into the upper gravel layer clearly indicates past liquefaction of the lower sand layer.

Therefore, in the present study, the soil amplification and the liquefaction potential of subsoil induced by medium earthquakes in both provinces are investigated. Although full initialization of liquefaction may not be the case, partial development of excess pore water pressure might cause damages to 1 - 2 stories housing which is usually built on ground or short piles. A preliminary study is therefore needed to survey the liquefaction susceptibility of the areas. Integration among field parameters, probabilistic study, and dynamic analytical results is used as a primary tool for further detail evaluation.

Figure 1.2 shows the general methodology adopted. There are three main information required in the procedure, including:

- (a) Subsurface information. Around 50 existing boring logs were collected from each province. The sub soils in both provinces are subject to wide variation. Nevertheless, layers of loose to medium dense sand are found at depths of 2 8 m in most of the area.
- (b) Laboratory determination of liquefaction resistance. Existing cyclic triaxial tests determining the liquefaction resistance of sand were used to obtain some effective stress parameters required in the effective stress analysis (Iai et al., 1992).
- (c) Existing liquefaction database (Liao and Whitman, 1986a). Since there is no liquefaction database existing for Thailand, the worldwide liquefaction database is used as a reference for determination of other related parameters.

Those three components shall be integrated to obtain a specific tool or guideline for indicating earthquake liquefaction potential in the studied area.

1.2 OBJECTIVE AND SCOPE

The major objectives of this thesis are:

- To quantify the potential amplification of earthquake ground motions in the city of Chiang Mai and Chiang Rai due to soil effects. The scope of the study is the selection of a range of peak rock outcrop accelerations and appropriate acceleration time histories to be used based on the seismicity of the region.
- 2) To develop simple and practical procedures for evaluating liquefaction risk. Charts similar to a semi-empirical chart such as that shown in Figure 1.3 are developed, but with lines corresponding to various liquefaction probability values.
- 3) To evaluate the liquefaction potential and susceptibility of sands in the city of Chiang Mai and Chiang Rai due to medium earthquakes (M = 5.5) based on probabilistic approach.

4) To develop guidelines and charts for analysis and evaluation of the generation of pore water pressure of sands in the city of Chiang Mai and Chiang Rai due to medium earthquakes corresponding to various probability values using finite element analysis. The parameters used in the analysis are obtained by back-fitting the calculated results with experimental data from undrained cyclic triaxial tests.

Scope and outline of the thesis are as follows:

The thesis mainly focuses on the evaluation of the potential amplification of earthquake ground motions, the liquefaction potential and susceptibility of sands, and the generation of pore water pressure due to medium earthquakes (M = 5.5) in the city of Chiang Mai and Chiang Rai. The evaluations were based on subsoil data, geology of the area, and the seismicity of Chiang Mai and Chiang Rai. Subsoil data were collected in the form of boring logs through standard penetration tests (SPT) from the consulting/engineering firms.

Chapter 2 of this thesis presents a review of the theoretical background about basic seismology, earthquake characteristics, Seismic response analysis, dynamic soil properties for seismic response analysis, soil amplification in earthquake engineering, and the methods for evaluation of liquefaction resistance of soils.

Chapter 3 describes the profile of the studied area.

Chapter 4 shows the evaluation of the amplification of earthquake ground motions in the city of Chiang Mai and Chiang Rai. Subsequently, the analysis results and discussion are presented.

Chapter 5 describes the catalog of liquefaction case studies which is used to develop liquefaction probability charts in this research. A description of the logistic regression method for evaluating the liquefaction resistance is presented and logistic models of liquefaction occurrence are then formulated.

In Chapter 6, the laboratory test results on undrained cyclic triaxial of sands were used to obtain some effective stress parameters required in the effective stress model. After the model parameters have been completely obtained, the finite element method is used to estimate pore water pressure. Finally, charts and guidelines for analysis and evaluation of the generation of pore water pressure of sands due to medium earthquakes corresponding to various probability values are presented.

A summary of the thesis, the conclusions drawn from the research, and suggestions for future work, are presented in Chapter 7.



สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 2

LITERATURE REVIEW

2.1 INTRODUCTION

In this chapter, the theoretical background and a review of the previous work about basic seismology, earthquake characteristics, seismic response analysis, dynamic soil properties for seismic response analysis, soil amplification in earthquake engineering, and evaluation of liquefaction resistance of soils are presented.

2.2 BASIC SEISMOLOGY

2.2.1 THE NATURE OF EARTHQUAKES

An earthquake is an oscillatory, sometimes violent movement of the Earth's surface that follows a release of energy in the Earth's crust. This energy can be generated by a sudden dislocation of segments of the crust, a volcanic eruption, or man-made explosion. Most of the destructive earthquakes, however, are caused by dislocations of the crust. When subjected to geologic forces from plate tectonics, the crust initially strains (i.e., bends and shears) elastically. For pure axial loading, Hooke's law gives the stress that accompanies this strain.

$$\sigma = E\varepsilon \qquad (\text{Axial Loading}) \tag{2.1}$$

As rock is stressed, it stores strain energy, U. The elastic strain energy per unit volume for pure axial loading is:

$$U = \frac{\sigma \varepsilon}{2}$$
 (Axial Loading) (2.2)

When the stress exceeds the ultimate strength of the rocks, the rocks break and quickly move or snap into new positions. In the process of breaking, the strain energy is released and seismic waves are generated. This is the basic description of the elastic rebound theory of earthquake generation. These waves travel from the source of the earthquake (known as the epicenter or focus) to more distant location along the surface of and through the Earth. Some of the vibrations are of high enough frequency to be audible, while others are of very low frequency with periods of many seconds and thus are inaudible.

2.2.2 EARTHQUAKE TERMINILOGY

The epicenter of an earthquake is the point on the Earth's surface directly above the focus (also known as the hypocenter). The location of an earthquake is commonly described by the geographic position of its epicenter and its focal depth. The focal depth of an earthquake is the depth from the Earth's surface to the focus. These terms are illustrated in Figure 2.1. Earthquakes with focal depths of less than approximately 60 kilometers are classified as shallow earthquakes. Very shallow earthquakes are caused by the fracturing of brittle rock in the crust or by internal strain energy that overcomes the friction locking opposite sides of a fault. Intermediate earthquakes, whose causes are not fully understood, have focal depths ranging from 60 to 300 kilometers. Deep earthquakes may have focal depths of up to 700 kilometers.

2.2.3 SEISMIC WAVES

Seismic waves are of three types: compression, shear, and surface waves. Compression and shear waves travel from the hypocenter through the Earth's interior to distant points on the surface. Only compression waves, however, can pass through the Earth's molten core. Because compression waves (also known as longitudinal waves) travel at great speeds (5800 m/s in granite) and ordinarily reach the surface first, they are known as P-waves (for "primary waves").

Shear waves (also known as transverse waves) do not travel as rapidly (3000 m/s in granite) through the Earth's crust and mantle as do compression waves. Because they ordinarily reach the surface later, they are known as S-waves (for "secondary waves"). Instead of affecting material directly behind or ahead of their lines of travel, shear waves displace material at right angles to their path. While S-waves travel more slowly than P-waves, they transmit more energy and cause the majority of damage to structures. The speed at which P-waves and S-waves travel varies with the stiffness of materials they travel through.

Surface waves, also known as R-waves (for "Rayleigh waves") or L-waves (for "Love waves"), may or may not form. They arrive after the primary and secondary waves. In granite, R-waves move at approximately 2700 m/s. The types of seismic waves are shown in Figure 2.2.

2.3 EARTHQUAKE CHARACTERISTICS

2.3.1 INTENSITY SCALE

The intensity of an earthquake is based on the damage and other observed effects on people, buildings, and other features. Intensity varies from place to place within the disturbed region. An intensity scale consists of a series of responses, such as people awakening, and movement of furniture. Although numerous intensity scales have been developed, the scale encountered most often in the world is the Modified Mercalli Intensity (MMI) scale, originally developed in 1902 by the Italian seismologist Mercalli and modified in 1931 by the American seismologists Harry Wood and Frank Neumann. The Modified Mercalli scale consists of 12 increasing levels of intensity (expressed as Roman numerals following the initials MM) that range from imperceptible shaking to catastrophic destruction. The lower numbers of the intensity scale generally are based on the manner in which the earthquake is felt by people. The higher numbers are based on observed structural damage. The numerals do not have a mathematical basis and therefore are more meaningful to nontechnical people than to those in technical fields. The Modified Mercalli intensity scale is shown in Table 2.1.

2.3.2 RICHTER MAGNITUDE SCALE

In 1935, Charles F. Richter of the California Institute of Technology developed the Richter magnitude scale to measure earthquake strength. The magnitude, M_L , of an earthquake is determined from the logarithm of the amplitude recorded by a seismometer. Adjustments are included in the magnitude to compensate for the variation in the distance between the various seismometers and the epicenter. Because the Richter magnitude is a logarithmic scale, each whole number increase in magnitude represents a ten-fold increase in measured amplitude.

Richter magnitude is expressed in whole numbers and decimal fractions. For example, a magnitude of 5.3 might correspond to a moderate earthquake. A strong earthquake might be rated at 7.3. Great earthquakes have magnitudes above 7.5. Earthquakes with magnitudes of 2.0 or less are known as micro-earthquakes are rarely felt by people. Several thousand seismic events with magnitudes of approximately 4.5 or greater occur each year and are strong enough to be recorded by seismometers all over the world. Earthquakes of this size and below have little potential to cause structural damage. The magnitude of an earthquake depends on the length and breadth of the fault slip, as well as on the amount of slip.

2.3.3 RICHTER MAGNITUDE CALCULATION

The Richter magnitude, M_L , is calculated from the maximum amplitude, A, of the seismometer trace, as illustrated in Figure 2.3. A₀ is the seismometer reading produced by an earthquake of standard size (i.e., a calculation earthquake). Generally, A₀ is 0.001 mm.

$$M_L = \log_{10} \left(\frac{A}{A_0} \right) \tag{2.3}$$

Equation (2.3) assumes that a distance of 100 kilometers separates the seismometer and epicenter. For other distances, the nomograph of Figure 2.4 and the following procedure can be used to calculate the magnitude. Due to the lack of reliable information on the nature of the Earth between the observation point and the earthquake epicenter, an error of 10 to 40 kilometers in locating the epicenter is not unrealistic.

- Step 1: Determine the time between the arrival of the P- and S-waves.
- Step 2: Determine the maximum amplitude of oscillation.
- Step 3: Connect the arrival time difference on the left scale and the amplitude on the right scale with a straight line.
- Step 4: Read the Richter magnitude on the center scale.
- Step 5: Read the distance separating the seismometer and the epicenter from the left scale.

Whereas one seismometer can determine the approximate distance to the epicenter, it takes three seismometers to determine and verify the location of the epicenter. The Richter scale ranges from zero to 8.9 (the largest recorded, Chile 1960). Because it is a logarithmic scale, an earthquake of 6.0 is 10 times more severe

than one of 5.0 and similarly, one of 7.0 is 100 times more severe. The Richter magnitude scale is shown in Table 2.2.

2.3.4 PEAK GROUND ACCELERATION

The peak (maximum) ground acceleration, PGA, is easily measure by a seismometer or accelerometer and is one of the most important characteristics of an earthquake. PGA can be given in various units, including ft/sec², in/sec², or m/s². However, it is most common to specify PGA in "g's" (i.e., as a fraction or percent of gravitational acceleration).

$$PGA = \left(\frac{a_{ft/sec^2}}{32.2}\right) \times 100\%$$
(2.4)

$$=\left(\frac{a_{in/\sec^2}}{386}\right) \times 100\%$$
(2.5)

$$= \left(\frac{a_{m/s^2}}{9.81}\right) \times 100\%$$
 (2.6)

Equation (2.7) (as determined by Gutenburg and Richter in 1956) is one of many approximate relationships between the Richter magnitude, M_L , and the PGA at the epicenter. The ground acceleration (in rock) will decrease as the distance from the epicenter increases, and for this reason, equations of this type are called attenuation equations.

$$\log_{10} PGA = -2.1 + 0.81M_L - 0.027M_L^2$$
(2.7)

Attenuation equations are very site dependent. Since Equation (2.7) was developed, newer studies have resulted in better correlations in different formats and

for many different locations, but they are based on limited data. Such studies are regularly incorporated into revisions of the seismic provisions of building codes.

2.3.5 SEISMOMETER AND ACCELEROMETER

Seismic waves travel through the Earth and are recorded on seismometers. A seismometer is the detecting and recording parts of a larger apparatus known as seismograph. Seismometers are pendulum type devices that are mounted on the ground and measure the displacement of the ground with respect to a stationary reference point. Since a seismometer usually records motion in only one orthogonal direction, three seismometers are needed to record all components of ground motion. Figure 2.3 illustrates a typical seismometer trace, known as a seismogram.

Note that while seismic activity usually continues for some time after the start of the earthquake, the major movement occurs in a concentrated period known as the strong phase. The longer the earthquake shakes, the more seismic energy is absorbed by buildings; thus, the duration of strong phase shaking greatly affects the damage inflicted. The peak acceleration is a spike that imparts no energy. The effective peak ground acceleration causes structures to move.

Seismometers record the varying amplitude of ground oscillations beneath the instrument. Sensitive seismometers greatly magnify these ground motions and can detect strong earthquakes occurring anywhere in the world. The time, location, and magnitude of an earthquake can be determined from the data recorded by seismometer stations. Since a seismometer is a spring-mass-dashpot device, it will magnify or distort earthquakes with frequencies in certain ranges. The ratio of actual damping to critical damping can be changed to minimize such distortion. Good seismometer design calls for a damping ratio of between 0.6 and 0.7 with a natural period of vibration smaller than the smallest period to be measured.

An accelerometer (accelerograph) is seismometer mounted in buildings for the purpose of recording large accelerations. For this reason, they are also known as strong motion seismometers. The large swings accelerometers record typically exceed the scale limits of most seismometers. An accelerometer located in a building does not run continually. It triggered by a P-wave and runs for a fixed period of time.

2.3.6 PREDOMINANT PERIOD AND DURATION

A single parameter that provides a useful, although somewhat crude representation of the frequency content of a ground motion is the predominant period, T_p . The predominant period is defined as the period of vibration corresponding to the maximum value of the Fourier amplitude spectrum. To avoid improper influence of individual spikes of the Fourier amplitude spectrum, the predominant period is often obtained from a smoothed spectrum. While the predominant period provides some information regarding the frequency content, it is easy to see from Figure 2.5 that motions with radically different frequency contents can have the same predominant period.

The duration of strong ground motion can have a strong influence on earthquake damage. Many physical processes, such as the degradation of stiffness and strength of certain types of structures and build up of pore water pressures in loose, saturated sands, are sensitive to number of load or stress reversals that occur during an earthquake. A motion of short duration may not produce enough load reversals for damaging response to build up in a structure, even if the amplitude of the motion is high. On the other hand, a motion with moderate amplitude but long duration can produce enough load reversals to cause substantial damage.

2.3.7 SITE PERIOD

The site (soil) period is now recognized as a significant factor contributing to structural damage. When a site has a natural frequency of vibration that corresponds to the predominant earthquake frequency, site movement can be greatly magnified. This is known as resonance. Thus the buildings can experience ground motion much greater than would be predicted from only the seismic energy release.

Determining the actual site period is no easy matter. Since the site period can be computed precisely from widely available formulas and still be grossly inaccurate.

2.4 DYNAMIC SOIL PROPERTIES

The nature and distribution of earthquake damage is strongly influenced by the response of soils to cyclic loading. This response is controlled in large part by the mechanical properties of the soil. Geotechnical earthquake engineering encompasses a wide range of problems involving many types of loading and many potential mechanisms of failure, and different soil properties influence the behavior of the soil for different problems. For many important problems, particularly those dominated by wave propagation effects, only low levels of strain are induced in the soil. For other important problems, such as those involving the stability of masses of soil, large strains are induced in the soil. The behavior of soils subjected to dynamic loading is governed by what have come to be popularly known as dynamic soil properties.

Soil properties that influence wave propagation and other low strain phenomena include stiffness, damping, poisson's ratio, and density. Of these, stiffness and damping are the most important; the others have less influence and tend to fall within relatively narrow ranges. The stiffness and damping characteristics of cyclically loaded soil are critical to the evaluation of many geotechnical earthquake engineering problems. Not only at low strains but because soils are nonlinear materials, also at intermediate and high strains. At high levels of strain, the influence of the rate and number of cycles of loading on shear strength may also be important. Volume change characteristics are also important at high strain levels.

2.4.1 SHEAR WAVE VELOCITY OR SHEAR MODULUS

In geotechnical earthquake engineering analysis such as a seismic site response analysis using the equivalent linear method and liquefaction analysis using finite element method, which will be explained later, one of the key input parameters is the shear modulus (G) at low strain ($\gamma \cong 5 \times 10^{-4}$ %), generally represented as G_{max}. Though G_{max} is difficult to accurately measure in the laboratory due to the effects of sample disturbance, it can be readily determined from the shear wave velocity (V_s) and the mass density (ρ) from the relationship G = $\rho \times V_s^2$. With the need to estimate values of G_{max} or V_s for the soil profiles for the northern part of Thailand, particularly Chiang Mai and Chiang Rai, several empirical correlations based on available field and laboratory measurements were reviewed.

Researchers have been attempting to formulate correlations between shear modulus or shear wave velocity and various geotechnical parameters for at least thirty years. These formulas have evolved from measurements made in both the field and laboratory, even though the accuracy of such correlations varies considerably.

In the following review, due to the difficulty in finding some of the original papers, most of the correlations were obtained from the excellent summary report by Sykora (1987). The correlations are presented in the units of the original paper when available, or in the units as presented by Sykora (1987), which are often U.S. Customary units. When any of the correlations are actually used as part of this research in subsequent sections, they are converted to the appropriate metric units.

2.4.1.1 CORRELATIONS BASED ON LABORATORY MEASUREMENT

Hardin and Richart (1963) conducted one of the first investigations of variables affecting V_s in soils. A resonant column testing device was utilize to load cyclically samples of Ottawa sands, crushed quartz sand, and crushed quartz silt. The

considered variables were effective confining pressure (σ_0), void ratio (e), moisture content, grain size distribution, and grain characteristic. The shear strain amplitude, however, was not examined and was kept constantly at 10⁻⁵ in/in for all tests. The confining pressure and void ratio were shown to apparently have the most influence on V_s. Based on the results of their study, Hardin and Richart developed empirical equations relating shear wave velocity to void ratio and effective confining with a reported accuracy within ± 10 %, and the equations are:

$$\sigma_0' < 2000 \text{ psf}; \quad V_s = (119 - 56.0e)(\sigma_0')^{0.30} \quad (\text{ft/sec})$$
(2.8)

$$\sigma'_0 > 2000 \text{ psf}; \quad V_s = (170 - 78.2e)(\sigma'_0)^{0.25} \quad (\text{ft/sec})$$
(2.9)

Seed and Idriss (1970) developed one of the initial simple relationship relating small shear strain modulus to undrained shear strength (S_u) based on an empirical study of 21 clays involving predominantly laboratory data. They proposed a linear relationship between G_{max} and S_u as follows:

$$G_{\max} = 2200 \cdot S_u \tag{2.10}$$

However, subsequent researchers (Trudeau et al., 1973; Koutsoftas and Fischer, 1980; and Egan and Ebeling, 1985) have shown that the ratio G_{max}/S_u , called normalized modulus, is not constant, but depends on undrained shear strength Su, overconsolidation ratio (OCR), plasticity index (PI), and effective confining stress (σ_0) .

Hardin and Drnevich (1972a, 1972b) examined the effect of various parameters affecting the stress-strain relations in soils in the strain range of 1% and less using results of the resonant column and cyclic simple shear testing. They concluded that strain amplitude, effective mean principal stress, and void ratio are very important factors, for both clean sands and clays. Furthermore, degree of saturation is very significant for clays. The parameters examined by Hardin and Dnervich (1972a) and the effect of these parameters on G_{max} are presented in Table 2.3.

Using the results of their earlier study, Hardin and Drnevich (1972b) subsequently developed a relationship for G_{max} as a function of void ratio, OCR, plasticity index, and mean effective stress, as shown in Equation (2.11).

$$G_{\max} = 1230 \cdot \frac{(2.973 - e)^2}{1 + e} \cdot OCR^k \cdot \sigma_m^{0.5}$$
 (psi) (2.11)

where σ_m is the mean effective stress in pounds per square inch (psi), and k is a dimensionless parameter that varies with plasticity index as shown in Figure 2.6.

Kokusho (1980) tested saturated samples of Toyoura sand using cyclic triaxial apparatus. From the test data, he proposed a relationship for G_{max} as a function of void ratio and effective confining stress (σ_0). The relationship is:

$$G_{\text{max}} = 8400 \cdot \frac{(2.17 - e)^2}{1 + e} \cdot \sigma_0^{+0.5}$$
 (kPa) (2.12)

Kokusho, Yoshida, and Esashi (1982) tested undisturbed samples of soft clay with plasticity index ranges from 40 to 85 using cyclic triaxial apparatus. They then presented the following equation as function of void ratio and effective confining stress (σ_0):

$$G_{\text{max}} = 141 \cdot \frac{(7.32 - e)^2}{1 + e} \cdot \sigma_0^{+0.6}$$
 (kPa) (2.13)

Knox, Stokoe, and Kopperman (1982) tested a 7-ft-cube sample of dry sand in a steel frame structure (true triaxial device) and concluded that, for shear wave propagating in a principal stress direction, shear wave velocity only depended on the stress in the direction of particle motion and that in the direction of wave propagation. Shear wave velocity was found to be independent of the state of stress in the third orthogonal direction. Sykora (1987) substituted equation by Knox, et, al. (1982) to equation by Hardin and Drnevich (1972b), Equation (2.11), then the result is:

$$G_{\max} = 1230 \cdot \frac{(2.973 - e)^2}{1 + e} \cdot OCR^k \cdot \sigma_a^{+0.25} \cdot \sigma_b^{+0.25} \cdot \sigma_c^{+0.0}$$
(psi) (2.14)

where σ_a is the effective stress in direction of shear wave propagation, σ_b is the effective stress in direction of shear wave particle motion, and σ_c is the effective stress in third (remaining) orthogonal direction, with all stresses being in units of psi.

Weiler (1988) developed the empirical relationship between G_{max} and undrained shear strength (S_u) of clay, measured in CU triaxial compression test, as shown in Table 2.4.

Dickenson (1994) presented the results of a study of the dynamic response of soft and deep cohesive soils during the 1989 Loma Prieta earthquake. He developed a nonlinear relationship between shear wave velocity determined from field testing and undrained shear strength for four cohesive soil units in the San Francisco Bay area: San Francisco Bay Mud, Yerba Buena Mud, Alameda Formation (marine), and Alameda Formation (oxidized). The relationship includes both the soft San Francisco Bay Mud as well as the deeper and stiffer old bay clays of the Alameda Formation. His proposed relationship is:

$$V_s = 18 \cdot S_u^{0.475}$$
 (ft/sec) (2.15)

where S_u , the undrained shear strength, has unit of psf. It should be noted that S_u determined from unconfined compression, unconsolidated undrained triaxial (TX), and consolidated undrained triaxial tests were used directly in Dickenson's equation. Field vane strengths were corrected using Bjerrum's plasticity-based correlation

(Bjerrum, 1972), and the direct simple shear (DSS) strengths were modified using an assumed relationship of $(S_u)_{TX} \cong 1.35(S_u)_{DSS}$. The relationship showed good agreement with the data, but Dickenson suggested that all such relationships be verified for specific soils by field testing.

2.4.1.2 CORRELATIONS BASED ON FIELD MEASUREMENT

The disturbance inherent in obtaining soil samples for subsequent laboratory testing makes it extremely difficult to measure values of low strain shear modulus representative of the in situ conditions, particularly for cohesionless soils. Therefore, in situ soil testing, such as the Standard Penetration Test (SPT) and the Cone Penetration Test (CPT) have become the primary methods of estimating V_s of soil in situ, with the exception of direct measurement. Schmertmann et al. (1978) concluded that a properly standardized SPT has a reasonable potential to evaluate shear waves velocity. The correlations based on field measurement have developed over the past twenty years in soil dynamic toward the estimation of the dynamic behavior of soil and the determination of representative constitutive relationships. The following suites of correlations represent part of that development.

(a) CORRELATIONS BASED ON DEPTH

The most generally reported shear wave velocity formulas are those that are a function of depth, however, these formula usually are presented for specific soils and locations (e.g., San Francisco Bay Mud). An example of such a relation is that proposed by Fumal (1978) for soil in the San Francisco Bay area. He collected and analyzed downhole seismic data from 59 sites throughout the San Francisco Bay region and determined a correlation for the variation of V_s with depth:

$$V_{\rm s} = 471 \cdot Z^{0.20}$$
 (ft/sec) (2.16)

where Z is depth in feet. Fumal's relationship applied not only to San Francisco Bay Mud, but to sand, stiff clay, and gravely soil as well.

(b) CORRELATIONS BASED ON SPT N-VALUE

Sakai (1968) postulated soil to be an elastic material, and used both SPT and plate bearing test results to evaluate Young's modulus (E). He then converted the Young's modulus to shear wave velocity by Equation (2.17).

$$V_s = \sqrt{\frac{E}{\rho} \cdot \frac{1}{2(1+\nu)}}$$
(2.17)

where ρ is mass density and v is Poisson's ratio. Sakai then substituted the values of Poisson's ratio ranging from 0.2 to 0.5 and the average strains that were developed during the plate bearing test to develop the following equation for sand:

$$V_s = (49 \ to \ 110)N^{0.5} \ (ft/sec)$$
 (2.18)

where N is the N-value obtained from the SPT. A similar relationship was developed by Kanai (1966) based on the results of over 70 microtremor measurements, almost entirely sands, to create the equation:

$$V_s = 62 \cdot N^{0.62}$$
 (ft/sec) (2.19)

Ohsaki and Iwasaki (1973) analyzed over 200 sets of V_s data, collected by predominantly the downhole method, throughout Japan. They developed correlations between shear modulus and uncorrected N-value, geologic age and soil type. Osaki and Iwasaki then presented the following equation for all soil types:
$$G_{\rm max} = 124 \cdot N^{0.78}$$
 (tsf) (2.20)

By assuming a constant unit weight of 112.4 pcf (\cong 1.80 t/m³), they presented the following relationship between V_s and uncorrected N-value:

$$V_s = 267 \cdot N^{0.39}$$
 (ft/sec) (2.21)

Ohsaki and Iwasaki (1973) also presented their analysis by classifying to geologic age, soil type. The parameters and correlation coefficients are compared and shown in Table 2.5, with the form of the equation as:

$$G_{\max} = a \cdot N^{b} \quad (\text{tsf}) \tag{2.22}$$

Ohta and Goto (1978) accumulated and analyzed almost 300 sets data from soil in Japan. The data analysis was carefully thought out regarding shear wave velocity, geologic age, depth, N-value, and soil types. They developed 15 empirical formulas, with only the 8 best-fit equations shown in Table 2.6, which involve Nvalue.

Imai and Tonuchi (1982) synthesized data from 400 sites throughout Japan, 1654 sets of data in all, and developed the following equation:

$$V_s = 318 \cdot N^{0.314}$$
 (ft/sec) (2.23)

Imai and Touchi also showed relations for shear wave velocity and shear modulus from SPT N-value for various soil categories as shown in Table 2.7.

Seed, Idriss, and Arango (1983) suggested using the following equation for sand and silty sand to access shear modulus, G_{max} , and shear wave velocity, V_s , using N-value:

$$G_{\rm max} = 65 \cdot N \qquad (\rm tsf) \tag{2.24}$$

$$V_s = 185 \cdot N^{0.5}$$
 (ft/sec) (2.25)

Sykora and Strokoe (1983) published correlations between the shear wave velocity and SPT N-value for cohesionless soils, and the relationship is:

$$V_s = 100.584 \cdot N^{0.29}$$
 (m/s) (2.26)

According to Dickenson (1994), the proposed formula by Ohta and Goto (1978) can be reasonably be used for sandy soils in the San Francisco Bay area. Dickenson (1994) then presented the following formula for cohesionless soil in the San Francisco Bay area:

$$V_s = 290 \cdot (N+1)^{0.30}$$
 (ft/sec) (2.27)

Based on the results of the above review of correlations, several were selected as appropriate for the northern area soils. These are presented in the later section.

2.4.2 MODULUS REDUCTION

Laboratory tests have shown that soil stiffness is influenced by cyclic strain amplitude, void ratio, mean principal effective stress, plasticity index, overconsolidation ratio, and number of loading cycles. The secant shear modulus, which will be described in Section 2.5.2, of an element soil varies with cyclic shear strain amplitude. At low strain amplitudes, the secant shear modulus is high, but it decreases as the strain amplitude increases. The locus of points corresponding to the tips of hysteresis loops of various cyclic strain amplitudes is called a backbone (or skeleton) curve, as shown in Figure 2.7; its slope at the origin (zero cyclic strain amplitude) represents the largest value of the shear modulus, G_{max} . At greater cyclic strain amplitudes, the modulus ratio G_{sec}/G_{max} drops to value of lass than 1. Characterization of the stiffness of an element of soil therefore requires consideration of both G_{max} and the manner in which the modulus ratio G/G_{max} varies with cyclic strain amplitude and other parameters. The variation of the modulus ratio with shear strain is described graphically by a modulus reduction curve, as shown in Figure 2.8.

In early years of geotechnical earthquake engineering, the modulus reduction behaviors of coarse- and fine-grained soils were treated separately (e.g., Seed and Idriss, 1970). Recent research, however, has revealed a gradual transition between the modulus reduction behavior of nonplastic coarse-grained soil and plastic fine-grained soils.

Zen et al. (1978) and Kokushu et al. (1982) first noted the influence of soil plasiticity on the shape of the modulus reduction curve; the shear modulus of highly plastic soils was observed to degrade more slowly with shear strain than did lowplasticity soils. After reviewing experimental results from a broad range of materials, Dobry and Vucetic (1987) and Sun et al. (1988) concluded that the shape of the modulus reduction curve is influenced more by the plasiticity index than by the void ratio and presented curves of the type shown in Figure 2.9. These curve show that the linear cyclic threshold shear strain is greater for highly plastic soils than for soils of low plasticity. This characteristic is extremely important, it can strongly influence the manner in which a soil deposit will amplify or attenuate earthquake motions. The PI = 0 modulus reduction curve from Figure 2.9 is very similar to the average modulus reduction curve that was commonly used for sands (Seed and Idriss, 1970) when coarse- and fine- grained soil were treated separately. This similarity suggests that the modulus reduction curves of Figure 2.9 may be applicable to both fine- and coarsegrained soils. The difficulty of testing very large specimens has precluded the widespread testing of gravelly soils in the laboratory, but available test data indicate that the average reduction curve for gravel is similar to, though slightly flatter than, that of sand (Seed et al., 1986; Yasuda and Matsumoto, 1993).

Modulus reduction behavior is also influenced by effective confining pressure, particularly for soils of low plasticity (Iiwasaki et al., 1978; Kokushu, 1980). The linear threshold shear strain is greater at high effective confining pressure than at low effective confining pressures. The effects of effective confining pressure and plasticity index on modulus reduction behavior were combined by Ishibashi and Zhang (1993) in the form:

$$\frac{G}{G_{\max}} = K(\gamma, PI) \cdot (\sigma_m)^{m(\gamma, PI) - m_0}$$
(2.28)

where

$$K(\gamma, PI) = 0.5 \left\{ 1 + \tanh\left[\ln\left(\frac{0.000102 + n(PI)}{\gamma}\right)^{0.492}\right] \right\}$$

$$m(\gamma, PI) - m_0 = 0.272 \left\{ 1 - \tanh\left[\ln\left(\frac{0.000556}{\gamma}\right)^{0.4}\right] \right\} \exp\left(-0.0145 PI^{1.3}\right)$$

$$n(PI) = \begin{cases} 0.0 & for \quad PI = 0\\ 3.37 \times 10^{-6} PI^{1.404} & for \quad 0 < PI \le 15\\ 7.0 \times 10^{-7} PI^{1.976} & for \quad 15 < PI \le 70\\ 2.7 \times 10^{-5} PI^{1.115} & for \quad PI > 70 \end{cases}$$

The effect of confining pressure on modulus reduction behavior of low- an highplasticity soils is illustrated in Figure 2.10. The influence of various environmental and loading conditions on the modulus ratio of normally consolidated and moderately overconsolidated clays is described in Table 2.8.

2.4.3 DAMPING RATIO

Theoretically, no hyteretic dissipation of energy takes place at strains below the linear cyclic threshold shear strain. Experimental evidence, however, shows that some energy is dissipated even at very low strain levels, so the damping ratio is never zero. Above the threshold strain, the breadth of the hyteresis loops exhibited by a cyclically loaded soil increase with increasing cyclic strain amplitude, which indicates that the damping ratio increases with increasing strain amplitude.

Just as modulus reduction behavior is influenced by plasticity characteristics, so is damping behavior (Kokushu et al., 1982; Dobry and Vucetic, 1987; Sun et al., 1988). Damping ratios of highly plastic soils are lower than those of low plasticity soils at the same cyclic strain amplitude, as shown in Figure 2.11. The PI = 0 damping curve from Figure 2.11 is nearly identical to the average damping curve that was used for coarse-grained soils when they were treated separately from fine-grained soils. This similarity suggests that the damping curves of Figure 2.11 can be applied to both fine- and coarse-grained soils. The damping behavior of gravel is very similar to that of sand (Seed et al., 1984a).

Damping behavior is also influenced by effective confining pressure, particularly for soils of low plasticity. Ishibashi and Zhang (1993) developed an empirical expression for the damping ratio of plastic and nonplastic soils. Using Equation (2.28) to compute the modulus reduction factor, G/G_{max} , the damping ratio is given by:

$$\xi = 0.333 \cdot \frac{1 + \exp\left(-0.0145 PI^{1.3}\right)}{2} \left[0.586 \left(\frac{G}{G_{\text{max}}}\right)^2 - 1.547 \frac{G}{G_{\text{max}}} + 1 \right] \quad (2.29)$$

The influence of various environmental and loading conditions on the damping ratio of normally consolidated and moderately overconsolidated soils is described in Table 2.9.

2.5 GROUND RESPONSE ANALYSIS

2.5.1 ONE-DIMENSIONAL GROUND RESPONSE

Geotechnical earthquake engineers are usually confronted with problems of determining the spatial and temporal variation of seismic motions in soil profile from a motion specified at a single point. Solutions to such problems, which are called as ground response problem, are crucial for liquefaction and soil-structure interaction analyses. Engineering analyses of ground response usually accept three assumptions: (1) ground motions developed near the surface of a soil deposit may be attributed merely to the vertical propagation of shear waves, (2) the ground surface, the interface between each layer, and the bed rock are essentially horizontal, (3) the material in each layer is homogeneous viscoelastic, but nonlinear. From these three assumptions, many researchers have developed computer programs to model one-dimensional wave propagation through soil (e.g. Schnabel et al., 1972). For one-dimensional ground response analysis, the soils and bedrock surface are assumed to extend infinitely in the horizontal direction.

Before describing of the ground response analysis, it is necessary to define several terms that are commonly used to describe ground motions. From Figure 2.12a, the motion at the surface of a soil deposit is the free surface motion. The motion at the base of the soil deposit (also the top of bedrock) is called a bedrock motion. The motion at a location where bedrock is exposed at the ground surface is called a rock outcropping motion. If the soil deposit was not present, as shown in Figure 2.12b, the motion at the top of bedrock would be the bedrock outcropping motion.

The nonlinear behavior of soil can be practically approximated by the equivalent linear method, as will be described below.

2.5.2 EQUIVALENT LINEAR METHOD

A typical soil subjected to symmetric cyclic loading as would be expected beneath a level ground surface far from adjacent structures, might exhibit a hysteresis loop of the type shown in Figure 2.13. This hysteresis loop can be described in two ways: first, by the actual path of the loop itself, and second, by parameters that describe its general shape. In general terms, two important characteristics of the shape of a hysteresis loop are its inclination and its breadth. The inclination of the loop depends on the stiffness of the soil, which can be described at any point during the load process by the tangent shear modulus, G_{tan} . Obviously, G_{tan} varies throughout a cycle of loading, but its average value over the entire loop can be approximated by the secant shear modulus:

$$G_{\rm sec} = \frac{\tau_c}{\gamma_c} \tag{2.30}$$

where τ_c and γ_c are the shear stress and shear strain amplitude, respectively. Thus G_{sec} describes the general inclination of the hysteresis loop. The breadth of the hysteresis loop is related to the area, which as a measure of energy dissipation, can conveniently be described by the damping ratio:

$$\xi = \frac{W_D}{4\pi W_s} = \frac{1}{2\pi} \cdot \frac{A_{loop}}{G_{sec} \gamma_c^2}$$
(2.31)

where W_D is the dissipated energy, W_S the maximum strain energy, and A_{loop} the area of the hysteresis loop. The parameters G_{sec} and ξ are often referred to as equivalent linear material parameters. For ground response analysis, they are used directly to describe the soil behavior. Because some of the most commonly used methods of ground response analysis are based on the use of equivalent linear properties, considerable attention has been given to the characterization of G_{sec} and ξ for different soils. It is important to recognize, however, that the equivalent linear model is only an approximation of the actual nonlinear behavior of the soil. The assumption of linearity embedded in its use has important implications when it is used for ground response analysis. It also means that it cannot be used directly for problems involving permanent deformation or failure; equivalent linear models imply that the strain will always return to zero after cyclic loading, and since a linear material has no limiting strength, failure cannot occur.

The dynamic behavior of soil cannot be analyzed by constant elastic modulus and constant damping because these parameters depend on strain level. Nevertheless, we can solve the intricate problem by applying the equivalent linear method (Seed and Idriss, 1970). A good approximation of the effects of soil nonlinearity on the response is modeled by the use of strain compatible shear modulus and damping ratio in a sequence of linear analysis through an iterative process.

In any ground response analysis, the equivalent linear method begins with a linear analysis using estimated soil properties, shear modulus (G) and damping ratio (ξ) , in each layer of the soil profile. This analysis yields complete time histories of shear strain, from which the effective shear strain amplitude is determined for each layer. The effective shear strain amplitude, which is function of the magnitude of earthquake, is typically taken as 65% of the maximum shear strain. Substituting in the calculated effective strain amplitudes, an improved set of shear modulus and damping ratio are obtained from suitable soil data curve, as described in Section 2.4.2 and 2.4.3. A new linear analysis is performed with these properties. The iterative process is terminated when the properties from two consecutive analysis, shear modulus and damping ratio, differ from each other by less than a specified tolerance (usually 5 to 10%). This process will usually converge in less than 5 iterations. The output of the

last iteration is taken as the final solution to approximate a nonlinear solution. Referring to Figure 2.14, the iterative procedure operates as follows:

Step 1: Initial estimates of G and ξ are made for each layer. The initially estimated values usually correspond to the same strain level; the low-strain values (0.001%) are often used for the initial estimate.

Step 2: The estimated G and ξ values are used to compute the ground response, including time histories of shear strain for each layer.

Step 3: The effective shear strain in each layer is determined from the maximum shear strain in the computed shear strain time history. For layer j:

$$\gamma_{eff \ i}^{(i)} = R_{\gamma} \cdot \gamma_{\max \ i}^{(i)}$$

where the superscript refers to the iteration number and R_{γ} is the ratio of the effective shear strain to maximum shear strain. R_{γ} depends on earthquake magnitude (Idriss and Sun, 1992) and can be estimated from:

$$R_{\gamma} = \frac{M-1}{10}$$

Step 4: From this effective shear strain, new equivalent linear value, $G^{(i+1)}$ and $\xi^{(i+1)}$ are chosen for the next iteration.

Step 5: Steps 2 to 4 are repeated until differences between the computed shear modulus and damping ratio values in two successive iterations fall below some predetermined value in all layers. Although convergence is not absolutely guaranteed, differences of less than 5 to 10% are usually achieved in three or five iterations (Schnabel et al., 1972).

The equivalent linear approach to one-dimensional ground response analysis of layered sites has been coded into a widely used computer program called SHAKE (Schnabel et al., 1972).

2.6 AMPLIFICATION OF EARTHQUAKE GROUND MOTIONS

Earthquake produces seismic waves that travel in every direction from the seismic source. A train of these seismic waves, when recorded by an instrument, is manifested as a record of earthquake ground motions. Basic characteristics of strong ground motions are influenced by the source, travel path, and modified by local soil conditions. In some cases, local soil conditions play a predominant role to the ground motion features. Therefore, the concept of soil amplification is defined as modification of the input bedrock ground motion by the overlying unconsolidated materials.

The phenomena were initially derived from the great differences in the degree of earthquake damage at places about the same distance from the epicenter. Early example can be come back to 1923 Kanto earthquake, Japan. Still further, it was shown by the strong earthquake in Caracas in 1967. The most notable example was 1985 Mexico earthquake in which there were the enormous differences in intensities of shaking and associated building damage in the different parts of the city. More recent two cases are the 1988 Armenia earthquake, Russia and the 1989 Loma Prieta earthquake, California, USA.

2.6.1 AMPLIFICATION FACTOR

Amplification factor is defined as the ratio between earthquake ground motion intensity on soil surface and that on rock surface. In soft surface layers, vibrations are amplified due to multi-reflection phenomena. In order to express the degree of magnification quantitatively, it will suffice the corresponding component vibrations of the earthquake wave in the base ground and the waveform at the ground surface and obtain the magnification of the latter over the former. In this regard, the formula proposed by Kanai (1957) is presented. Kanai (1957) has arrived at the formula below by combining the results of the actual measurements with the theoretical calculations:

$$A = 1 + \frac{1}{\sqrt{\left[\frac{1+k}{1-k} \cdot \left(1 - \left(\frac{T}{T_G}\right)^2\right)\right]^2 + \left(\frac{0.3}{\sqrt{T_G}} \cdot \frac{T}{T_G}\right)^2}}$$
(2.31)

- where A = soil amplification factor (i.e., ratio of amplitude of response wave at the ground surface and that of incident wave)
 - T = period of component vibration of seismic wave
 - T_G = predominant period of surface layer

$$k = (\rho_1 \beta_1) / (\rho_2 \beta_2)$$

- ρ_1 = density of soft surface layer
- ρ_2 = density of base layer
- β_1 = velocity of seismic wave in surface layer
- β_2 = velocity of seismic wave in base layer

The graphical form of this formula is shown in Figure 2.15 in which the following properties are recognized in the multiplication of the surface layer:

- 1. The amplification factor is largest when the period of the input wave coincides with the predominant period of the ground.
- 2. The ground whose predominant period is longer has a larger ratio of magnification.
- 3. The amplification factor of ground having very long predominant period becomes a constant value 2/(1+k) independent of the period of the input wave.

4. The soil amplification factor is more conspicuous for ground having a smaller value k than for ground having a larger one.

In fact, the amplitude of the amplification ground motion is a function of the shear wave velocity, the density and material damping, the thickness, water content, the geometry of the unconsolidated deposits and underlying rock, and where the surface of bedrock topography is irregular.

2.6.2 EFFECTS OF LOCAL SITE CONDITIONS ON GROUND MOTION

Case histories of ground response in Mexico City (i.e., 1985 Mexico earthquake), the San Francisco Bay area (i.e., 1989 Loma Prieta earthquake), and many other locations have clearly shown that local site conditions strongly influence peak acceleration amplitudes.

Comparisons of peak acceleration attenuation relationships for sites underlain by different types of soil profiles show distinct trends in amplification behavior (Seed et al., 1976b). Although attenuation data are scattered, overall trends suggest that peak accelerations at the surfaces of soil deposits are slightly greater than on rock when peak accelerations are small and somewhat smaller at higher acceleration level, as shown in Figure 2.16. Based on data from Mexico City and the San Francisco Bay area, and on additional ground response analyses, Idriss (1990) related peak accelerations on soft soil sites to those on rock sites, as shown in Figure 2.17. At low to moderate acceleration levels (i.e., less than about 0.4g), peak accelerations at soft sites are likely to be greater than on rock sites. In some cases, such as Mexico City in 1985 and the San Francisco Bay area in 1989, relatively small rock accelerations may cause high accelerations at the surface of soft soil deposits. At higher acceleration levels, however, the low stiffness and nonlinearity of soft soils often prevent them from developing peak accelerations as large as those observed on rock.

Local site conditions also influence the frequency content of surface motions and hence the response spectra (i.e., the variation of peak dynamic response of a single-degree-of-freedom system, for different values of its natural frequency or period, given a specified input transient motion) they produce. Seed et al. (1976) computed response spectra from ground motions recorded at sites underlain by four categories of site conditions: rock sites, stiff soil sites (less than 200 ft (61 m) deep), deep cohesionless soil sites (greater than 250 ft (76 m) deep), and sites underlain by soft to medium-stiff clays and sands. Normalizing the computed spectra (by dividing spectral accelerations by the peak ground acceleration) illustrates the effects of local soil conditions on the shape of the spectra, as shown in Figure 2.18. The effects are apparent: at periods above about 0.5 sec, spectral amplifications are much higher for soil sites than for rock sites. At longer periods, the spectral amplification increases with decreasing subsurface profile stiffness. Figure 2.18 clearly shows that deep and soft soil deposits produce greater proportions of long-period (low-frequency) motion. This effect can be very significant, particularly when long-period structures such as bridges and tall buildings are founded on such deposits (i.e., the dynamic response of long-period structures is increased if the local ground conditions are soft).

2.7 LIQUEFACTION

2.7.1 GENERAL

Liquefaction describes a phenomenon in which cyclic stresses produced by ground shaking induce excess pore water pressure in cohesionless soils of sufficient magnitude to cause partial or complete loss of shear strength of the deposits. These soils may thereby acquire a high degree of mobility, leading to damaging deformations. This phenomenon only occurs below the water table, but after liquefaction has developed, it can propagate upward into overlying non-saturated soil as excess pore water escapes. Liquefaction susceptibility under a given earthquake is related to the gradation and relative density characteristics of the soil, the in situ stresses prior to ground motion, and the depth of the water table as well as other factors.

It has long been known that the intensity of ground shaking during earthquake and its associated damage to buildings and other properties are affected by the geologic and ground conditions. The phenomenon of liquefaction has been reported in numerous earthquakes to cause landslides and damage to buildings, but it was more dramatically illustrated by widespread building damage and large landslides in the Niigata, Japan earthquake of 1964 and the Alaska, USA earthquake the same year.

Ambraseys (1988) compiled worldwide data from shallow earthquakes to estimate a limiting epicentral distance beyond which liquefaction has not been observed in earthquakes of different magnitudes (Figure 2.19). The distance to which liquefaction can be expected increases dramatically with increasing magnitude. While relationships of the type shown in Figure 2.19 offer no guarantee that liquefaction cannot occur at greater distances, they are helpful for estimation of regional liquefaction hazard scenarios.

Soil deposits that are susceptible to liquefaction are formed within a relatively narrow range of geological environments (Youd, 1991). The depositional environment, hydrological environment, and age of a soil deposit all contribute to its liquefaction susceptibility.

Since liquefaction requires the development of excess pore water pressure, liquefaction susceptibility is influenced by the compositional characteristics that influence volume change behavior. Compositional characteristics associated with high volume change potential tend to be associated with high liquefaction susceptibility. These characteristics include particle size, shape, and gradation.

For many years, liquefaction-related phenomena were thought to be limited to sands. Finer-grained soils were considered incapable of generating the high pore pressures commonly associated with liquefaction, and coarser-grained soils were considered too permeable to sustain any generated pore water pressure long enough for liquefaction to develop. Liquefaction of nonplastic silts has been observed (Ishihara, 1984, 1985) in the laboratory and the field, indicating that plasticity characteristics rather than grain size alone influence the liquefaction susceptibility of fine-grained soils. Coarse silts with bulky particle shape, which are nonplastic and cohesionless, are fully susceptible to liquefaction (Ishihara, 1993); finer silts with flaky or platelike particles generally exhibit sufficient cohesion to inhibit liquefaction. Clays remain nonsusceptible to liquefaction, although sensitive clays can exhibit strain-softening behavior (referred to as being contractive or flow) similar to that of liquefied soil.

Liquefaction susceptibility is influenced by gradation. Well-graded soils are generally less susceptible to liquefaction than poorly graded soils; the filling of voids between larger particles by smaller particles in a well-graded soil results in lower volume change potential under drained conditions and, consequently, lower excess pore water pressures under undrained conditions. Field evidence indicates that most liquefaction failures have involved uniformly graded soils.

2.7.2 CAUSES OF SOIL LIQUEFACTION

The basic cause of liquefaction of sands has been understood, in a qualitative way, for many years. If a saturated sand is subjected to ground vibrations, it tends to compact and decrease in volume; if drainage is unable to occur, the tendency to decrease in volume results in an increase in pore water pressure, and if the pore water pressure builds up to the point at which it is equal to the overburden pressure, the effective stress becomes zero, the sand loses its strength completely, and it develops a liquefied state.

In more quantitative terms, it is now generally believed that the basic cause of liquefaction in saturated cohesionless soils during earthquakes is the buildup of excess hydrostatic pressure due to the application of cyclic shear stresses induced by the ground motions. These stresses are generally considered to be due primarily to upward propagation of shear waves in a soil deposit, although other forms of wave

motions are also expected to occur. Thus, soil elements can be considered to undergo a series of cyclic stress conditions as illustrated in Figure 2.20, the stress series being somewhat random in pattern but nevertheless cyclic in nature.

As a consequence of the applied cyclic stresses, the structure of the cohesionless soil tends to become more compact with a resulting transfer of stress to the pore water and a reduction in stress on the soil grains. As a result, the soil grain structure rebounds to the extent required to keep the volume constant, and this interplay of volume reduction and soil structure rebound determines the magnitude of the increase in pore water pressure in the soil (Martin et al., 1975).

As the pore water pressure approaches a value equal to the applied confining pressure, the sand begins to undergo deformations. If the sand is loose, the pore water pressure will increase suddenly to a value equal to the applied confining pressure, and the sand will rapidly begin to undergo large deformations with shear strains that may exceed $\pm 20\%$ or more. If the sand will undergo virtually unlimited deformations without mobilizing significant resistance to deformation, it can be said to be liquefied. If, on the other hand, the sand is dense, it may develop a residual pore water pressure, on completion of a full stress cycle, which is equal to the confining pressure (a peak cyclic pore pressure ratio of 100%), but when the cyclic stress is reapplied on the next stress cycle, or if the sand is subjected to monotonic loading, the soil will tend to dilate, the pore pressure will drop if the sand is undrained, and the soil will ultimately develop enough resistance to withstand the applied stress. However, it will have to undergo some degree of deformation to develop the resistance, and as the cyclic loading continues, the amount of deformation required to produce a stable condition may increase. Ultimately, however, for any cyclic loading condition, there appears to be a cyclic strain level at which the soil will be able to resist any number of cycles of a given stress without further increase in maximum deformation (De alba et al., 1976). This type of behavior is termed "cyclic mobility" and it is considerably less serious than liquefaction, its significance depending on the magnitude of the limiting strain. It should be noted, however, that once the cyclic stress applications stop, if they return to a zero stress condition, there will be a residual pore water pressure in the soil equal to the overburden pressure, and this will inevitably lead to an upward flow of water in the soil which could have deleterious consequences for overlying layers. Ishihara (1996) described the meaning of cyclic mobility as a state where zero effective stress or initial liquefaction occurs momentarily whenever there is no applied shear stress (a large amount of shear strain can occur after the onset of initial liquefaction) but the effective stress regains with the application of shear stress. Mori, Seed, and Chan (1978) defined the initial liquefaction with limited strain potential or cyclic mobility as a condition in which cyclic stress applications cause limited strains to develop either because of the remaining resistance of the soil to deformation or because the soil dilates, the pore pressure drops, and the soil stabilizes under the applied loads.

Liquefaction of sand in this way may develop in any zone of a deposit where the necessary combination of in situ conditions and vibratory deformations may occur. Such a zone may be at the surface or at some depth below the ground surface, depending only on the state of the sand and the induced motions.

However, liquefaction of the upper layers of a deposit may also occur, not as a direct result of the ground motion to which they are subjected, but because of the development of liquefaction in an underlying zone of the deposit. Once liquefaction develops at a some depth in a mass of sand, the excess hydrostatic pressures in the liquefied zone will dissipate by flow of water in an upward direction. If the hydraulic gradient becomes sufficient large, the upward flow of water will induce a quick or liquefied condition in the surface layers of the deposit. Liquefaction of this type will depend on the extent to which the necessary hydraulic gradient can be developed and maintained; this, in turn, will be determined by the compaction characteristics of the sand, the nature of ground deformations, the permeability of the sand, the boundary drainage conditions, the geometry of the particular situation, and the duration of the induced vibrations.

2.7.3 FAILURE MECHANISMS FOR SLOPE GROUND

2.7.3.1 LATERAL SPREADS

Lateral spreads involve lateral displacement of large, surficial block of soil as a result of liquefaction of a subsurface layer (Figure 2.21). Displacement occurs in response to the combination of gravitational forces and inertial forces generated by an earthquake. Lateral spreads generally develop on gentle slopes (most commonly less than 3 degrees) and move toward a free face such as an incised river channel. Horizontal displacements commonly range up to several meters. The displaced ground usually breaks up internally, causing fissures, scarps, horsts, and grabens to form on the failure surface. Lateral spreads commonly disrupt foundations of buildings built on or across the failure, sever pipelines and other utilities in the failure mass, and compress or buckle engineering structures, such a bridges, founded on the toe of the failure.

2.7.3.2 FLOW FAILURES

Flow failures are the most catastrophic ground failures caused by liquefaction. These failures commonly displace large masses of soil laterally tens of meters and in a few instances, large masses of soil have traveled tens of kilometers down long slopes at velocities ranging up to tens of kilometers per hour. Flows may be comprised of completely liquefied soil or blocks of intact material riding on a layer of liquefied soil. Flows develop in loose saturated sands or silts on relatively steep slopes, usually greater than 3 degrees, as shown in Figure 2.22.

2.7.4 FAILURE MECHANISMS FOR HORIZONTAL GROUND

2.7.4.1 SAND BOILS

Liquefaction is often accompanied by the development of sand boils. During and following earthquake shaking, seismically induced excess pore pressures are dissipated predominantly by the upward flow of pore water. This flow produces upward-acting forces on soil particles, these forces can loosen the upper portion of the deposit and leave it in a state susceptible to liquefaction in future earthquake (Youd, 1984). If the hydraulic gradient driving the flow reaches a critical value, the vertical effective stress will drop to zero and the soil will be in a quick condition. In such cases, the water velocities may be sufficient to carry soil particles to the surface. In the field, soil conditions are rarely uniform so the escaping pore water tends to flow at high velocity through localized cracks or channels. Sand particles can be carried through these channels and ejected at the ground surface to form sand boils.

Sand boils are of little engineering significance by themselves, but they are useful indicators of high excess pore pressure generation. Shaking table (Liu and Qiao, 1984) and centrifuge (Fiegel and Kutter, 1992) tests have shown that pore water draining from the voids of the loose layers can accumulate beneath the less previous layers and form water interlayers, as shown in Figure 2.23. Sand boils can develop when the water interlayers break through to the ground surface. Some redistribution of soil grains is also likely to accompany the formation of water interlayers; specifically the sand immediately beneath the water interlayer may be loosened by the upward flow of water toward the interlayer. If such conditions develop beneath an inclined ground surface, the presence of the water interlayer and the reduced steady state strength of the loosened sand immediately beneath it can contribute to large flow deformations.

2.7.4.2 GROUND OSCILLATION

The occurrence of liquefaction at depth beneath a flat ground surface can decouple the liquefied soils from the surficial soils and produce large, transient ground oscillations. The surficial soils are often broken into block (Figure 2.24) separately by fissures that can open and close during the earthquake. Ground waves with amplitudes of up to several meters have been observed during ground oscillation, but permanent displacements are usually small. Prediction of the amplitude of ground oscillation at a particular site is very difficult; even detailed nonlinear ground response analyses can provide only crude estimates.

2.7.4.3 LOSS OF BEARING STRENGTH

When the soil supporting a building or other structure liquefies and loses strength, large deformations can occur within the soil which may allow the structure to settle and tip, as shown in Figure 2.25. Conversely, buried tanks and piles may rise buoyantly through the liquefied soil. For example, many buildings settled and tipped during the 1964 Niigata, Japan, earthquake. The most spectacular bearing failures during that event were in the Kawangishicho apartment complex where several fourstory buildings tipped as much as 60 degrees, as shown in Figure 2.26. Apparently, liquefaction first developed in a sand layer several meters below ground surface and then propagated upward through overlying sand layers. The rising wave of liquefaction weakened the soil supporting the buildings and allowed the structures to slowly settle and tip.

2.7.4.4 SUBSIDENCE AND SETTLEMENT

In many cases, the weight of a structure will not be great enough to cause the large settlements associated with soil bearing capacity failure described above. However, smaller settlements may occur as soil pore water pressures dissipate and the soil consolidates after the earthquake. These settlements may be damaging, although they would tend to be much less so than the large movements accompanying flow failures, lateral spreading, and bearing capacity failures. The eruption of sand boils is a common manifestation of liquefaction that can also lead to localized differential settlements.

2.8 EVALUATION OF LIQUEFACTION RESISTANCE OF SOILS

A number of approaches to evaluation of the potential for initial of liquefaction or liquefaction resistance have developed over the years. In the following sections, the most common of these approaches will be reviewed.

2.8.1 DETERMINISTIC APPROACH OR CYCLIC STRESS APPROACH

In the 1960s and 1970s, many advances in the state of knowledge of liquefaction phenomena resulted from the pioneering work of H.B. Seed and his colleagues at the University of California at Berkley. This research was directed largely toward evaluation of the loading conditions required to trigger liquefaction. This loading was described in terms of cyclic shear stresses, and liquefaction potential was evaluated on the basis of the amplitude and number of cycles of earthquake-induced shear stress. The general approach has come to be known as the cyclic stress approach.

Seed and Lee (1966) defined initial liquefaction as the point at which the increase in pore pressure is equal to the initial effective confining pressure (i.e., when $u_{excess} = \sigma_{3c}$ or when pore pressure ratio, $r_u = 100\%$). Because most of the early laboratory testing investigations were based on cyclic triaxial tests on isotropically

consolidated specimens (consequently, with complete stress reversal), initial liquefaction could be produced in both loose and dense specimens.

The cyclic stress approach is conceptually quite simple: the earthquakeinduced loading, expressed in terms of cyclic shear stresses, is compared with the liquefaction resistance of the soil, also expressed in terms of cyclic shear stresses. Application of the cyclic stress approach, however, requires careful attention to the manner in which the loading conditions and liquefaction resistance are characterized.

2.8.1.1 CYCLIC STRESS RATIO (CSR) AND CYCLIC RESISTANCE RATIO (CRR)

Calculation, or estimation, of two variables is required for evaluation of liquefaction resistance of soils: (1) the seismic demand on a soil layer, expressed in terms of CSR; and (2) the capacity of the soil to resist liquefaction, expressed in terms of CRR. The latter variable has been termed the cyclic stress ratio or the cyclic stress ratio required to generate liquefaction, and has been given different symbols by different writers. For example, Seed and Harder (1990) used the symbol CSR ℓ , Youd (1993) used the symbol CSRL, and Kramer (1996) used the symbol CSR_L to denote this ratio. To reduce confusion and to better distinguish induced cyclic shear stresses from mobilized liquefaction resistance, the capacity of soil to resist liquefaction is termed the CRR in this research.

2.8.1.2 EVALUATION OF CYCLIC STRESS RATIO (CSR)

The shear stresses developed at any point in a soil deposit during an earthquake appear to be due primarily to the vertical propagation of shear waves in the deposit. This leads to a simplified procedure for evaluating the induced shear stresses (Seed and Idriss, 1971). If the soil column above a soil element at depth h behaved as rigid body, the maximum shear stress on the soil element would be

$$\left(\tau_{\max}\right)_r = \frac{\gamma h}{g} \cdot a_{\max}$$
 (2.32)

where a_{max} is the maximum horizontal ground surface acceleration, g is the acceleration of gravity, and γ is the unit weight of the soil; see Figure 2.27(a). Because the soil column behaves as a deformable body, the actual shear stress at depth h, $(\tau_{\text{max}})_d$, as determined by a ground response analysis will be less than $(\tau_{\text{max}})_r$ and might be expressed by

$$\left(\tau_{\max}\right)_d = r_d \cdot \left(\tau_{\max}\right)_r \tag{2.33}$$

where r_d is a stress reduction coefficient with a value less than 1. The variations of $(\tau_{\text{max}})_r$ and $(\tau_{\text{max}})_d$ will typically have the form shown in Figure 2.27(b) and, in any given deposit, the value of r_d will decrease from a value 1 at the ground surface to much lower values at large depths, as shown in Figure 2.27(c).

Computations of the value of r_d for a wide variety of earthquake motions and soil conditions having sand in the upper 15 m. have shown that r_d generally falls within the range of values shown in Figure 2.28. It may be seen that in the upper 9 or 12 m., the scatter of the results is not great and, for any of the deposits, the error involved in using the average values shown by the dashed line would generally be less than about 5%. For routine practice, the following equations may be used to estimate average values of r_d (Liao and Whitman, 1986a):

$$r_d = 1.0 - 0.00765 z$$
 for $z \le 9.15$ m (2.34a)

$$r_d = 1.174 - 0.0267 z$$
 for $9.15 < z \le 23$ m (2.34b)

where z is depth below ground surface in meters.

Thus the assessment of the maximum shear stress developed during an earthquake can be made from the relationship

$$\tau_{\max} = \frac{\gamma h}{g} \cdot a_{\max} \cdot r_d \tag{2.35}$$

where r_d are taken from the dashed line in Figure 2.28 or Equation (2.34). The critical depth for development of liquefaction, if it is going to occur, will normally be in the depth covered by this relationship.

The actual time history of shear stress at any point in a soil deposit during an earthquake will have an irregular form such as that shown in Figure 2.29. From such relationships it is necessary appropriate to determine the equivalent uniform average shear stress. By appropriate weighting of the individual stress cycles, based on laboratory test data, this determination can readily be made. However, after making these determinations for a number of different cases it has been found that with a reasonable degree of accuracy, the average equivalent uniform shear stress, τ_{av} , is about 65% of the maximum shear stress, τ_{max} . Combining this result with the above expression for τ_{max} it follows that for practical purposes, the average cyclic shear stress may be determined by:

$$\tau_{av} = 0.65 \cdot \frac{\gamma h}{g} \cdot a_{\max} \cdot r_d$$
(2.36)
or equivalently:

$$CSR = \frac{\tau_{av}}{\sigma_{vo}} = 0.65 \cdot \frac{\sigma_{vo}}{\sigma_{vo}} \cdot \frac{a_{\max}}{g} \cdot r_d$$
(2.37)

where CSR = cyclic stress ratio caused by the earthquake, $\sigma_{vo} = \text{total overburden}$ pressure on sand layer under consideration, and σ_{vo} = initial effective overburden pressure on sand layer under consideration

2.8.1.3 EVALUATION OF LIQUEFACTION RESISTANCE (CRR)

A plausible method for evaluating cyclic resistance ratio, CRR, is to retrieve and test undisturbed soil specimen in the laboratory. Unfortunately, in situ stress states generally cannot be reestablished in the laboratory, and granular soils are extremely difficult to sample without disturbance. Hence, methods to characterize the soil in the ground rely heavily on in situ tests. Several field tests have gained common usage for evaluation of liquefaction resistance, including the standard penetration test (SPT), the cone penetration test (CPT), and field measurements of shear wave velocity (V_s). But the most commonly used method for determining the liquefaction resistance is to use the data obtained from the standard penetration test because it is relatively easy to use, the test is economical compared to other types of field testing, and the SPT equipment can be quickly adapted and included as part of almost any type of drilling rig.

Seed and his colleagues developed correlations between the SPT N-value and the cyclic stress ratio to cause liquefaction or cyclic resistance ratio during earthquakes of magnitude M = 7.5. The correlations, which are presented in Figure 2.30, were based on the observed response of sites during earthquake loading. Sites were considered to have liquefied based on observed surface features, such as sand boils. Lower bound curves separating liquefied from non-liquefied sites are shown in Figure 2.30 corresponding to various fines contents of the sands. It should be noted that the CRR curves in Figure 2.30 are valid only for magnitude 7.5 earthquakes. Scaling factors to adjust CRR curves to other magnitudes are addressed in the next section of this chapter. To compare the ground conditions at one site with those of another, it is necessary to standardize the measured penetration values to the standard driving energy level of 60% of the theoretical free-fall energy of the hammer, the rate at which the blows are applied, the borehole diameter, the rod lengths, and the effective overburden pressure of 100 kPa (1 tsf). Hence, the CRR curves presented in Figure 2.30 show the normalized SPT N-value, $(N_1)_{60}$. Corrections can be applied to the penetration test results to compensate for the testing procedures by using the following equation:

$$\left(N_{1}\right)_{60} = N_{m}C_{N} \frac{E_{m}}{0.60E_{ff}}C_{B}C_{R}C_{S}$$
(2.38)

where N_m = the measured penetration resistance, Blows/ft

- C_N = correction factor to normalize N_m to an effective overburden pressure σ'_{vo} of 100 kPa (100 kPa = 1 tsf = 10 t/m² = 1 atm)
 - $= \sqrt{\frac{10}{\sigma_{vo}}} \le 1.7 \text{ for } \sigma_{vo}^{'} \le 20 \text{ t/m}^2 \text{ (Liao and Whitman, 1986b)}$ $= \frac{2.2}{\left(1.2 + \frac{\sigma_{vo}^{'}}{10}\right)} \le 1.7 \text{ for } 20 < \sigma_{vo}^{'} \le 30 \text{ t/m}^2 \text{ (Kayen et al., 1992)}$
- E_m = the actual hammer energy which is the percent of the theoretical free-fall hammer energy; the value of E_m depends on the type of hammer used and on the standards of practice in different parts of the world such as anvil lifting mechanism and the method of hammer release (E_m is 0.6 for a safety hammer and 0.45 to 0.6 for donut hammer, Seed et al. (1985))
- E_{ff} = the theoretical free-fall hammer energy

- C_B = correction factor for borehole diameter (C_B = 1.0 for boreholes of 65 to 115 mm diameter, 1.05 for 150 mm diameter, and 1.15 for 200 mm diameter)
- C_R = correction factor for rod length (C_R = 0.75 for up to 3 m of drill rods, 0.8 for 3 to 4 m of drill rods, 0.85 for 4 to 6 m of drill rods, 0.95 for 6 to 10 m of drill rods, and 1.0 for drill rods in excess of 10 m)
- C_s = correction for samplers with or without liners (C_s = 1.0 for standard sampler and 1.1 to 1.3 for sampler without liner)

Although application of rod length correction factors mentioned above will give more precise $(N_1)_{60}$ values, these corrections may be neglected for liquefaction resistance calculation for rod lengths between 3 and 10 m because rod length corrections were not applied to SPT test data from these depths in compiling the original liquefaction case history databases. Thus rod length corrections are implicitly incorporated into the empirical SPT procedure.

2.8.1.4 MAGNITUDE SCALING FACTOR (MSF)

The liquefaction resistance or CRR curves in Figure 2.30 apply only to magnitude 7.5 earthquakes. To adjust the CRR curves to magnitude smaller or larger than 7.5, one have to use correction factor termed "magnitude scaling factor (MSF)." Several MSF values have been proposed by various investigators (i.e., Seed and Idriss (1982), Ambraseys (1988), Arango (1996), Andrus and Stokoe (1997), and Youd and Noble (1997)). The MSF in Table 2.10 (Seed and Idriss, 1982) is the routinely used values in engineering practice. In addition, the 1996 NCEER workshop (Youd and Idriss, 1997) recommended that the lower bound for MSF values can be defined by the following equation:

$$MSF = (M_w / 7.5)^{-2.56}$$
(2.39)

where M_w is the moment magnitude. It should be noted that the local magnitude M_L , the surface wave magnitude M_s , and moment magnitude M_w scales are reasonably close to one another below a value of about 7, as shown in Figure 2.31. Thus for a magnitude of 7 or below, any one of these magnitude scales can be used to determine the MSF. At high magnitude values, the moment magnitude M_w tends to significantly deviate from the other magnitude scales, and the moment magnitude M_w should be used to determine the MSF from Table 2.10 or Equation (2.39).

2.8.1.5 EFFECTS OF INITIAL STATIC SHEAR STRESSES AND HIGH OVERBURDEN STRESSES ON LIQUEFACTION RESISTANCE

There are two important limitations associated with Figure 2.30. The field data correspond to level ground conditions with no initial static shear stresses on horizontal planes and to effective overburden pressures less than 15 t/m² (150 kPa). Seed (1983) outlined procedures for making corrections when these conditions are violated.

The first estimate of the liquefaction resistance of a soil element in a dam or slope is determined using the in situ $(N_1)_{60}$ penetration resistance and the appropriate curve for critical conditions in Seed's liquefaction assessment chart (Figure 2.30). This resistance must then be corrected for deviations from the standard conditions of the database underlying the chart. A typical element in a slope, for example, will carry a static shear stress, τ_{st} , on the horizontal plane and therefore has an initial shear stress ratio $\tau_{st} / \sigma_{vo}^{\prime} = \alpha$. A correction factor, K_{α} , is established for various values of α by laboratory tests. Note that, as is commonly assumed in practice, initial static shear increases liquefaction resistance substantially. However, this increase applies to

resistance to cyclic mobility rather than to liquefaction, that is to non-contractive materials. The liquefaction resistance of dilative soils (moderately dense to dense granular materials under low confining stress) increases with increased static shear stress. Conversely, the liquefaction resistance of contractive soils (loose soils and moderately dense soils under high confining stress) decreases with increased static shear stress. Seed and Harder (1990) proposed a chart (Figure 2.32) for a variation of correction factor, K_{α} , with initial shear stress ratio, α . The values of K_{α} vary for different soils and should be evaluated on a site-specific basis whenever possible. In addition, the 1996 NCEER workshop (Youd and Idriss, 1997) recommended that the proposed chart for K_{α} should not be used by non-specialists in geotechnical earthquake engineering or in routine engineering practice.

For effective overburden pressures other than in the range 10-15 t/m² (100-150 kPa) and for sites that support heavy structures, a correction factor K_{σ} is used. Cyclically loaded laboratory test data indicate that liquefaction resistance increases with increasing confining stress. The rate of increase, however, is nonlinear. To account for the nonlinearity between liquefaction resistance or CRR and effective overburden pressure, Seed and Harder (1990) introduced the values of K_{σ} for many soils as shown in Figure 2.33. Increasing confining pressure can lead to a substantial reduction in resistance to cyclic loading. The values of K_{σ} vary for different soils and should be evaluated on a site-specific basis whenever possible.

2.8.1.6 FACTOR OF SAFETY AGAINST LIQUEFACTION

Once the cyclic stress ratio caused by the anticipated earthquake (Equation (2.37)) and the liquefaction resistance of the soils corresponding to earthquake of magnitude 7.5 (Figure 2.30) has been characterized, liquefaction potential can be evaluated. The factor of safety against liquefaction (FS) is defined as follows:

$$FS = \left(\frac{CRR_{7.5}}{CSR}\right) \cdot MSF \cdot K_{\alpha} \cdot K_{\sigma}$$
(2.40)

where CSR = calculated cyclic stress ratio generated by the earthquake shaking, Equation (2.37); $CRR_{7.5}$ = cyclic resistance ratio for magnitude 7.5 earthquakes determined from Figure 2.30; MSF = magnitude scaling factor determined from Table 2.10 or Equation (2.39); K_{α} = correction factor for sloping ground or for initial static shear stress determined from Figure 2.32; and K_{σ} = correction factor for effective overburden pressure determined from Figure 2.33.

If the factor of safety in Equation (2.40) is equal or less than 1.0, liquefaction is said to take place. Otherwise, liquefaction does not occur. In addition, it should be noted that the higher the factor of safety, the more resistant the soil is to liquefaction.

2.8.2 PROBABILISTIC APPROACH

There are many potential sources of uncertainty in both the loading and resistance aspects of liquefaction problems, and probabilistic approaches have been developed to deal with them. Uncertainties in cyclic loading can be evaluated using the standard probabilistic seismic hazard analyses (Cornell, 1968; Algermissen et al., 1982; Reiter, 1990). Uncertainties in liquefaction resistance can be treated in one of two general ways.

One group of methods is based on probabilistic characterization of the parameters shown by laboratory tests to influence pore pressure generation. Haldar and Tang (1979) characterized uncertainty in the parameters of the simplified cyclic stress approach or deterministic approach described in Section 2.8.1. Fardis and Veneziano (1982) used a similar approach with total stress and effective stress models. Chameau and Clough (1983) described pore pressure generation probabilistically using experimental data and an effective stress model. Each of these methods can compute the probability of liquefaction due to a particular set of loading

conditions. Their accuracy depends on the accuracy of the underlying liquefaction / pore pressure model and on how accurately the uncertainty of the model parameters can be determined.

An alternative method is based on in situ test-based characterization of liquefaction resistance (Christian and Swiger, 1975; Yegian and Whitman, 1978; Veneziano and Liao, 1984). This method, which will be presented in Chapter 5, use various statistical classification and regression analyses to assign probabilities of liquefaction to different combinations of loading and resistance parameters.



สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 3

PROFILE OF THE STUDY AREA

3.1 LOCATION AND TOPOGRAPHY

Chiang Mai, Thailand's second largest city with an area of 22,800 square kilometers and capital of the northern region, located about 700 kilometers north of Bangkok at latitude 18.80°N and longitude 98.98°E. A large part (69.31%) of Chiang Mai's land is covered by mountains and forests. These generally run in a north-south pattern through the province and give birth to several streams and tributaries (such as the Mae Jam, Mae Ngud, and Mae Klang) which in turn feed important rivers and irrigation canals (such as the Muang and Faay) which provide the water necessary to Chiang Mai's agriculture. Chiang Mai's largest and most important river is the Ping, which originates in the mountains of Chiang Dao and flows southward for 540 kilometers. It is along the banks of this river that Chiang Mai's flat, fertile valley area lies.

Chiang Rai, the northernmost province of Thailand with an area of 11,678 square kilometers, is approximately 785 kilometers from Bangkok at latitude 19.91° N and longitude 99.83° E. The province is situated on the Kok River basin. The average elevation is about 580 meters above sea level. Mostly mountainous, it reaches the Mae Khong River to the north and borders on both Myanmar and Laos. North Chiang Rai falls within the region known as the Golden Triangle, the area where the borders of Thailand, Myanmar and Laos converge.

The areas of investigation are shown in Figure 3.1.

3.2 SEISMICITY

Thailand lies in the interior of the Eurasian plate, with the boundary of the Burma microplate and Indian plate occurring to the west in the Andaman sea. Nutalaya et al. (1985) collected the database containing instrumental data of earthquakes from 1910 to 1989 within the region bounded by latitudes 5° N to 25° N and longitudes 90° E to 110° E, which includes Thailand, Indochina, and parts of Burma and China. These data were collected from several agencies which include the U.S. Geological Survey (USGS), the International Seismological Center (ISC) in U.K., and the Thai Meteorological Department (TMD). In addition, 12 seismic source zones within the region (zones A to L in Figure 3.2) have been identified on the basis of the spatial distribution of seismicity and regional seismotectonic structure. The maximum observed local magnitude (Richter scale, M_L) earthquake to occur within the eleven closest zones is shown in Table 3.1. As seen from Table 3.1, the maximum estimate M_L occurring within these zones ranges from 6.5 for the Northern Thailand Zone to 7.9 for the Tenasserim Range Zone.

Wanitchai and Lisantono (1996) proposed a seismic hazard map of Thailand as shown in Figure 3.3. As seen from Figure 3.3, Thailand is divided into various seismic zones according to the following criteria: zone 0 for PGA₀<0.025g, where PGA₀ is the peak ground acceleration (PGA) with a 10% probability of being exceeded in a 50-year period at any given site and g is the acceleration of gravity; zone 1 for 0.025g< PGA₀<0.075g; zone 2A for 0.075g< PGA₀<0.15g; zone 2B for $0.15g < PGA_0 < 0.20g$; zone 3 for $0.20g < PGA_0 < 0.30g$; and zone 4 for PGA₀>0.30g. These criteria are similar to those used in the 1988, 1991, and 1994 U.S. Uniform Building Code (UBC) zoning maps except that in the UBC, effective peak acceleration (EPA) is used instead of PGA. By its definition, EPA is the peak ground acceleration after the ground motion record has been filtered to remove the very high frequencies that have little influence upon structural response. However, it appeared that for PGA<0.3g there is no significant different between PGA and EGA. The zoning map in Figure 3.3 shows that some parts of the northern and western Thailand can be considered as moderate risk and moderately high risk zones equivalent to the UBC zone 2B and 3, respectively.

3.3 SOIL PROFILE

The soil underlying Chiang Mai and Chiang Rai consist of alternating layers of silty sand, clayey sand, sandy silt, and silty clay. Anantasech and Thanadpipat (1985) collected geotechnical investigation data of Chiang Mai City and proposed soil profiles of the city from north to south, and from east to west as shown in Figure 3.4. For Chiang Rai City, the general soil profile is shown in Figure 3.5. Examples of boring information in Chiang Mai and Chiang Rai are depicted in Figure 3.6 and 3.7, respectively. The sub soils in both provinces are subject to wide variation. Layers of loose to medium dense sand are found in most of the investigated boreholes. Those layers are distributed throughout the cities. Figure 3.8 summarizes the gradation of sands found in both provinces. Great variation of grain size distribution can be observed. The average diameter, D_{50} , of sands varies in the range of 0.2 to 1.5 mm.

3.4 INPUT MOTIONS FOR ANALYSIS

Due to the lack of strong motion records in Thailand, it is necessary to use the records from elsewhere. Three strong motions with different predominant periods used as input motions to the analyses are shown in Figure 3.9. These acceleration time histories are actual strong motion records recorded at rock sites; and selected from California earthquakes including the 1940 El Centro, 1989 Loma Prieta, and 1994 Northridge earthquakes. The recording station and estimated predominant period of the input motions are summarized in Table 3.2. The response spectra for 5 percent damping of the input motions are shown in Figure 3.10.

3.5 ESTIMATION OF SHEAR WAVE VELOCITY

Due to the lack of data of the shear wave velocity measured in the field, the empirical correlations based on available field and laboratory measurements were used in this study. There are many researchers proposed the empirical correlations of the shear wave velocity (or maximum shear modulus) based on SPT N-value and the undrained shear strength (S_u) from laboratory tests such as: Kanai (1966); Sakai (1968); Seed and Idriss (1970); Ohsaki and Iwasaki (1973); Ohta and Goto (1978); Imai and Tonouchi (1982); Seed et al. (1983); Sykora and Stokoe (1983); and Dickenson (1994). Even any empirical correlations inherently are site dependent. They, however, give very useful and reasonable guidelines for field engineers to evaluate the shear wave velocity of the in situ soil.

3.5.1 FORMULA FOR CLAY

The formula proposed by Dickenson (1994) was used to compute shear wave velocity of soft clay deposit in Chiang Mai and Chiang Rai. Dickenson (1994) presented the results of a study of the dynamic response of soft and deep cohesive soils during the 1989 Loma Prieta earthquake and developed a nonlinear relationship between shear wave velocity determined from field testing and undrained shear strength for cohesive soil. The proposed relationship is:

$$V_s = 68.7 \cdot S_u^{0.475}$$
 (m/sec) (3.1)

where S_u , the undrained shear strength, has unit of t/m². It should be noted that S_u determined from unconfined compression, unconsolidated undrained triaxial (TX), and consolidated undrained triaxial tests were used directly in Dickenson's equation.

For the estimation of the shear wave velocity for medium to stiff clay layer, the relationship, developed by Imai and Tonouchi (1982), for any soil type based on uncorrected N-value was used. The relationship is:

$$V_{\rm s} = 96.926 \cdot N^{0.314}$$
 (m/sec) (3.2)

3.5.2 FORMULA FOR SAND

There are two relationships used to estimate shear wave velocity for sand layer in the study area. For silty sand layer, the relationship proposed by Seed et al. (1983) was used. Seed et al. (1983) suggested using the following equation for sand and silty sand to assess shear wave velocity by using SPT N-value:

$$V_s = 56.388 \cdot N^{0.5} \quad \text{(m/sec)} \tag{3.3}$$

For sandy soils, the correlation between shear wave velocity and SPT N-value developed by Sykora and Stokoe (1983) was used, and the relationship is:

$$V_s = 100.584 \cdot N^{0.29}$$
 (m/sec) (3.4)

The SPT N-value used in Equation (3.3) and (3.4) is the corrected N-value, $(N_1)_{60}$, which is normalized to an overburden pressure of approximately 100 kPa (1 t/ft²) and a hammer energy ratio or hammer efficiency of 60%.

3.6 SUMMARY

In this chapter, location, topography, seismicity, and soil profile of Chiang Mai and Chiang Rai, were briefly presented. Subsequently, the strong motions with different predominant periods recorded at rock sites were described. Shear wave
velocity or shear modulus of subsoil in both provinces can be estimated using the empirical correlations based on available field and laboratory measurements (SPT N-value and undrained shear strength). The strong motions and the shear wave velocity described in this chapter are used as input to the analyses of soil amplification and generation of pore water pressure which is described in Chapter 4 and 6, respectively.



สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 4

AMPLIFICATION OF EARTHQUAKE GROUND MOTIONS

4.1 INTRODUCTION

Since the provinces in the northern and western parts of Thailand have been classified as a moderate seismic risk regions by the ministerial regulations on seismic resistant design, and there are several moderate earthquakes (5.2-5.5 on the Richter scale) have occurred in the northern part of the country during the past decades. In consequence, the Thai people are becoming increasingly concerned about seismic risk. Therefore, in order to describe some effects of earthquake ground motions in the mentioned areas, particularly Chiang Mai and Chiang Rai; one-dimensional ground response analysis using the equivalent linear approach is conducted.

4.2 COMPUTER PROGRAM SHAKE

The computer program SHAKE was written in 1970-71 by Dr. Per Schnabel and Prof. John Lysmer and was published in December 1972 by Dr. Per Schnabel and Prof. John Lysmer and H. Bolton Seed in report no. UCB/EERC 72/12, issued by the Earthquake Engineering Research Center at the University of California in Berkeley.

The soil profile is idealized as a system of homogeneous, visco-elastic sublayers of infinite horizontal extent; the idealized soil profile is shown in Figure 4.1. The response of this system is calculated considering vertically propagating shear waves. The algorithm on the program SHAKE (Schnabel et al., 1972) is based on the continuous solution to the wave equation (Kanai, 1951; Matthiesen et al., 1964; Roesset and Whitman, 1969; Lysmer et al., 1971), which was adapted for transient motions using the Fast Fourier Transform techniques of Cooley and Tukey (1965).

As described in the Section 2.5 of Chapter 2, an equivalent linear procedure (Idriss and Seed, 1968; Seed and Idriss, 1970) is used to account for the nonlinearity of the soil using an iterative procedure to obtain values for modulus and damping that are compatible with the equivalent uniform strain induced in each sublayer. Thus, at the outset, a set of properties (shear modulus, damping, and total unit weight) is assigned to each sublayer of the soil deposit. The analysis is conducted using these properties and the shear strains induced in each sublayer is calculated. The shear modulus and the damping ratio for each sublayer are then modified based on the applicable relationship relating these two properties to shear strain. The analysis is repeated until strain-compatible modulus and damping values are arrived at. Starting with the maximum shear modulus for each sublayer and a low value of damping, essentially strain-compatible properties are obtained in 3 to 5 iterations for most soil profiles.

The following assumptions are incorporated in the analysis (Schnabel et al., 1972):

- Each sublayer, j, is completely defined by its shear modulus, G_j; damping ratio, ξ_j; total unit weight, γ_{ij}; and thickness, h_j, these properties are independent of frequency. It is noted that the initially estimated values of G and ξ usually correspond to the same strain level; the low-strain values (0.001%) are often used for the initial estimate. In this study, the initial values of G were determined from the shear wave velocity (V_s) and the mass density (ρ) from the relationship G = ρ×V_s². The shear wave velocity can be calculated from the empirical correlations based on available field and laboratory measurements as previously described in Section 3.5 of Chapter 3.
- The responses in the soil profile are caused by the upward propagation of shear waves from the underlying rock half-space.
- The strain dependence of the shear modulus and damping in each sublayer is accounted for by an equivalent linear procedure based on an equivalent uniform strain computed in that sublayer. According to Idriss and Sun (1992), the ratio of

this equivalent uniform shear strain divided by the calculated maximum strain which is typically used from 0.4 to 0.75 based on magnitude earthquake (M) is evaluated by:

$$\text{Ratio} = \frac{M-1}{10} \tag{4.1}$$

The same value of this ratio is used for all sublayers.

4.3 MODULUS REDUCTION AND DAMPING RATIO

It has long been known that soil exhibits nonlinear behavior. This is particularly true when subjected to dynamic loading. Pioneering studies by Hardin and Drnevich (1972b) and Seed and Idriss (1970) using laboratory tests on soil such as the resonant column, cyclic simple shear, and cyclic triaxial, as well as shaking table tests and back calculated results, indicate that shear strain is the single most important parameter when looking at the variation of shear modulus and fraction of critical damping for the seismic response of different soil types. Seed and Idriss (1970) proposed that the variation of shear modulus with shear strain could be normalized by plotting the ratio of G/G_{max} (i.e., the ratio of shear modulus at a given strain, G, over the low strain shear modulus, G_{max}) versus shear strain, for different soil types (e.g., sand, clay, and gravel). The advantage to these normalized relationships for seismic site response analyses is that in many cases, only the G_{max} would need to be determined in the field from V_s testing; the variation of G with shear strain could be estimated based on soil type from the normalized relationships.

However, since laboratory testing to determine the variation of G with shear strain, γ , for soils in Chiang Mai and Chiang Rai was beyond the scope of this study, it was decided to use the latest normalized relationships available. For clays, Vucetic and Dobry (1991) combined the results of many previous studies to develop normalized relationships of G and γ as a function of plasticity index. Their

relationships for both variation of G and damping ratio (λ) as a function of γ are shown in Figure 4.2. For sands, the relationships proposed by Seed et al. (1984a) were used. Their relationships for G/G_{max} and damping as a function of γ are presented in Figure 4.3. Both the relationships by Vucetic and Dobry (1991) and by Seed et al. (1984) are in wide use for estimating shear modulus and damping variation with shear strain of soil for purpose of site response analysis.

4.4 RESULTS OF ANALYSES

One-dimensional dynamic ground response analyses were performed on eighteen sites within Chiang Mai City and fourteen sites within Chiang Rai City using the computer program SHAKE (Schnabel et al., 1972) for equivalent linear response. The examples of soil profiles of Chiang Mai and Chiang Rai are presented in Figure 3.6 and 3.7. Soil properties are defined by borings made for foundation design at the site and geologic references for the area. Soil type is indicated by Unified Soil Classification System group symbols.

Since one of unknown in the analyses is the depth to bedrock and no data was available to make a reasonable estimate of its shear wave velocity. However, for the purposes of seismic ground response analyses, the depth to bedrock itself is not as important as the depth to rock-like material that behaved essentially as bedrock (Lysmer et al., 1970), that is, material having a shear wave velocity on the order of 800 to 1200 m/sec. Lysmer et al. (1970) also found that the response at the ground surface was relatively independent of the shear wave velocity of the assumed rocklike material. Some effect was observed, however, on the assumed depth to rock-like material, and whether or not an intermediate rock-like layer was modeled between the bottom of the soil profile and the surface of the bedrock. This effect was mainly on the frequency content of the motion at the ground surface; the peak ground acceleration was relatively unaffected. Thus the depth to rock-like material in the analyses was assumed at the bottom of the soil profile which has the SPT N-value of more than 50 blows/ft with a shear wave velocity of 900 m/sec.

The seismic ground response analyses were conducted using the three seismograms described in Chapter 3 with bedrock accelerations scaled to 0.02, 0.05, 0.07, 0.10, and 0.30g. The results for the amplification of Chiang Mai and Chiang Rai are presented in Figure 4.4 and 4.5, respectively. It can be seen from the results that the amplification, which is defined as the ratio between the acceleration at ground surface and the acceleration at rock surface, decreases with increasing bedrock acceleration. It should be noted that for long duration and high strain ground motions, the amplification level would decrease because of the nonlinear behavior of soil. In addition, it is also apparent from Figure 4.4 and 4.5 that the amount of amplification is dependent on the predominant period of the input motion.

The variation of peak ground acceleration with the applied peak rock acceleration is shown in Figure 4.6 and 4.7 for Chiang Mai and Chiang Rai, respectively. They show that the peak ground acceleration increases with increasing amplitude of the base motion. Furthermore, the peak ground accelerations presented in Figure 4.6 and 4.7 are also plotted against peak rock acceleration along with observed values from the 1985 Mexico earthquake and the 1989 Loma Prieta earthquake; the results show that the range of calculated accelerations tends to be lower than observed in previous earthquakes, particularly for the Mexico earthquake which most of subsoil consist of soft soil deposits.

4.5 SUMMARY

In this chapter, the results of a study on the amplification of earthquake ground motion in the city of Chiang Mai and Chiang Rai, were illustrated. Three input

ground motion in the city of Chiang Mai and Chiang Rai, were illustrated. Three input motions with different predominant periods were used. Soil properties used as input to the analysis, specifically shear wave velocity, were estimated using existing correlations with field and laboratory data. Seismic ground response analyses were, then, conducted using the computer program SHAKE based on the equivalent linear method. Concerning amplification of the peak ground acceleration, it is apparent from the results in the study that the soils underlying the city of Chiang Mai and Chiang Rai have ability to amplify earthquake ground motions on the order of 1.5 to 3 times. The amount of this amplification depends on the actual soil properties and profiles or local site conditions, the level of acceleration in the underlying rock-like material, and the predominant periods of the input rock motion. In addition, it is also found that the amount of this amplification decreases with increasing bedrock acceleration because of nonlinearity of soils.



สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 5

EVALUATION OF LIQUEFACTION RESISTANCE BY PROBABILISTIC APPROACH

5.1 INTRODUCTION

Liquefaction occurs primarily in loose saturated sands. As seismic waves propagate through the soil, the structure of the soil is altered with a corresponding increase in the pore water pressure. As the pore water pressure increases, the effective stress decreases. If the effective stress is reduced to a point in which it is equal to zero, the soil loses its strength and liquefaction develops. Liquefaction is a function of soil type, relative density, age, fines content, and intensity and duration of the earthquake motion (Seed and Idriss 1971).

Several methods have been proposed to evaluate earthquake liquefaction potential. These methods range from purely empirical to highly analytical and require various degrees of laboratory and/or in situ testing. A common approach of the deterministic type is to use charts such as that shown in Figure 5.1. The horizontal axis measures the strength of the soil in terms of the corrected standard penetration resistance (N_1) , and the vertical axis measures the intensity of ground motion through the cyclic stress ratio (*CSR*). Points are drawn to represent cases of liquefaction (solid dots) and non-liquefaction (open circles), and a line is subjectively drawn to separate events of the two types. The line is then used as a deterministic classification criterion.

It is evident that there is no sharp demarcation between the two sets of observations in Figure 5.1, and that the probability of liquefaction actually varies in a continuous fashion as a function of soil strength and the intensity of ground motion. This conditional probability can be evaluated by using appropriate statistical procedures to analyze field data of the type in Figure 5.1.

The objective of this chapter is to use methods of statistical regression to quantify the probability of liquefaction by using the data catalog compiled by Liao and Whitman (1986a). In the context of the data catalog, the term "liquefaction" is defined as the surface manifestation of any phenomenon associated with a significant loss of shearing resistance in saturated cohesionless soils due to earthquake loading. No attempt is made to distinguish between liquefaction or cyclic mobility may occur at depth without exhibiting effects at the ground surface (Ishihara, 1985), such instances are classified here as cases of nonliquefaction. Note also that the results of this study are primarily applicable to sites with level ground conditions and are generally not valid for the evaluation of liquefaction of embankments or for sloping grounds.

5.2 LIQUEFACTION DATA CATALOG

The data catalog used in the statistical analysis comprise 278 case studies, with 114 cases representing sites that liquefied during earthquakes and 164 cases representing sites that did not liquefy. The catalog was compiled by Liao and Whitman (1986a) through the synthesis of eight previously published "source catalogs". The source catalogs are those of Whitman (1971), Seed et al. (1975), Yegian (1976), Yegian and Vitelli (1981), Xie (1979), Davis and Berrill (1981), Tokimatsu and Yoshimi (1983), and Seed et al. (1984b).

The catalog is based on 40 earthquakes, of which the earliest is the historical 1802 Niigata earthquake and the most recent is the 1981 Westmorland, California earthquake. Of the 278 cases, 120 are from Japan, 100 from California, 20 from China, and 38 from other locations in the world. The example of the data catalog is shown in Figure 5.2.

The variables, notations, and abbreviations of the various column headings in the data catalog (Figure 5.2) are described as follows:

- Entries of (-1) under any of the table headings indicate missing data.
- CASE (Case Code): Consists of two parts. The first two digits identify the earthquake. The last two digits identify the case associated with that earthquake.
- ECAT (Source Catalog Code): Indicates data source catalog.
 - 0 = data not reported in previous catalogs
 - 1 =Whitman (1971)
 - 2 = Seed, Idriss, and Arango (1975)
 - 3 = Yegian (1976)
 - 4 = Yegian and Vitelli (1981)
 - 5 = Xie (1979)
 - 6 = Davis and Berrill (1981)
 - 7 = Tokimatsu and Yoshimi (1983)
 - 8 = Seed, Tokimatsu, Harder, and Chung (1984)

9 = Liao(1986)

- LIQ1 (Liquefaction Code 1)
 - 0. = no lique faction.
 - 1. = liquefaction.
- LIQ2 (Liquefaction Code 2): Based on Tokimatsu and Yoshimi (1983).
 - .0 = no liquefaction.
 - .5 = marginal site, defined as a site located just outside the boundary of liquefaction zone.
 - .7 = moderate liquefaction, indicated by appearance of sand boils, but with minor ground or foundation movements.

1.0 = extensive liquefaction, indicated by sand boils and/or major ground or foundation movements.

- M (Richter Magnitude)
- H (Focal Depth): Units of kilometers.
- EP (Epicentral Distance to Site): Units of kilometers.

- DER (Distance to Energy Release): Unit in kilometers. Closest distance to fault rupture or to the zone of major energy release. A distance measure that is largely judgmental in many cases. If a measure of DER is not available, DER is set to EP.
- DUR (Duration of Shaking): Units in seconds.
- A (Peak Ground Acceleration at Site): Units of fraction of gravitational acceleration, g.
- CSR (Cyclic Stress Ratio): This quantity is calculated using the Seed and Idriss (1971) procedure, as described in Section 2.8.1.2 of Chapter 2.
- CSRN (Cyclic Stress Ratio for Magnitude 7.5 Earthquakes)
- ZW (Depth to Water Table): Units of meters.
- ZL (Critical Depth of Liquefaction): Units of meters.
- SIGT (Total Vertical Stress): Units of kg/cm².
- SIGE (Effective Vertical Stress): Units of kg/cm².
- N (SPT N-Value)
- N1 (SPT N-Value Corrected for Overburden Pressure): All of the values of N1 are calculated based on the Liao and Whitman (1986) correction factor.
- CE (Correction Factor for Sampling Equipment and Practices): See Equation (2.38) in Section 2.8.1.3 of Chapter 2 for details.
- N160 (Corrected SPT N-Value Normalized to the Overburden Pressure; and Sampling Equipment and Practices): See Equation (2.38) in Section 2.8.1.3 of Chapter 2 for details.
- FC (Fines Content): Units of percent. Fines defined as material passing No. 200 sieve.

For calculated CSR values in the catalog, it is noted that the primary variable affecting the value of CSR is the peak ground acceleration (a_{max}) , which can be obtained in several ways. In 127 out of 278 catalog entries, the peak acceleration is obtained from measurement at a nearby station that can mean a strong motion recorder located several kilometers away. In a few cases, a strong motion recorder is

actually close enough to be considered on site, which is indicated that the acceleration measurement is considered to be very accurate in representing the ground motion at the site. Other methods of estimating acceleration include performing a site response analysis with the input from a ground motion record some distance away, scaled to reflect inferred bedrock motion at the site of interest. In many cases, accelerations are calculated from earthquake attenuation relationships and/or correlations to an intensity damage scale (e.g., Modified Mercalli scale).

However, in many of the historical cases of liquefaction/non-liquefaction from California and Japan where the acceleration was not reported, or where the reported acceleration was suspected, accelerations were estimated from one of two attenuation relationships. For cases in California, the Joyner and Boore (1981) equation was used:

$$\log_{10}(a / g) = -1.02 - 0.249M - \log r - 0.00255r$$
(5.1a)

or

$$\frac{a}{g} = \frac{0.0955 \cdot 10^{0.249M}}{r \cdot 10^{0.00225 r}}$$
(5.1b)

where *M* is the moment magnitude, and $r = (d^2 + 7.3^2)^{\frac{1}{2}}$, in which *d* is defined as the closest distance (in kilometers) to surface projection of the fault rupture. For Japanese earthquakes, the relationship used is that due to Kawashima et al. (1984) for soft alluvium or reclaimed ground:

$$\frac{a}{g} = \frac{0.4109 \cdot 10^{0.262M}}{(R_{EP} + 30)^{1.208}}$$
(5.2)

where M is the Japanese Meteorological Association (JMA) magnitude and R_{EP} is the epicentral distance in kilometers.

5.3 REGRESSION METHOD FOR LIQUEFACTION DATA ANALYSIS

Each case study in the database can be represented through a binary variable Y which indicates whether liquefaction occurred (Y = 1) or did not occur (Y = 0) and a vector of explanatory variables $X = [X_1, X_2, ..., X_m]^T$, which includes ground motion and soil deposit characteristics. The problem is to use the available n observations $(X_1, Y_1), ..., (X_n, Y_n)$ to express the probability of liquefaction P_L as a function of X. Therefore, the regression technique called "the binary logistic regression" will be used for analyzing the database and can be written in the form:

$$\ln\left(\frac{P_{L}}{1-P_{L}}\right) = \beta_{0} + \beta_{1}X_{1} + \dots + \beta_{m}X_{m}$$
(5.3)

where P_L is defined as the probability that liquefaction will occur, $1 - P_L$ is the probability that liquefaction will not occur, X_k 's (k = 1, 2, ...,m) are various "explanatory variables" such as cyclic stress ratio and corrected SPT resistance, and β_k 's (k = 1, 2, ...,m) are regression coefficients to be obtained by fitting Equation (5.1) to data. Solving Equation (5.3) for P_L gives the following expression for the probability of liquefaction in terms of the variables $X_1,...,X_m$:

$$P_{L} = \frac{\exp(\beta_{0} + \beta_{1}X_{1} + \dots + \beta_{m}X_{m})}{1 + \exp(\beta_{0} + \beta_{1}X_{1} + \dots + \beta_{m}X_{m})}$$
(5.4a)

or equivalently:

$$P_{L} = \frac{1}{1 + \exp\left[-\left(\beta_{0} + \beta_{1}X_{1} + \dots + \beta_{m}X_{m}\right)\right]}$$
(5.4b)

Flexibility of the model is greatly enhanced if one allows $X_1,...,X_m$ in Equation (5.3) and (5.4) to be functions of the original explanatory variables rather than the variables themselves. For example, X_1 might be the logarithm of the cyclic stress ratio (*CSR*) instead of *CSR* itself. This option will be used in the analysis of liquefaction data.

Since procedures and details of the regression method to determine the regression coefficients (β_k 's) are beyond the scope of this study. However, they can be seen from other standard statistical textbooks such as: Dillon and Goldstein (1984); Johnson and Wichern (1988); Morrison (1990); Kleinbaum (1994); and Johnson (1998). The regression coefficients (β_k 's) in the study are obtained by using a well-known statistical computer program SPSS (SPSS, 1999).

5.4 LOGISTIC MODELS OF LIQUEFACTION BEHAVIOR

Models of the type in Equation (5.4b) may be used to quantify the probability of liquefaction once the variables $X_1, ..., X_m$ have been selected. The term "variable selection" refers here to the choice of both the physical factors that influence liquefaction and the functions of such factors that should enter the regression model as variables $X_1, ..., X_m$. For example, it may be important to decide not only whether a variable such as the cyclic stress ratio (*CSR*) should be among the explanatory variables, but also whether it should appear in Equation (5.4b) as $X_1 = CSR$, $X_1 = \ln(CSR)$, or in some other form. However, the models are limited to the use of variables recorded in the data catalog. Thus, potentially important parameters that are not reported or are difficult to quantify cannot at present be considered in variable selection. Performing a logistic regression analysis of the database yields the following probability equations:

For earthquake magnitude of 7.5:

$$\ln\left(\frac{P_L}{1-P_L}\right) = 9.119 - 0.243(N_1)_{60} + 3.458\ln(CSR_{7.5})$$
(5.5)

For earthquake magnitude of 5.5:

$$\ln\left(\frac{P_L}{1-P_L}\right) = 6.354 - 0.242 (N_1)_{60} + 3.450 \ln(CSR_{5.5})$$
(5.6)

where $(N_1)_{60}$ = the corrected SPT resistance which is normalized to an overburden pressure of approximately 100 kPa (1 t/ft²), a hammer energy ratio or a hammer efficiency of 60%, the borehole diameter, and the rod lengths; $CSR_{7.5}$ = the cyclic stress ratio generated at the site normalized to a magnitude of 7.5; and $CSR_{5.5}$ = the cyclic stress ratio generated at the site normalized to a magnitude of 5.5. The magnitude scaling factors published by Seed and Idriss (1982), which have been conventionally used in liquefaction hazard analyses, were used in this study.

Figure 5.3 and 5.4 show a set of probability curves defined by Equation (5.5) and (5.6), respectively. Also shown in Figure 5.3 and 5.4 is the deterministic criteria defined by Seed et al. (1985). The correlation of regression (equivalent to R^2) for Equation (5.5) and (5.6) is 0.637. The success rate in classification of liquefaction from both equations is greater than 80% for both liquefied and non-liquefied cases. Note that the probabilistic line at $P_L = 30\%$ well traces the deterministic criteria proposed by Seed et al. (1985) as shown in the figures. This result generally agrees with the findings of Youd and Noble (1997), and Toprak et al. (1999). The success rate is therefore determined based on $P_L = 30\%$ line.

5.5 SUMMARY

A statistical regression procedure has been used to derive models for calculating the probability of liquefaction as a function of earthquake load and soil resistance parameters. The data to which the models are fitted consist of 278 cases of liquefaction and non-liquefaction, obtained from worldwide liquefaction database.

Based on goodness-of-fit statistics and on considerations of usefulness in application, two models [Equation (5.5) and (5.6)] are recommended for the calculation of liquefaction probability. The models are based on the Seed and Idriss (1971) parameterization and employ the magnitude cyclic stress ratio *CSR* as a measure of earthquake load; and use the corrected/normalized SPT value $(N_1)_{60}$ as a measure of soil-liquefaction resistance.

As mentioned earlier, Chiang Mai and Chiang Rai are located in the seismic zone G (see Figure 3.2) which probable causes earthquake magnitude (M_L) of 5 to 6. Thus, Equation (5.6), which was developed for earthquake magnitude of 5.5, shall be used to study the generation of pore water pressure of sand layer in the areas. This subject is described in Chapter 6.

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 6

FINITE ELEMENT ANALYSIS OF PORE WATER PRESSURE IN SAND DEPOSITES DURING EARTHQUAKES

6.1 INTRODUCTION

The response of a saturated sand deposit to earthquake motions is very important and difficult problem of soil dynamics; and a completely satisfactory generalized solution is not yet available. The dynamic response, at least for loose to medium dense sands, is dominated by the effects of the progressive pore water pressure increases that develop during an earthquake. The resistance to deformation at any point in the sand deposit is a function of effective stress which in turn depends on the simultaneous rates of generation and dissipation of pore water pressure.

The purpose of this chapter is to illustrate the application of a numerical finite element scheme for development of guidelines and charts for analysis and evaluation of the generation of pore water pressure due to medium earthquakes (M = 5.5) corresponding to various probability values. The analytical procedure employed to quantify the pore water pressure is outlined in Figure 6.1. For each site, the minimum SPT N-value of sand at certain depth (from the ground surface to about 15 m) was selected in order to determine the value of cyclic stress ratio (*CSR*) from the probability curve (Figure 5.4). The maximum ground surface acceleration (a_{max}) for the specific probability values (P_L) of that site was then computed from the simplified equation [Equation (2.37)] proposed by Seed et al. (1983) as mentioned in Section 2.8.1.2 of Chapter 2. The pore water pressure was evaluated by performing trial and error of the scaled base acceleration. This was continued until the computed maximum ground acceleration was similar to the prescribed value given by the simplified equation. Note that the probability values (P_L) used in this study are 0.05, 0.10, 0.30, and 0.50. The computer program used for evaluating the pore water pressure is called "FLIP".

The parameters used in the analysis are obtained by back-fitting the calculated results with experimental data from undrained cyclic triaxial tests. The finite element method presented here is based on the effective stress analysis of a horizontally layered saturated sand deposit shaken by horizontal shear waves propagating vertically. The computer program and the model parameters of soils used in the analyses have been described in the following sections.

6.2 CONSTITUTIVE MODEL OF SOILS

6.2.1 BASIC CONSTITUTIVE EQUATIONS

As mentioned earlier, liquefaction of cohesionless soils has contributed to the failure of soil structure systems during earthquakes. The mechanisms of liquefaction have been studied extensively, and many soil liquefaction models have been proposed.

Available soil liquefaction models are based on either: (1) experimentally observed undrained stress paths during pore water pressure build up (Ishihara et al., 1975; Ishihara et al., 1976); (2) a correlation between pore water pressure response and volume change tendency of dry soils (Martin et al., 1975; Finn et al., 1977); (3) the formulation of pore water pressure response directly from observed data (Shibata et al., 1972; Seed et al. 1976a; Ishibashi et al., 1977; Sherif et al. 1978); (4) a plasticity theory in which the plastic volume change is related to pore water pressure build up (Mroz et al., 1978; Zienkiewicz et al., 1978); or (5) treatment of the soil as a two-phase medium (Liou et al., 1977; Blazquez et al., 1980). The soil model used in this study is an effective stress model proposed by Iai et al. (1992); and can be briefly described as follows:

The model is constructed within the framework of plasticity theory but it is quite different from the conventional type of plasticity models. First of all, the model is defined in strain space. Secondly, the concept of multiple shear mechanism is introduced for taking the effect of principal stress axes rotation into account. Thirdly, the dilatancy is treated as an additional volumetric strain component in such a manner as strains due to creep and temperature are treated in the constitutive equation. The behavior of soil under the plane strain condition is basically represented as a relation between effective stress and strain defined in terms of such vectors as

$$\left\{\sigma'\right\}^{T} = \left\{\sigma'_{x} \quad \sigma'_{y} \quad \tau_{xy}\right\}$$
(6.1)

$$\left\{ \boldsymbol{\varepsilon} \right\}^{T} = \left\{ \boldsymbol{\varepsilon}_{x} \quad \boldsymbol{\varepsilon}_{y} \quad \boldsymbol{\varepsilon}_{xy} \right\}$$
(6.2)

in which compressive stress and contractive strain will be assumed negative and the strain components will be given from displacements u and v in x and y directions by

$$\varepsilon_x = \frac{\partial u}{\partial x}, \quad \varepsilon_y = \frac{\partial v}{\partial y}, \quad \gamma_{xy} = \frac{\partial u}{\partial y} + \frac{\partial v}{\partial x}$$
 (6.3)

then the basic form of the constitutive relation is given by

$$\left\{ d\sigma' \right\} = \left[D \right] \left\{ d\varepsilon \right\} - \left\{ d\varepsilon_p \right\} \right) \tag{6.4}$$

in which

$$[D] = K \{ n^{(0)} \} \{ n^{(0)} \}^{T} + \sum_{i=1}^{I} R_{L/U}^{(i)} \{ n^{(i)} \} \{ n^{(i)} \}^{T}$$
(6.5)

In this relation, the term $\{d\varepsilon_p\}$ in Equation (6.4) represents the additional strain increment vector to take the dilatancy into account and is given from the volumetric strain increment due to the dilatancy ε_p as

$$\left\{ d\varepsilon_p \right\}^T = \left\{ d\varepsilon_p / 2 \quad d\varepsilon_p / 2 \quad 0 \right\}$$
(6.6)

and the first term in Equation (6.5) represents the volumetric mechanism with rebound modulus K and the direction vector is given by

$$\{n^{(0)}\}^{T} = \{1 \ 1 \ 0\}$$
(6.7)

The second term in Equation (6.5) represents the multiple shear mechanism. Each mechanism i = 1,..., I represents a virtual simple shear mechanism, with each simple shear plane oriented at an angle $\theta_i / 2 + \pi / 4$ relative to the x axis. A schematic figure is shown in Figure 6.2. The tangential shear modulus $R_{L/U}^{(i)}$ represents the hyperbolic stress strain relationship with hyteresis characteristics. The direction vectors for the multiple shear mechanism in Equation (6.5) are given by

$$\left\{n^{(i)}\right\}^{T} = \left\{\cos \theta_{i} - \cos \theta_{i} \sin \theta_{i}\right\} \quad (\text{for } i = 1, ..., I) \quad (6.8)$$

in which

$$\theta_i = (i - 1)\Delta\theta \qquad (\text{for } i = 1, \dots, I) \tag{6.9}$$

$$\Delta \theta = \pi / I \tag{6.10}$$

The loading and unloading for shear mechanism are separately defined for each virtual simple shear mechanism by the sign of $\{n^{(i)}\}^T \{d\varepsilon\}$.

6.2.2 PHYSICAL BACKGROUND OF THE MODEL

Physical meaning of the strain space multiple mechanism model can be found in the mechanics of assembly of sand particles (Iai et al., 1992). In the mechanics of assembly of sand particles, stress in sand as defined for continuum is given as a certain average of contact forces between the sand particles, which are assumed to be spheres. Before taking the average over all the contact forces in a representative volume element, the contact forces can be classified according to the directions. To define the directions, let us consider a plane of an arbitrary direction and find a class of pairs of contact forces and contact normals, both of which are parallel to the plane. (If the contact force is parallel to the contact normal, let us equally divide the contact force among all the relevant plane.) The average over this class of contact forces associated with the plane constitutes a partial stress contribution from "the virtual plane strain mechanism".

The contact forces within each plane can be further classified according to the directions within the plane. To define this level of directions, let us think of a class of contact normals of which direction is at an angle $\theta_i / 2$ relative to the reference axis appropriately defined within the plane. The contact forces can be further partitioned into its normal and tangential components as shown in Figure 6.3. Let us individually take an average of each component over the class of contacts to form a basic level of partial stress contribution.

To identify the nature of the basic level of partial stress contribution, let us consider a pair of those stress contributions associated with the contact normals being at right angle. Then it becomes evident that the pair of the normal components represents volumetric and compression shear stress contributions and the other pair of the tangential components represents a simple shear stress contribution. With the rotation of the reference axis with an angle $\pi/4$, the compression shear stress contribution associated with the contact normals at an angle $\theta_i/2$ becomes equivalent to the simple shear stress contribution associated with the contact normals at an angle $\theta_i/2 + \pi/4$. By summing up the simple shear stress contributions from the normal components at an angle $\theta_i/2$ and the tangential components at an angle $\theta_i/2 + \pi/4$, the basic level of shear stress contribution is defined for the *i*-th mechanism of "the virtual simple shear mechanism."

To summarize, the stress is composed of the stress contributions from the virtual plane strain mechanisms, which are in turn composed of the basic stress contributions from the virtual volumetric and simple shear mechanisms. All of these contributions are due to the sand particle displacements, which are closely related to the average strain as defined for continuum.

The above-mentioned physical background of the soil model can be confirmed through a series of manipulations with the tensors representing the relevant quantities (Iai, 1993).

In what follows, some of the details of the model will be presented in order to explain the meaning of the model parameters.

6.2.3 SHEAR MECHANISM

As mentioned earlier, each virtual simple shear mechanism is assumed to follow a hyperbolic stress-strain relation with hysteresis given by a rule similar to Masing's rule. Thus, the virtual tangent shear moduli are given for the initial loading by

$$R_{L}^{(i)} = \left[1 / \left(1 + \left|\gamma^{(i)} / \gamma_{\nu}\right|\right)^{2}\right] \left(Q_{\nu} / \gamma_{\nu}\right) \Delta\theta$$
(6.11)

in which Q_{ν} = virtual shear strength and γ_{ν} = virtual reference strain. The parameters Q_{ν} and γ_{ν} are not directly measurable but they can be readily determined from shear strength τ_m and shear modulus of sand at small strain level G_m by

$$Q_{v} = \tau_{m} / \left(\sum_{i=1}^{I} \sin \theta_{i} \Delta \theta \right)$$
(6.12)

$$\gamma_{\nu} = \left(\tau_m / G_m\right) \left[\left(\sum_{i=1}^{I} \sin^2 \theta_i \Delta \theta\right) / \left(\sum_{i=1}^{I} \sin \theta_i \Delta \theta\right) \right]$$
(6.13)

The derivation of Equation (6.12) and (6.13) are similar to that shown by Towhata and Ishihara (1985).

For incorporating the hysteresis, the Masing's rule is modified here in order to incorporate the ability to achieve realistic hysteresis loop instead of those given by Masing's rule. The approach for modifying the hysteresis loop is similar to that proposed by Ishihara et al. (1985).

From the above formulation, it becomes evident that the parameters necessary for specifying the hyperbolic relation are friction angle ϕ'_{f} and elastic shear modulus G_{ma} measured at the reference confining pressure of σ'_{ma} . These parameters determine the constant of the hyperbolic relation under the initial effective confining pressure of σ'_{mo} as follows:

$$\tau_{mo} = \left(-\sigma_{mo}\right) \sin \phi_{f}$$
 (6.14)

$$G_{mo} = G_{ma} \left(\sigma_{mo} / \sigma_{ma} \right)^{0.5}$$
(6.15)

$$\gamma_{mo} = \tau_{mo} / G_{mo} \tag{6.16}$$

in which τ_{mo} = shear strength, G_{mo} = elastic shear modulus, and γ_{mo} = reference shear strain.

6.2.4 VOLUMETRIC MECHANISM

In order to define the volumetric mechanism, the rebound modulus K and the volumetric strain due to the dilatancy ε_p in Equation (6.4) through (6.6) should be specified. The rebound modulus K at the initial confining stress $\left(-\sigma_{mo}^{\prime}\right)$ is given by

$$K_{0} = K_{a} \left(\sigma_{mo}^{'} / \sigma_{ma}^{'} \right)^{0.5}$$
(6.17)

Excess pore water pressure, which is directly related to the volumetric strain due to the volumetric strain due to the dilatancy ε_p is generated by the following rule. First of all, a state variable *S* is defined as a variable which is equivalent to σ'_m / σ'_{mo} under the undrained condition with a constant total confining pressure. This state variable is determined from the shear stress ratio $r = \tau / (-\sigma'_{mo})$ and the liquefaction front parameter S_0 , to be defined as a measure of cyclic mobility, to simulate the stress path in p' - q space as follows (see Figure 6.4):

$$S = S_{0} \qquad (\text{if } r < r_{3})$$

$$S = S_{2} + \sqrt{(S_{0} - S_{2})^{2} + [(r - r_{3})/m_{1}]^{2}} \qquad (\text{if } r > r_{3}) \qquad (6.18)$$

in which

$$\tau = \sqrt{\tau_{xy}^{2} + \left[\left(\sigma_{y}^{'} - \sigma_{x}^{'} \right) / 2 \right]^{2}}$$

$$r_{2} = m_{2} S_{0}$$

$$r_{3} = m_{3}S_{0}$$

$$S_{2} = S_{0} - (r_{2} - r_{3}) / m_{1}$$

$$m_{1} = \sin \phi_{f}$$

$$m_{2} = \sin \phi_{p} \qquad (\phi_{p} \text{ is the phase transformation angle})$$

$$m_{3} = 0.67m_{2}$$

Secondly, the liquefaction front parameter S_0 appearing in Equation (6.18) is defined as a function of the normalized plastic shear work w (i.e., $w = W_s / W_n$, where W_s = plastic shear work, and $W_n = \tau_{mo} \gamma_{mo} / 2$) as follows (see Figure 6.5):

$$S_{0} = 1 - 0.6 (w / w_{1})^{p_{1}} \qquad (if w < w_{1})$$
$$S_{0} = (0.4 - S_{1}) (w_{1} / w)^{p_{2}} + S_{1} \qquad (if w > w_{1}) \qquad (6.19)$$

in which S_1 , w_1 , p_1 , and p_2 are the material parameters which characterize the cyclic mobility of sands. Additional parameter c_1 is introduced for computing the plastic shear work; this parameter is introduced to explain the existence of the threshold limit in the shear stress or strain amplitude for generating excess pore water pressures. These parameters are determined by back-fitting to the test results obtained under the undrained cyclic loading condition.

When the effective stress analysis is conducted, the state variable S, given by Equation (6.18), is converted into the equivalent volumetric strain of plastic nature ε_p through the continuity condition, and then is substituted into Equation (6.4).

6.3 COMPUTER PROGRAM FLIP

The model presented in Section 6.2 is coded into the finite element computer program FLIP (Finite element analysis of LIquefaction Program) and is used in the

analysis of the generation of pore water pressure in this investigation. FLIP was developed by Iai et al. of Port and Harbour Research Institute, Japan. In the analysis, the equilibrium and mass balance equations for a porous saturated material are used with the undrained condition (Zienkiewicz and Bettess, 1982). For more details about FLIP, it can be seen from Iai et al. (1990, 1992).

6.4 LIQUEFACTION TESTS ON SANDS USING A CYCLIC TRIAXIAL APPARATUS

Laboratory testing of sands in the vicinity of the studied area were conducted by Gauchan (1984) at Asian Institute of technology, Thailand. The reconstituted samples were used and its physical properties are summarized in Table 6.1. Comparison of gradation between the samples and the sands in the studied area is shown in Figure 6.6. The samples were prepared into cylinder of 50 mm in diameter and 100 mm in height using dry deposition method and were classified into three groups (i.e., loose specimens (Dr = 45-50%), medium dense specimens (Dr = 55-65%), and dense specimens (Dr = 75-85%)). The samples were then consolidated under isotropic condition. For each relative density, undrained cyclic triaxial tests were carried out employing three values of effective confining pressures, that is 50, 100, and 200 kPa. As the cyclic load sinusoidal wave of 1 Hz was used as the input wave, the application of the cyclic loading was continued until 10% axial strain in double amplitude was observed. The results, plotted as the liquefaction resistance curves, are shown in Figure 6.7.

6.5 DETERMINATION OF MODEL PARAMETERS

As seen in the preceding descriptions (Section 6.2), the model has ten parameters; two of which specify elastic properties of soil, other two specify plastic shear behavior, and the rest specify dilatancy as shown in Table 6.2. These parameters were determined from the SPT N-values, which were corrected to the confining pressure of 100 kPa (1 tsf = $10 \text{ t/m}^2 = 1 \text{ ksc} = 1 \text{ atm}$) as often done in the liquefaction potential evaluation (Seed et al., 1985), and the results of the undrained cyclic loading tests described earlier. The details in determining the soil parameters were as follows.

The elastic shear modulus G_{ma} at a reference confining pressure of $\left(-\sigma_{ma}^{\dagger}\right)$ was determined by referring to the correlation between SPT N-values and shear wave velocities as previously described in Section 3.5 of Chapter 3. The reference effective confining pressure of $(-\sigma_{ma})$, which links elastic shear modulus to in situ confining pressure for each soil layer, was determined by referring to the K_0 condition in which $K_0 = 1 - \sin \phi_f$ for normally consolidated coarse-grained soils (Jaky, 1944; Brooker and Ireland, 1965) and $K_0 = 0.44 + 0.42 \left(\frac{PI(\%)}{100}\right)$ for normally consolidated finegrained soils (Massarsch, 1979). For overconsolidated soils, K_0 can be determined as a function of its value in the normally consolidated state using the relationship; $K_{0,OC} = K_{0,NC} (OCR)^{0.5}$ (adapted from Schmidt, 1966; Alpan, 1967; Schmertmann, 1975; Ladd et al., 1977). The use of the reference effective confining pressure permits a gradual increase in the elastic shear modulus in accordance with a gradual increase in confining pressure within each soil layer. Shear resistance angle $\phi_{f}^{'}$ was estimated by referring to a correlation with SPT N-values proposed by Peck, Hanson, and Thornburn (1953). The relative density Dr was estimated based on the SPT N-value using a relationship proposed by Gibbs and Holz (1957). The hysteretic damping factor for an infinitely large shear strain H_m was determined to be 30% by referring to the typical laboratory results summarized by Ishihara (1982).

The rebound modulus or the elastic tangent bulk modulus of soil skeleton K_a was determined by assuming that Poisson's ratio is 0.33. The phase transformation angle $\phi_p^{'}$, which separates dilative and contractive zones in p'-q space, was assumed to be 30 degrees for clean sand (FC < 5%) and 28 degrees for silty sand by referring to the typical value adopted for effective stress analyses of saturated sand (Ishihara et

al., 1989). The remaining five parameters p_1 , p_2 , w_1 , S_1 , and c_1 , which specify dilatancy of sand, were determined by the back-fitting to the liquefaction resistance curves of the sand as previously mentioned in Section 6.4. Figure 6.8 shows the comparison of shear stress ratio against number of cycle between laboratory test results and computed. The soil parameters determined in this manner are summarized in Table 6.3. More details in the back-fitting procedure as well as the definition of each parameter can be found in a paper by Iai et al. (1990).

6.6 RESULTS OF ANALYSES

Finite element analyses based on effective stress model were performed on twenty-nine sites within Chiang Mai City and seventeen sites within Chiang Rai City using the input parameters as mentioned above. From the analysis results, it can be found that the pore water pressure ratio in sand deposits increases with increasing probability of liquefaction (P_L). It should be noted that the pore water pressure ratio for clay layer is not resulted by FLIP.

The relationship between soil amplification and maximum pore water pressure ratio is presented in Figure 6.9 and 6.10. From the figures, it is apparent that the maximum pore water pressure ratio of sand layer in sand sites is greater than that in clay sites. In contrast, the amplification factor in sand sites is less than that in clay sites. Note that a sand site is defined as a site that most of soil types are sand. Similarly, a clay site is defined as a site that most of soil types are clay.

Figure 6.11 and 6.12 show the analytical results by plotting the cyclic stress ratio (*CSR*) for earthquake magnitude of 5.5 obtained from the logistic model as mentioned in Chapter 5 against the maximum pore water pressure ratio. The results show that the maximum pore water pressure ratio depends on the predominant periods of the input motion and tends to decrease with increase in the predominant period. This relationship can be used to evaluate the maximum excess pore water pressure of sand layer in the city of Chiang Mai and Chiang Rai.

Since Chiang Mai and Chiang Rai are located in the seismic zone G which probable causes earthquake magnitude (M_L) of 5 to 6 with maximum ground acceleration (a_{max}) of 0.2g (Figure 3.2 and 3.3). Factor of safety computed following Seed et al. (1983) at various values of P_L is then obtained and summarized in Table 6.4. At P_L of 5%, there are more than 80% of the sandy sites subject to a certain level of liquefaction susceptibility.

Figure 6.13 and 6.14 show the typical analytical results by plotting the maximum pore water pressure ratio against the maximum ground acceleration. The vertical line crossed at $a_{max} = 0.2$ g is drawn for reference. The points located on the left of this line represent sites where factor of safety is less than 1.0 (corresponding to those shown in Table 6.4). The maximum pore water pressure ratio for cases when $P_L = 5\%$ varies in the range of 0.1 - 0.8. Figure 6.15 and 6.16 show the maximum pore water pressure ratio for cases when $P_L = 5\%$ varies in the range of -2.5 m from ground surface. The pore water pressure ratio for cases when $P_L = 5\%$ varies in the range of 0.1 - 0.5. Although near surface sand may not experience liquefaction, factor of safety of shallow foundation can be greatly reduced due to decrease in effective stress.

6.7 SUMMARY

This chapter illustrates the use of the finite element method to investigate the pore water pressure in sand deposits in the studied area. Several important findings from this investigation are summarized below:

- 1) The pore water pressure ratio in sand deposits increases with increasing probability of liquefaction (P_L) .
- 2) The maximum pore water pressure ratio of sand layer in sand sites is greater than that in clay sites. In contrast, the amplification factor in sand sites is less than that in clay sites.

- 3) The maximum pore water pressure ratio depends on the predominant periods of the input motion and tends to decrease with increase in the predominant period.
- Although near surface sand in the studied area may not experience liquefaction, factor of safety of shallow foundation can be greatly reduced due to decrease in effective stress.



สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

CHAPTER 7

CONCLUSIONS AND RECOMMENDATIONS

7.1 CONCLUSIONS

In this thesis, the amplification of earthquake ground motion, liquefaction probability, and the generation of pore water pressure due to medium earthquakes in the northern part of Thailand, particularly Chiang Mai and Chiang Rai, were studied. Due to the lack of strong motion record in Thailand, three input motions recorded from elsewhere with different predominant periods were adopted. Soil properties used as input to the analyses, specifically shear wave velocity, were estimated using existing correlations with field and laboratory data. Seismic site response analyses were conducted using the computer program SHAKE to quantify the potential amplification of earthquake ground motions in the cities due to soil effects.

The binary logistic regression technique using the worldwide liquefaction database was used to form the probabilistic base correlation between cyclic stress ratio and the SPT resistance for evaluating the liquefaction probability. Subsequently, the finite element program based on effective stress model called FLIP was used to evaluate the generation of pore water pressure in sand deposits corresponding to various probability values. The laboratory test results on undrained cyclic triaxial of sands were used to obtain some effective stress parameters required in the effective stress model. The following conclusions are drawn from the study:

 The soil profile underlying the city of Chiang Mai and Chiang Rai has the ability to amplify earthquake ground motions 1.5 to 3 times. The amount of this amplification was found to decrease with increasing bedrock acceleration. This amplification depends on the local site conditions, the level of acceleration in the underlying rock-like material, and the predominant periods and characteristics of the input motion.

- 2) The models obtained from the probabilistic method are based on the Seed and Idriss (1971) parameterization and employ the magnitude cyclic stress ratio *CSR* as a measure of earthquake load; and use the corrected/normalized SPT value $(N_1)_{60}$ as a measure of soil-liquefaction resistance.
- 3) In the probability charts (Figure 5.3 and 5.4), the probabilistic line at $P_L = 30\%$ well traces the deterministic criteria for clean sand base curve proposed by Seed et al. (1985). Therefore, the success rate line which is drawn to separate events of liquefaction and non-liquefaction is determined based on $P_L = 30\%$ line.
- 4) The probabilistic procedure provides more statistical rigorous criteria for defining liquefaction resistance than was used in the original development of the deterministic or simplified procedure suggested by Seed and Idriss (1971). As with all empirical methods, however, the quality of the results are strongly dependent on the quantity and quality of the compiled input data.
- 5) With the proper input soil model parameters, the finite element method can be powerful and versatile analytical tool for studying the generation of pore water pressure in sand deposits due to earthquake shaking.
- 6) The pore water pressure ratio in sand deposits increases with increasing probability of liquefaction (P_L) .
- 7) The maximum pore water pressure ratio of sand layer in sand sites is greater than that in clay sites. In contrast, the amplification factor in sand sites is less than that in clay sites.
- 8) The maximum pore water pressure ratio depends on the predominant periods and characteristics of the input motion and tends to decrease with increase in the predominant period.
- 9) The charts correlating the liquefaction probability, estimated excess pore water pressure and peak ground acceleration proposed in this study can be used as a

simple tool for estimating excess pore water pressure in sand deposits due to earthquake shaking in the city of Chiang Mai and Chiang Rai.

- 10) Although near surface sand in the studied area may not experience liquefaction, factor of safety of shallow foundation can be greatly reduced due to decrease in effective stress.
- 11) The probabilistic procedure for evaluating liquefaction potential proposed in this study have the primary advantage for engineering applications is that the user can select an appropriate probability level of risk of occurrence for analyzing liquefaction hazard. Therefore, The developed method has the potential of becoming a practical tool for engineers involved in the assessment of liquefaction potential.

7.2 RECOMMENDATIONS FOR FUTURE RESEARCH

Several other subjects related in this research have been identified that need further investigation. The needs are summarized below:

- Additional strong motion records should be used as input motions to make the study more complete, including both actual seismograms from far-field sites and near-field sites; and synthetic seismograms with frequency contents or predominant periods matching the expected seismicity of the region.
- 2) The estimate of expected peak rock acceleration in Chiang Mai and Chiang Rai must be refined. This refinement would necessarily include definition and characterization of active faults that are likely to affect the cities.
- 3) The soil properties used in the analyses, particularly shear wave velocity and variation of shear modulus and damping with strain, must be better defined. This would include additional in situ testing for shear wave velocity and sophisticated laboratory testing for determining the variation of shear modulus and damping with shear strain. Although the published correlations and relationships used in

this study are often sufficient, it would seem prudent to develop relationships specifically for soil underlying Chiang Mai and Chiang Rai; the capital of the northern region of Thailand.

- 4) The depth to rock like material must be better determined through in situ testing to at least a depth of 160 m or more, in order to obtain the better analysis results.
- 5) The effects of several variables on the liquefaction probability, such as fines content (FC), gravel content (GC), and median grain size (D_{50}), were not included in the proposed models due to limitations in the data. They should be included in the probability models once the complete database is obtained.
- 6) In this study, the soil liquefaction resistance was measured by the standard penetration test (SPT). Though this test is relatively crude, it is one of the few measurements that provide a direct link between actual observations of liquefaction and soil properties. However, measurements from other in situ tests, such as the cone penetration test (CPT), the shear wave velocity test (V_s), and the Becker penetration test (BPT), are now being correlated to liquefaction performance. Logistic regression can also be applied to these measurements, once a significant database has been established.

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

REFERENCES

- Algermissen, S. T., Perkins, D. M., Thenhaus, P. C., Hanson, S. L., and Bender, B. L. 1982. <u>Probabilistic estimates of maximum acceleration and velocity in rock in</u> <u>the contiguous United States</u>. Open-File Report 82-1033, U. S. Geological Survey, Washington, D. C.
- Alpan, I. 1967. The empirical evaluation of the coefficients K_o and K_{oR} . <u>Soils and</u> <u>Foundations</u> 7, 1: 31-40.
- Ambraseys, N. N. 1988. Engineering seismology. <u>Earthquake Engineering and</u> <u>Structural Dynamics</u> 17: 1-105.
- Anantasech, C., and Thanadpipat, C. 1985. Characteristics and engineering properties of subsoils in intermontane basin. <u>Thailand Engineering Journal</u> 38, 2: 101-106 (in Thai).
- Andrus, R. D., and Stokoe, K. H., II. 2000. Liquefaction resistance of soils from shear wave velocity. <u>Journal of Geotechnical and Geoenvironmental Engineering</u> ASCE 126, 11: 1015-1025.
- Arango, I. 1996. Magnitude scaling factors for soil liquefaction evaluations. <u>Journal</u> <u>of Geotechnical Engineering</u> ASCE 122, 11: 929-936.
- Bjerrum, L. 1972. Embankments on soft ground. <u>Proceedings of the ASCE Specialty</u> <u>Conference on Performance of Earth and Earth-Supported Structures</u>, II, pp. 1-54. Purdue University.
- Blazquez, R. M., Krizek, R. J., and Bazant, Z. P. 1980. Site factors controlling liquefaction. <u>Journal of the Geotechnical Engineering Division</u> ASCE 106, GT7: 785-801.
- Brooker, E. W., and Ireland, H. O. 1965. Earth pressure at rest related to stress history. <u>Canadian Geotechnical Journal</u> 2, 1: 1-15.
- Castro, G. 1969. <u>Liquefaction of sands</u>. Harvard Soil Mechanics Series 87, Harvard University, Cambridge, Massachusetts.

- Castro, G. 1975. Liquefaction and cyclic mobility of sands. <u>Journal of the</u> <u>Geotechnical Engineering Division</u> ASCE 101, GT6: 551-569.
- Chameau, J. L., and Clough, G. W. 1983. Probabilistic pore pressure analysis for seismic loading. Journal of Geotechnical Engineering ASCE 109, 4: 507-524.
- Christian, J. T., and Swiger, W. F. 1975. Statistic of liquefaction and SPT results. Journal of the Geotechnical Engineering Division ASCE 101, GT11: 1135-1150.
- Cooley, P. M., and Tukey, J. W. 1965. An algorithm for the machine computation of complex Fourier series. <u>Mathematics of Computation</u> 19, 4: 297-301.
- Cornell, C. A. 1968. Engineering seismic risk analysis. <u>Bulletin of the Seismological</u> <u>Society of America</u> 58: 1583-1606.
- Davis and Berrill, J. B. 1981. Assessment of liquefaction potential based on seismic energy dissipation. <u>Proceedings of the International Conference on Recent</u> <u>Advances in Geotechnical Earthquake Engineering and Soil Dynamics</u>, 1, pp. 197-190. St. Louis.
- Day, R. W. 2002. <u>Geotechnical earthquake engineering handbook</u>. New York: McGraw-Hill.
- De Alba, P., Seed, H. B., and Chan, C. K. 1976. Sand liquefaction in large-scale simple shear tests. <u>Journal of Geotechnical Engineering Division</u> ASCE 102, GT9: 909-927.
- Dickenson, S. E. 1994. <u>Dynamic response of soft and deep cohesive soils during the</u> <u>Loma Prieta earthquake of October 17, 1989</u>. Ph.D. Dissertation, Department of Civil Engineering, University of California, Berkeley, California.
- Dillon, W. R., and Goldstein, M. 1984. <u>Multivariate analysis: Methods and applications</u>. New York: John Wiley.
- Dobry, R., and Vucetic, M. 1987. Dynamic properties and seismic response of soft clay deposits. <u>Proceedings of International Symposium on Geotechnical</u> <u>Engineering of Soft Soils</u>, 2, pp. 51-87. Mexico City.
- Egan, J. A., and Ebeling, R. M. 1985. Variation of small-strain modulus with undrained shear strength of clay. <u>Proceedings of 2nd International Conference on Soil Dynamics and Earthquake Engineering</u>, 2, pp. 27-36.
- Fardis, M. N., and Veneziano, D. 1982. Probabilistic analysis of deposit liquefaction. Journal of the Geotechnical Engineering Division ASCE 108, GT3: 395-417.
- Fiegel, G. L., and Kutter, B. L. 1992. <u>The mechanism of liquefaction in layered soils</u>. Report CR92.009, Naval Civil Engineering Laboratory, Port Hueneme, California.
- Finn, W. D. L., Lee, K. W., and Martin, G. R. 1977. An effective stress model for liquefaction. Journal of the Geotechnical Engineering Division ASCE 103, GT6: 517-533.
- Fumal, T. E. 1978. <u>Correlations between seismic wave velocities and physical</u> properties of near-surface geologic materials in the southern San Francisco bay region, California. Open File Report 78-1067, U.S. Geological Survey, Menlo Park, California.
- Gauchan, J. 1984. <u>Liquefaction tests on sand using a cyclic triaxial apparatus</u>. Master's Thesis, Asian Institute of Technology, Bangkok, Thailand.
- Gibbs, H. J., and Holtz, W. G. 1957. Research on determining the density of sand by spoon penetration testing. <u>Proceedings of the 4th International Conference on</u> <u>Soil Mechanics and Foundation Engineering</u>, 1, pp. 35-39. London.
- Gutenberg, B., and Richter, C. F. 1956. Earthquake magnitude, intensity, energy and acceleration (Second Paper). <u>Bulletin of the Seismology Society of America</u> 46, 2: 143-145.
- Haldar, A., and Tang, W. H. 1979. Probabilistic evaluation of liquefaction potential. Journal of the Geotechnical Engineering Division ASCE 105, GT2: 145-163.
- Hardin, B. O., and Drnevich, V. P. 1972a. Shear modulus and damping in soils: Measurement and parameter effects. Journal of the Soil Mechanics and <u>Foundations Division</u> ASCE 98, SM6: 603-624.

- Hardin, B. O., and Drnevich, V. P. 1972b. Shear modulus and damping in soils: design equations and curves. <u>Journal of the Soil Mechanics and Foundations</u> <u>Division</u> ASCE 98, SM7: 667-692.
- Hardin, B. O., and Richart, F. 1963. Elastic wave velocities in granular soils. Journal of the Soil Mechanics and Foundations Division ASCE 89, 1: 33-65.
- Iai, S. 1993. Micromechanical background to a strain space multiple mechanism model for sand. <u>Soils and Foundations</u> 33, 1: 102-117.
- Iai, S., Kameoka, T., and Matsunaga, Y. 1993. Numerical (Class A) prediction of Model No.1. <u>Verification of Numerical Procedures for the Analysis of Soil</u> <u>Liquefaction Problems (VELACS)</u>, pp. 109-127. Balkema, Rotterdam.
- Iai, S., Matsunaga, Y., and Kameoka, T. 1990. Parameter identification for a cyclic mobility model. <u>Report of Port and Harbour Research Institute</u> 29, 4: 57-83.
- Iai, S., Matsunaga, Y., and Kameoka, T. 1992. Strain space plasticity model for cyclic mobility. <u>Soils and Foundations</u> 32, 2: 1-15.
- Idriss, I. M. 1990. Response of soft soil sites during earthquakes. <u>Proceedings of the H. Bolton Seed Memorial Symposium</u>, 2, pp. 273-289. Vancouver, British Columbia.
- Idriss, I. M., and Seed, H. B. 1968. Seismic response of horizontal soil layers. Journal of the Soil Mechanics and Foundations Division ASCE 94, SM4.
- Idriss, I. M., and Sun, J. I. 1992. <u>SHAKE91: a computer program for conducting</u> <u>equivalent linear seismic response analyses of horizontally layered soil deposits</u>. User's Guide, University of California, Davis.
- Imai, T., and Tonuchi, K. 1982. Correlation of N-value with S-wave velocity and shear modulus. <u>Proceedings of 2nd European Symposium on Penetration</u> <u>Testing</u>, pp. 67-72. Amsterdam, Netherlands.
- Ishibashi, I., and Zhang, X. 1993. Unified dynamic shear moduli and damping ratios of sand and clay. <u>Soils and Foundations</u> 33, 1: 182-191.
- Ishibashi, I., Sherif, M. A., and Tsuchiya, C. 1977. Pore-pressure rise mechanism and soil liquefaction. <u>Soils and Foundations</u> 17, 2: 17-27.

- Ishihara, K. 1982. Evaluation of soil properties for use in earthquake response analysis. <u>Proceedings of International Symposium on Numerical Models in</u> <u>Geomechanics</u>, pp. 237-259. Zurich.
- Ishihara, K. 1984. Post-earthquake failure of a tailings dam due to liquefaction of the pond deposit. <u>Proceedings of International Conference on Case Histories in</u> <u>Geotechnical Engineering</u>, 3, pp. 1129-1143. University of Missourri, St. Louis.
- Ishihara, K. 1985. Stability of natural deposits during earthquakes. <u>Proceedings of the</u> <u>11th International Conference on Soil Mechanics and Foundation Engineering</u>, 1, pp. 321-376.
- Ishihara, K. 1993. Liquefaction and flow failure during earthquakes. <u>Geotechnique</u> 43, 3: 351-415.
- Ishihara, K. 1996. <u>Soil behaviour in earthquake geotechnics</u>. Oxford: Clarendon Press.
- Ishihara, K., Tatsuoka, F., and Yasuda, S. 1975. Undrained deformation and liquefaction of sand under cyclic stresses. <u>Soils and Foundations</u> 15, 1: 29-44.
- Ishihara, K., Lysmer, J., Yasuda, S., and Hirao, H. 1976. Prediction of liquefaction in sand deposits during earthquakes. <u>Soils and Foundations</u> 16, 1: 1-16.
- Ishihara, K. Yoshida, N., and Tsujino, S. 1985. Modelling of stress-strain relations of soils in cyclic loading. <u>Proceedings of the 5th International Conference on</u> <u>Numerical Methods in Geomechanics</u>, 1, pp. 373-380.
- Ishihara et al. 1989. Effective stress analysis of ground and soil structures. Proceedings of Symposium on Seismic Performance of Ground and Soil Structures, pp. 50-63. Japanese Society of Soil Mechanics and Foundation Engineering.
- Iwasaki, T., Tatsuoka, F., and Takagi, Y. 1978. Shear modulus of sands under torsional shear loading. <u>Soils and Foundations</u> 18, 1: 39-56.
- Jaky, J. 1944. The coefficient of earth pressure at rest. Journal of the Society of Hungarian Architects and Engineers 78, 22: 355-358 (in Hungarian).
- Johnson, D. E. 1998. Applied multivariate methods for data analysis. Duxbury Press.

- Johnson, R. A., and Wichern, D. W. 1988. <u>Applied multivariate statistical analysis</u>. 4th ed. New Jersey: Prentice Hall.
- Joyner, W. B., and Boore, D. M. 1981. Peak horizontal acceleration and velocity from strong-motion records including records from the 1979 Imperial Valley, California, earthquake. <u>Bulletin of the Seismological Society of America</u> 71, 6: 2011-2038.
- Kanai, K. 1951. Relation between the nature of surface layer and the amplitude of earthquake motions. <u>Bulletin of the Earthquake Research Institute</u> Tokyo University 29.
- Kanai, K. 1957. Semi-empirical formula for the seismic characteristics of the ground. Bulletin of the Earthquake Research Institute Tokyo University 35: 308-325.
- Kanai, K. 1966. Observation of microtremors, XI, Matsushiro earthquake swarm areas. <u>Bulletin of the Earthquake Research Institute</u> Tokyo University 44: 315-337.
- Kawashima, K., Aizawa, K., and Takahashi, K. 1984. Attenuation of peak ground motion and absolute acceleration response spectra. <u>Proceedings of the 8th World</u> <u>Conference on Earthquake Engineering</u>, 2, pp. 257-264. San Francisco, California.
- Kayen, R. E., Mitchell, J. K., Seed, R. B., Lodge, A., Nishio, S., and Coutinho, R. 1992. Evaluation of SPT-, CPT-, and shear wave-based methods for liquefaction potential assessment using Loma Prieta data. Proceedings of the 4th Japan-U. S. Workshop on Earthquake Resistant Design of Lifeline Facilities and Countermeasures for Soil Liquefaction, 1, pp. 177-204. Buffalo, New York.

```
Kleinbaum, D. A. 1994. Logistic regression. Springer-Verlag.
```

- Knox, D., Stokoe, K., and Kopperman, S. 1982. Effect of state of stress on shear wave velocity in dry sand. Geotechnical Engineering Report GR82-23, The University of Texas at Austin, Austin, Texas.
- Kokusho, T. 1980. Cyclic triaxial test of dynamic soil properties for wide strain range. Soils and Foundations 20, 2: 45-60.

- Kokusho, T., Yoshida, Y., and Esashi, Y. 1982. Dynamic properties of soft clay for wide range. <u>Soils and Foundations</u> 22, 4: 1-18.
- Koutsoftas, D., and Fischer, J. A. 1980. Dynamic properties of two marine clays. Journal of the Geotechnical Engineering Division ASCE 106, GT6: 645-657.
- Kramer, S. L. 1996. Geotechnical earthquake engineering. New Jersey: Prentice Hall.
- Ladd, C. C., Foott, R., Ishihara, K., Schlosser, F., and Poulos, H. G. 1977. Stressdeformation and strength characteristics. <u>Proceedings of the 9th International</u> <u>Conference on Soil Mechanics and Foundation Engineering</u>, 2, pp. 421-494. Japanese Society of Soil Mechanics and Foundation Engineering, Tokyo.
- Liao, S. S. C., and Whitman, R. V. 1986a. <u>A catalog of liquefaction and non-liquefaction occurrences during earthquakes</u>. Research Report, Department of Civil Engineering, Massachusetts Institute of Technology, Cambridge, Massachusetts.
- Liao, S. S. C., and Whitman, R. V. 1986b. Overburden correction factors for SPT in sand. Journal of Geotechnical Engineering ASCE 112, 3: 373-377.
- Lindeburg, M. R. 1998. Seismic design of building structures. 7th ed. Belmont, CA: Professional Publications, Inc.
- Liou, C. P., Streeter, V. L., and Richart, F. E., Jr. 1977. Numerical model for liquefaction. <u>Journal of the Geotechnical Engineering Division</u> ASCE 103, GT6: 589-606.
- Liu, H., and Qiao, T. 1984. Liquefaction potential of saturated sand deposits underlying foundation of structure. <u>Proceedings of the 8th World Conference on Earthquake Engineering</u>, 3, pp. 199-206. San Francisco.
- Lysmer, J., Seed, H. B., and Schnabel, P. B. 1970. <u>Influence of base rock</u> <u>characteristics on ground response</u>. Report No. EERC 70-7, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Lysmer, J., Seed, H. B., and Schnabel, P. B. 1971. Influence of base-rock characteristics on ground response. <u>Bulletin of the Seismological Society of America</u> 61, 5: 1213-1232.

- Martin, G. R., Finn, W. D. L., and Seed, H. B. 1975. Fundamentals of liquefaction under loading. <u>Journal of the Geotechnical Engineering Division</u> ASCE 101, GT5: 423-429.
- Massarsch, K. R. 1979. Lateral earth pressure in normally consolidated clay. <u>Proceedings of the 7th European Conference on Soil Mechanics and Foundation</u> <u>Engineering</u>, 2, pp. 245-250. Brighton, England.
- Matthiesen, R. B., Duke, C. M., Leeds, D. J., and Fraser, J. C. 1964. <u>Site characteristics of southern California strong-motion earthquake stations, part two</u>. Report No. 64-15, Department of Civil Engineering, University of California, Los Angeles.
- Mori, K., Seed, H. B., and Chan, C. K. 1978. Influence of sample disturbance on sand response to cyclic loading. <u>Journal of the Geotechnical Engineering Division</u> ASCE 104, GT3: 323-339.
- Morrison, D. F. 1990. <u>Multivariate statistical method</u>. 3rd ed. New York: McGraw-Hill.
- Mroz, Z., Norris, V. A., and Zienkiewicz, O. C. 1978. An isotropic hardening model and analytical methods in geomechanics. <u>International Journal for Numerical</u> <u>and Analytical Methods in Geomechanics</u> 2: 203-221.
- Nuttalaya, P., Sodsri, S., and Arnold, E. P. 1985. Series on seismology volume II-Thailand. <u>Southeast Asia Association of Seismology and Earthquake</u> <u>Engineering</u>, Bangkok.
- Ohsaki, Y., and Iwasaki, R. 1973. On dynamic shear moduli and poisson's ratio of soil deposits. <u>Soils and Foundations</u> 13, 4: 61-73.
- Ohta, Y., and Goto, N. 1978. Empirical shear wave velocity equations in terms of characteristics soil indexes. <u>Earthquake Engineering and Structural Dynamics</u> 6: 167-187.
- Peck, R. B., Hanson, W. E., and Thornburn, T. H. 1953. <u>Foundation Engineering</u>. New York: John Wiley & Sons.
- Reiter, L. 1990. <u>Earthquake hazard analysis–issues and insight</u>. New York: Columbia University Press.

- Roesset, J. M., and Whitman, R. V. 1969. <u>Theoretical background for amplification</u> <u>studies</u>. Research Report No. R69-15, Soils Publication No. 231, Massachusetts Institute of Technology, Cambridge.
- Sakai, Y. 1968. <u>A study on determination of S-wave velocity by soil penetrometer</u> test.
- Schnabel, P. B., Lysmer, J., and Seed, H. B. 1972. SHAKE: <u>A computer program for</u> <u>earthquake response analysis of horizontally layered sites</u>. Report No. EERC 72-12, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Seed, H. B. 1983. Earthquake-resistant design of earth dams. <u>Proceeding, Symposium</u> on Seismic Design of Earth Dams and Caverns, ASCE, pp. 41-64. New York.
- Seed, H. B., and Idriss, I. M. 1970. <u>Soils moduli and damping factors for dynamic response analyses</u>. Report No. EERC 70-10, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Seed, H. B., and Idriss, I. M. 1971. Simplified procedure for evaluating soil liquefaction potential. Journal of the Soil Mechanics and Foundations Division ASCE Vol. 97, No. SM9: 1249-1273.
- Seed, H. B., and Idriss, I. M. 1982. <u>Ground motions and soil liquefaction during</u> <u>earthquakes</u>. California: Earthquake Engineering Research Institute.
- Seed, H. B., and Lee, K. L. 1966. Liquefaction of saturated sands during cyclic loading. <u>Journal of Soil Mechanics and Foundations Division</u> ASCE 92, SM6: 105-134.
- Seed, H. B., Arango, I., and Chan, C. K. 1975. <u>Evaluation of soil liquefaction</u> <u>potential during earthquakes</u>. Report No. EERC 75-28, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Seed, H. B., Martin, P. P., and Lysmer, J. 1976a. Pore-water pressure changes during soil liquefaction. <u>Journal of the Geotechnical Engineering Division</u> ASCE 102, GT4: 323-346.
- Seed, H. B., Ugas, C., and Lysmer, J. 1976b. Site-dependent spectra for earthquakeresistant design. <u>Bulletin of the Seismological Society of America</u> 66: 221-243.

- Seed, H. B., Idriss, I. M., and Arango, I. 1983. Evaluation of liquefaction potential using field performance data. <u>Journal of Geotechnical Engineering</u> ASCE 109, 3: 458-482.
- Seed, H. B., Wong, R. T., Idriss, I. M., and Tokimatsu, K. 1984a. <u>Moduli and damping factors for dynamic analyses of cohesionless soils</u>. Report No. UCB/EERC-84/14, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Seed, H. B., Tokimatsu, K., Harder, L. F., and Chung, R. M. 1984b. <u>Influence of SPT</u> procedures in soil liquefaction resistance evaluations. Report No. UBC/EERC 84-15, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Seed, H. B., Tokimatsu, K., Harder, L. F., and Chung, R. M. 1985. Influence of SPT procedures in soil liquefaction resistance evaluations. <u>Journal of Geotechnical</u> <u>Engineering</u> ASCE 111, 12: 1425-1445.
- Seed, H. B., Wong, R. T., Idriss, I. M., and Tokimatsu, K. 1986. Moduli and damping factors for dynamic analyses of cohesionless soils. <u>Journal of Geotechnical</u> <u>Engineering</u> ASCE 112, GT11: 1016-1032.
- Seed, R. B., and Harder, L. F., Jr. 1990. SPT-based analysis of cyclic pore pressure generation and undrained residual strength. <u>Proceedings of H. Bolton Seed</u> <u>Memorial Symposium</u>, 2, pp. 351-376. University of California, Berkeley.
- Schmertmann, J. H. 1975. Measurement of in situ shear strength. <u>Proceedings of the</u> <u>ASCE Specialty Conference on In Situ Measurement of Soil Properties</u>, 2, pp. 57-138. Raleigh, N.C.
- Schmertmann, J. H., Smith, T. V., and Ho, R. 1978. Example of an energy calibration report on a standard penetration test (ASTM Standard D1586-67) drill rig. <u>Geotechnical Testing Journal</u> 1, 1: 57-61.
- Schmidt, B. 1966. Discussion on earth pressure at rest related to stress history. Canadian Geotechnical Journal Vol. 3, No. 4.
- Sherif, M. A., Ishibashi, I., and Tsuchiya, C. 1978. Pore-pressure prediction during earthquake loading. <u>Soils and Foundations</u> 18, 4: 19-30.

- Shibata, T., Yukitomo, H., and Miyoshi, M. 1972. Liquefaction process of sand during cyclic loading. <u>Soils and Foundations</u> 12, 1: 1-16.
- SPSS. 1999. <u>SPSS base 9.0 for windows user's guide and applications guide</u>. SPSS Inc.
- Sun, J. I., Golesorkhi, R., and Seed, H. B. 1988. <u>Dynamic moduli and damping ratios</u> for cohesive soils. Report No. EERC-88/15, Earthquake Engineering Research Center, University of California, Berkeley, California.
- Sykora, D. W. 1987. <u>Examination of existing shear wave velocity and shear modulus</u> <u>correlations in soils</u>. U.S. Army Engineering Waterways Experiment Station Misc. Paper GL-87-22.
- Sykora, D. W., and Stokoe, K. H. 1983. <u>Correlations of in-situ measurements in sands</u> with shear wave velocity. Geotechical Engineering Report GR83-33, The University of Texas at Austin, Texas.
- Tokimatsu, K., and Yoshimi, Y. 1983. Empirical correlation of soil liquefaction based on SPT N-value and fines content. <u>Soils and Foundations</u> 23, 4: 56-74.
- Toprak, S., Holzer, T. L., Bennett, M. J., and Tinsley, J. C., III. 1999. CPT- and SPTbased probabilistic assessment of liquefaction. <u>Proceedings of the U. S.-Japan</u> <u>Workshop on Earthquake Resistant Design of Lifeline Facilities and</u> <u>Countermeasures Against Liquefaction</u>, pp. 69-86. Multidisciplinary Center for Earthquake Engineering Research, Buffalo, New York.
- Towhata, I., and Ishihara, K. 1985. Modelling soil behaviour under principal stress axes rotation. <u>Proceedings of the 5th International Conference on Numerical Methods in Geomechanics</u>, 1, pp. 523-530.
- Trudeau, P. J., Whitman, R. V., and Christian, J. T. 1973. Shear wave velocity and modulus of a marine clay. <u>Boston Society of Civil Engineers</u> 61, 1: 12-25.
- Veneziano, D., and Liao, S. 1984. Statistic analysis of liquefaction data. <u>Proceedings</u> of the 4th ASCE Specialty Conference on Probabilistic Mechanics and Structural <u>Reliability</u>, pp. 206-209.
- Vucetic, M., and Dobry, R. 1991. Effect of soil plasticity on cyclic response. Journal of Geotechnical Engineering ASCE 117, 1: 89-107.

- Warnitchai, P., and Lisantono, A. 1996. Probabilistic seismic risk mapping for Thailand. <u>Proceedings of the 11th World Conference on Earthquake</u> <u>Engineering</u>, Acapulco, Mexico.
- Weiler, W. A. 1988. Small strain shear modulus of clay. <u>Proceedings of ASCE</u> <u>Conference on Earthquake Engineering and Soil Dynamics II: Recent Advances</u> <u>in Ground Motion Evaluation, Geotechnical Special Publication 20, ASCE</u>, pp. 331-335. New York.
- Whitman, R. V. 1971. Resistance of soil to liquefaction and settlement. <u>Soils and</u> <u>Foundations</u> 11, No. 4: 59-68.
- Xie, J. 1979. Empirical criteria of sand liquefaction: the 1976 Tangshan China earthquake. <u>Proceedings of the 2nd U. S. National Conference on Earthquake</u> <u>Engineering</u>, Standford University.
- Yasuda, N, and Matsumoto, N. 1993. Dynamic deformation characteristics of sands and rock-fill materials. <u>Canadian Geotechnical Journal</u> 30, 5: 747-757.
- Yegian, M. K. 1976. <u>Risk analysis for earthquake-induced ground failure by</u> <u>liquefaction</u>. Ph. D. Thesis, Massachusetts Institute of Technology, Cambridge, Massachusetts.
- Yegian, M. K., and Vitelli, B. M. 1981. Probabilistic analysis for liquefaction. Report No. CE-81-1, Northeastern University, Boston, Massachusetts.
- Yegian, M. K., and Whitman, R. V. 1978. Risk analysis for ground failure by liquefaction. <u>Journal of the Geotechnical Engineering Division</u> ASCE 104, GT7: 921-938.
- Youd, T. L. 1984. Recurrence of liquefaction at the same site. <u>Proceedings of the 8th</u> <u>World Conference on Earthquake Engineering</u>, 3, pp. 231-238. San Francisco.
- Youd, T. L. 1991. Mapping of earthquake-induced liquefaction for seismic zonation. <u>Proceedings of the 4th International Conference on Seismic Zonation,</u> <u>Earthquake Engineering Research Institute</u>, 1, pp. 111-147. Standford University.
- Youd, T. L. 1993. <u>Liquefaction-induced lateral spread displacement</u>. NCEL Technical Note N-1862, U. S. Navy, Port Hueneme, California.

- Youd, T. L., and Idriss, I. M., eds. 1997. <u>Proceedings of NCEER Workshop on</u> <u>Evaluation of Liquefaction Resistance of Soils</u>, State University of New York at Buffalo.
- Youd, T. L., and Noble, S. K. 1997. Magnitude scaling factors. <u>Proceedings of NCEER Workshop on Evaluation of Liquefaction Resistance of Soils</u>, pp. 149-165. State University of New York at Buffalo.
- Zen, K., Umehara, Y., and Hamada, K. 1978. Laboratory tests and in-situ seismic survey on vibratory shear modulus of clayey soils with different plasticities.
 <u>Proceedings of the 5th Japan Earthquake Engineering Symposium</u>, pp. 721-728. Tokyo.
- Zienkiewicz, O. C., and Bettess, P. 1982. Soils and other saturated media under transient, dynamic conditions; general formulation and the validity of various simplifying assumption. In Pande, G. N., and Zienkiewicz, O. C. (eds.), <u>Soil</u> <u>mechanics-transient and cyclic loads</u>, pp. 1-16. (n.p.): John Wiley & Sons.
- Zienkiewicz, O. C., Chang, C. T., and Hinton, E. 1978. Non-linear seismic response and liquefaction. International Journal for Numerical and Analytical Methods in <u>Geomechanics</u> 2: 381-404.

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย TABLES

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

Date	Magnitude	Center	Were Felt at
April 22, 1983	5.9	Kanchanaburi, Thailand	Bangkok, Western and northern parts
November 6, 1988	7.3	Southern of China (1,000 km from Bangkok)	Bangkok, Western and northern parts
September 29 – October 1, 1989	5.3 – 5.4 Several quakes	Western part	Bangkok, Western and northern parts
September 11, 1994	5.5	Phan District (Northern part)	Northern parts
January 22, 2003	7.5	Sumatra Island (1,000 km from Bangkok)	Bangkok
September 22, 2003	6.6	Burma (850 km from Bangkok)	Bangkok and Northern parts

Table 1.1. Examples of recent earthquakes felt in Thailand

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

Intensity	Observed Effects of Earthquake
Ι	Not felt except by very few under especially favorable conditions.
II	Felt only by a few persons at rest, especially by those on upper floors
	of buildings. Delicately suspended objects may swing.
III	Felt quite noticeably by persons indoors, especially in upper floors of
	buildings. Many people do not recognize it as an earthquake.
	Standing vehicles may rock slightly. Vibrations similar to the
	passing of a truck. Duration estimated.
IV	During the day, felt indoors by many, outdoors by a few. At night,
	some awakened. Dishes, windows, doors disturbed; walls make
	cracking sound. Sensation like heavy truck striking building.
	Standing vehicles rock noticeably.
V	Felt by nearly everyone; many awakened. Some dishes, windows
	broken. Unstable objects overturned. Pendulum clocks may stop.
VI	Felt by all, many frightened. Some heavy furniture moved. A few
	instances of fallen plaster. Damage slight.
VII	Damage negligible in buildings of good design and construction;
	slight to moderate in well-built ordinary structures; considerable
	damage in poorly built structures. Some chimney broken.
VIII	Damage slight in specially designed structures; considerable damage
	in ordinary substantial buildings, with partial collapse. Damage great
	in poorly built structures. Fallen chimneys, factory stacks, columns,
	monuments, walls. Heavy furniture overturned.
IX	Damage considerable in specially designed structures; well-designed
	frame structures thrown out of plumb. Damage great in substantial
	buildings, with partial collapse. Buildings shifted off foundations.
X	Some well-built wooden structures destroyed; most masonry and
	frame structures with foundations destroyed. Rails bent.
XI	Few, if any, masonry structures remain standing. Bridges destroyed.
	Rail bent greatly.
XII	Damage total. Lines of sight and level are destroyed. Objects thrown
	into air.

Table 2.1. Modified Mercalli intensity scale (Kramer, 19	96)
--	-----

Table 2.2. Rich	ter magnitude s	cale (M _L)	

01-01-01-01-01-01-01-01-01-01-01-01-01-0		
Magnitude M _L	Possible Effects	
0		
1	Normally only detected by instruments	
2		
3	Faint tramor causing little damage	
4	Frank tremor causing fittle damage	
5	Structural damage	
6	Distinct sheking loss well constructed building collense	
7	Distinct shaking, less wen-constructed bunding conapse	
8	Large buildings destroyed	
9	Ground seems to shake	

· · ·	IMPORTANCE TO ²				
Parameter	Mo	odulus	Damping		
(1)	Clean sands (2)	Cohesive soils (3)	Clean sands (4)	Cohesive soils (5)	
Strain Amplitude	v	v	v	v	
Effective Mean Principal Stress	v	v	l v	v v	
Void Ratio	v	v	v	l v	
Number of Cycles of Loading	Rb	R	v	v	
Degree of Saturation	R	v	L	U	
Overconsolidation Ratio	R	L	R	L	
Effective Strength Envelope	L	L	l L	L	
Octahedral Shear Stress	L	L	L	L	
Frequency of Loading (above 0.1 Hz)	R	R	R	L	
Other Time Effects (Thixotropy) Grain Characteristics, Size, Shape,	R	L	R	L	
Gradation, Mineralogy	R	R	R	R	
Soil Structure Volume Change Due to Shear Strain	R	R	R	R	
(for strains less than 0.5 %)	U	R	U	R	

Table 2.3. Parameters affecting shear modulus and damping of soils subjected to dynamic loading (Hardin and Drnevich, 1972a)

² V means Very Important, L means Less Important, and R means Relatively Unimportant except as it may affect another parameter; U means relative importance is not clearly known at this time. ^b Except for saturated clean sand where the number of cycles of loading is a Less

Table 2.4. Values of G_{max} / S_u (Weiler, 1988)

Diagticity Indox	Over	consolidation Ratio,	OCR
Plasticity muex	1	2	5
15-20	1100	900	600
20-25	700	600	500
35-45	450	380	300

Table 2.5. Parameters for Ohsaki and Iwasaki (1973) relationship between SPT N-value and V_s (Sykora, 1987)

		Param	eter	Correlation
Category	Groups	8	b	Coefficient
All data	 .	124	0.78	0.886
Geologic	Tertiary (Pliocene)	57.3	0.97	0.821
age	Diluvial (Pleistocene)	110	0.82	0.812
-	Alluvial (Holocene)	149	0.64	0.786
Soil	Cohesionless	66.3	0.94	0.852
type	Intermediate	121	0.76	0.742
	Cohesive	143	0.71	0.921
G/√σ	Sands			0.742

^DExcept for saturated clean sand where the number of cycles of loading is a Less Important Parameter.

0.719 0.721	on Correlative Parameters SPT N-value SPT N-value Soil Type	Equation No. I
0.719 0.721	SPT N-value SPT N-value Soil Type	1 2
0.721	SPT N-value Soil Type	2
0.784	SPT N-value Geologic Age	3
0.786	SPT N-value Geologic Age Soil Type	4
0.820	SPT N-v <mark>alue</mark> Depth	5
0.826	SPT N-value Depth Soil Type	6
0.848	SPT N-value Depth Geologic Age	7
0.853	SPT N-value Depth Geologic Age Soil Type	8
•	SPT N-value Depth Soil Type SPT N-value Depth Geologic Age SPT N-value Depth Geologic Age Soil Type	6 7 8

Table 2.6. Ohta and Goto (1976) relationships between SPT N-value and V_s (Sykora, 1987)

		Shear Wave Velo	city, fps	Shear Modulu	s, tsf
Category	No. of Data	Best-Fit Relation	Correlation Coefficient	Best-Fit Relation	Correlation Coefficient
Clay fill	63	$V_{\rm s} = 323 {\rm N}^{0.248}$	0.574	$G = 158 N^{0.557}$	0.582
Loam	64	$V_{2} = 430 \text{ N}^{0.153}$	0.314	$G = 229 N^{0.383}$	0.487
Sand fill	81	v = 301 N ^{0.257}	0.647	G = 145 N ^{0.500}	0.606
Tertiary clay and sand	108	v = 358 N ^{0.319}	0.717	$G = 209 N^{0.686}$	0,682
Alluvial clay	325	$v = 351 \text{ N}^{0.274}$	0.721	$G = 180 N^{0.607}$	0.715
Alluvial peat	17	$v = 209 N^{0.453}$	0.771	$G = 55.0 \cdot N^{1.08}$	0.769
Diluvial clay	222	$v = 420 \text{ N}^{0.257}$	0.712	$G = 257 N^{0.555}$	0.712
Alluvial sand	294	$v = 288 N^{0.292}$	0.690	$G = 128 N^{0.611}$	0.871
Diluvial sand	338	$V = 361 \text{ N}^{0.285}$	0.714	$G = 181 N^{0.631}$	0.729
Alluvial gravel	28	V = 247 N ^{0.351}	0.791	$G = 84.5 N^{0.787}$	0.798
Diluvial gravel	114	$v_{\rm s}^{\rm v} = 446 {\rm N}^{\rm 0.246}$	0.550	$G = 326 N^{0.528}$	0.552
All soils	1,654	$v_{\rm s} = 318 \ {\rm N}^{0.314}$	0.868	$G = 147 \text{ N}^{9.680}$	0.867

Table 2.7. Imai and Tonouchi (1982) relationships between SPT N-value and V_s (Sykora, 1987)

* Not adjusted for differences in energy efficiency between United States and Japanese SPT equipment and procedures.

Table 2.8. Effect of environmental and loading conditions on modulus ratio (at given strain level) of normally consolidated and moderately overconsolidated soils (modified from Dobry and Vucetic, 1987)

Increasing Factor	G/G _{max}
Confining pressure, σ_m	Increase with σ_m ; effect decreases with increasing
	PI
Void ratio, <i>e</i>	Increases with <i>e</i>
Geologic age, t_g	May increase with t_g
Cementation, c	May increase with <i>c</i>
Overconsolidation ratio, OCR	Not affected
Plasticity index, PI	Increases with PI
Cyclic strain, γ_c	Decreases with γ_c
Strain rate, $\dot{\gamma}$	G increases with $\dot{\gamma}$, but G/G _{max} probably not
	affected if G and G_{max} are measured at same $\dot{\gamma}$
Number of loading cycles, N	Decreases after N cycles of large γ_c (G _{max} measured
	before N cycles) for clays; for sands, can increase
	(under drained conditions) or decrease (under
	undrained conditions)

Table 2.9. Effect of environmental and loading conditions on damping ratio of normally consolidated and moderately overconsolidated soils (modified from Dobry and Vucetic, 1987)

Increasing Factor	Damping ratio, ξ
Confining pressure, σ_m	Decreases with σ_m ; effect decreases with increasing
	PI
Void ratio, <i>e</i>	Decreases with <i>e</i>
Geologic age, t_g	Decreases with t_g
Cementation, c	May decrease with <i>c</i>
Overconsolidation ratio, OCR	Not affected
Plasticity index, PI	Decreases with PI
Cyclic strain, γ_c	Increases with γ_c
Strain rate, $\dot{\gamma}$	Stays constant or may increase with $\dot{\gamma}$
Number of loading cycles, N	Not significant for moderate γ_c and N

Table 2.10. Magnitude scaling factor (Seed and Idriss, 1982)

Magnitude Scaling Factor (MSF)
1.43
1.32
1.19
1.08
1.00
0.94
0.89

Table 3.1. Maximum estimated M_L for seismic source zones in Thailand region (after Nutalaya et al., 1985 and modified by Warnitchai and Lisantono, 1996)

Zone	Name	Maximum M _L		
A	Arakan Coastal Area	6.75		
В	West-Central Burma Basin	7.40		
C	East-Cebtral Burma Basin	7.75		
D	Bhamo-Paoshan Area	5.96		
Е	Burma Eastern Highlands	7.30		
F	Tenasserim Range	7.90		
G	Northern Thailand	6.50		
Н	H North Indochina			
I South Yunnan-Kwangsi		8.38		
J Andaman Arc		7.20		
K Andaman Basin		6.50		

Table 3.2. Summary of input motions used in analyses

Earthquake	Year	Station	a_{max} (g)	T_p (sec)
Northridge	1994	Topanga	0.33	0.31
El Centro	1940	El Centro	0.34	0.68
Loma Prieta	1989	Yerba Buena Island	0.065	1.41



Table 6.1. Physical properties of sand used in cyclic triaxial test (Gauchan, 1984)

Gs	D ₅₀ (mm)	Cu	e _{max}	e _{min}	$\gamma_{\rm max}~(t/m^3)$	$\gamma_{\rm min}~(t/m^3)$
2.64	0.72	3.52	0.88	0.54	1.71	1.40

 Table 6.2. Model parameters (Iai et al., 1993)

Parameter	Type of Mechanism	Kind of the Parameter
K _{ma}	Elastic volumetric	Rebound modulus
G_{ma}	Elastic shear	Shear modulus
$\phi_{\!_f}$	Plastic shear	Shear resistance angle
ϕ_p	Plastic dilatancy	Phase transformation angle
H_m	Plastic shear	Hysteretic damping factor at large shear strain level
p_1	Plastic dilatancy	Initial phase of dilatancy
<i>p</i> ₂	Plastic dilatancy	Final phase of dilatancy
<i>w</i> ₁	Plastic dilatancy	Overall dilatancy
	Plastic dilatancy	Ultimate limit of dilatancy
<i>c</i> ₁	Plastic dilatancy	Threshold limit

 Table 6.3. Model parameters for dilatancy

Dr (%)	<i>w</i> ₁	<i>p</i> ₁	<i>p</i> ₂		S_{1}
45-50	38.5	0.7	0.4	1.0	0.005
55-65	9.5	0.6	0.5	1.0	0.005
75-85	18.8	0.6	0.8	1.0	0.005

Chiang Mai				Chia	ing Rai		
Site Factor of Safety			Site	Fa	actor of Saf	ety	
no.	$P_L = 5\%$	$P_L = 10\%$	$P_L = 30\%$	no.	$P_L = 5\%$	$P_L = 10\%$	$P_L = 30\%$
1	0.51	0.64	0.95	1	2.74	3.40	5.03
2	0.51	0.63	0.94	2	1.39	1.72	2.55
3	0.66	0.82	1.21	3	2.01	2.50	3.70
4	1.26	1.56	2.31	4	0.84	1.03	1.53
5	1.68	2.08	3.08	5	0.36	0.45	0.67
6	0.57	0.70	1.03	6	0.60	0.74	1.10
7	1.23	1.52	2.25	7	1.00	1.24	1.83
8	0.83	1.02	1.51	8	0.99	1.23	1.82
9	0.64	0.79	1.17	9	1.26	1.57	2.32
10	0.89	1.10	1.63	10	0.77	0.96	1.42
11	1.23	1.52	2.25	11	0.49	0.61	0.90
12	1.80	2.23	3.30	12	1.05	1.30	1.92
13	1.66	2.06	3.06	13	0.86	1.06	1.57
14	0.69	0.87	1.27	14	0.50	0.63	0.93
15	0.87	1.08	1.60	15	1.05	1.30	1.93
16	0.50	0.62	0.92	16	10.24	12.72	18.81
17	0.52	0.65	0.96	17	0.78	0.98	1.44
18	0.58	0.71	1.06				
19	1.42	1.77	2.61				
20	0.81	1.00	1.48				
21	1.09	1.35	1.99				
22	1.03	1.29	1.91				
23	1.95	2.43	3.59] []]			
24	0.85	1.05	1.55				
25	0.65	0.82	1.20	200			
26	2.04	2.53	3.74				
27 9	1.06	1.31	1.93				
28	0.45	0.56	0.83				
29	0.86	1.07	1.58				

 Table 6.4.
 Summary of the estimated values of factor of safety based on the procedure proposed by Seed et al. (1983)

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

FIGURES



Fig. 1.1. Evidence indicating the occurrence of liquefaction in the northern area of Thailand



Fig. 1.2. General study methodology adopted



Fig. 1.3. Example of deterministic method of liquefaction evaluation derived from empirical data for soils with $D_{50} > 0.25$ mm (Seed et al., 1983)



Fig. 2.1. Earthquake terminology (Lindeburg, 1998)



Fig. 2.2. Types of seismic waves: (a) P-wave; (b) S-wave; (c) R-wave; (d) L-wave (Kramer, 1996)



Fig. 2.3. Typical seismometer amplitude trace (Lindeburg, 1998)



Fig. 2.4. Richter magnitude correction nomograph (Lindeburg, 1998)



Fig. 2.5. Two hypothetical Fourier amplitude spectra with the same predominant period but very different frequency contents. The upper curve describes a wideband motion and the lower a narrowband motion



Fig. 2.6. Variation of k-coefficient with plasiticity index (Dickenson, 1994)



Fig. 2.7. Backbone curve showing typical variation of G_{sec} with shear strain



Fig. 2.8. Variation of the modulus ratio with shear strain



Cyclic shear strain, γ_c (%)

Fig. 2.9. Modulus reduction curves for fine-grained soils of different plasticity (Vucetic and Dobry, 1991)



Fig. 2.10. Influence of mean effective confining pressure on modulus reduction curves for (a) nonplastic (PI = 0) soil, and (b) plastic (PI = 50) soil (Ishibashi, 1992)



Fig. 2.11. Variation of damping ratio of fine-grained soil with cyclic shear strain amplitude and plasticity index (Vucetic and Dobry, 1991)



Fig. 2.12. Ground response nomenclature: (a) soil overlying bedrock; (b) no soil overlying bedrock



Fig. 2.13. Relationship between hysteresis loop and: (a) shear modulus; (b) damping ratio



Fig. 2.14. Iteration toward strain-compatible shear modulus and damping ratio in equivalent linear analysis



Fig. 2.15. Kanai's amplification factor for soft ground (Kanai, 1957)



Fig. 2.16. Approximate relationships between peak accelerations on rock and other local site conditions (Seed et al., 1976)



Fig. 2.17. Approximate relationship between peak accelerations on rock and soft soil sites (Idriss, 1990)



Fig. 2.18. Average normalized response spectra (5% damping) for different local site conditions (Seed et al., 1976)



Fig. 2.19. Relationship between limiting epicentral distance of sites at which liquefaction has been observed and moment magnitude for shallow earthquakes. Deep earthquakes (focal depths > 50 km) have produced liquefaction at greater distances. (Ambraseys, 1988)



Fig. 2.20. Cyclic shear stresses on soil element during ground shaking: (a) idealized field loading conditions; (b) shear stress variation determined by response analysis


Fig. 2.21. Lateral spreading adjacent to a river channel: (a) before earthquake; (b) after earthquake (Youd, 1984)



Fig. 2.22. Examples of flow failure caused by liquefaction and loss of strength of soils lying on a steep slope



Fig. 2.23. Formation of water interlayers in shaking table tests of Liu and Qiao, (1984)



Fig. 2.24. Ground oscillation: (a) before earthquake; (b) after earthquake (Youd, 1984)



Fig. 2.25. Example of structure tiled due to loss of bearing strength. Liquefaction weakens the soil reducing foundation support which allows heavy structures to settle and tip.



Fig. 2.26. Tilting of apartment buildings, Niigata (1964)



Fig. 2.27. Procedure for determining maximum shear stress, $(\tau_{\text{max}})_r$ (Seed and Idriss, 1982)



Fig. 2.28. r_d versus depth curves developed by Seed and Idriss (1971) with add mean-value lines plotted from Eq. (2.34) (Youd and Idriss, 2001)



Fig. 2.29. Time history of shear stresses during earthquake (Seed and Idriss, 1982)



Fig. 2.30. Plot used to determine the cyclic resistance ratio for clean and silty sands for M = 7.5 earthquakes (Seed et al., 1985)



Fig. 2.31. Approximate relationships between the moment magnitude scale M_w and other magnitude scales. Shown are the short-period body wave magnitude scale m_b , the local magnitude scale M_L , the long-period body wave magnitude scale m_B , the Japan Meteorological Agency magnitude scale M_{JMA} , and the surface-wave magnitude scale M_S (reproduced from Day, 2002)



Fig. 2.32. Variation of correction factor, K_{α} , with initial shear/normal stress ratio (Seed and Harder, 1990)



Fig. 2.33. Variation of correction factor, K_{σ} , with effective overburden pressure (Seed and Harder, 1990)



Fig. 3.1. Location of the study area



Fig. 3.2. Seismic source zones in Thailand and vicinity (Nutalaya et al., 1985)



Fig. 3.3. Map showing contours of peak ground acceleration (in units of acceleration of gravity) with 10% chance of being exceeded in a 50-year exposure time, and seismic zones for earthquake-resistant design (Wanitchai and Lisantono, 1996)



Fig. 3.4. Typical subsoil section in Chiang Mai Province (modified from Anantasech and Thanadpipat, 1985)

111 Top Soil, Thickness: 0 ~ 1.0 m

 $\overline{}$

Soft to Medium clay / Loose to medium dense sand Thickness: 1.5 ~ 8 m

Medium to stiff clay / medium dense to dense sand Thickness: 4.0 ~ 12 m

Very stiff clay / Very dense sand

Fig. 3.5. Typical subsoil profile in Chiang Rai province



Fig. 3.6. Examples of the soil profiles and soil properties collected from Chiang Mai



Fig. 3.7. Examples of the soil profiles and soil properties collected from Chiang Rai



Fig. 3.8. Grain size distribution of sands: (a) Chiang Mai; (b) Chiang Rai



Fig. 3.9. Acceleration time history of input motions



Fig. 3.10. Acceleration response spectra of input motions used in this study

จฬาลงกรณ์มหาวิทย



Fig. 4.1. One-dimensional idealization of a horizontally layered soil deposit over a uniform half-space (Idriss and Sun, 1992)



Fig. 4.2. Shear modulus and damping variation with shear strain for clays of varying plasticity (Vucetic and Dobry, 1991)



Fig. 4.3. Shear modulus and damping variation with shear strain for sands (Seed et al., 1984)



Fig. 4.4. Variations of amplification factors with base rock acceleration in Chiang Mai



Fig. 4.5. Variations of amplification factors with base rock acceleration in Chiang Rai



Fig. 4.6. Comparison between peak ground accelerations of Chiang Mai and observed results from 1985 Mexico and 1989 Loma Preita earthquakes (modified from Idriss, 1991)



Fig. 4.7. Comparison between peak ground accelerations of Chiang Rai and observed results from 1985 Mexico and 1989 Loma Preita earthquakes (modified from Idriss, 1991)



Fig. 5.1. Example of deterministic method of liquefaction evaluation derived from empirical data for soils with $D_{50} > 0.25$ mm (Seed et al., 1983)

CASE	BCAT	LIGI	LIQ2	н	H	EP	DER	DUR		CSR	CSRN	ZV	ZL	SICT	SIGE	N	N1	CE	N160	FC
0101	2	٥.	-t.	6.6	-1.	-1.	39.	20.	. 12	.14	-1.	. 9	6.1	-1.	. 68	6.	8.0	~1 .	-1.	-1.
0102	2	٥.	-1.	6.5	-1.	-1.	39.	20.	.12	. 14	-1.	.9	6.1	1.07	.65	12.	16.0	~1.	-1.	-1.
0201	2	ο.	-1.	6.1	-1.	-1.	47.	12.	.08	. 09	-1.	.9	8.1	-1.	. 55	6.	8.0	-1.	-1.	-1.
0202	2	0.	-1.	6.1	-1.	-1.	47.	12.	.08	.09	-1.	.9	6.1	-1.	. 55	12.	18.0	-1.	-1.	-1.
0301	2	1.	-1.	6.4	-1.	-1.	32.	76.	.35	.39	-1.	.9	13.7	-1.	1.30	17.	15.0	-1.	-1.	-1.
0301	4	1	1.0	7.9	-1.	30.	-1:	-1:	-1.	-2.	.364	.8	10.0	1.90	-1.	-2. 20.	20.2	1:	-1	-2.
0301	8	1.	-1.	7.9	-1	-1.	-1.	-1.	.32	.355	.375	.8	10.0	1.93	1.02	20.	19.5	1.30	25.5	0.
0302	24	1:	-1;	8.4 -2.	-1:	30.	32.	-1.	-1.	-1.	-1	2.0	9.1	-1.	-1.00	-2.	10.0		-1:	-2:
0302	8	1:	-1.0	7.9	-1:	-1:	-1:	-1:	.32	.31	.33	2.0	7:0	1.35	.86	15.	10.6	1,17	12.6	6 .
0303	2	8.	-1:	8.4	-1:	-1.	32.	75.	.35	.35	-1:	1.9	7.6	-i:	.85	19.	21.0 -1.	1:	-1:	-1:
0303	78	i:	1.0	7.9	-1:	-1 :	-1:	-1:	.28	.316	.275	1.9	6.0	1:14	.73	17. 17.	20.2 12.0	1,30	26.¢	3. 3
0304	2	1.	-1.	8.4	-1-	-1.	32.	75.	.35	.36	-1-	2.4	6.1	<u>-</u> i.	.75	16.	19.0	-1-	-1.	-1.
0304	7	i	1.0 -1.	7.9	-1:	-1.	-1	-1:	.28	.275	.246	2.4	5.0	1.00	.69	13. 13.	15.9 15.0	1.17	17.0	4
0305	3	ο.	-1.	8.4	-1.	-1.	-1.	-1.	.35	-1.	-1.	1.8	12.2	-1.	-1.	-i.	48.0	÷i.	-1.	-1.
0308	8	1.	-1.	B.4	-1.	55.	-1.	-1.	-1.	-1.	-1.	-1.	+1.	-1.	.74	-1.	16.4	-1.	-1.	-1.
0307	8	1.	-1.	8.4	-1.	51.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.45	-1.	20.3	-1.	-1.	-1.
						-		10.1	31											
0401		1.	-1.	7.5	-1.	79.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.02	-1.	10.0	-1.	-1.	-1.
0402		1.	-1.	7.5	-1.	60.	-1.	-1.	-1.	-1	-1	-1.	-1.	-1.	. 64	-1.	12.1	-1.	-1.	-1.
0404	Å	1.	-1	7.5	-1	37	-1	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.39	-1.	4.0	-1.	-1.	-1.
0405		1.	-1.	7.5	-1.	13.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.47	-1.	7.5	-1.	-1.	-1.
0501	3	1.	-1.	8.3	<u>-1</u> .	-1.	-1.	-1.	-1.	-1.	-1.	2.4	4.6	:ŀ	.60	16.	20.4	1.	-1.	-1.
0501	•	1.	-1.	-2.	26.	Ĵ	-1.	-1.	-1.	-1	-1	2.4	7.6	-1	-1.	16.	17.4	-1.	-1.	-1
0502	Ă	i :	-11	-2.	26.	-i:	- i :	-i:	-1.	-1.	-1 i	2.4	7.6	-1	-1.	16.	-1.	-1.	-i.	-1.
0503	3	1.	-1.	8.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	4.9	-1.	. 98	62.	62.2	-1.	-1.	-1.
0504 0504	3	1:	-1:	8.3	-1:	-1:	-1:	-1: -1:	-1:	-1:	-i:	1.5 1.5	4.6	-1:	-1.	24. 24.	33.3 -1.	-1.	-1:	-1:

Fig. 5.2. Example of liquefaction data catalog (Liao and Whitman, 1986a)



Fig. 5.3. Contours of equal probability of liquefaction (P_L) for magnitude, M = 7.5 with deterministic line by Seed et al. (1985)



Fig. 5.4. Contours of equal probability of liquefaction (P_L) for magnitude, M = 5.5 with deterministic line modified from Seed et al. (1985)



Fig. 6.1. Flow diagram for evaluation of pore water pressure generation



Fig. 6.2. Schematic figure for multiple simple shear mechanisms (pairs of circles indicate mobilized virtual shear strain in positive and negative modes of compression shear and simple shear) (Iai et al., 1992)



Fig. 6.3. Schematic figure of contact normal n_k , tangential direction t_k and contact force increment dP_k (Iai et al., 1993)



Fig. 6.4. Schematic figure of liquefaction front, state variable S and shear stress ratio r (Iai et al., 1992)



Fig. 6.5. Relationship between normalized plastic shear work w and liquefaction front parameter S_0 (Iai et al., 1992)



Fig. 6.6. Grain size distribution of sands: (a) Chiang Mai; (b) Chiang Rai



Fig. 6.7. Liquefaction resistance curves from undrained cyclic triaxial test (Gauchan, 1984)



Fig. 6.8. Test and computed liquefaction resistance curves

จุฬาลงกรณ์มหาวิทยาลัย



Fig. 6.9. Relationship between amplification factor and maximum pore water pressure ratio for Chiang Mai



Fig. 6.9. Relationship between amplification factor and maximum pore water pressure ratio for Chiang Mai (cont.)



Fig. 6.10. Relationship between amplification factor and maximum pore water pressure ratio for Chiang Rai


Fig. 6.10. Relationship between amplification factor and maximum pore water pressure ratio for Chiang Rai (cont.)



Fig. 6.11. Relationship between cyclic stress ratio for earthquake magnitude of 5.5 and maximum pore water pressure ratio for Chiang Mai



Fig. 6.12. Relationship between cyclic stress ratio for earthquake magnitude of 5.5 and maximum pore water pressure ratio for Chiang Rai



Fig. 6.13. Relationship between maximum ground acceleration and maximum pore water pressure ratio for Chiang Mai



Fig. 6.14. Relationship between maximum ground acceleration and maximum pore water pressure ratio for Chiang Rai



Fig. 6.15. Relationship between maximum ground acceleration and maximum pore water pressure ratio at GL -2.5 m for Chiang Mai



Fig. 6.16. Relationship between maximum ground acceleration and maximum pore water pressure ratio at GL -2.5 m for Chiang Rai

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

APPENDICES





APPENDIX A LIQUEFACTION DATA CATALOG

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

					0	ata	From	Previo	ously	Publ	ished	Cata	logs							
CASE	BCAT	LIQI	L192	M	H	EP	DER	DUR		CSR	CSRN	ZW	ZL	SICT	SIGE	NN	NI	CE	N160	FC
0101	2	٥.	-t.	6.6	-1.	-1.	39.	20.	.12	.14	-1.	. 9	6.1	-1.	. 65	6.	8.0	~1.	-1.	-1.
0102	2	ο.	-1.	6.6	-1.	-1.	39.	20.	.12	. 14	-1.	.9	6.1	1.07	. 65	12.	16.0	~1 .	-1.	-1.
			'																	
0201	2	٥.	-1.	6.1	-1.	-1.	47.	12.	.08	. 09	-1.	.9	8.1	-1.	. 55	6.	8.0	~1 .	-1.	-1.
0202	2	0.	-1.	6.1	-1.	-1.	47.	12.	.08	.09	-1.	.9	6.1	-1.	. 55	12.	18.0	-1.	-1.	-1.
0301	2	1.	-1.	8.4	-1.	-1.	32.	76.	.35	.39	-1.	.9	13.7	-1.	1.30	17.	15.0	1	-1.	-1.
0301	7	11	1,0	7.8	-1	-1.	-1	-1	.32	-1.	.364	.8	10.0	1.90	.98	20.	20.2	1.30	-1	õ.
0302	2	1.	-1.	8.4	-1.	-1	32.	75.	.35	.37	-1.	1.8	9.1	-1.	1.00	10.	10.0	~1.	-1.	-1.
0302	4	1	-1	-2.7.9	-1.	30.	-1	-1:	-1.	-1:	.317	2.0	7.0	1.33	-1.	10	11.i	-1	-1	-2.
0302	8	î.	-1.	7.9	-1.	-1.	-1.	-1.	.32	.31	.33	2.0	7.0	1.35	.86	19.	10.6	1.17	12.5	δ.
0303	24	8.	-1 -1:	8.4	-1:	30	32.	75.	.35	.35	-1;	1.9	7.6	-1:	.85	19.	21.0	1:	-1:	-1:
0303	78	1:	1.0 -1.	7.9	-1:	-1	-1:	-1:	.32	.316	.33	1.9	6.0	1:17	.73	17:	12.0	1.30	26.¢	3.
0304	2	1.	-1.	8.4	-1.	-1.	32.	75.	.35	.36	-1.	2.4	6.1	-1.	. 75	ſ٥.	19.0	<u>-1</u> .	-1.	-1.
0304	7	1:	1,0	7 9	-1:	-1.	-1:	-1:	.28	-1.	.246	2.4	5.0		. 69	13.	15.9	-1,	17	1
0305	•	1.	-1.	P.4	-1	-1	-1	-1.	.35	-1.	-1	1.8	12.2	-1.	-1.	-1.	48.0	-1	-1.	-1.
0308		1	-1.	8.4	-1	55	-1	-1.	-1.	-1	-1	-1.	-1.	-1.	.74	-1.	15.4	-1	-1.	-1.
0307	Ă	1	-1	8.4	-1.	51	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.45	-1.	20.3	-1.	-1.	-1.
0001	•	•																		
0401	6	1.	-1.	7.5	-1.	79.	-1.	-i.	-1.	-1.	-1.	-1.	-1.	-1.	.62	-1.	10.5	-i.	-1.	-1.
0402		1.	-1.	7.6	-1.	65.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.37	-1.	10.7	-1.	-i.	-1.
0403	6	e.	-1.	7.5	-1.	58.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.64	-1.	12.1	-1.	-1.	-1.
0404	8	1.	-1.	7.5	-1.	37.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.39	-1.	4.0	-1.	-1.	-1.
0405	8	1.	-1.	7.5	-1.	13.	-1.	-1.	-1.	-1.	-1,	-1.	-1.	-1.	.47	-1.	7.5	-1.	-1.	-1.
0501	3	1.	-1-	8.3	-1.	-1:	-1 ·	-1:	-1:	-1:	-1:	2.4	4.6	-1:	.60	16.	20.4	1:	-1.	-1
0502	3	1.	-1.	8.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	7.6	-1.	. 85	16.	17.4	-1.	-1.	-1.
0502	Ă	11	-1.	-2.	26.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	7.6	-1.	-1.	16.	-1.	-1.	-1.	-1.
0503	3	1.	-1.	8.3	-1.	-1.	-1·	-1.	*1.	-1.	-1.	2.4	4.9	-1.	. 98	62.	52.2	-1.	-1.	-1.
0504 0504	34	1:	-1: -1:	8.3	-1:	1	-1:	_ 1 :	-1:	-1.	-1;	1.5	4.6	-1:	.52	24.	33.3	1	-1:	-1:

CASE	BCAT	LIQI	LIQ2	H	Ħ	EP	DER	DUR		CSR	CSRN	ZW	ŽL.	SICT	SIGE	. N	N1	CE	N160	FC
0505	6	1.	-1.	8.3	-1.	48.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	. 48	-1.	8.7	-1.	-1.	-i.
0505	8	1.	-1.	8.3	-1.	48.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	- 1 ¹ .	. 69	-1.	5.5	-1.	-1.	~1.
0506	8	1.	-1.	8.3	-1.	184.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.43	-1.	10.3	-1.	-1.	~1.
0601	6	0.	-1.	6.9	-1.	51.	-1.	-1.	-1.	-1.	-1,	-1.	-1.	-1.	.74	-1.	15.4	-1.	-1.	-1.
0802	6	٥.	-1.	8.9	-1.	57.	-1.	-1.	-1.	-1.	-1,	-1.	-1.	-1.	. 45	-1.	20.3	-1.	-1.	~1 .
0701 0701	8	1:	0.7 -1.	7.9	-1:	-1:	-1:	-1:	.20	.163	.17	4.0	7.0	1.25	. 95	10.	10.3 10.0	1.09	11.0	10.
0702	7	1.	0.7	7.9	-1.	-1.	-1.	-1.	.20	-1.	. 168	4.0	8.0	1.44	1.04	1.	1.0	-1.	-1.	14.
0703	7	1.	0,7	7.9	-1.	-1.	-1.	-1-	.20	-1.	.130	4.0	4.3	.73	.71	2,2	2.7	, -1.	ā ¹ ė	22.
0703	7	•	-1.	7.9	-1.	-1.	-1.	-1.	20	-1	225	4.0	9.0	1 52	. 13	18.5	18.5	-1	-1	22.
0704	á	11	-i.	7.9	-1.	-1:	-11	-i:	.20	.225	.24	1.0	8.0	1.68	.85	18.	17.0	1.21	20.5	1:
0705 0705	78	1	0.7 -1.	7:8	-1:	-1:	-1:	-1:	.20	-1.	.187	1.0	5.0	.95 1.00	.65 .57	11.9 12.	15.0 15.5	1.09	16.5	δ. δ.
0708	7	0 .	0.5	7.9	-1.	-1.	-1.	-1.	.20	-1.	.162	3.0	5.0	. 95	.75	5.7	9 .7	-1.	-1,	20.
0705	в 7	0.	.6 0 E	7.9	-1.	-1.	-1.	-1.	.20	.10	18:	3.0	8.0	1.00	.78	o.	2.0	-1	-1	20.
0708		v. n	-1	7.9	-1.	114	-1	-1	-1	-1.	-1	-1	-1	-1	62	-1	10.6	-1	-1.	-1
0709	5	o.	-1.	7.9	-1.	102	-1.	-1.	-1.	-1.	-1	-1.	-1.	-1.	.37	-1.	10.7	-1.	-1.	-1.
0710	6	0.	-1.	7.9	-1.	105.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	. 54	-1.	12.1	-1.	-1.	-1.
0711	8	1.	-1.	7.9	-1.	96.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.39	-1.	4.0	-1.	-1.	-1.
0712	8	1.	-1.	7.9	-1.	77.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.47	-1.	7.5	-1.	-1.	-1.
0713	6	1.	-1.	7.9	-1.	72.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.42	-1.	6.5	-i.	-1.	-1.
0714	8	1.	-t.	7.9	-1.	75.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	. 47	-1.	15.1	-1.	-1.	-1.
										•										
0801	26	1:	-1:	6.3	-1:	$\overline{1}$	11. -1.	15. 15.	.20	.18	-1: -1:	4.6	7.6	-1:	-1:	-1.	-1:	-1:	-1: -1:	-1:
0801	6	1.	1.0	ē.3	-1.	11.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.06	-1.	2.9	-1.	-1.	-1.
	_			λI.		1.1	Ц		1/ .	Έ.	1.									
0901	6	1.	-1.	7.0	-1.	20.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	0.62	-1.	10.5	-1.	-1.	-1.
0902		1.	-1.	7.0	-1.	19.	-1.	-10	-1.	1	-4.	-1.	-1.	-1.	.37	-1	10.7	-1.	-1.	-1.
0004			-1	7.0	3	54.	5	-1	<u> </u>	127	-1	-1	-1	1.1	30	-1	4.0	-1	-1	-1.
0005			1	7.0	_1.	61. RE	01	-1		-1	-1	0	-1	-1		្នា	7.5	-1	-1	-1.
0.000	•	φ.	- 1 .	1.0	-1.	00.			•••	•••	••	• ·				• •		• •	•••	

CASE	BCAT	LIQI	L192	M	R	EP	DER	DUR		CSR	CSRN	ZV	ZL	SIGT	SIGE	N	N1	CE	N160	FC
1001	8	٥.	-1.	6.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	-3.	-1.	-2.	-2.	-2.	-1.	-1 -	-1.
1002	3	٥.	-1.	6.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	5.5	-2.	-1.	-2.	-2.	-2.	-1.	-1 -	-1.
1003	8	۰.	-1.	5.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	-3.	-1.	-2.	-2.	-2.	-1.	-1.	-1.
1004	3	٥.	-1.	6.3	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	-2.	-1.	-2.	-2.	-2.	-1.	-1 -	-1.
1005 1005	7 8	8:	-1:	6.3 6.3	-1: -1:	-1:	-1:	-1:	.20	.205	.163	-1. 1.8	-1. 8.2	1.5i	-1. .90	-1. 8.	7.0 8.0	1.00^{-1}	-1. 8,0	-1. 25.
1008	8	Ø.	-1.	8.3	-1.	-1.	-1.	-1.	.18	. 155	. 125	1.8	8.1	1.11	.71	7.	8.5	3.00	9.5	2.
1101 1101 1101	2 6	1. 1. 1.	-1. -1. -1.	7.0 -2 7.0		-1. -1. 8.	8. -1. -1.	30. -1. -1.	.25	.155	-1: -1: -1:	4.5	4.6	-1: -1: -1:	-1. -1. .77	-1. 9. -1.	-1. -1. 9.8	-1 -1 -1	-1. -1. -1.	-1: -1:
1102 1102 1102	256	1:	-1 -1 -1	7.0 7.0 7.0		-1. -1. 8.	8. -1. -1.	30. -1. -1.	.25	.155 -1. -1.	-1: -1: -1:	6.1 6.0 -1.	7.6	-1: -1: -1:	-1. -1. 1.17	-1. -1.	-1. -1. 3.8	-1 -1 -1	-1: -1:	-1. -1. -1.
1103 1103 1103	256	1.	-1. -1. -1.	7.0 7.0 7.0		-1. -1. 8.	8. -1. -1.	30. -1. -1.	.25	.28 -1. -1.	=1:	1.5 1.5 -1.	6.1 6.0 -1.	-1. -1. -1.	-1. -1. .69	-i. i. -1.	$\frac{-1}{-1}$, 1,1	-1. -1. -1.	-1: -1:	-1. -1. -1.
1201 1201 1201 1201	2478	1.	-1. 1.0 -1.	9.3 -2. 8.0 8.0	11111	-1. 165. -1. -1.	161.	70. -1. -1.	.03	.08 -1. -1. .18	-1. -1. .193 .195	1.5	4.0 -2. 5.0 5.2	-L. -1. .91 1.00	.50 -1. .61 .69	4 8. 8.	6,0 10,4 9,5	-1 -1. -1. 1.17	-1. -1. -1. 11.0	-1. -2. 10.
1202 1202 1202 1202	2 4 7 8	1. 1. 1.	-1. 1.0 -1.	8.3 -2. 8.0 8.0		-1. 165. -1. -1.	161. -1. -1. -1.	70. -1. -1. -1.	.08	.09 -1. -1. 225	-1. -1. .243 .24		2.4 -2. 3.5 3.7	-1.	.25 -1. .36 .40	-2. 1. 1.	2.0 -1 1.6 1.5	-1: -1: -1: 1.17	-1. -1. -1. 1.5	-1 -2 -1 27
1203 1203 1203	4 7 8). 1. 1.	ī10 -1.	8.0 8.0	1.	165. -1. -1.	-1. -1:	-1. -1. -1.	-1. .20 .20	-1. -1. .145	-1. .148 .166	2.5	-2. 3.0 3.0	-1. .51 .55	-1. .46 .49	-2.	2.5	-1. -1. 1.17	-1. -1. 3.0	-2. 30. 30.
1204 1204	78	1:	<u>1</u> .0 -1.	8.0		-1:	-1:	-1:	.18 .18	. 155	.161 .165	2.0 2.1	7.0	1.33	.83 .85	10. 10.	11.1 10.5	1.17	12.5	5. 5.
1301	7	ø.	1.0	7.3	-1.	-2.	-2.	-2.	.35	-1.	.392	.8	4.0	. 88	. 35	7.	11.2	-1.	-2.	38.
1302 1302 1302 1302	2 4 7 8	0. 0. 0.	-1. -1. 0.0 -1.	7.2 -2. 7.3 7.3		-1. 5. -1. -1.	6. -1. -1.	30. -1. -1. -1.	.30 -1. .35 .36	.32 -1. -1. .395	-1. -1. .388 .385	.8.9	7.0 -2. 8.0 8.2	-1. -1. 1.44 1.57	.70 -1. .72 .83	28. -2. 29. 29.	34.0 -1. 34.7 31.5	-1. -1. -1. 1.30	-1. -1. -1. 40,6	-1.
1303 1303 1303 1303	2478		-1. -1. 1.0 -1.	7.2 -2. 7.3 7.3	4444	-1. 5. -1. -1.	6. -1. -1.	30. -1. -1. -1.	.30 -1. .35 .35	.30 -1. -1. .29	-1. -1. .267 .28	8.4 3.7 3.7 3.7	7.0 -2. 7.0 7.0	-1. -1. 1.26 1.35	.95 -1. .93 1.01	18. -2. 19.	18.0 -1. 19.8 18.5	-1. -1. -1. 1.30	-1: -1: 24.0	-1:
1304 1304 1304 1304	2478	1:		7.9 -2. 7.3 7.3		-1. -1. -1.	6. -1. -1.	30. -1. -1. -1.	.30 -1. .40	.29 -1. -1. .395	-1. -1. .375 .38	1.2	3.0 -2. 4.0 4.0	-1 -1. 76 76	.38 -1. .48 .49	3. 2. 8. 8.	5.0 -1. 11.6 11.0	-1. -1. -1. 1.17	-1. -1. 13.0	-1. -2. 0.

CASE	BCAT	LIQI	1192	<u> </u>		P	DER	DUR		CSR	CSRN	ZW	ZL	81GT	81GE	<u> </u>	N1	œ	N160	FC
1305 1305 1305	247	1. 1. 0.	-1. -1. 0.5	7.2 -2. 7.3	-1. -1. -1.	-1. 5. -1.	-1. -1.	30. -1. -1.	.30 -1. .40	.33 -1. -1.	-1. -1. 450		6.1 -2. 6.0	-1. -1. 1.02	.60 -1. .52	-2. 8.	7.0 11.1	-1. -1. -1.	-1. -1.	-1. -2. 21.
1306 1306	7 6	8: 8:	0.5 -1.	7.3 7.3	-1 :	-1: -1:	-1:	-1:	:40 :40	.40	.451 .45	.9	7.5	1.31 1.42	. 65 . 75	20. 20.	25.2 22.5	1.30	29.0	0. 0.
1401	3	0.	-1.	5.4	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	17.1	-1.	1.50	22.	17.9	-1.	-1.	-1.
1601	3	٥.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	4.6	6.1	-1.	. 94	4.	4.1	-1.	-1.	-1.
1502 1502 1502	2 7 8	1. 1. 1.	-1 1 0 -1	5.5 5.5 5.3	-1: -1: -1:	$\stackrel{=1}{\stackrel{=1}{\overset{=1}{$	-1. -1.	18. -1. -1.	.18 .19 .19	.13 .10 .13	-1. -1. .085	2.4 -1. 2.4	3.0 -1. 3.0	-1. .60 .52	.50 .49 .46	7. 6. 5.	10.0 6.1 7.0	-1. -1. .75	-1. -1. 5.5	-1. 3. 3.
1503	8	٥.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.7	4.0	-1.	.68	14.	16.9	-1.	-1.	-1.
1504	3	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	4.6	-1.	.60	16.	20.6	-1.	-1.	-1.
1606	3	0.	-1.	5.6	-1.	~1.	-1.	-1.	-1.	-1.	-1.	2.4	7.6	-1.	. 85	16.	17.4	-1.	-1.	-1.
1508	/ 3	٥.	-1.	6.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.5	8.1	-1.	. 96	52.	62.E	-1.	-1.	-1.
1507	ំង	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.5	4.6	-1.	. 62	24.	33.3	-1.	-1.	-1.
1508	3	G.	-1.	8.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.5	6.1	-1.	. 63	6.	7.5	-1.	-1.	-1.
1509	\$	9 .	-3.	8.5	-1,	-1.	-1.	-1.	-1.	-1.	-1-	4.6	6.1	-1.	.94	20.	20.6	-1.	-1.	-1.
1510		0.	-1.	5.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	4.6	-1.	. 60	20.	25.7	-1.	-1.	-1.
1511	3	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.9	1.2	-1.	. 19	4.	9.2	-1.	-1.	-1.
1512	3	0.	-1.	6.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.		1.2	-1.	. 19	6.	13.9	-1.	-1.	-1.
1979		0.	-1.	0.0	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.2	3.7	-1.	. 41	5.	9.4	-1.	-1.	-1.
1014		v.	-1.		-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.2	4.3	-1.	.40		8.8	-1.	-1.	-1.
1010		v.	-1.	0.0 E E	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	3.7	-1.	.00		14.2	-1.	-1.	-1.
1617		•.	-1.	5.5	-1	-1.	-1.	-1.	-1.	-1	-1	3.0	5.9	-1	.04	10.	12.7	-1.	-1.	-1.
1518		0	-1	5.5	-1	-1	-1	-1.	-1	-1	-1	1.0	5.8	-1	58	18	20.0	-1	-1	-1.
1519	·	0.	-1.	6.6	-1	-1	-1.	-1	-1	-1.	-1.	1.6	3.7	-1.	.48	10	14.5	-1	-1	-1
1520	3	0.	-1	5.5	-1	-1.	-1.	-1	-1.	-1.	-1.	2.4	7.6	-1.	.85	3.	3.3	-1.	-1.	-1.
1521	3	ο.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	9.1	-1.	.97	Б.	5.1	-1.	-1.	-1.
1522	3	٥.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	5.8	-1.	.64	7.	8.7	-1.	-1.	-1.
1523	3	0.	-t.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	5.8	-1.	.64	δ.	6.2	-1.	-1.	-1.
1624	3	0.	-1.	6.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	3.0	-1.	.43	9.	13.8	-1.	-1.	-1.
1525	8	٥.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	4.8	-1.	. 65	Б.	6.8	-1.	-1.	-1.
1528	3	٥.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	4.8	-1.	. 60	8.	11.3	-1.	-1.	-1.

CASE	BCAT	L101	1102	<u> </u>	H	EP	DER	DUR	<u>^</u> -	CSR	CERN	ZV	ZL	SIGI	SICE	N	N1	Œ	N160	FC
1527	3	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	4.5	-1.	- 50	δ.	7.1	-1.	-1.	-1.
1528	3	0.	-1.	б.б	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	4.6	-1.	- 50	16.	21.3	-1.	-1.	-1.
1529	3	٥.	-1.	6.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.8	3.0	-1.	- 43	3.	4.6	-1.	-1.	-1.
1630	3	0.	-1.	6.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.1	2.7	-1.	.43	7.	10.5	-1.	-1.	~1.
1631	3	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	.6	1.5	-1.	. 18	13.	30.6	-1.	-1.	-1.
1533	3	0.	-1.	5.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.2	4.3	-1.	- 46	7.	10.3	-1.	-1.	-1.
1534	3	0.	-1.	5.5	-1	-1	-1	_1	-1	-1	-1	1.2	3.7	-1.	-41	ю. 10	7.8	-1.	-1.	-1.
	•	•.	•••	0.0	•.	•	•••		-	•••	•••	1.4	3.1	-1.	. 30	12.	20.2	-1.	-1.	-1.
1601	2	1.	-1.	8.4	-1.	-1.	113.	75.	. 15	.15	-1.	3.7	4.8	-1.	.76	6.	7.0	-1.	-1.	-1.
1502	2	1.	-1.	8.4	-1.	-1.	113.	76.	.15	.15	-1.	3.7	4.6	-1.	.75	8.	10.0	-1.	-1.	-1.
1603	2	ο.	-1.	8.4	-1.	-1.	113.	75.	.15	. 15	-1.	3.7	6.1	-1.	.90	15.	19.0	-1.	-1.	-1.
1604 1604	1 5	0. 0.	-1: -1:	-1. 8.4	-1;	-1:	113	-1:	.15	.135	-1:	3.7	7.0	-1:	-1:	10. 10.	-1: -1:	-1:	-1:	-1:
1605 1605	1 5	0 .	-1: -1:	-1. 8.4	- 1:	-1 :	113.	-1:	.15	.140	-1 :	3.7 3.5	7.9 8.0	-1: -1:	71:	35. 35.	-1 :	-1:	-1: -1:	-1:
1701 1701 1701	247	1.	-1. -1. 1.0	8.3 -2. 8.3	-1 -2 -1	142. -1.	97. -1. -1.	180. -1. -1.	.15 -1. .15	.18 -1. -1.	-1. -1. .210	.0	5.1 -2. 6.0	-1. -1. 1.14	- 55 - 1 - 54	-2. 5.	7.0 -1 4.9	-1 -1 -1	-1:	-1. 10. 40.
1702 1702 1702	2 4 7	1.	-1. -1. 1.0	8,3 -2, 8,3	-1 -2 -1	142.	97. -1. -1.	180. -1. -1.	.15 -1 .15	.15 -1. -1.	-1. -1. .166	$2.4 \\ -2.4 \\ 2.4$	8.1 -2 5.0	-1. -1. 1.14	.75 1 .73	-2. 6.	6.0 -1 4.2	-1:	-1. -1. -1.	-1. 10. 40.
1703 1703	2 4	0. 0.	-1:	8.3	-1:	145	113.	180.	-12	.146	-1:	3.1	7.6	-1:	~75	-2.	-2. -1.	-1: -1:	-1:	-1:
1704 1704 1704	24 7	1. 1. 1.	-1. -2. 1.0	8.3 -2. 8.3	-1. -2. -1.	128. -1.	89. -1. -1.	180. -1. -1.	.16 -1. .16	.185 -1. -1.	-1. -1. .224	.0 .0	5.1 -2. 5.0	-1:	.55 1. .54	10. -2. 10.	14.0 9.8	-1. -1. -1.	-1. -1. -1.	-1. 10. 10.
1705 1705 1705	247	1. 1. 1.	-1. -1. 1.0	8.3 -2. 8.3	-1:	-1: -1: -1:	58. 1. -1.	180. -1. -1.	.25 -1. .25	.25 -1. -1.	-1 -1 .25	$\frac{1.5}{-2}$ 1.5	6.1 -2. 6.0	-1. -1. 1.10	.65 1. .63	13. -2. 13.	16.0 11.9	-1: -1: -1:	-1:	-1: -1:
1705	1	1.	-1.	-1.	-1.	-1 .	-1.	120.	. 15	. 192	-1.	.0	4.6	-1.	~1.	18.	-1.	-1.	-1.	-1.
1801 1801	25	1:	-1:	7.5 7.6	-1:	2 <u>1</u> :	52. 52.	40. 40.	.18 .16	. 195	- <u>1</u> :	1:0	8.1 8.0	: 1:	.60 -1.	8. 8.	8.0 -1.	-1:	-1: -1:	-1:
1802 1802	25	1:	-1: -1:	7.5	$\begin{bmatrix} 1\\1 \end{bmatrix}$	-1:	52. 52.	40.	18 18	.195	-1: -1:	1.0	7.8	-1: -1:	.75 -1.	15. 8.	18.0 -1.	-1 :	-1:	- <u>1</u> :
1803 1803	25	0: 0:	-1: -1:	7.5	-1:	-1: -1:	52. 52.	40.	18	.195	- <u>1</u> :	1.0	8.1	-1:	.60 -1.	12.	18.0	-1:	-1:	-1.
1804	2	٥.	-1.	7.5	-1.	-1.	52.	40.	.18	.12	-1.	3.7	7.6	-1.	1.00	8.	8.0	-1.	-1.	-1.

CASE	BCAT	LIQI	L192	H	8	P	DER	DUR	٨	CER	CSRN	ZV	ZL.	BIGT	SIGE	×	N1	æ	N160	FC
1804	6	٥.	-1.	7.5	-1.	-1.	52.	40.	.16	-1.	-1.	3.5	7.5	-1.	-1.	8.	-1.	-1.	-1.	-1.
1805 1805	1 5	1:	-1:	71.5	-1 :	-1:	52.	20.	.16	.194 -1.	-1: -1:	1.0	8.1 8.0	-1: -1:	-1:	7:	-1:	-1:	-1:	-1:
1806 1805	1 5	8:	-1	-1. 7.5	4:	:1:	-1:	20.	.16 .16	.158	-1:	3.7	8.8 8.5	-1:	-1:	7:	-1: -1:	-1:	-1;	-1: -1:
1807 1807	1 5	8:	-1:	7.6	-1:	-1:	-1:	20.	.18	.194	-1:	1:0	6.1 6.0	-1:	-1: -1:	12. 12.	-1:	-1 :	-1 :	-1 :
1806	3	٥.	-1.	7.5	-1.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	12.2	-1.	-1.	-1.	55.0	-1.	-1.	-1.
1809	4	٥.	-1.	7.5	-2.	51.	-1.	-1.	-1.	-1.	-1.	.9	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
1810	1	1.	-1.	7.5	-2.	51.	-1.	-1.	-1.	-1.	-1.	.9	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
1811	1	1.	-1.	7.5	-2.	51.	-1.	-1.	-1.	-1.	-1.	.9	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
1812	1	1.	-1.	7.5	-2.	51.	-1.	-1.	-1.	-1.	-1.	.9	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
1814	7	1.	-1.	7.5		-1	-1.	-1.	-1.	-1	170		-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
1814	á	i .	-i.	7.5	-i:	-i:	-11	-i:	.16	.ið	.18		7.0	1.35	.74	8.	9.0	1.09	10.0	2:
1815 1815	78	8: 0:	0.5 0.5	7.5 7.5	-1:	-1:	-1:	-1 :	:16	-1.	.170	1.0	7:0	1.33	.73 .74	12	14.2 13.6	1.09^{-1}	15.0	2:
1816 1816	7	8:	0.0 -1.	7.5 7.5	-1:	-1:	-1:	-1:	.18	-1. .18	.151	2.0	7:0	1.31	.81 .83	18. 18.	20.3 19.5	1.2i	23.5	2.
1817 1817	7 8	1:	1.0 -1.	7.6 7.6	-1:	-1 :	-1:	-1:	:10	-1 .18	.168	1.0	10.0 10.1	1.90 1.93	1.00	10. 10.	10.0 10.0	1.09	10.5	2:
1818 1818	7 8	8:	0.5 0.5	7.5 7.5	-1:	-1:	-1:	-1:	:16	-1. .18	.168	1.0	10.0 10.1	1.90	1.00	18. 18.	16.0 15.5	1.09	17.0	2:
1819 1819	7 8	8.	0.0 -1.	7.5	-1:	-1:	-1 :	-1;	:18 :18	.185	.134 .185	2.0	10.0	1.88	1.08	20. 20.	19.1 19.0	1.2i	-1. 22.5	2:
1820 1820	7 8	1:	1.0 -1.	7.8	-1:	-1:	-1:	-1:	.16	-1. .2i	.205	:0	1.3	.82	.39 .40	ł	8.2 8.0	1.09	-1 6.5	10:
1821 1821	7	8:	0.0 -i.	7.5	-1:	-1:	-1: -1:	-1:	.16 .18	-1 195	.181 .195	1.3	8.0 6.1	1:17	.67 .68	27. 27.	33.5 32.0	1.21	-1. 38.5	8.
1822 1822	7 · 8	8:	0.0 -1.	7.5	- 1:	-1:	-1:	-1 :	.16	-1. .165	.128	2.5 2.4	8.0 6.1	1.17	.70 .81	12. 12.	14.0 13.0	1.09	14.8	8:
1823 1823	78	1:	1,0 -1.	7.5 7.5	-1 :	-1:	- 1:	=1:	.16 .16	.185	.177 .185	:8	4.5	.88 .88	:47 :48	6.	8.7 8.5	-1. 1.09	9.0	8:
								γ												
1901	8	0.	0.0	4.9	-1.	-1.	-1.	-1.	-1.	-1.	-1.	2.4	17.1	-1.	1.50	22.	17.9	-1.	-1.	-1.
2001 2001	2	ł	: ::	6.3 8.3	-1:	-1 :	58. -1.	15. -1.	.13	.085	-1. .07	.9	.9	-1. .16	.165	3. 3.	5.0	-1. .50	-1 3.5	-1:
2101		1.	-1.	8.3	-2.	8.	-1.	-1.	-1.	-1.	-1.	1.8	-2.	-1.	-1.	7.	-1.	-1.	-1.	-2.
2102	4	1.	-1.	6.3	-2.	8.	-1.	-1.	-1.	-i.	-1.	1.4	-2.	-1.	-1.	7.	-1.	-1.	-1.	20.

CASE	BCAT	L101	LIQ2	M	Ħ	EP	DER	DUR		CSR	CSRN	ZW	ZL	SIGT	SICE	N	N1	Œ	N160	FC
2201 2201 2201 2201 2201 2201 2201 2201	1 2 4 5 6 7 8	1. 1. 1. 1. 1.	-1. -1. 1.0 -1. -1. -1.	-1. 7.8 -2. 7.8 7.8 7.9 7.9	-1. 20. -1. -1. -1.	-1. -1. 283. 160. -1.	-1. 161. 160. -1. -1. -1.	20. 45. -1. 45. -1. -1.	.19 .18 -1. .18 -1. .20 .20	.237 .205 -1. -1. -1. -1. -1.	-1. -1. -1. -1. -1. -1. -1. -1. -214 .22	.9 1.0 1.0 -1. 1.0	5.8 4.6 -2. 4.5 -1. 4.0 4.0	-1. -1. -1. -1. -1. -1. -1. -1. -1. -1.	-1. .50 -1. .28 .48	13. -2. -1. 5.	-1. 9.0 -1. 5.8 7.3 7.0	-1. -1. -1. -1. -1. -1. -1.	-1. -1. -1. -1. -1. -1. -1. 8.0	-1. -2. -1. -1. 20. 20.
2202 2202 2202	1 2 5	0. 0. 0.	-1. -1. 0.0	-1. 7.8 7.8	-1: -1:	-1 -1 172	-1. -2. -1.	20. 45. 45.	.21 .21	.205	-1: -1:	1.5 1.5 1.6	3.7 3.0 3.0		-1 .40 -1.	18. 15. 15.	23.0 -1.	-1. -1. -1.	-1. -1. -1.	-1. -1. -1.
2203 2203	1 5	1.	-1. -1.	-1. 7.8	-1^{1}_{1} ;	172.	-1: -1:	20.	.21 .21	-1.	-1: -1:	1.5 1.5	3.7	-1:	-1. -1.	6. 5.	-1 -1	-1. -1.	-1: -1:	-1: -1:
2204 2204	25	0.	-1:	7.8 7.8	-1:	172.	-2: -1:	45.	.21	.23	-1:	1:0	3.7	-1:	.40 -1.	14. 14.	21 -1	-1:	-1:	-1 :
2205 2205	25	1:	-1	7.8 7.8	: <u>1</u> :	172	-1:	45. 45.	21	.185	-1: -1:	1.5	2.1 3.5	-1:	.40 -1.	6. 6.	9. -1.	-1:	-1: -1:	-1: -1:
2205	4	٥.	-i.	7.8	-2.	160.	-1.	-1.	-1.	-1.	-1.	1.0	-2.	-1.	-i.	-2.	-1.	-1.	-1.	-2.
2207 2207 2207	· 4 7 8	0. 0.	-1. 0,0 -1.	7.8 7.9 7.9	-2: -1: -1:	160. -1. -1.	-1:		-1. .23 .23	-1. -1. .215	-1. .222 .23	1.5	-2. 6.0 6.1	1.14	-1. .74 .78	-2. 28. 28.	-1. 33.i 31.0	-1. -1 1.21	-1. -1. 37.5	5.
2208 2208 2208	4 7 8	0. 0.	-1. 0.0 -1.	7.8 7.9 7.9	-2. -1. -1.	160. -1. -1.	4	-1:	-1. .23 .23	-1.	-1. 248 255	1.3	-2. 4.0 4.0	-1. .76 .78	-1 40 40	-2. 16. 18.	-1. 23.4 22.5	-1. -1. 1.21	-1. -1. 27.5	-2. 5. 8.
2209	4	٥.	-1.	7.8	-2.	160.	-1.	-1.	-1.	-1.	-1.	1.5	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-2.
2210	4	ο.	-1.	7.8	-2.	160.	-1.	-1.	-1.	-1.	-1.	1.5	-2.	-1.	-1.	-2.	-1.	-i.	-1.	-2.
2211	4	۰.	-1.	7.B	-2.	160.	-1.	-1.	-1.	-1.	-1.	1.0	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-2.
2212	4	٥.	-1.	7.8	-2.	160.	-i.	-1.	-1.	-1.	-1.	1.3	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-2.
2213 2213	7 8	1:	1.0 -1.	7.9	=l:	-1: -1:	-1:	-1:	.23	-1.	270	.5	4.0	.78	.42 .42	6. 6.	9.1 9.0	1.09	9.6	5.
2301	4	٥.	-2.	6.1	50.	-1.	-i.	-1.	-1.	-1.	-1.	8.0	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
2302	4	٥.	-2.	6.1	60.	-1.	-1.	-1.	-1.	-1.	-1.	8.0	10.0	-1.	-1.	47.	-1.	-1.	-1.	-1.
2303	4	٥.	-2.	6.1	60.	-1.	-1.	-1.	-1.	-1.	-1.	2.0	5.3	-1.	-1.	10.	-1.	-1.	-1.	-1.
2304	4	0.	-2.	6.1	50.	-1.	-1.	-1.	-1.	-1.	-1.	2.0	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
2305	4	٥.	-2.	8.1	60.	-1.	-1.	-1.	-1.	-1.	-1.	3.5	-2.	-1.	-1.	-2.	-1.	-1.	-i.	-1.
2300	4	٥.	-2.	6.1	50.	-1.	-1.	-1.	-1.	-1.	-1.	3.0	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1 .
2501	4	٥.	-1.	7.1	-1.	-1.	-1.	-1,	-1.	-1.	-1.	3.7	7.	-1.	-1.	12.	-1.	-1.	-1.	-1.
2601 2601	27	1:	-1. 1.0	5.5	-1: -1:	-1: -1:	8. -1.	15. -1.	40	. 28 -1	.237	4.6	5.1 5.0	đi	1.00	2.	2.0	- <u>f</u> :	- f:	-1 10.

Liquefaction data catalog compiled by Liao and Whitman (1986a) (cont.)

CASE	BCAT	LIQI	L192	M	н	P	DER	DUR		CSR	CSRN	ZW	ZL	SIGT	SIGE	N	N1	CE	N160	PC
2601	8	1.	-1.	6.6	-1.	-1.	-1.	-1.	.45	.325	.28	4.8	8.1	1.14	.98	2.	2.0	.75	1.5	50 .
2602 2602 2602	2 4 7	1. 1. 1.	-1. -1. 1.0	8.6 5.6 6.6	-1. -2. -1.	-1. -1.	8. -1. -1.	15. -1. -1.	-1. -1.	.16 -1: -1:	-1. -1. .190	18.8 -2. 15.5	16.8 -2. 16.5	-1. -1. 3.22	3.30 -1 3.22	24 . -2 . 24 .	11.0 7.4	-1. -1. -1.	-1. -1. -1.	-1. -1. 50.
2603 2603	7 8	1:	1.0 -1.	8.8 8.8	-1:	-1:	-1: -1:	-1:	.45 .45	.345	.287	3.0 3.0	6.0	1.10 1.10	.88 .88	9:	6.9 9.5	-1. .76	71.0	20. 20.
2604	3	٥.	-1.	6.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	8.3	19.8	-1.	3.42	15.	8.1	-1.	-1.	-1.
2605	8	0.	-1.	8.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	18.3	25.9	-1.	3.88	32.	16.2	-1.	-1.	-1.
2605	3	O.	-1.	6.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	14.3	25.9	-1.	2.89	42.	24.7	-1.	-1.	-1.
2607	3	٥.	-1.	6.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	1.5	4.6	-1.	.48	8.	11.8	-1.	-1.	-1.
2608	3	o .	-1.	8.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	16.8	19.8	-1.	3.43	21.	11.3	-1.	-1.	-1.
2609	3	٥.	-i.	8.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	10.7	15.2	-1.	2.28	30.	19.9	-1.	-1.	-1.
2610	3	ο.	-1.	6.6	-1.	-1.	-1.	-1.	-1.	-1.	-1.	10.7	16.8	-1.	2.61	25.	16.0	-1.	-1.	-1.
2611	3	٥.	-1.	8.8	-1.	-1.	, ≁1.	-1.	-1.	-1.	-1.	13.7	21.3	-1.	3.25	39.	21.6	-1.	-1.	-1.
2701	4	0.	-1.	7.3	<mark>-1</mark> .	-1.	-1.	-1.	-1.	-1.	-1.	3.1	-2.	-1.	-1.	-2.	-1.	-1.	-1.	-1.
2801	6	ο.	-1.	7.3	-1.	90.	-1.	-1.	.075	-1.	-1.	1.0	8.0	-1.	-1.	15.	-1.	-1.	-1.	-1.
2802	6	٥.	-1.	7.3	-1.	90.	-1.	-1.	.075	-1.	-1.	1.0	12.0	-1.	-1.	20.	-1.	-1.	-1.	-1.
2803 2803	5 8	8:	-1: -1:	7.3 7.3	-1 :	90. -1.	-1:	-1:	.075	.095	.095	2.0	8.0 8.0	1.14	-1 .73	9.5 9.5	ii :	-1. 1.00	11.0	-1:
2804	Б	٥.	-1.	7.3	-1.	90.	-1.	-1.	.075	-1.	-1.	2.0	12.0	-1.	-1.	15.5	-1.	-i.	-1.	-1.
2805 2805	5	8:	- 1:	7.3	-1:	104.	-1:	-1:	.075	.095	-1. .09	2.0	13.0 13.0	2.58	-1. 1.48	14.5 14.5	11.6	1.00	11.5	-1:
2806	5	1.	-1.	7.3	-1.	95.	-1.	-1.	.075	-1.	-i.	1.5	5.2	-1.	-1.	5.5	-1.	-1.	-1.	-1.
2807 2807	5 8	0. 1.	-1: -1:	7.3	-1:	95. -1.	-1:	-1:	.075	-1 .10	-1 .10	1.6 1.6	6.2	1.22	.76	6. 6.	6.5	-1. 1.00	6.5	-1;
2808	5	٥.	-1.	7.3	-1.	67.	-1.	-1.	.075	-1.	-1.	1.5	7.0	-1.	-1.	8.	-1.	-1.	-1.	-1.
2809 2809	5 8	0. 1.	-1: -1:	7.3 7.3	-1:	53. -1.	-1:	-1 :	.15 .20	-1. .21	-1 .20	1:8	10.5	1.8i	-1. .94	!! :	11.0	1.00	-1. 13.5	-1 :
2810 2810	5	ł	-1:	7.8	-1:	60. -1.	-1:	-1:	.075	.1o	.095	2.0	3.0	-1. .68	.48	8. 6.	-1. 8.5	-1. .75	6.5	-1:
2811 2811	5 8	1:	;	7.3	- 1:	-1:	-1:	-1:	.15	-1.	-1 195	2.0	10.0 10.0	1.98	1.1é	9 9	8.0	1.00	-1. 8.0	-1 :
2812 2812	68	ł	-1:	7.3 7.3	- 1:	55. -1.	-1:	-1:	:15 :20	.195	-1 .1\$	2.0	10.3 10.3	2.04	1.21	8:	a.o	1.00	-1. 8.0	-1: -1:
2813	8	1.	-1.	7.3	-1.	-1.	-1.	-1.	.13	.135	.13	1.5	9.1	1.79	1.03	8.	8.0	1.00	б.Б	67.
2814	8	1.	-1.	7.3	-1.	-1.	-1.	-1.	.20	.21	.20	1.5	8.2	1.61	.94	13.	13.5	1.00	13.5	48.

Liquefaction data ca	atalog compi	led by Liao	and Whitman ((1986a) (cont.)

CASE	BCAT	LIQI	L102	. N	R	EP	DER	DUR		CSR	CSRN	ZV	ZL	SICT	SIGE	N	N1	æ	N160	FC
2815	8	ο.	-i.	7.3	-1.	-1.	-1.	-1.	.10	. 105	.10	1.6	8.2	1.61	.94	9.	9.0	1.00	9.0	-1.
2901 2901	7 8	1. 1.	1.0 -1.	7.5	-i:	-1:	-1:	-1:	.135 .13E	-1 i j	.169	1.5	10.4	1.42	.29 .88	8. e.	9.3 5.0	-1.75	-1. 5.0	3. 3.
2902 2902	7 8	8:	0.5 0.5	7.5	-1:	-1:	-1:	-1: -1:	.135 .135	.135	.135	2.0	4.8	. 58 . 56	.35 .34	8.	9.3 12.0	-1 .75	9.0	3. 3.
2904	7	1.	1.0	7.5	-1.	-1.	-1.	-1.	. 135	-1.	.126	3.3	7.0	1.03	.64	14.	12.7	-1.	-1.	3.
2905	8	1.	0.5	7.5	-1.	-1.	-1.	-1.	.135	. 18	.18	2.4	11.6	1.67	.91	15.	15.6	.75	11.5	3.
2905	8	Ο.	-1.	7.5	-1.	-1.	-1.	-1.	.135	.15	.15	-2.	10.7	1.40	.73	16.	18.5	.75	14.0	3.
3001 3001	5 8	1:	-1: -1:	7.8 7.6	-1:	43. -1.	=1:	4:	.15	18	-1 .17	1:4	2.3	-1. .43	-1. .33	11 11:	-1 18.0	-1 .75	13.6	-1 :
3002 3002	6 8	1 :	-1: -1:	7.8 7.6	-1:	25. -1.	-1: -1:	-1:	.30	-1.	405	1.0	7:0	1.34	-1 .74	4:	-1 4.6	1.00	-1. 4.5	-1:
3003	7	1.	-1.	7.8	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-1.	-i.	-1.
3004	7	1.	0.7	7.8	-1.	-1.	-1.	-1.	. 20	-1.	.192	. 6	11.0	2.20	1.30	-1.	7.0	-1.	-1.	Б0.
3005	7	٥.	0.5	7.8	-1.	-1.	-1.	-1.	. 20	-1.	.192	. 6	11.0	2.20	1.30	-1.	8.0	-1.	-1.	50.
3005	8	ο.	-1.	7.6	-1.	-1.	-1.	-1.	. 50	.405	.42	3.0	5.3	1.00	.77	30.	33.5	1.00	33.5	10.
3007	8	1.	-1.	7.8	-1.	-1.	-1.	-1.	.35	. 38	.395	. 9	5.3	1.04	.60	17.	21.5	1.00	21.5	20.
3008	8	1.	-1.	7.6	-1.	-1.	-1.	-1.	. 22	.165	. 175	3.7	5.4	. 99	.82	20.	22.0	1.00	22.0	3.
3009	8	1.	-1.	7.8	-1.	-1.	-1.	-1.	. 20	.155	.16	1.5	2.0	.37	.32	10.	16.5	.75	12.5	12.
3010	8	1.	-1.	7.8	-1.	-1.	-1.	-1.	. 13	.14	.145	1.2	6.1	1.19	.70	10.	11.5	1.00	11.6	12.
3011	8	1.	-1.	7.6	-1.	-1.	-1.	-1.	. 20	.22	.23	.9	6.1	1.20	. 68	9.	11.0	1.00	11.0	-1.
3012	8	0.	-1.	7.6	-1.	-1.	-1.	-1.	.07	.06	.05	4.0	9.1	1.74	1.22	13.	11.6	1.00	11.5	1.
3013	8	1.	-1.	7.6	-1.	-1.	-1.	-1.	. 20	.205	.21	1.5	7.0	1.3/	.82	11.	12.0	1.00	12.0	•
3201	8	1.	-1.	7.4	-1.	-1.	-1.	-1.	.20	.165	.16	4.6	8.2	1.45	1.08	9.	8.5	.75	6.0	20,
3203	8	1.	-1.	7.4	-1.	-1.	-1.	-1.	. 20	.165	.16	6.7	11.9	1.50	1.04	12.	11.5	.75	8.5	-1.
3204	8	ο.	-1.	7.4	-1.	-1.	-1.	-1.	. 20	.20	. 195	1.2	3.7	.64	.40	14.	21.5	.75	16.0	4,
3205	8	0.	-1.	7.4	-1.	-1.	-1.	-1.	. 20	.155	.15	2.1	3.0	. 64	.44	14.	19.5	.75	14.5	з.
3206	8	1.	-1.	7.4	-1.	-1.	-1.	-1.	. 20	.195	.19	1.8	5.2	. 91	. 58	6.	8.0	.76	5.0	51.
3301	7	1.	1.0	7.0	-1.	-1.	-1.	-1.	. 25	i 41.	.344	1.0	7.0	1.33	.73	-1.	1.2	-1.	-1.	90 (
	λĽ			<u>NI</u> N			ה ה			Γ		Γ.								
3401 3401	8	0.	0.0 -1.	6.7	-1. -1.	-1: -1:	-i:	-1:	.10	3 -1: .1:	.092	1.0	6.4	1.21	.68	10.	12.0	1.09	13.0	Ő.

CASE	RCAT	1.101	1.102	ж	R	FP	DER	DUR		CSR	CSRN	7₩	ZI.	STOT	SIGE	м	NS	Œ	N160	FC
3403 3403	7 8	0. 1.	0.7 -1.	8.7 8.7	-1:	-1:	-1:	-1:	.12	.135	.117 .115	.5 .6	3.3 3.4	.63 .64	.42 .37	5. 5.	8.1 8.0	1.00	-1. 8.0	Б. Б.
3404 3404	7 8	0. 0.	0.0 -1.	6.7 6.7	4:	-1:	-1:	-1:	12	.115	.098	1.3	3.3 3.4	. 63 . 64	. 42 . 43	7:	10.6 10.0	1.00	10.0	4.
3405 3405	7 8	0. 0.	0.0 -1.	8.7 8.7	-1:	-1:	-1:	-1:	:12 :12	-1 .12	.100 .105	1.8	8.3 5.5	95 1.01	.60 .64	2:	2.6 2.5	-1. 1.00	-1 2.5	60. 60.
3405 3406	7 8	8. 0.	0.0 -1.	8.7 8.7	-1:	-1:	-1:	-1:	.12	-1. .13	.109	.9	4.3	.82 .82	.48 .48	11. 11.	15.C 15.0	1.00	-1. 15.0	0. 0.
3408 3408	7 8	8:	0.0 -1.	6.7 6.7	=1:	-1: -1:	=1:	-1:	.12	-li	.094	1.7	4.3 4.3	.82 .82	.55 .68	4:	Б.4 5.0	-1. 1.00	-1. 5.0	10. 10.
3409 3409	7 8	0 .	0.0 -1.	6.7 6.7	-1:	-1:	-1:	-1:	:12 :12	.116	.098	$1.3 \\ 1.2$	3.3 3.4	.63	.42 .43	13. 13.	19.7 19.0	1.12^{-1}	-1 21.5	7:
3410 3410	7 8	8:	0.0 -1.	6.7 6.7	-1:	-1: -1:	-1:	: <u></u> ;	.12	.145	.125	.3	4.3	.82 .82	.42 .42	8. 8.	13.7 12.0	1.00	12.0	12.
3412 3412	7 8	0. 0.	0.0 -1.	6.7 6.7	-1 :	=1:	-1:	-1:	:14	.11	089	4.3 4.3	6.3 6.4	1.05	.85 .87	9.	9.9 9.5	1.00	-1. 9.5	Б. Б.
3413 3413	78	0.	0.0 -1.	6.7 6.7	-1:	-1:	-1:	- <u>1</u> :	:14	.13	:111	2.4	6.3	$1.11 \\ 1.17$.72 .77	8. 8.	9.6	1.00	9.0	4:
3414 3414	7 8	0. 0.	0.0 -1.	0.7 5.7	-1:	- 1:	-1:	-1:	:14	.095	.080	3.0	3.3 3.4	.61	.58 .61	11: 11:	14.8 13.5	1.00	13.5	5. 5.
3417 3417	78	8:	0.0 -1.	5.7 8.7	-1:	-1:	-1:	-1:	:14	.11	.093	2.5	4.0	.78	.61 .51	6. 6.	7.8 7.5	1.09	-1. 8.0	10. 10.
3419 3419	7 8	8. 8.	0.0 -1.	8:7 8:7	-1:	-1:	:1:	-1:	:14	-1. .12	.100	2.6	5.0 5.2	1.00	.70	9.	10.9 10.5	1.09	-1. 11.5	20. 20.
3421 3421	78	8:	0.0 -1.	6.7 8.7	-1 :	-1:	-1:	-1:	:14	.125	.105	2.5 2.4	6.0 6.1	1:14	.79 .81	12. 12.	13.7 13.0	1.09	-1. 14.6	3. 3.
3424 3424	8	8:	0.0 -1.	6.7 6.7	-1:	-1:	-1:	-1:	.12	-1: .1:	.098	1.4	4.0	:78	.50 .52	1:	5.7 5.5	1.09	-1. 6.0	10. 10.
3601 3501	7 8	0. 1.	0.0 -1.	7:4	-1: -1:	-1:	-1 -1	-1:	.20	-1. .22	. 208	1.0	8.3 6.4	1.21	.68 .68	10. 10.	12.3 12.0	1.09	-1 13.0	0. 0.
3502 3502	7 8	8: 0:	∩.0 -1.	7:4	-1: -1:	-1:	-1:	-1:	.32 .32	-1.	.314	:0	3.3	.63	.39 .40	19 19	29.8 28.5	1.12^{-1}	31.5	4:
3503 3603	7 8	0. 1.	0.7 -1.	7:4 7:4	-1:	-1;	-1:	-1:	.32 .32	-1 .355	.350	.5 .6	3.3 3.4	.63	.42 .37	Б. Б.	8.1 8.0	1.00	8.0	3. 5.
3504 3504	7 8	0. 1.	0.0 -1.	7:4	- 1:	-1:	-1:	-1:	.32	-1. .305	.292	1.3	3.3 3.4	.63	.42 .43	7:	10.6 10.0	1.00	-1. 10.0	4:
3505 3505	7 8	0. 1.	0.0 -1.	7:4	- 1:	-1;	-1: -1:	$\begin{bmatrix} 1\\ -1 \end{bmatrix}$.24	.235	.224	1.8 1.8	5.3 5.5	.95 1.01	.60 .64	2.	2.6	-1. 1.00	-1. 2.5	60. 60.
3506 3506	78	9: 1:	0.0 -1.	7:4	-1:	-1: -1:	-1 -1	\mathbb{T}_1^1	.24 .24	.255	- 245 . 25	.9	4.3 4.3	.82 .82	.48 .48	11. 11.	15.8 16.0	1.00	15.0	8:
3507 3507	7	0 :	0.0 -1.	7:4	-1:	-1: -1:	-1:	-1:	.24 .24	.22	.205	2.2	5.3 5.5	1.05	.68	20 20	24.6 23.0	1.12	 25.0	0. 0.
3508	7	ο.	0.0	7.4	-1.	-1.	-1.	-1.	.24	-1.	.210	1.7	4.3	.82	.56	4.	5.4	-1.	-1.	10.

Liquefaction data catalog compiled by Liao and Whitman (1986a) (cont.)

CASE	BCAT	LIQI	L192	M	Я	EP	DER	DUR		CSR	CSRN	Z₩	ZL	SIGT	81GE	м	N1	CE	N160	FC
3508	8	1.	-1.	7.4	-1.	-1.	-1.	-1.	.24	.215	.215	1.8	4.3	.82	. 58	4.	Б.О	1.00	5.0	10.
3509 3509	7 8	0. 1.	0.0 -1.	7.4 7.4	-1:	-1:	-1:	-1: -1:	.24	-1. .23	.219	1.3	3.3	.63 .64	.42 .43	13. 13.	19.7 19.0	1.12	21.5	; ;
3510 3510	7 8	0. 1.	0.0 -1	7:4	-1:	-1:	-1:	-1:	.24	.29	.281	.3 .3	4.3	.82 .82	.42 .42	8. 8.	12.1 12.0	1.00	-1 12.0	12. 12.
3511 3511	7 9	0. 0.	0.0 -1.	7:4	-1:	-1;	-1:	-1:	.24 .24	-1.	.241	1.3	7.3	1.39 1.41	.79 .80	17 17	19.4 18.5	1.12^{-1}	-1. 21.0	17 17
3512 3512	7 8	0. 1.	0.0 -1.	7:4	-1: -1:	-1: -1:	-1: -1:	-1:	.24	. 185	.172	4.3	6.3 6.4	1.05	.85 .87	9. 9.	9.9 9.5	1.00	9.5	5. 5.
3513 3513	7 8	0. 1.	0.0 -1.	7:4	-1:	-1: -1:	-1:	-1:	.24	.225	.214	2.4	6.3 6.4	1:17	.72	8. 8.	9.6 9.0	-1. 1.00	-1 9.0	4:
3514 3514	7 8	0. 1.	0.0 -1.	7.4	-1 :	-1:	-1:	-1:	.28	. 186	.179	3.0 3.0	3.3 3.4	.61 .63	.58 .61	11. 11.	14.6 13.5	1.00	13.5	5. 5.
3515 3515	7 8	0. C.	0.0 -1.	7:4	$\begin{bmatrix} -1 \\ -1 \end{bmatrix}$	-1:	-1;	-1:	.28	.236	.223	3.0 3.0	6.0 6.1	1.12 1.17	.82 .57	23. 23.	25.7 24.0	1.2i	29.0	8:
3516 3516	7 8	<u>.</u>	0.0 -1.	7.4	-1: -1:	-1:	-1:	-1:	.24	.215	.202	2.5	6.0 6.1	1:14	.79 .82	10. 10.	11:8	1.09	-1. 12.	10. 10.
3517 3517	7 8	0. 1.	0.0 -1.	7:4	-1:	-1:	-1 :	-1:	.24	-1: .19	.180	2.5	4.0	.78	.61 .61	8. 8.	7.8 7.6	1.09	-1. 8.0	10. 10.
3518 3518	7 8	0 .	0.0 -1.	7:4	-1:	-1:	-1:	-1:	.24	.225	.208	2.5	7.0 7.0	1.33	. 88 . 89	21. 21.	22.0 22.0	1.21	26.5	δ. δ.
3519 3519	7 8	0. 1.	0.0 -1.	7:4	-1:	-1:	-1:	-1:	.24	.205	.193	2.5	5.0 5.2	1.00	.70 .72	8.	10.9	1.09	11.5	20. 20.
3520 3520	7 8	0 .	0.0 -1.	7:4	-1:	-1:	-1:	-1:	.24	-1.	.187	2.5	4.5	.85	. 65	10. 10.	12.0 12.0	1.00	13.0	26. 25.
3521 3521	7 8	0. 1.	0,0 -1.	7:4	-1: -1:	-1:	-1 -1	-1:	.24	-1.	.202	2.5	6.0	1:14	.79	12. 12.	13.7 13.0	1.01	14.5	3. 3.
3522 3522	7 8	0 .	0.0 -1.	7:4	-1:	-1 -1	-1:	-1: -1:	.24	-1.	.209	2.4	6.0 6.1	1.09	.73 .77	15. 15.	17.5	1.2	20.0	11: 11:
3523 3523	7 8	8.	0.0 -1.	7:4	-1:	-1; -1;	-1	-1.	.24	-1.	.188	3.6 3.7	7.0	1.30	.93	17	17.7	1.2	20.5	12.
3524 3524	7 8	0. 1.	0.0 -1.	7:4	-1 -1:	-1 -1	-1 -1	-1:	.20	.168	.183	1.4	4.0	.76	.50 .52	4	5.7 5.6	-1. 1.0	6.0	10. 10.
3525 3525	7	0 .	0.0 -i.	7:4	: 1:	-1: -1:	-1 -1	-1 :	.20	.205	.195	1.4	5.0 5.1	1:13	.68 .71	15 15	18.8 17.8	-1 1.2	21.0	10 10
	_						Ц	1		٤.			Ϊ.							
3901 3901	8	0. 0.	-1,	6.6	-1. -1.	-11	-1	-10	.78	.67	.575	1.8	3.7	.69	.61	28	38.0	1.08	39.6	25
3902 3902	8	8: 0:	0.0 -1.	8.8 8.6	-1:	-1	-1	-1:	.60	-1.	449 575	1.8	4.0	.74	.51	1	5.6	1.06	-1 1.6	25. 29.
3903 3903	8 7 8 8	0.	0.0	5.6 5.6	-1 :	-1 -1	-1	-1:	.60	-1.	.449 .595	1.8 1.8	4.0	.74	-52 57	11 13	15.3 16.8	1.00	17.6	19. 37.

CASE	BCAT	LIQI	1102	. н	R	P	DER	DUR		CSR	CSRN	ZW	ZL	SIGT	SICE	N	N1	œ	N160	FC
3904 3904	7 8	1:	0.7 -1.	6.6 6.6	-1:	:1:	-1:	-1:	.20 .24	265	.205	23	2.0 1.8	.38 .36	20	3. 3.	5.7 5.6	-1 .79	-1. 4.5	56. 80.
3905 3905	7 8	1:	0.7	6.6 6.6	-1:	- 1:	-1:	-1:	.20	-1.28	.209	:3	5.0 4.3	.95	47	11.	10.1 15.0	-1. 1.05	18.0	34. 18.
3918	8	1.	-1.	6.8	-1.	-1.	-1.	-1.	. 20	. 16	. 135	2.1	3.4	. 62	. 50	2.	2.5	1.05	3.0	75.
3919	8 -	Ο.	-1.	8.8	-1.	-1.	-1.	-1.	.20	.13	. 115	2.1	2.3	. 41	.39	11.	18.5	.79	13.0	30.
3925	8	1.	-1.	6.6	-1.	-1.	-1.	-1.	. 51	. 39	. 335	1.5	2.1	.39	.33	3.	Б.О	.79	4.0	31.
4101 4101	;	0 .	0.0 -1.	6.1 6.1	- <u>1</u> :	-1 :	-1:	-1:	:10	-1.	.086	1.0	6.0 6.1	1.C8 1.10	.58 .59	5. 5.	6.6 6.5	-1 1.09	-1. 7.0	13. 13.
4102 4102	7 8	1 .	0.7 0.5	8.1 8.1	-1:	-1;	-1:	-1:	.145	-1. .17	.143	-1.9	6.i	1.10	-1	-1. 6.	6.6 6.5	-1. 1.09	7.0	13. 13.
4103 4103	7 8	8.	0.0 -1.	8.1 8.1	-1:	-1:	-1:	:1:	.10	-1 .105	.094	1.0	14.0 14.3	2.52 2.59	1.08 1.25	1:	3.0 3.5	1.09	-1. 4.0	27 . 27 .
4104 4104	7 8	1:	0.7 0.5	6.1 6.1	-1:	-1:	-1:	-1:	.145	.166	.165	-1. .9	-1. 14.3	2.59	1.08 1.25	1:	3.8 3.5	-1 1.09	-1. 4.0	27 27
:						(
4204	8	٥.	-1.	5.8	-1.	-1.	-1.	-1.	.21	.235	. 165	. 3	1.8	. 36	.21	3.	δ.δ	.79	4.5	80.
4205	8	٥.	-1.	5.6	-1.	-1.	-1.	-1.	. 21	.245	. 175	.3	4.3	.85	. 45	11.	15.0	1.05	16.0	18.
4218	8	1.	-1.	6.6	-1.	-1.	-1.	-1.	. 20	. 18	.11	2.1	3.4	. 62	. 50	2.	2.5	1.05	3.0	76.
4219	8	٥.	-1.	5.6	-1.	-1.	-1.	-1.	. 20	.13	. C95	2.1	2 3	.41	. 39	11.	16.5	.79	13.0	30.
4225	8	1.	-1.	5.6	-1.	-1.	-1.	-1.	.28	. 28	. 20	.9	4.9	. 98	. 58	9.	11.0	1.05	11.5	40.
4228	8	1.	-1.	5.6	-1.	-1.	-1.	-1.	. 32	.28	. 185	2.7	4.3	.79	.64	δ.	8.5	1.05	6.5	92.
4227	8	٥.	-1.	5.6	-1.	-1.	-1.	-1.	. 09	. 07	. 05	1.5	2.1	.39	.33	3.	δ.0	.79	4.0	31.

จฺฬาลงกรณมหาวทยาลย



APPENDIX B

BOREHOLE DATA USED IN THIS STUDY

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย APPENDIX B.1 CHIANG MAI

สถาบันวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย

GROUND WATER OBSERVATION.	1	١	N.A.C. BO	RING LOG		BORING NO.	1
DATE TIME EL. of HOLE EL.of WATER						SURFACE ELV.	-0.20 m.
	LOC	ATIO	N. SUANP	RONG HOSPITAL		DATE START	
24 HR.AFTER BORING. 0.78 M.	1			CHIANGMAI		DATE FINISH	
			•	- LIQUID LIMIT.		∆One half Unconfined Compressive Strength	TOTAL-
SOILS DESCRIPTION	OFILE TVDE	×	STANDARD PENETRATIO	O PLASTIC LIMIT.		INSITU VANE SHEAR	DENSITY.
×.	SOIL PF	DEPTH	DL 0140 / 57	X NATURAL MOISTURE CON	TENT.	SENSITIVITY	γd.γw ●-⊖
		0.00	30 60	20 40 60	80	1 2 3	1/M.°
Fill soil		0.00			Ť		
Very loose grey and brown clay- ey very fine sand. SC	7 4 -5S	.1	3	-0-20			e o
Medium to very stiff light grey and brown silty clay, occasional very fine sand. CL	-55. -55.	3 4 5	13 19			e	• • • • • • • • • • • • • • • • • • •
Dense light grey and brown 4.50 clayey very fine sand.	ST SS.	6 5.00.	34	×0	and a constant		
Have to your stiff light men	-ss.	7	41	0×-0	-	Δ	••
yellowish brown and brown silty clay, trace of very fine sand.	-ss.	8				Δ	•••
CL	-55.	9	25			Δ	•••
Dense light grey very fine sand. SM	H-F-SS.	10,00	49				•
Very stiff light grey and yellow ish brown silty clay, trace of very fine sand. CL	SS.	11	31	×o-•		· · · · · ·	•
Very dense grey very fine sand. SM	SS1	12	60				•
Very stiff light grey and brown silty clay. CL16.00	IH L	13 15.00	21		+	·····	•
Very dense brownish grey and brown very fine to fine sand.	- 55: 	14				· · · · · · · · · · · · · · · · · · ·	P
SM		15	9	8 • X			† •
สถาเ	- 55.	16 20,00	10	2 X A	5		
21.45 END OF BORING.	111 ss	17	13	2 • ×			
จฺฬาลงก	าวี	ณ	มท	13,115	ſ	ลย	

GRO	UND W	ATER OBS	ERVA	TIO	N.					0.5	0.00	NC	10	~			BOR	RING	NO.		0	
DATE	TIME	EL. of HO	LEE	L.of	WATER			N	N.A.	.C. E	ORI	NG	LOC	a l			SUF	FACE	ELV.	-	-0.20	m.
										SUZ	ANPRO	NG	HO	SPIT	AL		DAT	ESTA	ART			
24 HF	AFTER	BORING.	0.7	75	М.	L	DCA	TIO	N.		CH	IIANG	MAI				DAT	E FIN	NISH			
				-	15.5	=	-	r	-						-		4.0				-	
										٠		• LI	QUID	LIMI	T.		Comp	pressi	ve Stre	ngth.		
							z										= pea	akD	remold	led	DEN	TAL-
sc	DILS D	ESCRI	PTI	ON		FILE	TYP	s	PEN	ETRA	TION.	O-P	LAST	IC LIN	AIT.		® One	e half	Pocket	EAH.		
						PRO	LE	Ĩ				~	1	NATU	RAL		Pen	etrom	eter R	dg.	70	
						OIL	AME	EPT				^ м	OIST	URE	CONT	ENT.	• SE	NSITI	VITY		•	-0
						S	ŝ	0	BL	ows	FT.			%				K	SC.		Τ/	м. ³
								0.00	-	30 0	60	20) 4	0 6	0 8	0	1		2 3		-1	2
	F	ill soil			-0.80						1											
Soft	grey v	ery fine	san	xdy (clay.	4	SS.1		4			0.0					1					••
fine sa	nd.	grey cla	ayey	ver	2.00	79	-55.2		R.	21		0										-
ey very	y fine	sand. S	h br	own	clay-	11	-SS.4		(11			xo	-				- 14	1. and 10.		-		•
Madia		nich lie			-]	11	SS.5		22			xc	7	•••••								•
ey very	fine	sand.	nc g	rey	CLAY-	11																
		SC				31	SS.6	5.00	22			Þ	•					•2 C				•
L			_		-5 80	31						1										
					5.00		-CS.7						-			1.00				-		••
Verv	stiff	to hard	bro	wni	sh					10												
light	grey a	nd yello	wish	bro	wn	K	55.8			1.0		¢						Δ	6-			••
silty sand.	clay,	occasion	at w	ery	fine	K				V								1	1			
		CI.					-SS.9		23	1		0					-	K -				•
		CD				V		10.00	-	X		1						1				
							55.10			145		x										•
								1	24													
Hamil	hora	ton fi		-	12.00		-\$5,11				64	0×										
hard	DLOWIT	ML-OL	e so	uniy	Clay.					1/	T	1										
Voiru	stiff	light on	nvie	h h	13.20	Y.	SS D		26	V		1										
and bro	wn sil	ty clay,	000	asio	onal	H	55.12					1	•					l				T
very 1	une san	a.					CC 13	15.00	- 25													
	CL-	ML & CI	-				55,15															T
					16.15	THH				1	70	1			2							
Very	dense	light gr	ey a	and I	brown		35,14				1	X			-	1		1.1993.3.10				P
very fi	ine to	fine san	d, o	xcas	sional						17	i I	-	-								
coarse	sand a	nd grave	e1.				-SS.15				1 14	X	1.00									1
		SM				111																
		(Chinada)					-SS.16	20,00		-	175	-×		•	****							•
										1		1										
		<u> a </u>			21.45	HH	SS.17				82	×										• • •
1	END	OF BOR	RING	•																		

จุฬาลงกรณ์มหาวิทยาลัย

BORING LOG						BOR	ING	NO.			1	BH-1	_			ELE	ATIC)N (m) :			
LOCATION : BUT REAL PROVIDENT	-				-	DEP	THU	n) N	-			5.45			8	DAT	ESTA	RTED		-	-3.1	/07
LOCATON					-	100		E.	<u> </u>						8	DAT	EEDI	ISHER	, .	_	12/04	197
· · · · · · · · · · · · · · · · · · ·	—		T	1	T	\vdash		E	-						-	DAL	6 FIR	IS HEL		T	7204	797
	2	8		9	(m	3	SDT.	NVA	TITE			MO	IST	AL RE				91			10	NIT
SOIL DESCRIPTION	L) H	HC	OHI	TE	ERY		Chl	ows/	ft.)			co	NTE	NT			e.	sam)	,		WE	IGHT
SOIL DESCRIPTION	DEPI	RAP	E E	AME	COV		(0.						(%)				1.				04	
		0		1	RE	10	2	3	0 40		20	40	6	0 80		1	2	3	4		16 1	18 20
		İTT	PA			ΤĨ					T	T	_					1	T		T	11
MEDIUM DENSE CLAYEY FINE TO MEDIUM	1.0		SS	1	25		211			_	ę											
SAND, BROWN			SS	2	30		Q	16			8		*									6
(SC) 2.00	2.0		SS	3	25		A	19		_	1	9		_		-		+			-	- °
			SS	4	30		1	14			9											
	3.0		SS	5	35		012			-	1	+				-			+	+	+-	1
	40		-	-	1		1.				1											
	4.0	1	PA				1			-	V					-		1	1		\top	
MEDIUM DENSE FINE TO MEDIUM SAND,	50		SS	6	30		9	18			þ											
BROWN							1				1											
(SM, SP-SM)	6.0		PA	-			-11			_	_	-	_	_	-	_		_	-	_	\vdash	
			SS	7	40		9	17			91								1			
	7.0		PA				1			-		-			-	-		-+	-	+-	+	+
			55	8	30		1	13			6											
8.50	8.0		-		-		T +					-	-	-	-			-	+	-	1	
	9.0		PA																			
LOOSE FINE TO COARSE SAND, BROWN			SS	9	35	9	9				0											
(SP-SM) 10.00	10.0						1			_		-		_		_	_	-	_	1	-	
			PA				1															1
	11.0		SS	10	32	-	-	9	24	-	9			-	_		-+	-	-	+	+	+-+
			PA					1			1					1						
SAND WITH CRAVEL CREV	12.0		SS	11	30		-	p	27	-	6	+	-	-	-	1		-	+	+	+	1
(SP-SM)	13.0							1	1		11											1
			PA				1	1														
	14.0		SS	12	30		9	18			0		_					_				-
	-						1	1														
15.45	15.0		PA	13	15		-	_		-	1	-	•	-		-	-	-+			+	-
END OF BORING REFILSAL		1001010	1						4	20			-									
KEI USAL							-	-		-		-	-	-		_	-	+	-	+	-	
							1						-					1				
																	_				-	
																						1
6000							+	-		_	-	+	_	-	-	-	-			-	+-	++
						-+	+	-	+	-		-+		+	-			-+		+	+	
A = POWER AUGERING HA = HAND AUG	ERING	3	wo	= WA	SHO	UT					ST = S	SHEI	BY	UBF	_	-		SS = S	SPLIT	SPOOR	4	<u> </u>
ARTY CHIEF: SARADECH CH. MADE BY : VIPAV	VEE O	3.	GEO	LOGI	ST :	UDON	MPOI	RN C	н	1	FILE	1	BCP-	CH1				DISK	: 9	1 CHL	NGN	MAI
	-	1	-		-	1	17	1	-		-	-	-									

PROJECT : BCP "ISSignides 2 Intersting" LOCATION : 0.1304 0.1304 111 SOIL DESCRIPTION (E) BOUL DESCRIPTION (E) BOU	A SS SS SS PA SS PA SS PA SS PA	ON ETAWVS 1 1 2 3 4 5 6 7	30 30 38 32 38 40			N VA E N VA lows/1 17 3 14 3 16	LUE t)	2	2 000000	NA MC CC	UTUR DISTU DISTU (%)	AL RE NT	1		(m) STAR FINIS (t/so	TED SHED du qm.)	4		3 12 /04/9 TOT. UN WEIC (V/cu.	7 AL T HT 2.0
SOIL DESCRIPTION	COHLEW PA SS SS SS PA SS PA SS PA	ON ETAWVS	40 40 KECOVERY (cm)			N VA	LUE t) 0 40	2	200000	NA MC CC	ATUR DISTU DISTU DINTE (%)	AL RE NT 0 80	1		STAR FINIS (V/so	TED SHED Su qm.)	4		104/9' TOT. UN: WEIG (U/cu.	7 AL T HT m) 20
SOIL DESCRIPTION	COHLISW PA SS SS SS PA SS PA SS PA	ON 27dWVS	40 BECOVERY (cm)	10		E N VA lows/1 17 3 14 3 16	LUE t)	2	2000000	NA MC CC	ATUR DISTU DISTU (%)	AL RE NT	1		2	Su gm.)	4	1.6	104/9' TOT. UN WEIC (Vcu.	AL T (HT 20
SOIL DESCRIPTION	PA SS SS SS PA WEHOD PA SS SS PA SS PA SS PA SS PA SS PA	ON 310WVS 1 2 3 4 5 6 7	30 30 30 32 38 32 38 40 40 40 40 40 40 40 40 40 40 40 40 40			N VA lows/1 0 3 17 3 14 3 16	LUE ft)	2	2000000	N/A CCC 0 44	ATUR MISTU DNTE (%)	AL RE NT 0 80		1	2 2	Su gm.)	4	1.6	UN WEIG (V/cu.	AL TT (HT m) 20
10 20 30 40 50 BROWN (SM, SP-SM) 70 80 90	PA SS SS SS SS PA SS PA SS PA SS PA	1 2 3 4 5 6 7	30 30 38 32 38 38 40		00000000	17 3 14 3 16			000000											
10 20 30 40 50 BROWN (SM, SP-SM) 70 80 90	SS SS SS SS SS SS SS SS PA SS PA SS PA SS PA	1 2 3 4 5 6 7	30 30 38 32 38 38 40		000000000	17 3 14 3 16			0000-0-0-											
20 30 40 50 BROWN 60 (SM, SP-SM) 70 80 90	SS SS PA SS PA SS PA SS PA SS PA SS PA	2 3 4 5 6 7	30 38 32 38 38 40		00000000	3 14 3 16		_	000-0-0-			_	+	-	_	-			_	_
20 30 40 50 50 80 90	SS PA SS PA SS PA SS PA	4 5 6 7	38		000000	14		_	00-0-				\dagger	+	+	+	+-		+	+
30 40 50 50 BROWN 60 (SM, SP-SM) 70 80 90	PA SS PA SS PA SS PA	5 6 7	38		0 0	16		_	0			1	- 1	1						
MEDIUM DENSE FINE TO MEDIUM SAND, BROWN (SM, SP-SM) 70 80 90	SS PA SS PA SS PA	5 6 7	38 40		0 0	16		_	0		_		. 1							
MEDIUM DENSE FINE TO MEDIUM SAND, BROWN (SM, SP-SM) 70 80 90	PA SS PA SS PA	6	40		0		-	-											1	
MEDIUM DENSE FINE TO MEDIUM SAND, BROWN (SM, SP-SM) 70 80 90	PA SS PA SS PA	6	40		0				+		_		-	_	+	+	+-		-	+
MEDIUM DENSE FINE TO MEDIUM SAND, BROWN 60 (SM, SP-SM) 70 80 90	PA SS PA	7	40		- 1	17			Y											
BROWN 60 (SM, SP-SM) 70 80 90	PA SS PA	7			11		-		-		-	-+	+	+	+	+	+		+	+
(SM, SP-SM)	SS PA	7			1															
70 80 90	PA		40		0	13				p				1						
80				-	+			-	-		-	-+	+	+	+		+-	\vdash	+	+
90	SS	8	30		1	18			6											
9.0					1				1			-	1	1	-	-	1		1	T
	PA				11						_			-	-	_			_	-
	SS	9	35		9	16			Î											
10.0	PA			-	1		-	-	-		-	-	-	+		+	+	+	+	+
110	SS	10	35			2	27		0											
11.50						V			-/								1			
12.0	PA				_				4			-	_	_	_	_	-		+	_
DENSE FINE TO COARSE SAND TRACE OF	SS	11	30				9	33	p											
	PA			+	-	-	1	-	$\left \right $	_	-	+	+	+	+	+-	-		+	+
(SP-SM) 14 0	SS	12	30				31		0							1				
									1				T							
15,45 15.0	PA	12	10	-	-	_	Y	_	1		_	-	-	-	-	-	+		-	
END OF BORING REFUSAL	00	15	10				I	38	0											
				-	-		-	1			-	1	+	+	+	+	+-		+	+
																1				
				-	-	-	-	-	-		-	+	+	+		+-	-		+	+
91					1			-				-	1	1		1	1			
สการเรา		q			-		-		-			-			_	_	-		_	-
		_						-				_	1		+	_				
A = POWER AUGERING HA = HAND AUGERING	WO	= WA	SHO	UT	(PO)	PNO	¥	-	ST =	SHE	BCP	CH2		-	S	S = SP	9/1	CHIAN	0 M	I
THE OTHER OF THE RADED I. TEATED U.	0.00			5205				-					-		12					

	~			-	BOR	ING	NO.	81	1-1	_			GF	ROUN	DEI	.EV.(m.)			
BORING LOO PROJECT อาคาร 7 ชั้น หลังโรงแรมเ LOCATION กำเภอเมือง จังหวัดเชียงให	ร์ ชียง ม่	ใหม่	<u>กูคำ</u>		DEP COO	TH (RD.	m.)_	1	2.4	5			DA DA	TE S	VED STAR	WL: (TED	m.) _ 27/ 27 /	- 2.0 8 / /8 /	0 94 94	
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	SAMPLING	10)	S P blow 20	T − N s/ft 30) 40	P	20 4	Wn (%) 0 6	LL 		Su O UC X FV 1 2	(1/0 T 4 T 2 3	n ²) ∆ PP ⊡ TV 4		1 (1) 1,6	f m ³	.0
VERY STIFF CLAY, BROWNISH GREY (CL)	1 _		PA SSE SSE SS PA	1 2 3		9-20	5 24	32	- 9	91										9 79
LOOSE CLAYEY FINE TO MEDIUM SAND, GREY (SC) 5.50	4 -		PA SS	4		10	\$3			•									6	
MEDIUM DENSE SILTY MEDIUM SAND, GREY (SM) 7.00	6 -		SS WO	6		0	20		0											
MEDIUM DENSE CLAYEY SAND, BROWN (SC)	8 - 9 -		WO	8			23	1	0 0		1									
. 12.45	10 -		SS WO	9			50												22	1
END OF BORING		<u>-1.</u>																		
ลถาบ		6						0						0			+			
จุฬาลงก		6	J	9										6						

BORING L PROJECT : <u>ธนาการกรุงศรีอยุธยา</u>	OG สาขาโชตน	1		- 194-55	6	orin Epth Oori	G NC (m.)).) : ; ;	12	BH-	-1			GR WA DA		D ELE LEVE	EV (m EL (m) ED	N) : :	-1.3	0
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	(i	SP blow	T-N s/ft) 30 4	10	۲ 20	<u>ال</u>	wn %) 60	LL 80		S (t/sq) UC (FV1 2	u j.m) T 4 S	∆ P4 ⊡ T\ 4	;	TO UI WE (t/cu 1.6 1.	TAL NIT IGHT (1 u.m.) 8 20
LOOSE CLAYEY SAND WITH GRAVEL BROWN (SC) VERY LOOSE SILTY FINE SAND, GI (SM) LOOSE TO MEDIUM DENSE CLAYEY S TRACE OF GRAVEL, BROWN AND GRI (SC) HARD FINE SANDY CLAY, GREY AND BROWN (CL) DENSE CLAYEY SAND, GREY AND BROW (SC) VERY STIFF FINE SANDY CLAY, GR (CL) MEDIUM DENSE TO DENSE CLAYEY S TRACE OF GRAVEL, GREY AND BROW (SC) END OF BORING	1.00 REY 2.00 2.00 2.00 2.00 3.0 4.00 4.		A 00 4 00 ± 00 ≥ 00 ≥ 00 ≥ 00 ≥ 00 ≥ 00 ≥	1 2 3 4 5 6 7 8 9 10 II					338											
		4	W) = 1	VAS		IT		-	ST	1	HE	LBY	TUR	E	SS	1=5	PUT	SP	
PARTY CHIEF : SAYAN K DRILLER : SA	DAWIIT Y	- 1	GE	010	GIS	T · M	ATT/		×	0	DAE	TMA	N.	~	-	TY	PIS	T ·	VC	

PROJECT : <u>ธนาคารกรุงศรีอยุธยา สาขาโชตน</u> LOCATION : <u>อ.เมือง จ.เชียงใหม่</u> SOIL DESCRIPTION	un			-	COOF					_	-		-1		LEVEL	(m)		.50	
SOIL DESCRIPTION	8			-		RD.	:						D/	ATE S	NISHE	0		6/	95 95
	DEPTH (m.) GRAPHIC	METHOD	SAMPLING	RECOVERY	10	SF (blov 20	PT-N ws/ft) 30 4	40	F	Р <u>Г</u> 0 40	wn (%)	LL 60		(t/s O UC X FV 1 2	Su q.m) T (2 T (2 2 3	A PP TV 4	V (1.	TOT. UNI VEIG T (VCU. 6 1.8	AL T HT m.) 2.0
MEDIUM CLAYEY SAND WITH GRAVEL, BROWN (SC) 1.00 LOOSE SILTY FINE TO MEDIUM SAND, BROWN (SC) 2.00 MEDIUM DENSE TO DENSE CLAYEY SAND, LI-GREY, BROWN (SC) 4.00 HARD CLAY TRACE OF SAND, BROWN (CL) 5.50 VERY STIFF TO HARD FINE TO MEDIUM SANDY CLAY, GREY AND BROWN (CL) 9 10.00 VERY DENSE CLAYEY SAND, GREY AND BROWN (SC) 11.50 DENSE SILTY CLAYEY FINE TO MEDIUM SAND, GREY (SM-SC) 12.45 END OF BORING		20 초 14 전 12 12 12 12 12 12 12 12 12 12 12 12 12	A 1 2 3 4 5 0 6 7 7 8 0 9 0 1																
	0	+	wo =	WA	ASH	OUT	r.	1		ST =	SH	ELB	YT	JBE	S	S = S	PLIT	SPC	20
PARTY CHIEF : SAYAN K. DRILLER : SARAWUT	Υ.	+	GEOL	.00	IST :	NA	TARO	LK	1	DR/	FTN	AN	: 5	5	T	PIST	•	VS.	_

BORING LOG PROJECT : สถานีบริการน้ำมันบางจาก หจเ LOCATION : อ.เมือง จ.เชียงไหม่	า.เชียงใ	หม่ ท	<u>วิช</u> เ	BORING NO. : DEPTH (m.) : 18ชชิศิตร์ :	BH - 1 15.45	GROUND ELEV (m) WATER LEVEL (m) DATE STARTED DATE FINISHED	-0. 90 22/3/91 22/3/96
SOIL DESCRIPTION	DEPTH (m.) GRAPHIC LOG	METHOD	SAMPLING	SPT-N (blows/ft) 10 20 30 40	PL wn LL (%) 20 40 60 80	Su (Ưsq.m) OUCT △ PP XFVT ⊡TV 1 2 3 4	TOTAL UNIT WEIGHT 1 (t/cu.m.) 16 18 20
SOFT FINE TO MEDIUM SANDY CLAY, GREY (CH/SC) 1.00 MEDIUM DENSE CLAYEY - SILTY FINE TO MEDIUM SAND, GREY (SC,SM) 2.70 STIFF FINE SANDY CLAY, GREY (CL) 4.00 MEDIUM DENSE FINE TO COARSE SAND WITH CLAY, GREY (SC)	2 3 4 5 6 7 8 9 10 11	PA SS SS SS SS SS SS WO SS		Q 3 Q 13 Q 13 Q 13 Q 13 Q 13 Q 14 Q 16 Q 15 Q 15 Q 19 Q 19 Q 19 Q 19 Q 19 Q 21			
MEDIUM DENSE TO DENSE CLAYEY- SILTY FINE TO COARSE SAND WITH CLAY , GREY (SC,SM) 15.45 END OF BORING	12- 13- 14- 15- -	W0 SS 11 W0 SS 11 W0 SS 11	2	¢ 15 0 25 40 0	6		
PA = POWER AUGERING HA = HAND AUGERIN	G	wo	= WA	ASH OUT	ST = SHELBY T	UBE SS = SPL	

BORING LOG PROJECT : อาการที่พักคนไข้ 30 เดียง LOCATION : ก่ายกาวีละ อ.เมือง จ.เชียงใ	ไหม่					BORING DEPTH (COORD	NO : (m) :	8 1; 	H-1 3.9	5			GR WA DA1 DA1	OUNC TER TE ST TE FIN	D ELE LEVE TARTE	V (m) L (m) ED	:	-1.50 4/0 5/0	0 5/9 5/9
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	(bl	SPT-N lows/ft	40	F	PL	wn (%) 60	LL 80	C X	S (t/sc) UC (FV 2	5u (.m) T (T 3	∆ PP ⊡ TV 4	1	TOT UN WEIC 1 (t/cu.	AL IT 3HT .m.)
STIFF FINE SANDY CLAY, BROWN (CL-ML) 1.00 LOOSE SILTY-CLAYEY FINE TO MEDIUM SAND, BROWN (SM-SC, SP-SM) 2.50 MEDIUM FINE SANDY CLAY, BROWN (CL/SC) 4.00 LOOSE TO MEDIUM DENSE SILTY SAND, GREY (SM, SP, SP-SM) 13.00 VERY DENSE SILTY SAND/GRAVEL, GREY (SP-SM) 13.95 END OF BORING			PA SS PA SS PA SS WO SS WO SS WO SS WO SS WO SS WO SS WO SS WO SS SS WO SS SS WO SS SS WO SS SS SS SS SS SS SS SS SS S				18 13 0 2 ⁻¹ 0 24 55		1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0										
PA = POWER AUGERING HA = HAND AUGERIN	G		wo)=1	WA	SHOL	Т	t	s	T =	SHE	LBY	TUE	BE	S	S = S		SPC	
PARTY CHIEF : SAYAN K. DRILLER : SANIT S.	Ì.]	GE	OLO	OGI	ST : P	۱.	Ζ	D	RA	TM	AN :	₩ĸ		Τ	PIST	:t	p	

PROJECT : อาการศาล มทบ. และ จทบ. LOCATION : มทบ.33 ก่ายกาวีละ อ.เมือง จ	.เชี	ยงให	าม่			BORI DEPT	NG N TH (m RD.	o :) : ;	Bł 15	1- 1 .45					GROU WATE DATE DATE	JND E ER LE STAF	elev Vel (1 Rted Shed	(m) : m) : :	- 4	1.00 /8/9)/8,) 95 /95
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	SF (blov 20	PT-N ws/ft 30	40		PL	wn (%)	0 8		(L O L X F 1	Su /sq.n JCT TVT 2	n) 	PP TV	۲ ۱.6		AL T HT m.) 20
CONCRETE SLAB 0.07 MEDIUM DENSE CLAYEY FINE TO MEDIUM SAND, LI-BROWN (SC/CL) 1.00 LOOSE SILTY FINE TO MEDIUM SAND, LI-BROWN (SM, SP-SM) 4.00 VERY STIFF CLAY, DARK GREY (CL) 8.50 MEDIUM DENSE TO DENSE SILTY FINE TO COARSE SAND, YELLOWISH BROWN (SM, SP-SM) 13.00 DENSE SANDY GRAVEL, GREY, BROWN (GP-GM) 15.45 END OF BORING									32												
PA = POWER AUGERING HA = HAND AUGERIN	G	-	w	0 = 1	WA	SH	DUT	-	1	H	ST :	= SH	ELE	BY T	UBE	+	SS =	SP	LIT	SPO	ON
PARTY CHIEF : KOBKIT T. DRILLER : SERM J.		10	G	OL	OGI	ST :	PI		57		DR/	FT	MAN	1:w	к	T	TYP	IST	: 1	JS	

BORING LO PROJECT : ก่อสร้างทางแยกต่างระดับเ LOCATION : อ.เมือง จ.เชียงใหม่	DG าลุ่มที่	37-1	1				BORI DEP1 COO	ing f Th (m RD.	40. n.) :	: AB : 4	H- 9. 6	1				GRC WA DAT	TER TE ST TE FI	D ELI LEVE TART NISH	EV (EL (m ED ED	m): 1): :	-14	0.70 /7/ /7/) /95 /95
SOIL DESCRIPTION	1	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	Si (blo 20	PT-N ws/f	N t) 40		PL 20	wi (%	n L) 50 8	<u>.</u> 0	G X 1	(t/s4) UC (FV	Su q.m) T T 2 3	∆ F 0 1 4	< *	1 W (1		AL IT SHT m.) 20
MEDIUM DENSE SILTY FINE TO COAN SAND, BROWN (SM) LOOSE TO MEDIUM DENSE FINE TO	RSE 1.50	1		PA SS PA SS PA SS	1 2 3		4	Ø 1 4	5	4	000												+
COARSE SAND, BROWN (SP,SM)	3.50	3-4-		84 55 90 55 90	4		-		8		ł									_			
MEDIUM TO VERY STIFF CLAY TRACI FINE TO MEDIUM SAND, GREY AND BROWN (CL,CH)	8	6 -		SS WO ST WO	6				8			0 0						1	7.00	5 0	,		0
		8-		ST WO	2							0						15	.43		, +		
MEDIUM DENSE TO DENSE CLAYEY/S FINE SAND, GREY AND BROWN	LTY	10- 11- 12-		SS ST WO SS	738		a	~	4	30	4									_			
(SM, SC/CL)	4.00	13 - 14 -		WO SS	9		+	-	1	30	6								_	_			+
LIGHT GREN (GP-GM) 1. VERY DENSE SILTY SANDY GRAVEL,	5.00	15 - 16 -	0 8 0	\$\$ ₩0	10				0 2	28	0						_	_		_			+
LIGHT GREN (GP-GM)	8.50	17	8	wo ss	12				×	2/0								_		_		$\frac{1}{1}$	
(CL) 2 PA = POWER AUGERING HA = HAND AU	0.50 JGERIN	20 - G		SSO W	13 4 0 =	WA	SH	4	r			ST	= SI	HEL	BY	TUE	BE	S	iS =	SP		SPC	
PARTY CHIEF : KERKSIN I. DRILLER : SAI	NEH K.		6	G	LOL	G	51	ľ	3	1		UR	Ar I	MA	14:1			E	TPI	01	W	S	

BORING LOG PROJECT : ก่อสร้างทางแยกต่างระดับกลุ่มที่ 37–1 LOCATION : อ.เมือง จ.เชียงใหม่							BORING NO : <u>ABH - 1</u> DEPTH (m.) : <u>49.65</u> COORD. :							GROUM				XOUND ELEV (m) : ATER LEVEL (m) : TE STARTED : ATE FINISHED :			- 0.70 14/7/95 17/7/95			
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	상품 SPT-N (blows/fi 권 10 20 30			SPT-N (blows/ft) 10 20 30 40			PL wn LL (%) 20 40 60 80					Su (t/sq.m) OUCT △ PP XFVT ⊡TV 1 2 3 4			PV	TOTAL UNIT WEIGHT T _t (t/cu.m.) 16 1.8 2.0			
VERY LOOSE TO LOOSE SILTY SAND DARK GREY (SM)	, 21- 22.00		₩0 SS	14		10	12							_	_		_					+		
DENSE TO VERY DENSE SILTY SAND GRAVEL, LIGHT GREY	Y 23-	4 0	WO SS	15				50/	2				-	_										
(GP-GM) 24.00 VERY STIFF TO HARD CLAY, GREY	25-		ss wo	16			V	ø	34		0	_	_	_		-	_			-	-			
(CH)	28.00 28		wo	17							10		-											
VERY DENSE SILTY GRAVEL, SILTY SAND, LIGHT-GREY (SP-SM,GP-SM)	29		SS WO SS WO SS	18 19 20				50/ 60/ 50/	6		P •		_	_										
	32		wo ss wo ss	21	N. N. N.			50/	a, 15	0		_		_										
	34- 35- 36-		wo ss wo	23				50/	50			_		_										
3 VERY DENSE SILTY SAND/GRAVEL, BROWN	7.00 37 - LIGHT 38-	t o	ss ₩0 ss	24				63/ 5C/	10					_	•			+				+		
	39- 40-	0	WO SS TC	26		SHO		63/	50	0	ST =	SH	ELF	37.	TUB	E	. Si	S =	SPI	JTS	SPO			
PARTY CHIEF : KERKSIN I. DRILLER : SA	DRILLER : SANEH K. GEOLOG						IST : DRAFTMAN						1 : Y	WK TYP			YPIS	T: ws						

BORING LOG PROJECT : ก่อสร้างทางแยกต่างระดับกลุ่มที่ 37-1 LOCATION: อ.เมือง จ.เชียงใหม่							: <u>AB</u> : <u>4</u> :	<u>- 1</u> 9.65				GROUND ELEV (m) WATER LEVEL (m) DATE STARTED DATE FINISHED				-0.70 14/7/95 17/7/95			
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	SPT (blows		PL (wn %) 60	60 60	Su (t/sq.m) ⊙uct △ pp XFVT ⊡TV 1 2 3 4				TOTAL UNIT WEIGHT T ₁ (Vcu.m.) 1.6 1.8 2.0				
VERY DENSE SILTY SAND/GRAVEL, LIGHT BROWN (SM/GM) 49.65 END OF BORING	41			2Z 23 30 31 32			50/4" 50/4" 50/6" 50/6"												
ลถาบน	0				C					2									
PA = POWER AUGERING HA = HAND AUGERI PARTY CHIEF : KERKSIN I. DRILLER : SANEH K	PA = POWER AUGERING HA = HAND AUGERING WO = W PARTY CHIEF : KERKSIN I. DRILLER : SANEH K. GEOLOGI						1	t	ST =	SHE	LBY	WK	BE	SS	LIT SPOON				
BORING LOG PROJECT : ก่อสร้างทางแยกต่างระดับกลุ่มที่ LOCATION :อ.เมือง จ.เชียงใหม่	37-1	BORING NO. ;A DEPTH (m.) ; COORD. ;	BH-2 31.95	GROUND ELEV (m) WATER LEVEL (m) DATE STARTED DATE FINISHED	- 2.40 12/7/95 21/7/95														
--	--	---	-----------------	---	--														
SOIL DESCRIPTION	DEPTH (m.) GRAPHIC LOG METHOD SAMPLING	상공 SPT-N (blows/ft) 10 20 30 4	PL wn LL (%)	Su (t/sq.m) ⊙∪CT Δ.PP X.FVT ⊡TV 1 2 3 4	TOTAL UNIT WEIGHT T _t (t/cu.m.)														
MEDIUM DENSE CLAYEY FINE TO COARSE SAND WITH GRAVEL, YELLOWISH BROWN (SC) 3.00 VERY STIFF CLAY, GREY AND BROWN (CL) 8.30 DENSE SILTY FINE TO MEDIUM SAND, DARK GREY -(SP-SM)	PA SS 1 SS 2 PA SS 2 PA SS 3 SS 4 SS 4 SS 6 WO SS 6 WO SS 6 WO SS 6 WO SS 6 WO SS 6 WO SS 6 WO SS 7 SS 7 WO SS 6 WO SS 6 SS 7 WO SS 6 SS 7 WO SS 6 SS 7 SS 7 WO SS 6 SS 7 SS 6 WO SS 6 SS 7 SS 7 SS 7 SS 6 WO SS 6 SS 7 SS 7 SS 7 SS 7 SS 6 SS 8 SS 6 SS 8 SS 8	ρ 15 10 17 11 19 12 19 13 19 14 19 15 10 17 11 19 12 10 12 10 12 11 13 12 13 13 14 14 15 15 15 16 17 17 11 13 12 14 15 15 15 16 15 17 11 14 15 15 15 16 15 17 15 18 15 19 15 19 15 10 15 10 15 10 15 10 15 15 <		13.69 0															
12.45 DENSE SILTY SAND GRAVEL, LIGHT GREY (GP-GM) 15.45 HARD CLAY, YELLOWISH BROWN (CL) 17.00 CLAYEY GRAVEL (GC) 18.00 HARD CLAY TRACE FINE SAND, BROWN AND GREY	12 - 35 9 13 - 6 W0 14 - 6 SS 10 15 - 6 SS 11 16 - 7 W0 17 - 8 W0 17 - 8 W0 18 - 5 SS 12 19 - 5 SS 13 19 - 7 W0	4 4 9																	
PA = POWER AUGERING HA = HAND AUGERING PARTY CHIEF : KERKSIN I. DRILLER : SANEH K.	20 SS 14 W0 3 W0 = GEOL	WASH OUT	ST = SHELBY 1	TUBE SS = SPI															

BORING LOG PROJECT : ก่อสร้างทางแยกต่างระดับกลุ่มที่ LOCATION: อ.เมือง จ.เชียงใหม่	37-	1				BORING DEPTH COORD	3 NO. (m.)	:A <u>B</u> : <u>3</u> 1 :	H-2	5			9 × 6 6	ATER ATE ST ATE FIN	D ELE LEVE TARTI	EV (r EL (m ED ED	n):): :	- 2 12 21	2.40 /7/ /7/	<u>'95</u> '95
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	(b	SPT-I olows/1	N ft) 40		PL	wn (%)	LL 80		S (t/sc O UC X FV 1 2	би q.m) т т 2 з	∆ P ⊡ T 4	P V	T W (t 1.6		AL T HT n.) 20
DENSE SILTY FINE TO MEDIUM SAND AND CLAYEY FINE SAND, LIGHT GREY (SM,SC) 24.10 HARD CLAY, DARK GREY	21 22 23 24 25-		WO SS WO SS WO	15			4	2 A												
(CL) 26.00 VERY DENSE SILTY FINE TO COARSE SAND, LIGHT GREY (SM)	26- 27- 28- 29-		SS WO SS WO SS WO	18 19 20			5	31	200											
VENG DENSE SILTY SAND/GRAVEL, LIGHT GREY (SM/GM) 31.95 END OF BORING	30-	000	SS WO SS	21			5	0/4	00	, 										
สถาบัน																				
PA = POWER AUGERING HA = HAND AUGERI	NG		W	0 =	W	ASHO	UT		+	ST	= SH	ELB	Y TI	UBE	S	S =	SP		SPC	NON
PARTY CHIEF : KERKSIN I. DRILLER : SANEH	۲.	1.5	G	EOL	.OG	IST :				DR/	AFTI	MAN	WW H	¢	T	YPI	ST	W	S	

	GEOTECHNICAL ENGINEERING DEPARTMENT OF CIVIL EN CHIANG MAI UNIVE	ELABORATOR NGINEERING RSITY	ť	Subsoi	il Profile	
Client: Project: Locatio	บริษัท ธำรงก์พัฒนาการ อาคารเรียนและปฏิบัติการมนุษย์ศาสตร์ n คณะมนุษย์ศาสตร์ มหาวิทยาลัยเซียงใหม่		Date: Water Water	15/12/98 Level: initial (m.) -20.3 Level: final (m.) -7.0	Job No. Bore Hole: Sheet No. :	461/41 BH1 1 of 2
Elevation (m.) Profile	Soil Desciptions	SPT N-Va	ue	 Natural Water Content (%) 	Undrain Streng (t/r	ed Shear (th, Cu (n2)
0	Ground Surface at Elevation +0.04 m.	0 10 20 30 4	0 50 (0 10 20 30 40 50 60	0 10	20 30
	SILTY SAND (SM): yellow brown and grey brown changing to light grey with yellow brown mottled at bottom of layer, fine to medium sand, non plastic clay, LOOSE to MEDIUM.	o o				
-5.0. 	SANDY SILTY CLAY (CL): grey brown with yellow brown mottle, medium to high plasticity, STIFF to VERY STIFF, fine to medium sand. Occasional lenses of clayey sand wih traces of mica			PL LL	0	0
-10					0	
-12						
·14	สกาบับก					
-18						
-20	M 10/11/92				IJ	









G G	EOTECHNICAL ENGINEERING	LABORATOR	Y	1.1.1.1	i a and		
rofile	DEPARTMENT OF CIVIL ENG	GINEERING			Subsoi	l Profile	?
E C	CHIANG MAI UNIVERS	SITY					
Client:	บริษัท ธำรงค์พัฒนาการ		Date:	L	8/12/98	Job No.	461/41
Project:	อาการเรียนและปฏิบัติการมนุษย์ศาสตร์		Water 1	Level: initia	l (m.) -16.7	Bore Hole:	BH3
Location	: คณะมนุษย์ศาสตร์ มหาวิทยาลัยเชียงใหม่		Water I	Level: final	(m.) -9.7	Sheet No. :	2 of 2
i				o Natura	al Water	Undrain	ed Shear
	Soil Desciptions	SPT N-Va	lue	Co	ntent	Streng	gth, Cu
				(%)	(t/i	m2)
-20	<u> </u>	50 60 70 80 5	0 100 0	10 20 30	40 50 60	0 10	20 30
0,0	CLAYEY SAND (SC); light grey, medium			-		-	F
.0	to coarse sand, MEDIUM to DENSE, slightly	to		J			
25	plastic, traces of fine to medium		<u> </u>	- 11 -			6
-22.	size angular shape gravel, traces of mica.	·					· · • • • •
*o	Occasional thin layers of stiff to					÷	1
18	very stiff sandy clay.			- /		1 I	
-24 0				- /	3	1 -	1
00		6					
-24.9				PL	LL		
1	SILTY CLAY (SC) agrey with light yellow			10		-	
-26	brown, medium plasticity, STIFF.			þ			
				6		[i	
-28	End of Borehole at Elevation 27.42 m.						
1 1							
1 1						1	
				-		-	
-30					· · ·		
				-			
				.		-	
1 C - 3		ł		-			
-32					···· •		
				-			
				-	;	-	
				-		-	
-34							
				-	1.15	T i	
				-		Г	1
				-	1		
-36	สภายยา			311		1	
1				-	0	1	1
				-			
		1			1	4 F 3	:
-38		2 61 6 19 2		19713		12	
	YN I DYNII d b i				1.14		
-40							

L					LOG (DF E	801	RIN	IG	No). I	3H-1					
PI	RO.	IEC.	г:	ι	เกรนด์ริมปิง สูง 4 ชั้นและใต้ดิน 1 ชั้น		L	OCA	TIO	N :		ถ.ฟ้าอ่า	ນ ຈ.ເ	ชียงใ	หม่		
CI	LIE	NT:															
O DEPTH,m	SAMPLE No.	TYPE OF SAMPLE	SAMPLE DIST	RECOVERY	DESCRIPTION OF MATERIAL		GRAPHIC LOG	0 × △	Nai Pla Liq	tural V stic Li uid Lin (%) 40	Vater (imit nit 60 E	<i>Content</i>	0	O Su △ Su x Op 2 □ SP	(UC) (Fv) 5/2 (t/ 1.5 T ,N (E	● Su ▲ Su m ²) 5 7 Slow/ft) 40	'(UC) '(Fv) 7.5 60
H			L		1.	.00 m.								2 			
F	01 02 03	55 55 55			Silty fine SAND but fine to medium grain @ SS-3, brown, loose. (SM)	.80 m.		G	9		-			⊒ 4 □ 9 □ 8			
Ħ	04	SS	F		Clayey fine SAND some medium sand, loose. (SCI	grey,			b					4.			
5	05	SS			Silty CLAY trace fine sand, brownish grey, medium stiff. (CL)	.50 m.			-	-	-		_	de			
Ħ	06	55				.00 m.			1						222		
F	07	SS	-		Silty CLAY trace fine sand, greyish brown, very stiff. (CL)				~						6		
10	08	SS							0						21		
H	09	SS	E		1	0.50 m.		4	1						030		
	10	SS	5		Medium to coarse SAND but medium s grain @ SS-9, some gravel @ SS-11 &	and		•							D 24		
15	11	SS			grey, medium to dense. (SM-SP)	20		1							1 26		
H	12	SS			1	6.50 m.		•		-					6	36	
	13	55			Fine to medium SAND trace coarse sar and gravel, grey, very dense, (SM-SP)	nd		9									50/4
				Π					-		-	+	+		-		1400
Ĥ	15	SS				9.95 m.	17.61	6	+								73 0
111111411					Silty CLAY some fine sand and root, brown. (Top Soil) (B) Fine to medium sandy CLAY, brown, m stiff. (CL) (C) Silty CLAY trace fine sand, grey, medium stiff. (CH) 1	nedium	2	1				9					
7		ົ	6		BORING STA	ARTED.	10/0	8/94		RIG.	ACK	ER	m	-1.3	м.	AFTER	24 HRS. BORING.
		7			BORING FIN	ISHED.	10/0	8/94		FORE	MAN	SR.	JOB	No.	\$820		

								NO.	DIF2				
PRO TIT	JEC.	1:		แกรนครมบง สูง 4 ชนและ โคคิน 1 ชน	L	DCA	TION :	,	ถ.ฟาฮาม	จ.เชียง	เหม		
SAMPLE No.	TYPE OF SAMPLE	SAMPLE DIST	RECOVERY	DESCRIPTION OF MATERIAL	GRAPHIC LOG	O × △	Natura Plastic Liquid (%	Wate Limit Limit () 60	r Content 80 100	O Si △ Si × Q	u (UC) u (Fv) p/2 (t/n 2.5 PT ,N (Bl	● Su' ▲ Su' f) 5 7. low/ft)	(UC) (Fv) .5
				(A) 1.00 m.						·			
01 02 03	SS SS SS			Silty CLAY some fine sand, brown, medium stiff.			9			4 5	115		
04	SS			Silty fine SAND, brownish grey, loose. (SMI)			þ			03			
05	SS			Silty CLAY trace fine sand, brownish grey, medium stiff.			×	-		- 45			
06	SS			6.00 m.							919		-
07	SS	-		Silty CLAY trace fine sand, greyish brown, stiff. (CH))							18		
08	SS			Clayey fine to medium SAND, grey, medium dense. (SC) ¹		-					13		
09 10	ss ss			GRAVEL and SAND mixture, trace clay pocket, grey, medium dense. (GM-GU)		1		-			19		
11	ss	23		13.50 m. Medium to coarse SAND some gravel, grey, loose. (\$9)*		-							
12	SS		-	GRAVEL and SAND mixture trace clay pocket, grey, medium dense. (GM)#		9					5125		
13	SS			(C) 17.30 m.		->		-			15		
14	SS			9		1					030		
15	SS			SAND and GRAVEL mixture, li-grey, dense		-							5
16	SS					-			++				50/6
17	SS	38		22 95 m		\Box							
				(A) END OF BORING Silty CLAY some fine sand and root, prown. (Top Soil) Fine sandy CLAY, brown, medium to stiff. (Jac			d o d		15				
	6	1		Silty CLAY some fine sand, dark grey, stiff (CL)	08/00	8/94	RIG	. AC	KER	WL1.0	95 M.	24 FTER BO	4 HRS ORING
				BORING FINISHED.	09/0	8/94	FOR	REMAN	SR.	JOB No.	4820		

						LOG OF B	801	RIN	IG	No). B	H-3				
PR	oJ	ECT	:	L	กรนด์ริมปิง สูง 4 ชั้นและใ	ต้ดิน 1 ชั้น	L	OCA	TIO	N :	ຄ	.ฟ้าย่าม จ	เชียงให	าม่		
CL	E	T:														
O DEPTH,m	SAMPLE No.	TYPE OF SAMPLE	SAMPLE DIST	RECOVERY	DESCRIPTION OF	MATERIAL	GRAPHIC LOG	0 × 4	Nat Pla: Liqu	tural W stic Lin uid Lin (%) 40 0	Vater C mit nit 50 B	Content 0 100 -	O Su △ Su × Qp 2 □ SP	(UC) (Fv))/2 (t/i 1.5 T ,N (E	● Su' ▲ Su' m ²) 5 7 Blow/ft) 40	(UC) (Fv) .5 60
H					(A)	1.00 m.										
	01	SS		-	(8)	1.50 m.	14		Q				Q8			
H	02	SS	-		Clavey fine SAND (C)	2.20 m.		-	-	-			48			
1	~				(D)	3:28 m	X	_	1	_			14'			
-	04	SS		-	Silty fine SAND trace to so	me medium			R				d a			
-					Sund, Diomin, 10056. (300)*	4.50 m.			1				11	1		
5	05	SS	0,0		Silty CLAY trace fipe sand, medium stiff	grey,			×	-			45-			
-						6.00 m.	1/		11							
-	06	SS					11		10				1 5	19		
-					Silty CLAY trace fine sand,	brownish			1		-			<u>∧</u>	-	
_	07	SS			grey, stiff to very stiff. (C	7.	11	-	-	-				126		
-							V//							X		
_	08	SS	50		Clayey fine SAND trace to	some medium	15		6				q	15		
10					sand, grey, medium dense.	(SCF 10.50 m		-			-		+		+	
	09	SS	2.4			10.50 11.	ΠĤ	6	-					122		-
-														1		
	10	SS	20						1-	-				16	1	
_								1	-		-		+			
-	11	SS	1										1 4.			
-					Medium to coarse SAND so	me gravel, trace		T					1			
4	12	22	100		to some fine sand, grey, m	edium to		+					-	100		
	"	00			rely delise. (and of p			ſ						1953		
4		~~														L
	13	33		1				19	-	-	-		1	-		161
4								1	-	-	-				-	11-
	14	55		-				P .			1					1066
0	15	SS		_		19.58 m.		6								50/3·
-						F BOKING		-	-							
-	1				(A)	Dennie					1		-			
4	1				Silty CLAY some fine sand	with root,					1					
					(B)			-	-	-	1		1	1	-	1
-					Clayey fine to coarse SANI brown, loose, (SC)) and gravel,		-	-	-						
					(C)											
5					brown, medium stiff. (CL)	greyish										
					(D) Silty CLAY trace fine cond	brown			-	+	-		1		+	1
			1		etiff (CI)	510 WII,			1	_			h	1		
					หาลงอย	BORING STARTED.	10/0	8/94		RIG.	ACK	R	VL2.3	35 M.	AFTER B	4 HRS. ORING.
					161711	BORING FINISHED.	10/0	8/94		FOREN	MAN .	SR. J	08 No.	4820		

GRC	UND W	ATER OBSER	NOITAVE	N.			V	V.A.C	. BO	RIN	IG LOG			BORI	NG NO.	1
DATE	TIME	EL. of HOLE	EL.of V	VATER		-			CULTURE	DO	10 110 000			SURF	ACE ELV.	
					10	CA	TIOI	N.	SUANP	ROI	NG HOSPI NGMAI	TAL		DATE	START	
24 HF	R.AFTER	BORING.	2.51	М.										DATE	FINISH	
sc	DILS C	ESCRIP	TION.		IL PROFILE	MPLE TYPE NO	ертн., м.	STA PENE	• NDARD TRATIO	Ν.	LIQUID O- PLASTIC X NI MOISTU	LIMIT. C LIMIT. ATURAL RE CONT	ENT.	△One I Compr ■ peak INSITU © One Pene ● SEN	nalf Uncon essive Stre u vane Stre vane Stre half Pocke trometer F	fined- ength. ded IEAR. DEN: tt - Rdg. Yd,
					SC	S	ä	BLO	NS / FT			0/0			KSC.	T/N
							0.00	30	60		20 40	60 8	0	1	2	3 1.
Soft silty of Very brown i very fi Dens very fi gravel Harri	medium clay. stiff silty c. ine san e greyi ne san . SC	to hard lin a dark grey CL to hard lin lay, occas d and grave ish brown o d, trace o	rish bro ght gre sional el.CL clayey of	0.20 own -1.65 ey and /2.50 /3.00		-55.1 -55.2 -55.3 -55.4 -CS.5	5.00-	8 252	•39 57					8		•
and br	own sil	ty clay, d and gray	occasio vel.	onal		SS. 7	5.00		39		-	ľ			8	
	CI	& ML-OI		-8,30		-SS. 8		25								
Dens clayey gravel	e grey: very f	ish brown Tine sand, SC	and bro occasi	own Lonal	2222	-55.9 -55.10	10.00-		48	-	×					
Hard clay.	to ve	ry stiff b DL & CL-M	rown s: L	ilty	Ħ	-SS. 11 -SS. 12	No No	25	45		×			•	∆ 8	•
				15.00		-SS.13	15.00-		48		×					
Dens yellowi very fi	e to vi .sh broi ine san	ery dense wn and bro d, occasio	grey, wn silt onal co	ty barse		-SS 14			82	1	*					
sand a	nd grav SM	el. 1 & SW-SM				-55.15			90		*	2				
					9 0 9 0 9 0	-SS.16	20.00		9	3/	×					
	END	OF BORT	NG.	21.45		SS.17				73	×					
				Q		Å										
				۱	1	6	25.00	1			วิก	15				
							5				9					

GROUND WATER OF	BSERVATION.			W.A.C.	BORI	NG LOG		BORING	NO.	
		LO	CATIO	N.	CH	ONG HOSPI IANGMAI	TAL	DATE ST	TART	
24 HR.AFTER BORING	a. 3,13 M.							DATE P	INISH	
			NO.	•			LIMIT.	∆One half Compress	Unconfined- sive Strength.	TOTAL-
SOILS DESCR	RIPTION	SOFILE	E TYPE	STAND	ARD- ATION.	O- PLASTIC	LIMIT.	SOne hal	ANE SHEAR. 1 Pocket -	DENSITY.
		SOIL PI	DEPTH	DI OWE	/	X MOISTU	RE CONTENT.	SENSIT	TIVITY	γd,γw ●-Ο
			0.00	30	60	20 40	60 80		2 3	1 2
Top so Soft dark grey s occasional very fir	ilty clay, ne sand. CL	404	S. 1 S. 2	2		2-4				
Very stiff to har yellowish brown and clay, occasional v	2.15 rd light grey, brown silty ery fine sand.		55.3 55.4 55.5	26 4	0			4	0	• • •
Medium light gre clayey very fine s SC	y and brown ^{4.50} and.	2225	S.6 5.00	26		*	•			•0
Hard grey and bro of very fine sand.	wn silt, trace CL 6.90	AL S	S.7 S.8	56	7	×			0	••
Dense to medium very fine sand, co sand.	grey and brown ccasional coarse		S.9	-23		×		×		•0
. SM & SP	-SM		S.10		4	****				•
Dense yellowish h very fine sand. SM Hard grevish bro	wn silty clay,	S	S.11	634		*	.	Δ		•
trace of very fine ML-OL &	sand.	5	IS.13 15.00	4	5	9.				
· .	16.20	- s	IS 14		136	×				- 100
Very dense brown fine to coarse san gravel.	d, occasional	5	S.1 5		110	×				
SM, SW-SM	& GP-GM	S	S.16 20.00		98	*				
END OF	BORING.				0,					
6	เลาเ		25.0	1	11	161	ากร			

จุฬาลงกรณ์มหาวิทยาลัย

GRO	DUND W	ATER OBSE	RVATIO	N.			,		~	POP			0			BO	RING	NO.	1		
DATE	TIME	EL. of HOL	EL.of V	WATER				N.A	.0. 1	BUN	INC	1 10	G			SUF	RFAC	EELV			
				_	1.				PRO	V. PC	DLIC	EOF	FICE	Ξ		DAT	TE ST	ART			
24 HF	R.AFTER	BORING.	2.85	М.			110	N. (CHI	ANGM	AI					DAT	TE FI	NISH			
sc	DILS	DESCRIP	TION.		SOIL PROFILE	SAMPLE TYPE NO	DEPTH., M.	ST PEN BL	AND. NETRA	ARD- ATION.	• • ×	LIQUI PLAS	D LIM TIC LII NATU TURE 0	IT. MIT. RAL CONT	ENT.	∆One Comp ∎ pea INSI1 ⊗One Per ● SE	e half I pressi ak I U VA e half hetrom NSITI	Uncon ve Stre remol NE SH Pocke neter F	fined- ength. ded tEAR. tEAR. et - Rdg.	TC DEI Yc	DTAL- NSITY. d, y w
							0.00		30	60		20 4	40 6	50 8	30	1		2 :	3	1	. 2
Ver brown occasic	y stiff and bro mal ver	to medium ownish grey ry fine san cl.	greyist silty d.	_0.50_ 1 clay,		5.01 500 500 500 500 500 500 500 500 500 5		27	31-		8							P	Δ	•	+0 -0 +0 +0
Me fine to sand c	edium tr fine so and gro SM	o very dens Ind, occasio Ivel. & SP-SM.	se grey nal coo	_550. Y very Irse		+507 +507 +508 +509 +510 +511 +512 +514	5.09- 10.09- 15.09-	21	47 • 30 • 52 • 59												
	END	OF BORING	5	18.45		-55.15	_			60-	*	-								\square	+
		র	ń	e Y	9		20.00	Л	e	9					Ā						

จุฬาลงกรณ์มหาวิทยาลัย

GROUND WATER OBSERVATION]		v		NG LOG	BORING NO.	
DATE TIME EL. OF HOLE EL. OF WATER	·					SURFACE ELV.+0.20	m.
	LO	CAT	101	VOV LAI S	QUARE BRANCH	DATE START 4 / 10 / 3	19
24 HR.AFTER BORING. 0.13 M.				AMIPHUE M	ICANO, CITILINO MAI	DATE FINISH 57 107 5	
				•	- LIQUID LIMIT.	∆One half Unconfined- Compressive Strength.	OTAL
	Щ	PEN	3	STANDARD	O- PLASTIC LIMIT.	■ peak□remolded INSITU VANE SHEAR. DE	NSITY.
SOILS DESCRIPTION	PROF	LE TY	H. M.	PENETRATION.	NATURAL	One half Pocket - Penetrometer Rdg.	
	SOIL	AMP	DEPT!		* MOISTURE CONTENT.	SENSITIVITY	•-0
		-	0.00	BLOWS / FT.	% 20 40 60 80	KSC. 1	T/M. ³
Fill soil		-					i i
Stiff to hard yellowish brown and	Is	SS.1		e ¹²	- 0 x •	e	•••
brown silty clay, occasional very fine sand.	As	SS2		-26	- 0	. 8	••
	As	E.Z		57	• •	ο Δ	•••••
Very stiff to hard light arey vellow and							
brown silty clay, occasional very fine	1 s	S.4 4	5.00	4 37 - · ·	- 0 ; • • • • • •	••••••••••••••••••••••••••••••••••••••	•••••
CL							
		5.5		1		۰.	
Very dense to dense light grey clover	735 S	S.6	1				-•0
very fine sand.	553						
9.60	755-5	S.7	10.00	46		1 1 10 A	••
Dense light grey very fine sand.	+I+	8.22	10.00	32	- ×	x 7	•••
ѕм	+ 7						
Very dense brown and brownish yellow	s	5.9	16	51	*		•••
sand SM & SP-SM		510		EE	ί.		
14.10		55.10	11		Î		
nard light grey and brown silty clay, occasional very fine sand.	Is	s.n 1	15.00	72	- 0ו	φ	•••
CL							
END OF BORING	rZ-s	S:12		● 68	0		• •0
, Southand							
~~					M		
() () () () () () () () () ()		12	20.00				
สกาเ			٦	9/1919	15005		
			d	I L L			

จุฬาลงกรณ์มหาวิทยาลัย

GROUND WATER OBSERVATION.			V	V.A.C	. вс	DRI	NGI	.OG			BO	RING	NO. ((2)	
DATE TIME EL. OT HOLE EL.OT WATER				VO		10			ANICI		SUP	TE ST	ART	+0.20	1/30
24 HB AFTER BORING 0.20 M	LC	CA	TIOI	N. AN)F N	ALIANO		ENG	ΜΔΙ	DA	TE FI	NISH	6/10)/ 39
			-					, 0111		1.0.4					
					•		- LI	QUID L	IMIT.		∆On Com	e half i pressi	Uncon ve Stre	fined- ength.	
	щ	E NO		STA	NDAR	D.					■ pea	akD		ded	DENS
SOILS DESCRIPTION	OFIL	ТҮР	ź	PENE	TRATI	ON.	O- PL	ASTIC	LIMIT		@One	e half	Pocke	et -	
	PR	IPLE	ΞĦ				×	NA	TURAL	TENT	Per	netron	neter F	Rdg.	Yd.)
	SOII	SAN	DEP	BLOW	NS / I	ET	MIC	1510	NE COM	NIENI.	• SE	NSITI	VITY		TIM
			0.00	30	60		20	40	60	80	-	1	2	3	1
Fill soil 0.50	舅	_				_				1		•			Ť
Stiff dark brown silty clay.	1	-55.1		•10-			0								
CL 2:00	155	cc 2													
fine sand. sc 300	555	33.2		25			N.								
5.00	1	EZ2		4	8		- 07	•					1 990, 19 00 , (41), (4)		2.11.00
Very stiff to hard brownish yellow,	1	-					i								
brown and light grey silty clay, occasional		SS.4	5.00		41		OX	•							
CL															
	1	·SS.5			54.		- 4	•							
7.40-	4		2.4		1		i								
Very dense to dense light grey and	355	·SS6	1		1	63	X	•							
brown clayey very fine sand.	25		7				1								
- SC	555	·SS.7		- 4	1		1	•		1		and a			
Dense light grey very fine sand.	-I-I-		40.00				1								
SM		33.8	.66.	R	1		X					to a life		++	
Very dense hownish vellow very fina	ŧ.	0.23	12		1		1								
to fine sand. SP-SM 12.85	Į.	22.7	101	5	1		ĩ							1.005	
11-11-11-11-11-11-11-11-11-11-11-11-11-	1	5510	20		1	64	al								
riard brownish yellow and light grey silty clay, occasional very fine sand	1	55.10			Π	~	1							8	
	1	SS11	-15.00			69	OX								
CL	1	00.11													
16.95	1	CS.12					-0*			_					
END OF BORING															
			20.00												
		1	-20.00				1								
ิลกาย	9	1			2	91									
. D.I. D		0	0.1		1	Ц				0					

จุฬาลงกรณมหาวทยาลย

BORING LOG						BOF	UNG	NO.	:			BH-	1		_	ELE	VATI	ON (#	n)	1				
PROJECT : SELF SUPPORT TOWER					.	DEP	TH (m)	:			25.4	5		_	GW	L (m)			: .		-3.20		
LOCATION : A MUANG, CHIANGMAI				_	.	cod	ORD.	N	:				_		-	DAT	TE ST.	ARTE	D	: .	_			
							-	E	:		1			-		DAT	EFIN	ISHE	D				-	_
		8		ä	î							N	ATU	RAL								TOT	TAL	
	Ĕ	10 K	DOH	NA	RY (c		SPT	NVA	LUE			M	OIST	URE			·	Su				UN	Ш	
SOIL DESCRIPTION	EPTH	APH	MET	MPL	OVE		(6	lows/	ft.)			c	ONT	ENT			a	/sq m	1)			WEI	GHT	
	a l	18		SA	REC		о 17 ла		20 10				(%)								(Va	1m.)	
	+	1111	PA			1	0 2	0 3	0 4	0	-	1	T	50 1	BO	-		2 3	4		- 1	6 1.	8 2	0
HARD FINE TO MEDIUM SANDY CLAY, GREY	1.0		SS	1	10				9	38	9													
(CL) 1.5			SS	2	20				0	38	9		T	T	1									
	2.0		SS	3	10		D	17			9	1-	-	_	_	1					_			
			SS	4	15	9	8					9												
STIFF CLAY, GREY	3.0	-	22	5	25	-	0	-	-		-		+	+	+-	+-	-	-			-	-		0
(CL)	40			-		Ĭ	Í					1												
45			wo			1			-			11	T	1	1									
	5.0	Π	SS	6	20	0	8				_	9		L									_	-
LOOSE CLAYEY FINE SAND, GREY	1		wo									11												
(SC)	6.0		wo	2	25	-	0		-	-	-	1	-	-		-	-	-	-	-		-	-	
6.7	2	Ш	22		25	9	•					ľ												
MEDITIM DENSE FINE TO MEDITIM SAND. GREY	1.0		wo				1	1			-	1-	\uparrow	1	1	\mathbf{T}		-			4			
(SP-SM)	8.0		SS	8	10		6	16			G	[_			
8.4	2	-					7					X												
STIFF FINE SANDY CLAY, GREY	9.0		wo			-/	-	-		-		1	-	-	-	-	-	-			_		-	-
(CL/SC) 9.6	2		SS	9	30	٩	8					1												
	10.0	4	wo			-	-		-			1	+	+	+	\vdash	-					-		
	11.0		SS	10	10	4		0	29		9							-						
· ·		1				1		T			Π	Γ												
2	12.0	<u>\</u>	wo					11			1		1	-		1		1				_		-
			SS	11	10			19	24		9	1												
MEDIUM DENSE TO DENSE SAND WITH GRAVEL,	13.0	4	wo	0		-	-	1		-	H	┝	-	+	-	-	-	-		-	-			-
GREY (TP. SM)	140		SS	12	10		d	19			4													
(01-014)	-							N			IT	\top		1		1								
	15.0		wo											-										-
			SS	13	10			d	28		19													
	16.0	4	wo					-	A	_		-	-	-	+-	-	-		-		_	_	-	-
			88	14	15				46	1				1										
177	11.0	1			-			-			17	-	+	+	+	-	-	-			-			\vdash
11.0	18.0	m	wo						V															
		1111	SS	15	30			9	26		1	2												9
VERY STIFF CLAY, GREY	19.0	4111					_	1		-		-		-	-	1_	-	-	-	-				4
(CL)	19		wo		20	3		11	20					1										1
L DIDI L	20.0	4///	WO	16	30	-	-	19	28	-	H	-	-	-	-	+	-	-		\square	-	-		F
PA= POWER AUGERING HA = HAND AUGERIN	 9	щ	wo	= W/	SHO	UT		1/	-	-	ST	= SH	ELBI	TUE	E		0	SS	= SPL	IT SP	OON		L	
		_					_		-	-					-	_	_		_	_	_		-	

)	BORING LOG	1					BOR	UNG	NO.	;			BH-	1		-	ELE	VATIO	ON (n	a.)	:			
PROJECT : SELF SUI	PPORT TOWER					~	DEP	TH (I	m)	÷		_	25.4	5		-	GW	L (m)			:		3.20	
LOCATION : A. MUAN	IG, CHIANGMAI						coc	RD.	N	:			·			20	DAT	E STA	ARTE	D	3			
									E	3						-	DAT	EFIN	ISHE	Ð	:			
SOIL DESC	RIPTION	hebru / ~ /	DEFIN (m.)	METHOD	SAMPLE NO.	RECOVERY (cm)		SPT- (bl	N VA	LUE fl.)			N. M	ATUR OISTI ONIE (%)	URE INT			a	Su /sq.m	.)			TOT UN WEIC (t/cu	AL T DHT m)
TTEF CLAY, GREY		+			+	\vdash				0 4		T	0 4						3	4		1.6		3 2 0
(CL)	20.40	21	0	wo																				
MEDIUM DENSE CLAYEY F	INE TO MEDIUM SAND,			SS	17	30			0	22		þ												
GREY (SC)		22	0	wo			\vdash	-		\mathbf{A}		\vdash	-	-	-	-		-	_	-		-	-	-
	22:	50	1	SS	18	30				53	1		ł											
		F			1																			
VERY DENSE SILT SAND, GI	REY	24	0	wo	'							L.												_
(SM)				SS	19	30			-	52		Î												
	25.	25	0	SS	20	30		-	-	56	-		-	-		-			-	-		-	+	-
END OF BORING	2.07	-	813		-	-					1													
							Π																	
								_			_	_						_				_	_	_
								-	-	-	-	-	-				\square	-				+	+	+
					1																			
					10			_		_	_	_						_	_	_		-	-	_
					1						-	-				-		-	-	-		-+	+	+
								-	-	_								-				-	-	-
								-	-													+	+	+
			1			1				-														
																					T	T	T	T
								+	-	-	_	_	_	-	_		_	-		_	-	+	-	+
													-	-		1								
								+		+	-			-		-		-	-	1		+	+	+
								T																
		4						+	-	+	-	_	_	-	_		-	-	-+	-	-	+		
		0			0		0					5			4									
A	. 6																							
A=POWER AUGERING	HA = HAND AUGERIN	G		wo	= W/	SHO	UT				_	ST=	SHE	LBY	TUBE				SS =	SPLI	T SPO	NOK		
PARTY CHIEF: SAMALB.	MADE BY : DUSSADEE	C.		GEC	LOG	ST :	PANY	AL				FILE	:	SUPI	PORT	1		4	DISK	:	9/1 C	HIANO	G MA	L

BOF PROJECT : โรงแรมคิเล LOCATION : อ.เมือง 3	RING LOG อมเพลส 2 จ.เชียงใหม่					L 1	BOF DEF	RING PTH D	(m.)	: :	21	ин- 7. О	- 1 5				GROU WATE DATE DATE	UND ER LE STAV	elev Vel (1 Rted Shed	(m) m)	1641 m111	0.4	0 6/9	95
SOIL DESCRI	PTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	; (bi	SPT lows	'-N ≤/ft)	0	1	PL	wn (%)	LI		(t O t X F 1	Su /sq.n UCT FVT 2	n) Δ 3	PP TV	,	TO UN WEI 1 (t/cu	TAL NIT GH 1.m.	т) о
MEDIUM SANDY CLAY, B (CL) LOOSE SILTY FINE SAN (SM) MEDIUM FINE SANDY CL. (CL) STIFF CLAY WITH FINE (CL) MEDIUM DENSE CLAYEY I SAND, BROWN (SC/CL) VERY STIFF TO HARD F SANDY CLAY, BROWN (CL/SC) VERY STIFF CLAY WITH GREY (CL)	ROWN 2.00 D, GREY 2.50J AY, GREY 4.00 SAND, GREY 5.00 FINE TO MEDIUM 6.00J INE TO MEDIUM 8.50 FINE SAND,			PA SS SS SS PA ST SS WO	- 2 3 4 - 5 6 V7 8 9 20		0 0000	4 8 6 7 5 6	2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	8 8	38								18	5				
VERY STIFF TO HARD FI BROWN (CL) HARD CLAY, BROWN (CL) DENSE CLAYEY FINE TO BROWN (SC/CL) VERY STIFF CLAY, GREY (CL) VERY DENSE SILTY FINE (SM)	12.00 INE SANDY CLAY, 14.00 16.00 MEDIUM SAND, 18.00 19.30 SAND, GREY	12 - 1 13 - 1 14 - 1 15 - 1 17 - 1 18 - 1 19 - 1 20 - 1		₩0 \$\$\$ ₩0 \$\$\$\$ ₩0 \$\$\$\$\$\$\$\$\$\$\$ ₩0 \$	II I2 3 13 14 15 16					8 8 4 6 8	39 39 9	To a b a							14.	*4 (
PA = POWER AUGERING PARTY CHIEF : SONGSAK S	HA = HAND AUGERIN DRILLER : BOONSON	G G C	C P	W G	O =	WA	SH	OL	T	A.I	7	1	ST =	SH	ELB	Y T : 5	UBE 5		SS =	SF	: 1	SPO	ЮС	N

во	RING LOG						BOR	IING N TH (m.			3H- 27. C	- I)5			_GF	ROUN	ND EL R LEV	.EV (EL (n	(m): n):	-0). 4)
PROJECT : เริ่งแรมค LOCATION : อ.เมือง	อมเพรล 2 จ.เชียงใหม่					-	000	RD.	:	_					- DA DA	TE S	INIS	ted Hed	:	30	0/6	/95
SOIL DESCR	RIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	SF (blov 20	PT-N ws/ft) 30 √	10	2	PL 0 40	wn (%)	LL 80		(t/s O U (X F V 1	Su sq.m CT /T 2 3		PP TV	1 W (1		AL IT 3HT .m.)
VERY DENSE SILTY FI	NE SAND, BROWN	21 -		₩0 SS	17			+	50	8	4		+	+		-			-	+	4	+
(SM)	22.50	22 -		₩0 55	18			+	50		H	$\left \right $	+	\dagger	+	\uparrow	\vdash	\square		\dagger	+	+
VERY DENSE SILTY FIN SAND, GREY	NE TO MEDIUM	23-		wo ss	19				50					1						1		1
(SM)	25.00	25 -		wo				-					4	+	1					\downarrow	+	+
VERY DENSE GRAVELLY BROWN	SAND, GREY,	26-		ss ₩0	20		_	-	50	4	þó		+	+	╞	╞	-	_		+	+	+
(SP-SM)) 27.05	27-		55	31	H		+	- 50,	12-	þ-	H	+	+	+	+		\square		+	+	+
- (ลี 1	ลาบัน																					
PA = POWER AUGERING	HA = HAND AUGERIN	1G		w	0 =	w	ASH	OUT				ST =	SHI	ELBY	TU	BE		SS =	= SPI		SPO	NOC
PARTY CHIEF :	DRILLER :			G	EOL	.OG	IST	:			1	DRA	FTN	IAN :				TYP	IST :	W	S	

BORING LOG						BORI DEP1	ING N TH (m RD.	ю. : ц :		8H- 28.	- 2 70				GRC WAT	XUND TER LE	elev Evel Rte	V (m . (m)	ı) : . : .	25/	6/9	
LOCATION: อ.เมือง จ.เชียงใหม่									_						DAT	E FINI	SHE	D	: }	26/	6/9	!
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	МЕТНОD	SAMPLING	RECOVERY	10	Si (blo 20	PT-N ws/ft 30	1) 40		PL	wr (%	n L) 50 8	L_ 0	0 X 1	Su (t/sq.i UCT FVT 2	u m) 2 3	4 L	;	TO UI WE 1 (t/cu	TAL NIT IGH 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	T)
SOFT FINE TO MEDIUM SANDY CLAY,		1	PA																			
GREY (CL/SC) 1.00 MEDIUM CLAY WITH FINE TO MEDIUM	1'-		SS	2		٩	5	+	\dagger	1	F	1			H	+	\dagger	$^{+}$	+	+	H	-
SAND, GREY (CL) 1.80 MEDIUM DENSE SILTY FINE TO MEDIUM	2 -	4	SS SS	3		À		4	+	19	6	┝	-	_	\vdash	+	+	+	+	+	\square	
SAND, GERY (SM) 2.00	1.		PA			V					L	L						1			Ц	
GREY (CL/SC) 3.40		K	SS	-	٦	R				1	ť										ŀ	
MEDIUM DENSE SILTY FINE TO MEDIUM	4-		10	6		T	V		T	T	T	T				T	T	T	T	Π	Π	
SAND, GREY (SM) 4.50	5 -	$\left\ \right\ $	WO	0	+	+	q	9	-	┢	9	\vdash	\vdash	-		+	+	+	+	+	$\left \right $	9
VERY STIFF CLAY, BROWN	6 -	1	ST	17		4	1	1	-	1	¢	-	4	_		+	12	ЦВ	4			5
(CH)	-		WO	H				028	3		ľ								L			9
7.50	l'	Ш	SS	8	_			1	36													
DENSE CLAYEY SAND TRACE GRAVEL, BROWN (SC) 8.50	8-		WO		F	+	1	1	T	Ħ	1			-	+	+	$^{+}$	t	+	Ħ		
MEDIUM DENSE CLAYEY SAND, BROWN	9-		SS	9	-	+	-	4	+	1	-		\vdash	-	+	-	+	+	+	\mathbb{H}		-
(sc) 10.00	0-		WO			-	1	Y	L					_	-	1	1	1	\downarrow			
			SS	ю	-				39		6											9
HARD CLAY, BROWN			WO		T	T	T	V	T		T				T	T	T	T	T	Π		
(CL) 12.00	-12-		SS	11	-	+	6	125	+	1	L			+	+	+	+	+	+	Н		6
CLAY, BROWN	13 -		wo		+	+	-		1	Н				4	+	+	+	+	╀	\square		+
(CL) 13.50	14		SS	12	-			82		ļ												ø
VERY STIFF TO HARD CLAY, BROWN	1		WO	2		Τ		N									2					/
(СЧ)	15_		SS	13	1	+	+	47	19		9			1	+	┢	f		Ť	Η	đ	
(Ch)	ю -		₩O		┢	+	+	+	#	-		-		-	+	+	+	+	╀	Н		-
	17_		SS	14					44		þ				1	\perp		1				9
.17.80			wo						A	Ι,	ľ											
DENSE FINE SANDY SILT, GREY	18 -		SS	15		T	T	46	6	ø					T	T	T	T	Т	Π		
(ML/SM) 19.50	19_		wo		d-	+	+	+	1	H			-	+	+	+	+	+	+	Н	-	_
VERY DENSE SILTY SAND, GREY	20-		SS WO	16	٦.	+	+	50	10	90	-			4	-	+	+	+	+	Н		_
(SM) PA = POWER AUGERING HA = HAND AUGERIN	IG		W	0=1	WAS	SHO	JUT	1	1	1	ST =	SH	IELE	BY 1	UBI		SS	= 5	PLIT	SP	100	4
PARTY CHIEF : SONGSAK S DRILLER : BOONSON	GC	1	GE	OL	OGIS	ST :			97		DRA	FT	MAN	1: 5	s	T	TY	PIS	т:	WS		-

PROJECT : โรงแรมคิ LOCATION: อ.เมือง	DRING LOG เอมเพรส 2 จ.เซียงใหม่						BORIN DEPTI COOF	NG NC H (m.) D.). : : :	8	H-	2				GRC WA DAT DAT	DUNI TER TEST TEFIN	D ELE LEVE TART	EV (m EL (m) ED ED	n):):	25/ 26/	/6/ 6/	9
SOIL DESC	RIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	(SPT blows	Γ-N s/ft) 30 4	0	2	PL	wn (%) 0 6	L 0 6	۲ <u>۱</u>	0 X 1	S (t/sq UC FV1 2	би 1.m) Т 4 Г 1 З	∆ Pr ⊡ T\ 4	•	TC U WE (t/c		(L F n.) 2.0
VERY DENSE SILTY S. (SM) VERY DENSE GRAVELL COARSE SAND TRACE	AND, GREY 25.50 Y MEDIUM TO SILT, GREY	21 - 22 - 23 - 24 - 25 -		₩0 55 ₩0 55 ₩0 59 ₩0 59 ₩0	17				50, 50, 50,	80.00													
(SP-SM VERY DENSE SANDY G GREY (GP-GM END OF BORING	27.00 RAVEL TRACE SILT 28.70	27		55 W0 55	22				50/	9 0													
PA = POWER AUGERING	HA = HAND AUGERING		9/	we) = V	VAS	HOL	JT			s	T = :	SHE	ELB'	YTU	JBE		SS	= Si	PLIT	SP	004	2
PARTY CHIEF.			T			-	-			T	-		-				T	-		-	LIC	2	1

BORING LOG PROJECT : อุบาคารไทยพาณิชย์ สาขาย่อยนวร	117			BOR DEP COO	ing no. Th (m.) RD.	: :	8H- 5. 16	1			GROUN WATER DATE S	ID ELI LEVE TART	EV (m) EL (m) ED		-04	10 /94
LOCATION: อ้าเภอเมือง จึงหวัดเชียงไหม	8			╞			Г			_	DATE F	INISH	ED	T	TOT	TAL
SOIL DESCRIPTION	DEPTH (m.) GRAPHIC U	METHOD	RECOVERY	10	SPT (blows 20 3	-N /ft) 0 40	20	rL w (9	m Ll 6) 60 8	0	(Us OUC X FV 1	Su q.m) 7 2 3			UN WEIC 1 (t/cu.	IT 3HT .m.) 2.0
VERY SOFT TO SOFT CLAY, WITH FINE SAND, BROWN, DARK BROWN (CL)	2	PA SS SS SS SS SS SS SS SS SS SS SS SS SS	1 2 3	0-0-0-	2			9-9-9		_						+
MEDIUM CLAY, DARK BROWN (CH) 4.50	4	SS WO	4	9	5			0					1	T	\square	+
VERY STIFF CLAY, DARK BROWN (CH) 5.90	5 -	SS WO	5		B	26		1	4					╞		-
DENSE CLAYEY FINE TO MEDIUM SAND, DARK BROWN (SC) 7.30	7 -	wo	6.			44 0	9	-	$\left \right $	-	+		_	╞	$\left \right $	╬
STIFF TO VERY STIFF CLAY, DARK BROWN	8-	ss wo	7		ø	25										4
(CL) 10.00	10	SS WO	8	a	9			Å		_	<u> </u> .		_	╀	$\left \right $	_
MEDIUM DENSE SILTY SAND, GREY, BROWN (SP-SM) 13.50	11	SS WO SS WO	9		0 0	25										
VERY DENSE SANDY GRAVEL, BROWN	14 -	wo	-	H	+	55	90	+		+	+		•	T		+
END OF BORING		50	15			5077	4									
PA = POWER AUGERING HA = HAND AUGERIN	NG	wo) = W	ASH	OUT		s	T = 8	HEL	BYT	UBE	8	S = 5	SPLF	T SPC	NOC

BORING Lo PROJECT : <u>ธนาคารไทยพาณิชย์ สาขา</u> LOCATION : <u>อำเภอเมือง จังหวัคเซีย</u>	OG ย่อยนวรั งใหม่	123				-	BOR DEP COC	ING TH (I IRD.	NO. m.)	::	- E	3H- 5. (· 2 X0				GRC WA DAT DAT	DUNI TER TE ST TE FII	D ELI LEVE TART NISH	EV (EL (m ED	(m) : n) : :	-(3.	/12	0 /9 /9	4
SOIL DESCRIPTION		DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	S (bk 20	PT ows	-N (/ft)	0	2	PL 0 4	wn (%)			0 X 1	S (t/sc UC FV	Su q.m) T T 3		PP TV	(1.1		AL 3H .m.)	т) о
VERY SOFT CLAY, DARK GREY (CL) MEDIUM CLAY WITH FINE SAND, D. BROWN (CL) VERY STIFF CLAY, DARK BROWN (CL) DENSE CLAYEY FINE TO MEDIUM S. DARK BROWN (SC) VERY STIFF CLAY, DARK BROWN (CH) MEDIUM CLAY, DARK BROWN (CH) MEDIUM DENSE TO DENSE FINE TO MEDIUM SAND, DARK BROWN (SP-SM) VERY DENSE SANDY GRAVEL, BROWN (GP-GM) END OF BORING	1.50 ARK 4.50 5.90 AND, 7.30 9.00 10.00 10.00	1 - 2 - 3 - 3 - 3 - 3 - 3 - 3 - 3 - 3 - 3		2 33 4324 € 13 € 13 € 13 € 13 € 13 € 13 € 13 € 1			a d	77	a a a	23	31														
PA = POWER AUGERING HA = HAND A PARTY CHIEF : pk DRILLER : SA		- - - -		W	0 =	WA	IST :	OU PA	T	A			ST =	SH	ELI	BY 1	TUE	E	s	S =	SP		SPC		N

K. ENGI	NEI	ERI	NG	c	101	ISI	UL'	ТА	N	TS	(co	, L	ТI).					_		
BORING LOO PROJECT อาคารบ้านพักข้าราชการ (มห LOCATION อำเภอเมือง จังหวักเชียงให	G บ.	33)			B Di C	ORIN EPTH DORI	IG NI 1 (m D	0 .)	81	-1 ∙45					BROU DBSE DATE	ND E RVEE STA	ELEV WL	(m. (m. 0 2 0 2) <u>-0</u>) -0 1 / 1	0.90 0./	94	_
SOIL DESCRIPTION Fill 0.40 21, ASING N: 2	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	с (ь ю	PT lows	- N / ft 50 4) 10	P	PL F	Wn (%			S O U X F 1	u († / CT VT 2	(m ²) △F □T 3 4	*P *V		1 (1/) 6 1.	m ³) 8 2.	0
STIFF CLAY, BROWNISH GREY (CL) 1.50 SOFT TO MEDIUM FINE SANDY CLAY, BROWNISH GREY (CL) 3.30	2 -		PA SA SA SA SA SA SA SA SA SA SA SA SA SA	1 2 3 4	0.0	8 6 3				ŀ	99-09										_	
VERY STIFF CLAY, GREY (CH)	4 -		PA SS PA	5		0	16				đ									0	_	
MEDIUM DENSE TO DENSE FINE TO MEDIUM SAND, GREY (SP-SM)	6 - 7 - 8 - 9 -		SS WO SS WO SS WO SS	6 7 8 9			021	4 0 0	33	0 0 0 0												
12.45 END OF BORING	12 -		wo SS	10				42	0	0					+						_	
สถาบั	2																					
จุฬาลงก	3	6		0											6						_	

SOIL DESCRIPTION FILL 3 H., ADVIL: N: 5 STIFF CLAY WITH SAND, BROWN (CH) 1.60 MEDIUM DENSE CLAYEY MEDIUM SAND, BROWN (SC) 2.00 3 4 LOOSE TO MEDIUM DENSE SILTY SAND (DENDE SILTY SAND AT SS-9) GREY, BROWNISH GREY (SM, SP-SM) 8 9 	In the second seco	0 € 12 0 € 12 0 € 12 0 12 12 12 12 1 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	SAMPLING I SAMPLING 1 5 9 9 23 4 2 2 4 2 2 2 4 2 2 3		PT www./10 0 300 18 18 18	N ft) 0 40				L))	Su (0 O UC X FV' 1 2	2) PP	(1)
STIFF CLAY WITH SAND, BROWN (CH) 1.60 MEDIUM DENSE CLAYEY MEDIUM SAND, BROWN (SC) 2.00 3 - 4 5 - 6 - 5 - 6 - 9 - 8 - 8 - 8 - 8 - 9 - 9 - 10 - 9 - 10 - 9 - 10 - 9 - 10 - 10 - 10 - 10 - 10 - 10 - 10 - 10		PA SS 88 SS ₩0 SS %0 SS ₩0 SS	1 2 3 4 5 6 6	9	18 020 18			4					
13.50 ¹³		SS WO SS WO SS	8 9 10		14 0 8	031	0 0						
VERY DENSE SILTY FINE SAND, 14 GREY (SM) 15.00 HARD CLAY, GREY (CL) 16 16.95		ss wo ss wo ss	11 12 13		4	31			+				

K. ENGIN	VEE	ERI	NG	CC	ON	ISUL	TAN	T	5 (со	., Ľ	۲D.						
BORING LOC PROJECTอาการบ้านพักกองพันสัตว์ต่าง เ IOCATION อำเภอแม่ริม จังหวัคเชียงให:	3 กส.1	⁄∩บ.			BC	ORING N OPTH (m OORD	o E .) _ 13	3H- 3. 9:	3			GRO OBS DAT	DUND SERVE	ELEV D WL ARTEI	(m.) (m.) 0 22 0 22	-9. / IC	80 0/94	4
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	SAMPLING RECOVERY	LICALEN	SPT { blows 10 20 3	- N /ft) 50 40		PL 1	Wn (%	LL 	c ×	Su(t UCT FVT	/m ²) Δ F □ T 3 4	*	(ĩ, t∕m 1.8	3)
STIFF TO VERY STIFF FINE TO MEDIUM SANDY CLAY, BROWN (CL) 6.00 MEDIUM DENSE CLAYEY SAND, BROWN (SC/CH) 7.50	1 - 2 - 3 - 4 - 5 - 6 - 7 -		PA SSASS PA SS PA SS PA SS PA SS PA SS PA	4		9 15 9 15 9 10 9 10 9 13			ad 0 0									0
VERY STIFF SANDY CLAY WITH GRAVEL, BROWNISH GREY (CL/SC) 8.50 MEDIUM DENSE TO DENSE CLAYEY SAND, GREYISH BROWN (SC)	8 - 9 - 10 - 11 - 12 -		SS PA SS PA SS PA SS PA	7 8 9 10		32	3											
END OF BORING			55			3	9 &											

BORING LOG	ź					BOR	UNG	NO	1			BH-	1		-	ELE	EVAT	ION (r	m.)	1	_		
IOCATION : 8.เมือง ช.เรียงไหม่					-	COC	DRD.	AII) N	10 10			10.4	, 		-	DA	TE ST	ARTE	Ð			7/10/	96
					-			I		_					_	DA	TE FR	NISHE	ED		2	7/10/	96
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm.)		SPT- (b)	-N V.	ALUE (fl.)	:		N M C	ATUI OIST ONTI (%	RAL URE ENT)			(Su Usq.m	1)			TOT UN WEI (1/ct	TAL TT GHT 1THO
VERY SOFT FINE TO MEDIUM SANDY CLAY,	1		PA			10	0 2	20 3		10	T	20 4		50 1	BO		1	2 3	Γ		1	6 1.	8 2.0
(CL/SC) 1.50	1.0		SS	1	25	9	2	\vdash	-	1	Q	\mathbf{T}	t	1	-	-	1	t	1		-	\square	
	2.0		SS ST	2	25 25	ò	7	\vdash	\vdash	-	9	Þ		$\left \right $	\vdash	-	-		16	577			_
VERY STIFF CLAY WITH FINE SAND, BROWN	3.0	-	PA	2	25	\square	-	-		-		+	-	+	-	-	-	-	112	0.1		\vdash	21
(CL)	4.0		PA	3	25							9	-	-	-	-	-		10	.72 (
5.00	5.0	Щ	PA									1	1		-				L				
LOOSE CLAYEY-SILTY FINE SAND, BROWN	6.0		WO SS	3	25	9	8				-	Ļ			_	_							
7.00	7.0					X																	
VERY SHIFF FINE SANDY CLAY, BROWN	8.0		wo ss	5	40		0	18				9											
(CL) 8.80	9.0	111	wo									V											
	10.0		SS	6	40		0	14															
MEDIUM DENSE TO DENSE FINE TO MEDIUM	11.0		SS	7	30		-		2	36	9			-						\square			_
(SP)	12.0		wo	8	25		0	19				-				-				\square		\vdash	+
13.50	13.0		wo	9	25		-	17			1									$\left \right $			+
	15.0		wo				1				Π												
MEDIUM DENSE SAND WITH GRAVEL, BROWN (SP-SM)	16.0		SS	10	25			9	22		9		6										
	17.0		WO SS	11	25		2	15				9											
18.00			wo		1			1			1	1											
DENSE SILTY/GRAVELLY SAND, BROWN (SM/GM) 18.45	18.0		SS	12	25	-+	-		6	35	6											\top	1
END OF BORING			0				_					-		-	-				\vdash	$\left \right $	-	-	+
ิลลาบ						8	-														_		1
- POWER AUGERING HA - HAND AUGERING			WO	= WA	SH O	π					ST -	SHE	LBY	TUB	E			SS =	SPL	IT SPO	DON	MAI	
ARTY CHIEF: SAYAN K. MADE BY : PACHAREE	С.	_	GEO	LOGI	ST :	PANY	AI		-	2	FIL	E :	UNI	DER-1	-	-		DISK	<u>c:</u>	y CH			

BORING LOG	,					BOR	NGI	NO.	:			BH-3	2		1	LEVA	TION (n)	: .		-	
PROJECT : UNDERPASS AQUQUINTAIQUIA	5					DEPI	TH (n	n)	٠.			15.43	5		ľ	SWL (n	1) TADTE	n	<u> </u>	27/	.80	_
LOCATION : 0.1104 0.1984 1111						000	RD.	N	•		-				ľ	JAIES	TAKIE		•	200	0/90	
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm.)	1	SPT-J (blo	E N VA ows/1	tue			N. M	ATUR OISTU ONTE (%)	AL TRE NT		1	Su (t/sq n	n)		2//1	VEIGHT	T) 20
MEDIUM FINE SANDY CLAY, BROWN (CL)	1.0		PA SS	1	30	9	7				q					1	1			$\overline{+}$	+	
2.0	2.0		SS	2	30	4	5				Ì		_		-		1			+	_	9-
SINF TO VERY STIFF CLAY WITH FINE SAND, GREY, BROWN (CL)	3.0	•	SS PA SS PA ST	3 4 1,2	30 25 25	0	10				٩ 	000						10	48 0		-	0
4.8	50	Щ	ST	3	25		_					ø	-		\downarrow	-	+	-		+	-	2.25
MEDIUM DENSE CLAYEY-SILTY FINE SAND,			SS	5	30			9	26			ì										1
GREY, BROWN (SC-SM) 6.0	6.0		SS	6	25		-	6	23	-	d	-	+	\vdash	-+	-	+-	-	$\left \right $	-	+	+
	7.0		wo	2	26		_	7	1	24	1		-		+		+				+	
medium dense to dense sand, brown	9.0		wo	,	2.5			1	Ž	34	Ť						1					
(SP-SM)	10.0	2	wo	8	25			9	26						_	_	-	-		+	_	+
. 128	11.0		ss wo	9	30		_		9	32	0											
/	13.0	2	ss wo	10	30				0	37	0	_										
DENSE TO VERY DENSE SAND WITH GRAVEL, GREY, BROWN (SP-SM)	14 (2	SS	11	25				9	38	0				+	-	+		\square		+	-
15.4	5	2	SS	12	25				5	2 2			F				T					
สถาย				1	1	8																
PA = POWER AUGERING HA = HAND AUGERIN	G		wo	- W.	ASH	TUC					ST	= SH	ELBY	TUBE		Q	SS	- SPI	IT SP	OON	MAI	
PARTY CHIEF: SAYAN K. MADE BY : PACHARE	EC.	1	GEO	DLOG	IST	PAN	YA I			-	FIL	E :	UN	DER-2		-	DI	SK :	9 CH	LANG		

BORING LOG PROJECT : สถานีบริการน้ำมันบางจาก BCP LOCATION : อ.เมือง จ.เชียงใหม่	-51	เชเชี	ยงแ	สน	-	BOR DEP COC	TH (DRD	NO m.)	:. :. :.		BH 15	-1				GRO WAT DATE DATE	UND ER LE E STA E FINI	elev Evel (Rted Shed	(m) : (m) :	-1 21 21	.60 /3/ /3/	96 96
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	(bl	SPT ows	-N /ft)	0	2	⊃L 0 4	wn (%)	LI 1 0 80		(0 X 1	Su USQ. UCT FVT 2	ı m) Ω 3	PP TV 4	T W (t	OTA UNI EIGI T ₁ /cu.n	ιL Γ ΗΤ n.) 20
MEDIUM FINE TO MEDIUM SANDY CLAY, GREY (CL) 1.00 SOFT FINE TO MEDIUM SANDY CLAY, GREY (CL) 2.80 STIFF TO VERY STIFF FINE TO MEDIUM SANDY CLAY, GREY (CL) 5.80			PA SS SS SS SS PA SS WO SS WO	1 2 3 4 5 6		0000	0 4 4 V 0	11			0000										22	21
MEDIUM DENSE CLAYEY FINE TO MEDIUM SAND, GREY (SC) 7.00 STIFF TO VERY STIFF FINE TO MEDIUM SANDY CLAY, GREY (CL)	6- 7- 8- 9- 10- 11- 12-		55 W0 55 W0 55 W0 55 W0 55 W0 55 S5 W0 55 S5 W0	7 8 9 10			0 0 0	15	26				+									
13.50 MEDIUM DENSE SILTY FINE TO MEDIUM SAND, GREY (SM) 14.50 VERY DENSE SILTY FINE TO MEDIUM SAND, GREY (SM) 15.45 END OF BORING			₩0 \$\$ ₩0 \$\$	12					29													
PA = POWER AUGERING HA = HAND AUGERIN	G		W	0 =	WA	SH	OL	T		-	5	ST =	SH	IELE	BY '	TUB	ε	SS	= SP	LITS	SPO	ON
PARTY CHIEF : SAYAN K. DRILLER : LERTCHA		10	G	EOL	OG	ST	: (JC.			C	DRA	FT	MAN	1:	PN.		TYP	PIST	:WS		_

BORING LOG PROJECT : เรือนอำอังหวัดเรียงใหม่ แห่งใหม่						BORING NO.	BH-1 16.00	ELEVATION (m.) GWL (m)	-0.10
LOCATION : ค.ร้างเผือก อ.เมือง จ.เชียงใหม่						COORD. N :	-	DATE STARTED	31/10/96
						E :	-	DATE FINISHED	01/11/96
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm.)	SPT-N VALUE (blows/ft.) 10 20 30 40	NATURAL MOISTURE CONTENT (%) 20 40 60 80	Su (Vsq.m.) 1 2 3 4	TOTAL UNIT WEIGHT (t/cu.m.) 1.6 1.8 2
	1.0								
	2.0								
GARBAGE	3.0		PA						
2	4.0								
	5.0								
6.00	6.0	13111							
VERY STIFF FINE SANDY CLAY, GREY (CL/SC)	7.0		SS	1	40	9 17			
7.50	8.0		wo	2	40	9 29	9		
	9.0		wo						
	10.0		SS	3	30	40 0	9		
MEDIUM DENSE TO DENSE CLAYEY FINE SAND, BROWN	11.0		wo	4	35	Q 19	0		
(8C,8C/CL)	12.0		wo						
	13.0		22	2	40	410	Ĭ I I I		
13.50	14.0		SS	6	25	53	80		
VERY DENSE SILTY FINE TO MEDIUM SAND, BROWN	15.0		wo						
(SM) 16.00	16.0		WO	1	20	50/8"	Ĩ		
END OF BORING ROCK			SS	8	Ţ.	50/0			
สถาบ	9			9		21915	203		
PA = DOWER ALIGERING HA = HAND ALIGERING			wo	= WA	SHO	UT .	ST = SHELBY TUBE	SS = SPLIT SI	POON
PARTY CHIEF: SAYAN K. MADE BY : DUSSADEE C	2		GEC	DLOG	IST :	PANYAI	FILE : CHANG-1	DISK 9 C	HIANG MAI

BORING LOG						BORING	3 NO. (m)	: .			BH-2				ELEV GWL	(m))N (m.)) : g	-	-1.5	30	
LOCATION : ค.ร้างเพื่อก อ.เบื้อง จ.เรียงใหม่						COORD	N							-	DAT	ESTA	RTED	:	1	01/1	1/96	
							F			-					DAT	E FIN	ISHED	:	-	02/1	/96	_
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm)	SP1 (1	L F-N VA blows/	LUE ft.)	0	2	MC CC	ATUR DISTU DISTU DISTU DISTU DISTU	AL JRE NT		1	(U 2	Su (sq.m.) 3	4		T 1 W. (L 1.6	OTAL UNIT EIGH /cum 1.8	.) 2.0
THE TO HARD CLAY WITH FINE SAND.	+	İIRI			-	Ť	T	ľ					Ē	Ē			T		1	Τ	Τ	Τ
ROWN, REDDISH BROWN	1.0		PA								_	_	-	-			_		4			_
(CH)		.]].	SS	1	40		0.22	1			1											ĭ
2.00	2.0		SS	3	40		+	18	38	q	y	-	-	-				+	+	+	+	+
DENSE CLAYEY FINE TO MEDIUM SAND,	3.0		PA	-																		
(\$C)			SS	4	40			19	40	9												
4.00	4.0		PA					K			Y	-		-		_			+			1
VERY STIFF FINE TO MEDIUM SANDY CLAY,			SS PA	5	40		9	25			19											Ĭ
BROWN (CL/SC) 5.00	5.0	10	SS	6	40		16	23	-	0	1-	1	\vdash	\mathbf{t}				-	1	+	+	0
	6.0		PA			1	11															
			SS	7	35		9	24		9												
	7.0							-		-	1-	-	-	-				-+	+			+
MEDIUM DENSE TO DENSE CLAYEY FINE TO			wo	0	10			0 30			1											ł
MEDIUM SAND, BROWN	8.0		00	0	40		+	N	-	-	V	-	1	\vdash				-	1	-	-	T
(ac)	9.0		wo	-					1		1											
			SS	9	40				49	9												
	10.0	2					-		K		1	-	-	+	-	_			-	-+-		
10.5	9	m	wo	1.0	-			Vn			1											9
VERY STIFF FINE SANDY CLAY, GREY	11.0	41	- 20	10	40		+	1			Ť	+-	+-	1-						+	1	1
(SC/SC)	0 12.		wo					1														
DENSE CLAYEY FINE TO MEDIUM SAND,	-		SS	11	40			42	9		9											Î
BROWN	13.	0							1		1	-	-	-	-	-		-	-		+	+
(SC) 13.5	0		WO	112	26	$\left \right $		11	22		1											
DENSE SILTY FINE TO MEDIUM SAND, GREY,	14.	2	00	12	135	++	-	11	1	ľ	1-	1	1	+		1			-	+		+
BROWN (SM) 14.3	15	.III	wo	1				V			N	1										
VERY STIFF SANDY CLAY, GREY, BROWN			SS	13	35		1	27			9											Ŷ
(CL)	16.	0					+	1-	-	-	1	+		+	-	-	-		-	-	-	1
16.5	9	Щ	- wo	14	10			0.30		1	1											9
	17.	<u> </u>	00	14	40		+	1		+ Y	+	1	-	+	1	+					1	1
DENSE TO VERY DENSE CLAYEY FINE SAND,	18	0	wo						N		N											1
YELLOWISH BROWN			SS	15	40			53		6	0											Ó
(SC)	19	0					_	-	1	-	+	-	+-	+-		+	+	-	-		+	-
19.5	0		SS	16	-	191		50/0		0												
END OF BORING ROC	K					+	+	T	Г	t	t	-	1	t	1	1	1				1	
PA = POWER AUGERING HA = HAND AUGERIN	IG	_	wo) = W	ASH	OUT			-	ST	= SH	ELB	YTU	BE		6	SS =	= SPL	IT SP	OON		
PARTY CHIEF SAYAN K MADE BY DUSSADE	EC.		GE	OLO	JIST	PANYA	I		_	FI	LE	CH	ANG	-2	_	_	DIS	K :	9 CI	IANG	MAI	
จุพาสงก																						

BORING LOG PROJECT : เรือนข้าอังหวัดเรียงใหม่ (แห่งไหม่)						BORING DEPTH (NO.	1 1			BH-3	3			elev Gwl	/ATIC)N (m) :	-	-2.4	45	
LOCATION : ค.ข้างเพื่อก อ.เมือง จ.เรียงไหม่	-					COORD	N	;							DAT	E STA	RTED	, :		03/1	/96	
							E								DAT	E FIN	ISHEI	: 0	-	03/11	/96	
SOIL DESCRIPTION	DEPTH (m.)	ORAPHIC LOG	METHOD	SAMPLE NO	RECOVERY (cm)	SPT (t	-NV	alue ft.) 0 4	0	2	NA Ma	ATUR DISTU DNTE (%)	AL JRE ENT)	0	1	Q/ 2	Su sqm.) 3)		T T W (V 1.6	DTAL JNIT EIGHI cum) 1.8	Г 2.0
DENSE CLAYEY SAND, BROWN						Ť	T	Ī	Ī			Ī	Ī				Ì	T	+	T	T	Τ
(SC)	10		PA SS	1	40		\vdash	47	0	. 9		-	-	-		-	-	+	+	+	9	+
VERY STIFF FINE TO MEDIUM SANDY CLAY,	2.0		SS	2	40		Q	25		9	į					-	_	_	-	_	9	1
BROWN (CH) 2.00			SS	3	40				480	9												
DENSE CLAYEY FINE TO MEDIUM SAND,	3.0		PA	_			-		K	\downarrow			-				_	+	+		+	-
BROWN			SS	4	40			10	35	١٩												V
(SC) 4.00	4.0	h	00	-	10		1	175	-	+	5	-		-				-+	+		+	5
VERY STIFF FINE SANDY CLAY, BROWN		FIE	WO	,	40		19	25		11												I
(CL/SC) 5.00	5.0	HH	88	6	35		19	28		6	-	-	-	-		-	-	+	+		+	+
MEDIUM DENSE CLAYEY SAND, BROWN (SC) 5.80	60		wo	-				Y		$ \rangle$												
HARD FINE TO MEDIUM SANDY CLAY, BROWN	0.0		SS	7	35			2	32		0											0
	7.0		wo				1	1-		-	-	-	1	-		-		-	+	-	+-	t
SAND CREV	80		SS	8	45	0	(17			10												9
(SC/CL)	0.0	11	-				T	K		11		1	1									T
8.80	9.0	Ш	wo					1	K										_			
VERY DENSE SILTY FINE TO MEDIUM SAND,			SS	9	45			55	\triangleright	99												
GREY (SM) 10.00	10.0	L.,					-	1	1	<u> </u>	1	-	-				_	-	-	_	+	+
			wo	-			1				N											
	11.0		SS	10	40	9	14	-	-	-	9	-	-		\vdash			-+	+	+	+	+
			wo				X															
MEDIUM DENSE TO DENSE CLAYEY FINE	12.0		SS	11	40		6	26		-	6	┼─	-						+	-	+	+
SAND, GREY, BROWN	1.20		-				1				V											
(30,30/01)	13.0		wo				1	1	1		1	1	1						1		T	T
	14.0		SS	12	35		1	40 }	5	6												9
							T	Y														1
14.80	15.0	h	wo				V					1	-						_		+	4
STIFF CLAY WITH SAND, GREY			SS	13	40	a	13			9											0	'
(CH)	16.0		1		1		P		-	\mathbf{H}	-	-						-	+		+	+
16.50		ШÜ	100	-	116				1													
	17.0		22	14	15		-	33/0	-	Ť	-	-		-					-		+	+
VERY DENSE SILTY SAND WITH GRAVEL,			wo		i			1														
BROWN, GREY	18.0	1	SS	15	-		+-	50/1		0	+	1							-	+	+	+
END OF BODING BOCK		212122	SS	16	-			50/0	• •	ò												
สถาย	9			S		9	0		5				4									
PA = POWER AUGERING HA = HAND AUGERING	۶		wo	= W/	ASH C	UT			-	ST	= SH	ELBY	TUB	E		0.	SS =	SPLI	TSPO	ON		
PARTY CHIEF: SAYAN K. MADE BY DUSSADEE	c	-	GEO	DLOG	IST	PANYA	1	-	-	FIL	E	CH	ANG-	3	-		DISH	K : :	9 CH	ANG N	IAI	

BORING LC	G						BOR	UNG	NO.	:	_		BH-	4		_	ELP	VAT	ION (a	n)	:	_	-		
PROJECT : ISOUTION PRISON IND (INA)	เพม) เ				_	-	DEP	TH (m)				10.0	0		-	GW	L (m)	DTE	T	÷.,		-1.00	~	
LOCATION : W. NIMON S.LEGY UIEGVINA					-	-	100	JRD.	N		2					-		IF21	AKIE	Ð	1			-0 	_
SOIL DESCRIPTION		DEPTH(m)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm.)		SPT- (bl	E NVA	: LUE ft.)	0		N M C	ATUI OIST ONTI (%	RAL URE ENT)	-	DAT	(1	Su Vsq.m	1) 1)		04	TOT UN WEI (Veu	AL TT OHT Im)	
DOSE CLAYEY FINE TO MEDIUM SAND,							1				ľ			Ī	Ť	Ī-	-	İ	Ī						-
BROWN		1.0		PA												-									
(SC/CL)	1.50		лн	SS	1	40	9	5				19													J
STIFF TO VERY SHIFF FINE SANDY CLAY,		2.0		SS	3	40	Ť	Y	022			Į	1	\vdash	\vdash	+	+	+			\square				
(CL)	3.00	3.0		PA					1																
				SS	4	35				9	34	9													
		4.0		PA	-	176				-	_	H	-	-	-	+	+	-						_	
		5.0		PA	1,	35				40 0		Ĭ													
		5.0		SS	6	40				53	2	00	1-	-	\vdash		+	1	1						
DENSE TO VERY DENSE CLAYEY FINE TO		6.0		PA												L									
MEDIUM SAND, GREY, BROWN				SS	7	30			-	0/7*	(ġ													
ദ്രാ		7.0		wo			\vdash	-			-	$\left \right $		-	-	+	-	-	-		\square			+	-
		80		SS	8	30				62															
		0.0													1	\uparrow									
		9.0		wo													_								
				SS	9	30				55	0	o d													
	10.00	10.0		SS	10	-		-		50/0*		<u> </u>	-	-	-	+	\vdash	-			\vdash	+	-	+	
END OF BORING	RUCA								1	1															
										-	_				-		-	-				\square		-	_
			4																						
			1	1				-	-			-	-	\vdash	-	1-	1						-	\neg	
		1	2														Γ								
	1							_	_	_					-	-	-	-	-				+	_	
10										-				P											
								-	-	-		-	-		-	1	-	-						-	
							\vdash	-	_				-	-		-	-			_	-	-	_	-	
	0																								
	0	0			0		0					6				T	1								_
01011											2					1			-						
A= POWER AUGERING HA= HAND AUG	ERING		_	wo	= WA	SHO	UT				-	ST =	= SHE	LBY	TUB	E			SS=	SPL	SPO	NOC	MAI		
MADE BI DUSS	ADEE C.	-	17	GEC		. 101	PAN	IAI	-	-		FILA		Chi	-110-	7	17		10100		100				_

B PROJECT : เรือนอำจังห	BORING LOG วัดเรียงไหม่ (แห่งไหม่)						BORING NO. DEPTH (m)		1	BH-:	5		-	ELE	EVAT	ION ((m.)	:	_	-:		
LOCATION : ค.ร้างเพื่อก	อ.เมือง จ.เรียงใหม่					-	COORD. N		_				-	DA	TE ST	ART	ED	:	0	4/11/	96	
						-	E :						_	DA	IE FI	NISH	ED	:	c	5/11/	96	
SOIL DESCR	LIPTION	DEPTH (m)	GRAPHIC LOG	METHOD	SAMPLE NO.	RECOVERY (cm.)	SPT-N VALUE (blows/fl.) 10 20 30 4	0	20	N. M C	ATUR DIST ONTI (%)	CAL URE ENT	30		1	Su Vsq.n 2	n.) 3 -	4	1	TO UN WEI (I/ci .6 1.	TAL NTT GHT um) 8 2	.0
		1.0																				
		2.0																				
		3.0		PA																		
GARBAGE		4.0																				
		5.0																				
	6.00	6.0																				
MEDIUM DENSE CLAYEY I	FINE TO MEDIUM	70		SS	1	40	φ 28		9													
VERY STIFF CLAY, BROWN	NAC N	1.0		wo	2	40	0 28		1	2							T				0	
(CH)	8.80	9.0		wo					1	1												1
DENSE CLAYEY FINE TO M (SC)	MEDIUM SAND, GREY	10.0		SS	3	40	930		ð								Γ					
		11.0		wo ss	4	35	45	9	0													
		12.0		wo																		
DENSE TO VERY DENSE SI	LTY SAND WITH	13.0		SS	5	35	G 31		9													
(SM,SP-SM)		14.0		SS	6	40	48	Y	•													
		15.0		wo	7	20	50/09				_											
	16.23	16.0		WO	8	30	50/1*	_Ĭ		_	2										_	
END OF BORING					-			-	-	-											_	
								+	-													
	์ถาเ	9			q	Л	91911	3	1			4										_
A = DOWED ALLEDDIG	HA - HAND ANATOPIC		-		- 11/4	en e		-			DV		0		_	80	ent	Ter	001			_
ARTY CHIEF: SAYAN K	MADE BY DUSSADEE C.			GEOI	LOGI	ST :	PANYAI	I	TLE	SHE	CHA	NG-5	,		9	DISI	K	9/1 0	CHIAI	IG M	AI	
จุฬา	เลงก	5			Y	9	าหา					2			6							

E	ORING LOG						BOR	RING	NO.	:	_		BH-7	1			ELEV	OITA	N (m.) :	_			1
PROJECT : ISOUTION	มศารยงเหม (แหงเหม) อ.เบือง อ.เรียงใหม่					-	DEF	TH (m) N	1			10.00	,			GWL	(m)	PTED	-	-	-1.1	0	
DOCATION . H.INHON	0.1004 0.1004110					-							·			•	DATE	551A			-	05/11	190	_
SOIL DESCR	PTION	DEPTH(m)	GRAPHIC LOG	METHOD	SAMPLE NO	RECOVERY (cm.)		TTZ	E NV/	tue			N/ MC CC	ATUR DISTO DNTE (%)	AL		DATE	(V	Su sqm)			U WE	VTAL NTT IGHT	_
MEDIUM FINE TO MEDIU	M SANDY CLAY,	┢				-					T	H	4				Ť	Ť	,	Ť	+	1.0	T I	
GREY, BROWN		1.0		PA															_					
(CL/SC)	1.50			SS	1	40	٩	5					19											
		2.0		SS	2	30		٩	15	6	1 79		12	-		_		-+	+		+		+	1
		30		PA	-	33					N		Ĩ											Ĭ
		30		SS	4	40				53	\rightarrow	50	t		-			1	+	+	+	+	\mathbf{t}	9
		4.0		PA							1													
				SS	5	40			33	9		Ý												9
		5.0		PA					1	1_								-	-	-	+			
MEDIUM DENSE TO VERY	DENSE CLAYEY FINE			PA	6	35			p	25		9								1				
TO MEDIUM SAND, GREY		6.0		55	7	40		6	17			6	-					+	+	+	+		+	6
(SC'SC/CT)		70		-				Ĩ																
				wo														1	1	T	+	1		П
		8.0		SS	8	40		9	17			9								_				
					-				1															
		9.0		wo		10			-1									-+	+	-	+	-		H
				55	9	40			1	21		1	í						1					Ĭ
	10.50	10.0		wo				1	-	-	-	-		-			-	+	+	+	+	+	+	1
*****		11.0	m	SS	10	40		11					9										d	
					20			1													T			
STIFF TO VERY STIFF CLA	Y, GREY, BROWN	12.0		wo		-	_	1					\square				_	-	-		+	_	-	4
(CH)				SS	11	45		0	18				9											0
	10.00	130		wo	3			-	-			-	\parallel				-	+	+	+	+	+	-	
	13.50	14.0	1111	SS	12	35				45	P		li l											
								-					1					1		1	T	1		
DENSE TO VERY DENSE SI	LTY FINE TO	15.0		wo										-		1	_		\downarrow		1			
MEDIUM SAND, GREY, BR	OWN			SS	13	25				45	9	0												
(aMI)	16.50	16.0		wo				-		_	1					-	-	-	+	+	+	+	-	-
END OF POPINC	10.50			SS	14					50/0*														
END OF BORING											_								+	+	1	1		
																					_			
								_	_	-				_		-		+	+	+	+	+-	+	-
	การ	9			9		e			4					9									
			0	d				-	C		U					1		1		-		1		
PA = POWER AUGERING	HA = HAND AUGERING			wo	= WA	SHO	UT					ST=	SHE	LBY	TUBE		_	1	SS = 5	SPLIT	SPOO	ON		
PARTY CHIEF: SAYAN K	MADE BY DUSSADEE			GEO	LOG	ST :	PAN	YAI	_	_		FILI	E :	CHA	NG-7			1	DISK	: 9/	/1 CH	IANG P	IAN	


SCALE	LEGENS	LOCATION SAENG TAWEE HOTEL BORING NO. BH-I	FIELD MOISTURE CONTENT	BLOWS PER FOOT	UNCONFINED COMPRESSIVE STRENGTH
M FT	1 1	SROUND ELEVATION		10 5 20 25	.5 1 19 2 25
		Lotenie filled.			
· .		Top 301	:0 12		0.450 1.5
Γ.			1.0.12	1 6	0.400
-2	· · · ·	Macium grey- mottled			
-		stown sandy ckay.		1	
	1		1575	1 \	0.360
	· ·		5.15	3	8.25
F .	J.,			1.1	
-4				()	
1			1630		0.875
Г. ·		Shiff whitish grey sandy		1 13	
-5 -		clay.	. /		
+ -					
L-6 .			4.75		L217
				13	
Γ.	0.21				
-7	0:0				
-			5.17		2.400
La ·	1			20	
	- 0.00				
Г					
-9			-14.91	. /	2519
Ļ.	···· · · ·		IGN A	15	
L					
~	1. 0.0.		STATA		
5	13.		12.00000		
H- u				. 15	
-	-500	very stiff whitish gray			
1		sandy cidy;			1 11
Γ2	1	1		28	
-	14	Some of bor hole 245 m.		1	
1		291			1
-	1				l í .
	1		1		
1	-				
-	-				
-	-			770	
1					
Г	7				

DEPTH, M. ELEVATION, M.	SAMPLE NO	TYPE OF SAMPLE	SAMFLE DIST RECOVERY.	DESCRIPTION OF	MATERIAL	O Natura 1 Pkristic 2 Uquid 1 20 40	Water Umit Umit (%) 50	Content - 30 VOO	0 3u/	2 2 2 3 5 7 1 5 7 1 1 1 1 1 1 1 1 1 1 1 1 1	04/2 94 5 0 7) 0 30	
10.0				Sandy clay, som Top soil.	e root.		i				:	
	1	ss	- 01	Silty-sandy cla	Y, Drown &	9	;		QQ		1	
	2	SS		li-gray, medium	to stiff.	6		·	di			
	2	-			(CL)			<u> </u>				
	1	55	Π	Clayey fine to	coarse sand,	Ti			1	$\langle $		
5.0	4	ss	12.02	grayish brown,	dense.	6	1		0	2	÷	
					(SC)	/ 1	1	1		1	1	
	5	ss	2	Medium to coars	e sand, some	RI	;			ġ 📃		
	6	4		gravel, white,	(SP)				1	1		
	0	55		Silty clay, tra	ce of fine				1.7		1	
	7.	ss	-	to very stiff.	errow, scill			. 1	toit			
10.0			IT						1			
	8	SS	2.2	•		0			ļģņ		L	
					(CL)				11		<u> </u>	
	5	SS		Sandy clay to c medium sand, ye	layey fine to llowish brown	LAT	-A .		192			
	10	ss	En 31	medium dense.	1.5 100 6	1201				1	:	
15 0						1			1			
	11	ss					1	: 1	1.			
					(CL-SC)				1-1-			
	12	SS		Silty clay, tra	ce to some	-9-1			1-7-		<u> </u>	
	113	53	100	very fine sand,	brwon, stiff	1	~	_	1:4:			
	1		ĨT						17	i.		
20.0	14	ss										
					(CL) :				1. 1			
	15	SS	221	Medium to coars	e sand, some		1		Q L	Į į		
	16	SS		dense.	a m-gray,	0				~	1	
		-	Π			T III			+			
	17	SS	AR		(SM-SP)	6		+		···· -+ · 占··		
25.0				END OF E	BORING.							
	TEO	10		CASE DUATIONIC			i	1	1 1	<u>!</u>		
WL	ILR.	LE!		W.S. OR WD				BORING S	MARTED.	Feb. 20	,81	
WL.			3.C.	R. A.C.R.	SUL LEST	ITD	MI .	RIG	Acker	FOREMAN	SK	
WL 0.30 m.24 HRS. AFTER						CO,LID.				APPROVE	D. VVN	
	80	RIN	3.		BANG	5.2	JOB NO. 590 A SHEET. 1/1					

BANGKOK. JOB NO. 590 A SHE

SCALE		LOCATION CHIENG MAL	FIELD	BLOWS	UNCONFINED	Wet Unit
	LEGENO	BORING No. 1	MOISTURE	PER	STRENGTH	Wt. • Gm/cc.
M FT		GROUND ELEVATION	10 20 10 40	10 20 30 40	1 ×2'cm2 4	. 1 . 2 . 3 . 4
1		Too soul Q30 M				
-		Dense to measure ught brown				
L	1	clayey fine to medium sand	1 16	; 30		7 2.14
		some coarse soud				
Γ.					20	
-3 -	1	3.40 м				1 2 0 5
			23	1 25		2.00
- 4		Hard brownish light grey clay				
L	<u>.</u>	some fine sand.				
	<u> </u>	5.00 M.	26	31	/ 3.33	1.99
	1977				1	
		Very dense to medium light			1	
1-6 -	1				, i	
		greyish brown clayey fine		51	€ 0.907	2.05
-7 -		to medium sand.				
L -					i	
-		MODE	15	129	1.079	2.08
-a -						
Γ -						
-9 .	1	Hard reddish brownish light				
	1 e		23	4	2 3.886 Y	2.06
1_10 -		grey clay some decompose			1	
	<u> </u>	gravel			į.	
Γ		groven	23		7650	201
F"			25	/**	5.005 7	2.0.
		II.DU M.			į	
-12					1	
L -		Very stiff greyish brown fine	22	. 18	2.162	2.01
-13 -		sandy clay.				
- T					i	
Γ -		13.85 M				
-14 .			24	20	. 1.2.402	2.02
+ -		Hard reddish light brown clay			Ň	
-15 -		trace of fine sand.			ľ,	
L.		<u></u>	28	> 38	4.514	1.96
Γ· · ·	TL	Bottom of bore hole 15.45 M				

LOCATION CHIENG MAI	FIELD MOISTURE	8LOWS PER FOOT 2.0.40 50 80	UNCONFINED COMPRESSIVE STRENGTH 0.5 HOCTS 20	WET UNIT WT GM/OC 20 22 2.4 25
JADUND ELEVATION Jose brown dayey fine to -edium sands 2.60 m -dia reddish prown sitty day Sor e fine sand 4 i0 m Medium to dense light brown Copey fine to corse sand 660 m Very shift light prown exity fine sandy cay. Medium light brown dayey fine to corse sand some grave I3.70 m Medium light brown dayey sity fine sand I8.70 m. Dense to very cense light brown re to accorse sand some gravei Bartom of bore note 2445 m	QCC_P = 20 + 4C 17 19 12 26 20 22 19 13 21 24 17 24 24 21 5 0 9	5 20.40 50 60 60 5 33 42 17 25 22 13 10 14 46 40 92/12" 130/10"	0.5 H0CHS 20 0.254 390 1.299 1.299 1.299 1.299 0.309 0.268 0.509 0.411 0.744 1.957 0.509 0.411 0.744 1.957 0.336	20 22 2.4 26 0 2.35 199 200 2.19 202 2.13 2.09 199 2.05 2.05 2.25 199 2.05 2.08 1.56 1.64
สถาบั	นวิทย	บริก	าร่	

Su	άLΕ	LEGENO	LOCATION THA PARE, CHIENG MAI. BORING No. BH-1	FIELD MOISTURE	BLOWS PER FOOT	UNCONFINED	••••
M	FT		GROUND ELEVATION	10 20 30 40	10 20 30 40	1 2 3 4	
– .	_		MEDIUM DARK BROWN ORGANIC SILTY CLAY		┉┷┷╼┺╌┸╌┖╶╄╌╧╶╧╶┖╴		
\vdash	_						
Ē.	-						·
[-		LOOSE DARK GREY,				
L	-		WHITE AND BROWN	15.40	ə 6	0.148	
F	. –	· · ·	CLAYEY COARSE SAND		· ·		
-2	_	•	SOME GRAVEL.				
+ .	_			11.67		0.327	11 a.
F	-				8	\backslash .	
F	-				1 .		
-3		<u> </u>					
F				9 19.59		2.599	
Ľ	_		STIFE TO VERY STIFE		- 14	· · · · / ·	
Γ.	-		LIGHT CREY AND				
	-		AROWN CLAY				
L	_		unoun ceat.	16.40	• 16	4.900	
+	_						
-0	_			1362 2000			· .
+	-		MEDINA WHITE CLASES	20.65		Laio	
\mathbf{F}	-	i.	COARSE SAND		- 17		• • • ·
F					•		
- 6		T					tan s
Γ	_			11.21.1.1.			
E	-	V	BOTTOM OF BORE HOLE				• • •
L	-						· · · · ·
1	_				. 8-		
F.				•			
F	-				1	· · · ·	

SCALE LEGENO	LOCATION SAN PA KHOY. BORING No BH-1 GROUND ELEVATION	FIELD MOISTURE CONTENT IO 20 30 40	BLOWS PER FOOT 5 IO IS 20	UNCONFINED COMPRESSIVE STRENGTH 1234	:
	LOCATION SAN PA KHOY. BORING No BH-1 GROUND ELEVATION Top Soil. LOOSE Brown Medium to. Coarse Sond. Medium Brown Clay. Stiff Dark Brown Clay. Medium Brown Fine Sandy, Clay. Stiff Dark Grey Clay.*	FIELD MOISTURE CONTENT 10 20 30 40 35.60 28.45 20.48	BLOWS PER FOOT 5 IO IS 20 1 I I I I 8 8 10 15	UNCONFINED COMFRESSIVE STRENGTH 1234	
	J		Ū		

SCA	LE	LEGENO	LOCATION CHIANG MAI BORING No. BH-I	FIELD MOISTURE CONTENT	BLOWS PER FOOT	UNCONFINED COMPRESSIVE STRENGTH	:
M	FT		GROUND ELEVATION	10 20 30 40	20 40 60 60	5 1 15 2	1
1.		12	Tup soil 0.30 m.			ĺ	1
Γ.							
F			Soit to medium corr			•	1
F			brown fine sondy cloy				i
-2		<u>v</u>	silty clay and pocket	y 25.39	e 4	e 0.407	
F		- V	sand				
-3	_	V					
-	_			31.59	5	: 0.520	
4	_	v v					
F		v					
- 5			Medium light gray cloy	2517	B		
L			and some fine sond	\$ 2.0.11	10	0.855	
			at 5.00 m.				
F°.							
F				20.23	28	2.200	
-7			Very stiff grouteb links				
-			brown ciev to 9 40'm				
- 8			city 10 3.40 m.	20.40	27		
-				0 100 0			
- 9					18		
L				22.26	1	· ·	
Lin						- 4. -	
				Sinza a			
Γ			Madium to desce a lite	(CO) (O) (9)	4 18		
Γ"	٦		light grey fine to medua	22.47			
F	٦		and and seems seed				1
- 12	1		sono ond coorse sond.		6 33		
F	- 1	-	Battam of how had to de	17.55			
-13	٦	4	Bollom of bore hole 12.45 m			1	
F.	٦						1
F	1	-					
L	٦		SA.				
L	٦						l i
	. 1						
<u> </u>	1					:	

MEREVIAT CHS	SPT - STANDARD PENETRATION IEST	T	1	2	O LL % Not. W % S PL %
PA - POWER AUGER	DB - DIAMOND BIT	8		61110	O UNCON ST. 1/m 2 A SPT - BLS/-
ST - SHELSY TUBE	GROUND ELEV 4.	IIIC	HI	9 11	* WET UNIT WT 1/m3
S3 - SPLIT SPCON		RAP	DEI	Fute	
SOIL DI	ESCRIPTION	1		SAM	
3012 0		TT.	1 4		
		k)	- 1	SS	
Dark brown to	brown medium sandy	1	-	SS	
barn brown co	orona, accrea sandy	1.1		22	
CLAY, to claye	v with some decom-	1.			
		1.	2	SS	+ ≻
posed gravel.	(SC-CL)	1.31	2-	SS	
		11.	-	cc	
		1.2	-	33	
		1.	-	SS	
		11.1	-	SS	1+++ 1++221++++++++++++++++++++++++++++
		1.1	4-	60	
		11.1	-	55	
Brown to yello	w and gray, very	1.1	_	SS	
		11	_	SS	
sciff sandy CL.	AY. (SC)	1.1	_		
		1.1	6-	WO	<u>╶</u> ╞╪┇┝╗╧╎╎╎┊┍╎╎╎╎╎╎╎╎╎╎╎╎╎
		V.I		SS	
		1./		WO	
		Vi	-	wu	
		11	-	SS	ε
			-	WO	<mark>┨<mark>┝╋┇╎╊┙╎╊┙┇╷┆╄╋╢┢╡┆╏╪╎╞┢╎╡</mark>╛╬</mark>
White and eray	wary dance claver	1.	8-		
white and gray	, very dense crayey		-	22	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
SAND. (SC)			-	UO	
		•.:	-	-	
			_		<mark>────────────────────────────────────</mark>
			10-	SS	
End	of Boring	21	12		
			-		<mark>╶╶┥╞╶╶╒╞╶╶╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴╴</mark>
			: -		╊ <mark>╋╋╗╗╗╗╗╗╗╗╗╗╗╗</mark>
			-		┝ ┥┥┍┥┥╞╔╴╪╞┊┊╡┥┍┍┥┥╹
			-		
			_		
			_		
					╘┽╧┟╧┼╪╪╪┼
					<u>╞╋╡╞┟┧┇┼┝┯┧╏┟╎┾┍┞╞</u> ╆┟┠╎╎┟┾┝┝┝
					╒┱┼┼┟┟┟┟┟╎╎╎╎╎╎┝╎┝╎┝╎┝╎┝┝
					┟╉╪┟╡┫╎╪╋╪╅╪╪┼┊┼┽╞╄╪╪╪╪┾╵╪┾┾
			-		
			-		<u>┝╋┺┶┥┥┥</u> ╪╫╪┥┥┥┥╧┝┝┥╎┥┥ <mark>╞╞┾╞</mark> ╎╎

ser			LOCATION CHIENGMAL	5151.0	61 Out	1	1 :
		FOEM		MOISTURE	PER	COMPRESSIVE	
		Leocie	BORING No BH-I	CONTENT	FOOT	STRENGTH	
	FT	<u> </u>	GHOUND ELEVATION	0 20 30 40	10 20 30 40	5 1 15 2	
- 1	-	X	Top soil 1.50 πι				
- 2 - 		· · ·	Madium dark brown silly clay.	40.70	0214	0.336	
- 4	-	× v × v	Very soft dark grey. silty clay.	21.27	o 2	/ / / 0.274	
- 6		v v v	Medium dark grey slity clayey, fine sand.	20,48	. 16		
н в 1 9 9	-	• • •	Loose to medium dark gre fine sand tocoarse sand and some gravel.	13.58	8		
- 11	1 1 1 1 1		Dense light grey coarse	7.96	• 31		
- 13 -	1 1 1	а. А.	sond ond some gravel.	694	38		
- 14 - - 15				6.70	41		
-	1	4	Bottom of bore hole 15.45 m.	1	40 6		

APPENDIX B.2 CHIANG RAI

GROUND WATER OBSERVATION.	WAC BORING LOG	BORING NO.					
DATE TIME EL. of HOLE EL. of WATER		SURFACE ELV0.36 m.					
	LOCATION, AMPHOE MUANG, CHIENGRAI	DATE START					
24 HR.AFTER BORING. 0.26 M.		DATE FINISH					
SOILS DESCRIPTION	U U U STANDARD E E E PENETRATION. ○ PLASTIC LIMIT.	AOne half Unconfined- Compressive Strength • peakgremoided INSITU VANE SHEAR. DENSITY.					
		Penetrometer Rdg. Yd, Yw					
,	BLOWS/FT. %	KSC. T/M. ³					
	0.00 30 60 20 40 60 80	1 2 3 1.2					
Medium reddish brown silty clay. (Fill soil) Medium reddish brown clayey very fine sand. SC Medium dark brown silty very fine sand. SM Very stiff brownish grey and grey silty clay.	55.1 55.2 55.3 55.4 7 55.4 7 55.4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7						
ML-OL 6.00 Very stiff light grey silty clay, trace of very fine sand and gravel. ML-OL 7.50 Very stiff to stiff yellowish	-55.7 -55.8 -31						
occasional very fine sand. ML-OL	55.9 55.10 55.10 55.11 55.11						
Hard brownish grey silty clay, occasional very fine sand and gravel. ML-OL 15.00	-55.12						
Very dense brownish grey cement ed silty very fine sand, trace of gravel. SM		•0					
Hard brownish grey and reddish brown very fine sandy clay, occasional decomposed rock.ML-OL 19.95 END OF BORING.	55.15 *						
SI II							

GROUND WATER OBSERVATION.	Τ					BORING NO. (1)				
DATE TIME EL. of HOLE EL.of WATER	2		W.A.	C. BOH	ING LOG	SURFACE ELV.				
	1.00	TIO				DATE START 11 / 7 / 39				
24 HR.AFTER BORING. 1.65 M.		110	N.AM	ipnoe m	luang, CHIENG RAI	DATE FINISH 13 / 7 / 39				
				•		ΔOne half Unconfined- Compressive Strength.				
SOILS DESCRIPTION	FILE		STA		O- PLASTIC LIMIT.	peak premolded TOTAL- INSITU VANE SHEAR. DENSITY.				
	DIL PRO	EPTH. N			X NATURAL MOISTURE CONTENT.	Penetrometer Rdg. SENSITIVITY				
	S S	ā	BLO	WS / FT.	0%	KSC. T/M. ³				
Fill soil		0.00	30	60	20 40 60 80	1 2 3 1 2				
Medium to stiff brownish red silty clay. ME-OL & CH	SS.1		€ ●10	1.	x 0.0	•				
	-SS.3	5.00-	22		X	•••				
Medium to dense grey and brown very fine to coarse sand.				33		· · · · · · · · · · · · · · · · · · ·				
SW-SM, SM & SP-SM				34						
	8.22	10.00-	- 25			•				
Medium brown and light grey clayey very fine sand. sc 1330-	9.22 SS.9	2	25		Xo					
Stiff to very stiff light grey and red dish grey silty clay.	SSI	15.00	24			A				
· CL	TIT SST		29			▲ → → → → → →				
Dense to very dense brownish yellow and yellow silty very fine sand		20.00		60	*	• • • • • • • • • • • • • • • • • • •				
SM	SS.1			90	• ×	•••				
END OF BORING	THESSIG				1 1 1 1	••				
0		2500-								
			þ							
สถ่าเ	9	ĥ	1/1	6 9	そわれち					
010116	1.00	-0	1.1		2 0 1 1 1 0					

GRO	OUND W	ATER OBSER	VATION.			1	M A	C 8		NG	1.00	2			BOF	RING	NO. (2)			
DATE	TIME	EL. of HOLE	EL.of WATER				V.A.	U. E		NG	200	а 			SUF	FACE	ELV.				
				10		тю	ν Δι	nnh	ne m	m	1 CF	IFN	GR/	T	DAT	E STA	ART "	1417	/ 39)	
24 HF	R.AFTER	BORING.	0.89 M.		5,								DAT	DATE FINISH 15/7/ 39							
							•			- LIQUID LIMIT.					∆One half Unconfined- Compressive Strength.						
sc		ESCRIPT	ION	ILE	YPE N		ST			O-P	LAST		NT.		pea INSIT	U VAN	NE SH	EAR.	DEN	ISIT	Y.
		2001111		PROF	LET	н. м	1 214		non.		N	ATUP	RAL		© One Pen	e half i etrom	Pocke eter R	t- Idg.	20	1 24	
				SOIL	SAMF	DEPT				^ N	OIST	UREC	ONTI	ENT.	• SE	NSITI	VITY		•	-0	_
00				-	0.00	BLC 3	0 e	FT.	2	0 40	%) 60	0 8	0	1	2	SC.		T/	M. ⁰	-	
Stiff	f to med	Fill soil	yellow silty	Ø	-SS.1		•7-				-×.•										_
Mod	ium bro	very the su	2.50	TA	-SS.2 -SS 3		5			C	-		17 14		6						
' silty v	ery fine	sand.	ish yealow	1 1	-55.4	5.00					×			1.000							
		5М	6.00		-555	5.00-	18				-										_
Medi to coar	um to de se sand	ense grey very	y very fine		-SS.6	m		42.		×											
Very	dense bi	rown and grey	9.00 very fine to		-SS.7		- 1 - st-	55		×					·						
Har	d brown	and light or	www.silty.clay		823-	10.00					0×	• -								-	
	e brown	CL CL	cy sitty city.	1	-SS 9	2.3		9 39			ó						!				
Den	se light	grey and brow	13.50- wn clayey	55	-SS:10	-		30													
very ti	ne sano	sc	omposed rock.	755	-SS 11	15.00-	-	36-		xo	-						0	n (a. 1941), a 1		+	
Very trace of	very fir	grey and brown ie sand. CL	silty clay, 17.10-		-SS.12	100	29	1		-0-	>						8				
sand.		SM		3	-SS:13	1			62	×											
Har	d brown	ish red silty	clay.		-SS.14	20.00		47			0 X	•					8				
		CL & MI-OL			-SS.15		-	52			- CA			-			\$!				
	END	OF BORING	22.95	1010	-SS.16	215		1	78		•x						8				
		C				25.00-							h							*	
			2																	Α.	
					ŝ	1											-				
															1						
				6	1		2	0	a												
	a a citi				0	6		Ē.													
															- 0						



GROUND WATER OBSERVATION.	W.A.C. BORING LOG	BORING NO. 2
DATE TIME EL. of HOLE EL.of WATER		SURFACE ELV.
	LOCATION. CHAINGRAI PRACHANUROK	
24 HR.AFTER BORING. 1.13 M.	CHAINGRAI	DATE FINISH 31-0CT-43
		∆One half Unconfined- Compressive Strength.
		INSITU VANE SHEAR.
SUILS DESCRIPTION		One half Pocket - Penetrometer Rdg.
		SENSITIVITY
	Ø Ø BLOWS / FT. %	KSC. T/M. ³
Filled Soil 035		
Soft to stiff light grey and brown silty clay, trace of very fine sand. ML-OL Loose dark grey silty very fine sand. SM & SP Medium to dense light grey very fine to fine sand, coccasional coarse sand. SM 19.95 END OF BORING.	951 47 952 953 955 956 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 958 959 958 9512 958 9512 958 9514 958 9514 958 9514 958 9514 958 9514 958 9514 958 9514 958 9516 9516 9516 9517 9516 9518 9516 9519 9516 951	
ลถาบ		
จฬาลงก	รณมหาวทยา	ลย

GROUND WATER OBSERVATION.	WAS BORING LOS	BORING NO.
DATE TIME EL. of HOLE EL.of WATER	W.A.C. BORING LOG	SURFACE ELV.
	RACHANUKROK HOSPITAL	DATE START
24 HR.AFTER BORING. 0.48 M.	CHIENGRAI	DATE FINISH
SOILS DESCRIPTION	U U U U U U U U U U U U U U	∆One half Unconfined-Compressive Strength. TOTAL-DENSITY. ■ peak□remolded INSITU VANE SHEAR. ØOne half Pocket - Penetrometer Rdg. Yd, Yw ● SENSITIVITY ●-O KSC. T/M. ³
Concrete pavement / 0.15		
Veey loose grey clayey very fine sand. SC	$\begin{array}{c} & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & $	
END OF BORING.	- D3'10 - D3'10 - D1'10	

GROUND WATER OBSERVATION.	WAC BOBING LOG	BORING NO.
DATE TIME EL. of HOLE EL.of WATE		SURFACE ELV.
	RACHANUKROK HOSPITAL	DATE START
24 HR.AFTER BORING. 0.58 M.	CHIENGRAI	DATE FINISH
SOILS DESCRIPTION	UNDER CONTENT.	AOne half Unconfined- Compressive Strength ● peakgremolded INSITU VANE SHEAR ©One half Pocket - Penetrometer Rdg. ● SENSITIVITY KSC. T/M. ³
	0.00 30 60 20 40 60 80	1 2 3 1 2
Concrete pavement 0.1 Medium brown silty clay, trac of very fine sand. CL 2.0 Loose light greyish brown and brown clayey very fine sand. SC 4.2 Medium yellowish brown and light grey very fine sand. SM 6.8 Medium light grey silty clay. CL 8.5 Medium light grey and brown clayey very fine sand. SC 10.4 Medium grey clayey very fine sand. Stiff to hard light grey, reddish and yellowish brown silt clay. CL Stiff to hard light grey, reddish and yellowish brown silt clay. CL		
สถาเ	25.00-	

K. ENGI	NEE	ERI	NG	С	ONSU	JLT	ΓΑΝ	тs	С	0.,	LI	D.				۰ 		
BORING LOO PROJECT : KRUNGTHAI BANK: CHIANG	G R#	\12	BRĄ	NCH	BORIN DEPTH COORD	G NC I (m.))	81 15	4-1 .45			GRO OBS DATI	UND ERVE	ELE D WI	V. (m.)) /3.	5.00 /91	
LOCATION: A. MUANG, CHIANG RAT	[_					_				DAT	E FII	NISHE	D <u>3</u>	/3/	91	
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	RECOVERY	S (ы Ю 2	PT -	N (ft) 0 40	PI 1	0 40	₩n ↔) ↔)	LL 	0 × 5	Su(1 UCT FVT 10) PP TV 20		Υ ₁ 1/π 6 1.8	2) 2.0
STIFF CLAY TRACE OF FINE SAND, DARK GRAY	1-	•	PA SS PA		PI				9	1			_				0	
(SC) 2.50	3-		SS PA SS	2	0	14		R	P	_	-							
	4 -		wo SS	4	0	12		1										
MEDIUM DENSE SILTY FINE TO MEDIUM SAND, GRAY	6 -		SS	5		16				+	+			+	+-		+	+
law and	7-		WO	6		13							1	T	T			
(SM, SP-, SP-SM)	8-		wo					Π			•						1	
	9-		SS	7		17								-				
	10-		SS	8		18												
12.00	12-		wo															
	13-		wo	9	20		46 6	0										
DENSE TO VERY DENSE SILTY FINE TO COARSE SAND, GRAY TO DARK GRAY	14-		SS	10			62	00			_		_	\downarrow	ŀ		_	
(SM) 15.45	15-		WO					\prod										
END OF BORING			33		· .		56	• •		_	·		_	_	-		_	
สถาบ	9		99				5	2			\$	$\left \right $	-	+			_	-
ลหาลงภ	5		o L	9	9			q		6	$\frac{1}{1}$			e				\top
9	d	b																_

K. ENGI	NEE	ERI	NG	C	ON	st	JL.	ГА	N	гs	(co	., 1	LT	D.					• •		
BORING LOC PROJECT KRUNGTHAI BANK: CHIANG LOCATION A. MUANG, CHIANG RAI	G RA	12	BRA	NCH	BO DE CO	RIN(PTH ORD	3 N((m	D .)		B 15	H-2	2		-	GR OB DA1 DA1	OUN SER TE S	D EI VED STAF	WL WL RTEI	(m. (m. 0)) 3/3	3.80)
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	SAMPLING	-	S (bk	PT -	- N / ft 1) PO	P	20 4	Wn (%	1) 60 8	1	•	Su OUC FV	(1/) T T	m ²) △ P □ T 5 2	P V O		¥, (1/1	m ³) 8 2.0
VERY STIFF CLAY, BROWNISH DARK GRAY AND DARK GRAY (CL) 1.80	1-		PA SS PA SS PA	1		012	> 20				P	1	*									0
LOOSE TO MEDIUM DENSE SILTY FINE TO COARSE SAND, GRAY, BROWN AND BROWNISH GRAY	4 - 5 - 6 -		SS WO SS WO	3 4 5	0 0	7 9	1															-
(SP-SM)	7-		WO	6	de	2	14				9											
9,00	9-		WO SS WO	7				/-//	8													
DENSE TO VERY DENSE CLAXEY AND SILTY FINE TO COARSE SAND, LI-GRAY. TO DARK CRAY	11-		SS WO SS	8			-	0	30 9									_				
(SC, SM)	13-		wo ss wo	10			_	0	37	0		2		_		_		_		-		
15.45 END OF BORING	15-		SS	11				-	040	-0					_	-				_		
ปไปสด องชาวองเว	L'I								3 0 0							80						
9 9	d	Ь		0	·	-	_		d								9			_	-	

K. ENGI	NEI	ERI	NG	С	ON	ISU	JL	TAI	NT	S	С	0.,	LT	D.								
BORING LOO PROJECT แฟลดนายทหารขึ้นประทวบจุ LOCATION อ.เมือง เชียงราย	G 80	ครอ	; บครั	3	BC DE CO	PTH PTH	GN(I(m)	D .)	BI	H-1 9.2				GRO OBS DAT DAT	OUNC SERV TE S	VED TAR	WL TED	(m.) (m.)	14	-0. -2. /6/	.13 .80 /86	
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	с (ы ю 2	PT -	- N /ft) k0 40	,	PL + 20	() 40	₩n ↔ %) 60	LL -1 80	o x	Su (OUC' (FV)		m ²) △ PI □ T1 3 2(1,	¥, (+/ 6 1.	m ³) 82.	.0
LATERITIC CLAY (FILL), REDDISH BROWN 1.00			PA																			
MEDIUM SILTY CLAY, BROWN (CL) 2.00	2-	11	SS	1	×	8				*	24					^					0	
	3-		PA	2	X	P	11		-	0	_	•	+-					_	_	_	_	
MEDIUM DENSE FINE TO COARSE SAND, BROWN	4-	1	wo	3	x	ł	15		-	0		+	 .				-	-	_	_	_	
(SP-SM)	6-		wo	4			15			0											_	
7.00	7-		wo		-							+	-			_	_	_		_	_	
HARD CLAY, DARK BROWN AND GREY	8-	V	ss	5			5	0/9	-0	4	1	+	+			-	-	+	-		-	_
9.28	9-	12	SS		+		5	0/5	. •	0	rt	+	+	-			-	-			-	
END OF BORING	-		2		F	-			+		+						_				-	
·																						
A.	-								-	-	-		-		_		_	_	_	_	_	
	-				-				+	-	+	-	-						-	_		
															_				-			
สถาบ้	9	9			0			4		4												
DIDIIL		0					-	_			-		0					_				_
จฬาลงก	5	6		9					+	-	4	P	1					_			_	
9	-				F				+	+	+											_

K. ENGI	NEE	ERI	NG	С	0	N	su	Ľ	ГА	N	тs	(C).,	L	.T	D.									
BORING LOC PROJECT :แฟลตนายทหารชั้นประทวนจุ LOCATION: อ.เมือง เรียงราย	3	0 61	อบค	ร้ว	-	BOR DEP COC	TH RD.	(m.	D .)	B)	H-:	2 38				_	GR OB DA DA	OUN SEF TE TE	ID I IVE(STA FIN	ELE W	EV.	(m.) (m.) 1:	3/6 3/6	-0. 2.7 /8	08 5 6 6	
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	` (IC	SF blo	- T -	- N (ft)	i i	F	20	40	(n 5- %) 60	L 	L 1 0	2	Su O UC C F V	(† / ===================================	/m. Δ	2) PF	,	1	1 (+/	1 m ³)
LATERITIC CLAY (FILL), REDDISH BROWN 1.00		//	PA																							
STIFF SILTY CLAY, BROWN (CL) 2.00	2-	11	SS	1	×	9	9					p	1	\downarrow						2					0	
	3 -		PA	2	×	0	8				6		•	+					-		_	_				
LOOSE TO MEDIUM DENSE FINE TO COARSE SAND, BROWN AND	4-		wo											+	_			-	-	-	-	_			_	-
GREY (SP-SM, SC)	5-		ss wo	3	×	4		>						+	-		-		-		-	_				-
	6-		SS	4	×	-	6	18			-									+	+	-				+
	7-		wo	5	X			L	28																	
8.50 VERY STIFF CLAY, BROWN	9_	Z	wo	6	×			19														_				
(CH) 10.00	10-	K	SS	7	×		4	10	1	-	•	k	Ļ	-	_					+		_				
HARD CLAY, BROWN	11 -		55	8	×	-	_	5	0/7	9	•	H	-	+	-	_				+	+	_				-
(CH,CL) 12.38	12-	2	SS	9	2		-	5	0/9		-	Ð		+	-			_	-	+	-	_	_			-
END OF BORING	-						-							+	-				-	+	+	_	_			-
	-													-	-		-		+	+	+					\vdash
	-												T							T						
	٩		٦٩			2	Ġ		N		6				9											
			0			_						-	L									_				
	2	6				9	8													+	+	_			-	-
9.						-	_							+	-					+	+	_				-

BORING LOG PROJECT แม่กกวัลเล่ย์ รีสอร์ต Location อ. เมือง จ. เชียงราย	;				BO DE CO	RING PTH ORD	(m.)	B1	H-1	5		 	GROU OBSE DATE DATE	IND I RVEI STA	ELEN D WL ARTE	/.(m .(m D	.) .) 29 29	2.00) /1	9
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING		S (bla	от- ws/ 03	- N (ft) Ю 4	0	P 1 2	0 4	Wn 0 (%		0 I X I	Su(1. JCT EVT 2		PP TV 4		1 (1/ 6 1	m.3)
MEDIUM TO STIFF FINE TO MEDIUM SANDY CLAY, REDDISH BROWN (CL;CL/SC) 4.00	1 1 X 2 3		PA SS PA SS PA SS WO	1 2 3	0 0	5				·	4									0	
MEDIUM DENSE SILITY FINE TO COARSE SAND TRACE GRAVEL, GREY (SM,SP-SM)	5 -		SS WO SS WO	4		01	17			0 0											
8.00 STIFF CLAY, GREY (CL)	- 8 9 - 9 -		ss wo ss wo	67		0 0	15			•		-1	-								
DENSE TO VERY DENSE SILTY FINE TO COARSE SAND TRACE GRAVEL, GREY (SM)	11 - 12 - 13 -		ss wo ss wo ss	899				0 43	36	0 0 0											
15.45	15 -		wo	11			•	331		0											
END OF BORING	25																				

K. ENGI	NEE	ERI	NG	C	ONSI	JLTA	N	rs (co.,	LT	D.					
BORING LOC PROJECT แม่กกวัลเล่ย์ รีสอร์ต LOCATION อ. เมือง จ. เชียงราย	3				BORIN	G NO I (m.) _)	B	H-2 5.45			GROUN OBSER DATE DATE	ID ELE IVED W START FINISH	EV. (m.) (L (m.) ED ED	-2.0 28 / 9 28 / 9	0 / 92 / 92	
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	S (b)	PT - N lows / ft 20 30) 40	PL 	Wn 0 (%) 40 60	LL -1 . 80	Su OU XF\ 1	(1/m T Δ T 0 2 3	2) PP TV 4	1 (1/ 1,6 1	1 m ³) .6 2.0	0
SOFF FINE TO MEDIUM SANDY CLAY, REDDISH BROWN (CH/SC) 2.70 STHFF CLAY, GREY (CH) 4.40 LOOSE CLAYEY FINE TO MEDIUM SAND, PALE BROWN (SC) 7.50 MEDIUM DENSE SAND TRACE SILT AND GRVFL, GREY (SP-SM) 8.90 SOFT CLAY, GREY (CL) 10.00 DENSE SHITY FINE TO COARSE SAND TRACE GRAVEL, GREY (SM, SP-SM) 15.45 END OF BORING	1 - 2 - 3 - 4 - 5 - 6 - 7 - 8 - 9 - 10 - 11 - 12 - 13 - 14 - 15 - 14 - 15 - 14 - 15 - 14 - 15 - 16 - 16 - 17 - 16		PA SS PA SS PA SS PA SS PA SS PA SS PA SS PA SS VO SSS VO SS SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS SS VO SS SS SS VO SS SS SS SS SS SS SS SS SS SS SS SS SS	1 2 3 4 5 6 7 7 8 8 9 9		3 925 366 03									0	
จุฬาลงก	U o	50		9							16					

K. ENGI	NEI	ERI	NG	C	ONS	UL	TAP	T	's (co.,	LT	D.						
BORING LO PROJECT แม่กกวัลเล่ย์ รีสอร์ต LOCATION อ. เมือง จ. เชียงราย	G				BORI DEPT COOF	NG NG H (m D	0 .)	B	H-3 5.45			GROU OBSE DATE DATE	UND E ERVED STAF	WL (RTED	m.) m.) 27 27	-2.0	0 / 92 / 92	-
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC	METHOD	SAMPLING	(10	SPT blows	- N /ft) 50 40		PL	Wn (%)	LL 	0 X 1	Su(1/ UCT FVT 2 ;	m ²) △ PP ⊡ TV 3 4		1 (†/ 1,6 1	t m ³) .8 2.0	>
SOFT TO MEDIUM CLAY REDDISH BROWN	1.		PA SS PA	1	93				9									
2.8 STIFF CLAY, GREY, REDDISH	0 3		SS PA SS	2 3	97	0		-	H				-		+	+	C	
BROWN (CH) 4.4 MEDIUM DENSE SILTY FINE	0 4 · 5 ·		wo ss	4		13		+	4						+			
(SM) 6.0	6		wo ss	5		A	932	-	0		_							
DENSE SAND WITH GRAVEL, GREY	8		ss wo	6			034	•	-									
(SP-SM)	9		ss wo	7		/	9 32		0						+			_
MEDIUM DENSE SAND WITH GRAVEL, GREY (SP-SM) 12.0	11.		ss wo	8	6	13			-				-					
DENSE SILTY FINE TO MEDIUM SAND, GREY	13		ss wo	9			033		0	2	-				+			_
(SM)	14		wo	10			44		•		+							
END OF BORING					0		45								+			-
61 61 FL								0										
พพ าตงเ	0	6		3		1			+									_

K. ENGI	NEE	ERI	NG	C	ONSULTANI	rs co., lt	D.	
BORING LOC PROJECT แม่กกวัลเล่ย์ รีสอร์ต LOCATION อ. เมือง จ. เชียงราย	3				BORING NOB DEPTH (m.)I COORD	H-4 5.45	GROUND ELEV.(m. OBSERVED WL (m. DATE STARTED DATE FINISHED) <u>- 2.00</u> 26/9/92 26/9/92
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	SAMPLING BECOVERY	SPT - N (blows/ft) 10 20 30 40	PL W _n LL + + + + + + + + + + + + + + + + + + +	Su(t/m ²) ⊙UCT △PP XFVT ⊡TV 1 2 3 4	₹ ₁ (1/m ³) 1,6 1,8 2,0
STIFF TO VERY STIFF CLAY, GREY, BROWN (CL,CH) 5.50 MEDIUM CLAY, DARK GREY (CH) 7.00 MEDIUM DENSE FINE TO COARSE SAND, GREY (SP-SM) 8.50 VERY STIFF CLAY, BROWN, REDDISH BROWN (CH)	1 - 2 - 3 - 4 - 5 - 6 - 7 - 8 - 9 - 10 - 11 -		PA ST PA ST PA SS PA SS WO SS WO SS WO SS WO SS WO	1 2 3 1 2 3 4 5 6	0 16 010 0 21 0 24			
13,00 MEDIUM DENSE TO DENSE SILTY FINE TO MEDIUM SAND, GREY (SM) 15,45 END OF BORING	12 - 13 - 14 - 15 -		wo ss wo ss	7	9 26 9 23 9 32			
จุฬาลงก		6		3				

K. ENGI	NEE	ERI	NG	С	ON	ISU	JLI	AN	тs	C	:0.,	LI	D.						
BORING LOO PROJECT แม่กกวัลเล่ย์ รีสอร์ต LOCATION อ. เมือง จ. เชียงราย	3				BC DE CC	PTH ORD	G NO. (m.)	·	BH-	5			GRC OBS DAT DAT	OUND SERVE E ST	ELE D WI ARTE	v. (m. . (m. :D :D) <u>-</u> 2 30/ 30/	.30 9 9	/ 92
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERT	SI (blo 10 2	PT - ows/	N ft) 040	P 1 2	0 4	Wn (%) 0 60	LL 	O X 1	Su(t UCT FVT 2	/m ² △ 3) PP TV 4	(¥, 1/1	n ³) 3 2.0
MEDIUM FINE TO MEDIUM SANDY CLAY, GREY, REDDISH BROWN (CL) 2.00 LOOSE SILITY/CLAYEY FINE TO MEDIUM SAND, GREY, BROWN (SC/CL,SM) 4.00 MEDIUM FINE TO MEDIUH SANDY CLAY, GREY (CL) 5.50 SOFT FINE TO MEDIUM SANDY CLAY, GREY (CL/SC) 7.00 LOOSE SILITY/CLAYEY FINE TO COARSE SAND, GREY, BROWN (SC,SM) 10.00 MEDIUM DENSE TO DENSE SILITY FINE TO MEDIUM SAND, GREY, BROWN (SM) 15.45 END OF BORING	1		PA SS SS PA SS PA SS SS PA SS SS PA SS SS PA SS SS PA SS SS PA SS SS SS SS SS SS SS SS SS SS SS SS SS	1 2 3 4 5 6 7 8 9 10 11		5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		26			+								
จุฬาลงก	(The second sec	6		9								ŗ							

K. ENGIN	NEE	ERI	NG	С	ONSULTAN	TS CO., LT	D.	
BORING LOC	;				BORING NOB DEPTH (m.)IS COORD	8H—I 3.45	GROUND ELEV.(m. OBSERVED WL (m. DATE STARTED DATE FINISHED))150 12 / 1 / 92 12 / 1 / 92
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	SAMPLING RECOVERY	SPT - N (blows/ft) 10 20 30 40	PL W _n LL + + + + + + + + + + + + + + + + + + +	Su(1/m ²) ⊙UCT △ PP XFVT ⊡TV 1 2 3 4	¥ ₁ (1/m ³) 1,6 1,8 2,0
MEDIUM CLAY, DARK GREY (CL) 1.00	1 -		PA	1		0		0
STIFF TO VERY STIFF CLAY, TRACE FINE SAND, DARK GREY, BROWN, YELLOWISH BROWN (CL, CH) (CL, CH) 7.00 LOOSE CLAYEY FINE TO COARSE SAND, GREY (SC/CL) 8.50 DENSE CLAYEY FINE TO COARSE SAND, WITH GRAVEL, GREY (SC) 10.00 VERY DENSE SILTY SANDY GRAVEL, GREY	2 - 3 - 4 - 5 - 6 - 7 - 8 - 9 - 10 - 11 -		PA SS PA SSS PA SS SS PA SS SS PA SS PA SS PA SS PA SS SS SS PA SS SS SS SS SS SS SS SS SS SS SS SS SS	2 3 4 5 6 6 7 8 8 9	09 010 014 020 016 09 016 09 046 0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			
(GP-GM) 11.50 VERY STIFF TO HARD CLAY, GREY, YELLOWISH BROWN AND BROWN (CL) 18.45 END OF BORING	12 - 13 - 14 - 15 - 16 - 17 - 18 - 18 -		PA SS PA SS PA SS PA SS PA SS PA SS	10 11 12 13	45 0 50 0 50/2" 9 400			
	-							

K. ENGI	LE	KI	10		HOULIAN	13 CO., LI		
BORING LOC	3				BORING NO.	BH-2	GROUND ELEV.(m	.)
					DEPTH (m.)	15.15	OBSERVED WL (m	1/1/02
PROJECT OVERBROOK HOSPITAL					COORD		DATE STARTED	11/1/92
LOCATION A.MUANG, CHIANGRAI		-					DATE FINISHED	1
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING RECOVERY	SPT - N (blows/ft) 10 20 30 40	PL Wn LL → → → + (%) 20 40 60 80	Su(1/m ²) ⊙UCT △PP XFVT ⊡TV 1 2 3 4	τ₁ (t/m ³) 1,61,82,0
MEDIUM CLAY, DARK GREY (CL) 1.00			PA					
	1']		SS	1	¢10	9		Q
	2-		PA					
			SS	2	99			P
STIFF TO VERY STIFF CLAY,	3 -		PA			+ + + + + + + + + + + + + + + + + + +	┟╸┼╸┼╸┼╶┼┈	
TRACE SAND AND WITH GRAVEL,		•	55	3	015	9		9
GREY, REDDISH BROWN AND	4 -		PA					
YELLOWISH BROWN			PA		0 22	PT		
(CL)	5 -	•	SS	5	023		┟╴┼╶┼╌┼╌┼╌	
			PA	tΓ		T		
	6 -		SS	6	919			
7.00			-	Π				
7.00	7-		PA					
			SS	7	38 0	6-1		
DENSE TO VERY DENSE CLAYEY	8-		PA					
FINE TO COARSE SAND, WITH	9-							
GRAVEL, GREY AND DARK BROWN	1.		SS	8	62	• •		
10.00	10 -		PA					+ + + -
VERY DENSE SILTY GRAVELLY								
SAND, GREY (SM) 11.00	11-		SS	9	52 9	P9	┝┼┼┼┼	
			PA					
	12 -		22	10			┟┼┼┼┼	++++
HARD FINE TO MEDIUM SANDY	12		55	H	033	7		
CLAI, BROWN TO DATA BROWN	13 -		PA				+ + + + + + + + + + + + + + + + + + +	╏┼┼┼┼
(CL/SC, CL)	-		SS	μ.	50/8"	• •		
	14-		PA					
15.15	15-	111	SS	12	50/6"	99-1		
END OF BORING								
		4						
สกาย	9 -		9			404	+ + + + + + - + - + - + - + - + - + -	+ + + +
6N 6 1° 1 L			5					
	-		-			┟┼┟┼┼		++++
a M 1 a 1 a	-						-	┟┼┼┼
	Q.	0						
Ч. – – – – – – – – – – – – – – – – – – –	-				+ + + + + + + + + + + + + + + + + + +	+++++		
	10000		S	1 1.				

K. ENGIN	NEI	ERI	NG	CC	ONSULTAN	TS CO., LT	D.	
BORING LOC	3				BORING NOE DEPTH (m.)I	3H-3 6.50	GROUND ELEV. (m. OBSERVED WL (m	.) .)i.20
PROJECT OVERBROOK HOSPITAL					COORD.	<u>.</u>	DATE STARTED	13/1/92
LOCATION A. MUANG, CHIANGRAT	-	-	T	-			DATE FINISHED	13/ 1/ 92
SOIL DESCRIPTION	DEPTH	GRAPHIC	METHOD	SAMPLING RECOVERY	SPT - N (blows/ft) 10 20 30 40	PL Wn LL (%) 20 40 60 80	Su(t/m ²) ⊙UCT △PP XFVT ⊡TV 1 2 3 4	τ ₁ (1/m ³) 1,61.82.0
MEDIUM CLAY, DARK GREY (CL) 1.00			PA					
STIFF CLAY AND STIFF GRAVELL CLAY, GREYISH BROWN AND REDDISH BROWN (CL,CL/SC) 2.70	Y 2	•	SS PA SS PA	1	911 99			0
	3 -		SS PA SS	3	013			
STIFF TO VERY STIFF FINE SANDY CLAY, GREY AND BROWN	5 -		PA SS PA	5	021	14-1		
(CL,CL/SC)	7 -		PA	6	Ø 17	\$ ·		φ
8,00	8 -		PA		015	1 9 -1		
MEDIUM DENSE CLAYEY FINE TO COARSE SAND, WITH GRAVEL,	9 -		SS PA	8	925	9		
GREY AND BROWN (SC)			SS	9	018	q		
VERY STIFF TO HARD FINE TO	12 -		SS	10	0,20	He-1		
COARSE SANDY CLAY, TRACE GRAVEL, GREY, BROWN AND DARK BROWN	13 -		PA					
(CH) 14.50	14 -		PA		50/8 6			
VERY DENSE SILTY FINE TO COARSE SAND, WITH GRAVEL, DARK BROWN (SP-SM) 16.50	15 -		PA	12 -	50/8" (
END OF BORING		6						
จุฬาลงก	. The	6		3				
	1							

_{PROJECT : ธนาคารไทยพาณิชย์ สาขาแม่กร _{:ocation : อ.เมือง จ.เชียงราย}}	1					BOR	ING I TH (r	NO. m.)	:. ;	16.	H- 95	5			_ GI W		ID EL	EV (m	n):	-0.7	70
	124					c00	RD.		: .			_			_ D/	ATE S	TART	ED ED	:	20/	6/95 6/95
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	10	S (blo 20	PT- ows	-N /ft)	,	f	<u>рг</u>	wn (%)	LL 80		(t/s © U(X FV 1	Su q.m) 7 7 2 3		P V	TC U WE (t/c)TAL NIT IGHT (1
VERY SOFT SANDY CLAY TRACE GRAVEL, BROWN (CL) 1.00	1 -	2	PA SS	-		Q	2	_			F	ø									
OOSE CLAYEY SAND WITH GRAVEL, BROWN (SC) 1.50	2.		SS SS SS	2 3 4		9	9	12	_	_	9 0 0	-		_	+	\downarrow			4	+	
OOSE AND MEDIUM DENSE CLAYEY SAND	3 -		FA	5		Y	6	16	_	_	9	-		+	+	+		+	+	+	++
(SC) 4.50	4 -		₩0 SS	6			10	19		-	9			+	+	\top					
(SM) 5.20	6.		WO	7			1														
(CL)	7 -	- - -	33 WO	ŕ			9	12				0		_	+	-		+	_	_	
7.80 TEDIUM DENSE CLAYEY FINE SAND, BROWN (SC) 8.60	8 -		SS WO	8			\$	17	-	-	<u>_</u>	\$		+	+	+		+	+	-	
CCL) 10.00	9.	V	SS	9	-	a	5	-				0		1				1		T	
STIFF CLAY WITH SAND, GREY	11-		SS	10			6	13			ł	6 1							_	1	
(CL)	12.		WO SS	11			a	3		_		ø		_	+	+		+	+	+	
12.00	13-		₩0 SS	12					6	35	7	-		-	+	+		+	+	+	
SROWN	14.		wo						1	\mathbf{v}	T				T					1	
(SM)	16-		w0	13			-	_	44	0				_	\downarrow	-			\downarrow	_	
END OF BORING			SS	14			_	_	47	9	9	-		+	+	+		+	+	+	
								_		_	-			+	+	+		-			
			2												1	-					
PA = POWER AUGERING HA = HAND AUGERIN	NG	1	N	10 =	W	ASH	OL	JT			-	ST	= SH	ELB	YT	JBE	s	S =	SPL	IT S	2004
PARTY CHIEF : SONG SAK S DRILLER : BOON SON	IG (: 0	G	EOL	.OG	IST	: 1	J8.	0	1	1	DR/	FT	AN	: 5	S	T	YPI	ST :	N.	5

BORING LOG						BORIN	ig N	o :		BH-	2			_	GRO		ELEV	(m)		2 60	
pROJECT : ธนาคารไทยพาณิชย์สาขาแม่กรณ LoCATION : อ.เมือง จ.เชียงราย	<u>ú</u>				-	COOR	-1 (m. D	; ;		0. 3				_	DATE	STAF	RTED	m) : :	21	/6/	/95 '95
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECOVERY	(SP blow 20	T-N vs/ft) 30	40	2	PL	wn (%)	LI 0 80		(t O t X F 1	Su /sq.n UCT FVT 2	n) Δ 3	РР ТV 4	(1.6		AL IT SHT m.) 20
LOOSE CLAYEY FINE TO MEDIUM SAND, BROWN (SC) 1.00 VERY LOOSE CLAYEY SAND WITH GRAVEL,	ı .		PASSASS	1		φ 5 φ 4		-	-	q	10		_	_	_	+	-			+	+
BROWN (SC) 2.00 VERY LOOSE TO LOOSE CLAYEY SAND,	2.		SS SS PA	3		08	-	+		9	¢	4	_	-	+					+	+
BROWN (SC) 4.50	3 -		ss wo	5		93				9					-	T				-	T
MEDIUM DENSE SILTY SAND, BROWN (SM) 5.50	5 -		SS WO	6		2	18	-		╞	0		-	+	+	+	$\left \right $	$\left \right $		+	+
VERY STIFF CLAY WITH SAND TRACE GRAVEL, BROWN	6 -		SS WO	7		9	15	-			9		_								1
(CL) 8.20	8-		SS WO	8			818	-	-	1		-	-	_	+	+	-		+	+	+
MEDIUM DENSE TO DENSE SILTY FINE TO MEDIUM SAND, BROWN AND GREY	9-		SS WO	9				0	32							+					
· (SM)			SS	10	-	_		0	31	9		_	-	_	+	_	-		+	+	+
13.00	12-		SS	11	-	+	d	25	-	9			+	+	+	+			+	+	+
DENSE TO VERY DENSE SILTY SAND	14 -		SS WO	12	-			47	0	0			_	_	-		_			-	T
(SM)	15 -		SS	13	-	+	-	47	0	0		-	+		+				+	+	+
16.95	16-		wo ss	14	-			54													
END OF BORING												_	+	+	+	+	-		+	-	+
	3				9																+
PA = POWER AUGERING HA = HAND AUGERIN	IG		w	0 =	WA	SHC	UT			5	ST =	SH	ELE	BY T	UBE		SS	= SP		SPO	ON
PARTY CHIEF : SONGSAK S DRILLER : BOONSON	s c		G	EOL	OGI	ST:	UB		1	τ	ORA	FTN	AN	1:5	S	T	TYP	IST	1	WS	

BOF	RING LOG						BOR	RING	NO.	:			BH-	1			ELEV	ATIC)N (m.)) :	-			
PROJECT : BCP Hon. Hou	เส้กปีโครเลียม						DEF	TH (m)	•			10.5	5			GWL	(m)			_	-3.	2	
LOCATION : อ.เมือง จ.เชียงรา	าย						cod	ORD.	N	:			•				DAT	E STA	RTED	- 1	_	07/04	/97	
									E	:			•			_	DAT	EFIN	ISHED		_	07/04	/97	
SOIL DESCRIPTI	ION	DEPTH (m)	ORAPHIC LOG	METHOD	SAMPLE NO.	ECOVERY (cm)		SPT- (b)	-N VA lows/	LUE ît.)			N M C	ATUE OISTI ONTE (%)	DRE DRE ENT			¢	Su 'sq.m.)			T T W (V	DTAL JNIT EIGHI (cu.m.)	r >
						*	1	0 2	20 3	0 4	0	2	0	10 e	50 8	0	1	2	3	4		1.6	1.8	2.0
VERY STIFF FINE SANDY CLAY	Y, BROWN		ŀ.	PA																				
(CL/SC)	1.00	1.0	ĮĮ	SS	1	25	~	2	15		-	2	-	+	+	-			+	-+-	+	+	+-	+
	38/N	20		SS	3	28	J	4				2												
(CL/SC)		2.0		SS	4	35	0	6					9	t						1	T	T	1	
	3.00	3.0		PA			1						Щ_	-				_	_	_	+	_	-	-
STIFF FINE SANDY CLAY, BRO	WN			SS	5	35	Ŷ	8					9	1										
(CL/SC)	4.00	4.0	 	wo				-	-	-			-	+		-		-	-+	+	+	+-		+
MEDIUM DENSE SILTY SAND, I	LI-BROWN AND	50		CS	6	32		6	13			6												
(SM)		5.0						1	t	-		T	1		-				1		T			1
	6.00	6.0		wo					1		_			_				_	-		_		_	-
				SS	7	30			1	230		9												
DENSE TO VERY DENSE SILTY	SAND,	7.0		wo					-			+	-	+	+	-		-	+	+	+	-	+	+
YELLOWISH BROWN		80		SS	8	20				50	19.	50												
(314)	8.50	0.0							1		-	\square		1					-		T	1	1	
VERY DENSE SANDY GRAVEL,	BROWN	9.0		wo															_		_			
(GP-GM)	9.00			SS	9	20				50	/8" (19												
VERY DENSE SILTY SAND, LI-E	BROWN	10.0	4	wo			-	-		-	-		-	-	-	-		-	-+	-+	+	+	+-	+-
(SM)	10.55			SS	10	5				50	1/2" (
LID OF BORING												Γ			1									
													-	_				_	_	-	_		_	
			4	1															1					
			1	1			-		-		-	-	-	+	+	-		-	-	-	+	+	+	+
		1	10	1		1							1											
			-										-	-				-		-	+	-	-	+
														10										
							-		-				+	-	+-	-		-	-	-	+	-	1	1
									_			_			-	-				_	+	_	-	+
									-	-		-	+	-	-	-		-	-	+	+		-	-
		9			19		e								1									
611	b I _ I U			G							0													
PA = POWER AUGERING H	A = HAND AUGERING	+		wo	= W.	ASHO	TU					ST	= SH	ELBY	TUB	E		-	SS =	SPLIT	SPO	ON		
ARTY CHIEF SURADECH CH MA	ADE BY : PACHAREE	C.		GEO	DLOG	IST :	NAT	TAPI	HON	_	-	FIL	E :	HU.	AYSA	K1		_	DISK	: 1	0 CH	LAING	KAI	

		3	a.)	ON (n	VAT	ELE				BH-2	1		÷ .	NO.	RING	BO						G	G LOC	ORING	B	-
-3.40		1			L (m)	GW		_		0.65			•	m)	PTH (DE							ครเลขม	ทวยสถาบเครเ	BCP Hon.	JECT :
06/04/97			D	ARTE	ESI	DAT							• .	N	ORD.	co								ขงราย	0.1104 0.19	ATION :
06/04/97	00	3	D	AISHE	E FI	DAT				•			:	E												
TOTAL UNIT				Su	,			AL RE	TUR	MO			LUE	N VA	SPT		ERY (cm.)	LE NO.	THOD	HIC LOG	H(m)			INTION	SOT DESCE	
(l/cum)	1.0	4		2 3	1		0	0 8	(%)) 6() 40	20	,	0 4	0 3	0 3	1	RECOV	SAMP	ME	GRAPI	DEPT			IF HON	SOIL DESCR	
																			PA			r I	BROWN	E SAND, BRO	SE SILTY FU	DIUM DEN
-+		+	-		-			-	-	-	S	-	25	13	0	_	20	1	22	111	1.0	1.00			(SM/ML)	
											II			13	1.	6	30	3	22			. п		BROWN	ANDY CLAY,	F FINE SA
		+	-		-				-		-2	-		-	6	J	30	4	SS		2.0	1.50	13		(CL)	
																		-	PA		30		N	Y, BROWN	E SANDY CLA	DIUM FINE
		\vdash	-		-		-		-	5	-			-	7	q	25	5	SS		3.0				(CL,CL/SC)	
											1				N				-		40	3.80				
		T																	wo				DIUM	E TO MEDIU	SE SILTY FU	DIUM DEN
		i_									ę		28	2			35	6	SS		5.0		EY	ND LI-GREY	SH BROWN A	D, REDDIS
											1											5.50			(SM)	
											1								wo		6.0					
											9	5.0	50/				20	7	SS							
		1			_							_									7.0					
											11								wo							
	\square	+				-					9	6' C	51/	-			15	8	SS		8.0		N	LI-BROWN	SILTY SAND,	Y DENSE
																		-0							(SM)	
\rightarrow		+	-		-		-		-	-		-						-	wu		9.0					
											YI	4° C	50				10	9	SS							
-+-+-		+	-	-	-					-	+				-	-			wo		10.0					
											1		52				10	10	SS			10.65	1	Name of Street, or other	and the second second second second second second second second second second second second second second second	
		+	-	-			-		-	-	-		1			-									ING	OF BORL
																2			19	2						
		\top	-		-					-		-		-	1			219								
																		10		_						
		T																		2						
																				27						
		-					-											1	2	-		-				
		+-	-	-	-		-	-	-		-	-		-		-										
		+	-	-	•	-				-	-	_		-	-											
																-										
	+	+	-	\vdash				-		-	-	-		-	-	-										
		1																				_				
		1										-		-								9				
								4			1		14			D		Q				a 1				
NON	POON	LTT SJ	= SPJ	SS =			E	TUBE	LBY	SHE	ST=					UT	SHO	= W/	wo			ERING	ND AUGER	HA = HAND	GERING	OWER AUC
LAING RAI	CHAIN	10 0	K :	DIS	0		K2	YSAN	HUA	:	FILE			ION	TAP	NAT	IST :	LOG	GEO		2	AREE C	PACHA	MADE BY : P.	SURADECH CH	CHIEF S
																		Ĩ				0	0 <	0.0	19-	0
	CI	10 C	K SPI	SS = DIS	6	1	E K2	TUBI	HUA	SHE	ST =		1	ION	TAPI	UT	IST :	= WA	WO GEC	1		ERING	ND AUGES	HA = HAND . MADE BY : P.	GERING SURADECH CH	POWER AUC

GROUND WATER OBSERVATION.	WAS BOBING LOS	BORING NO. 1
DATE TIME EL. of HQLE EL.of WATER	W.A.C. BORING EOG	SURFACE ELV.
	CHAINGRAI PRACHANUKROK	DATE START
24 HR.AFTER BORING. 1.75 M.	CHAINGRAI	DATE FINISH 31-OCT-43
SOILS DESCRIPTION	UNDER STANDARD PENETRATION PENETRATION VIEW MOISTURE CONTENT. BLOWS / FT. % 0.00 30 60 20 40 60 80	AOne half Unconfined- Compressive Strength. ■ peak□remolded INSITU VANE SHEAR. ® One half Pocket - Penetrometer Rdg. ■ SENSITIVITY KSC. 1 2 3 1 2
Medium dark brown silty very fine sand. SM Medium to stiff dark brown 360 and reddish brown silty clay, trace of very fine sand. ML-OL Soft dark grey silty clay, trace of very fine sand. ML-OL Very stiff grey silty clay, trace of very fine sand. ML-OL Medium to very dense light grey silty very fine to fine Sand, cocasional coarse sand and pea gravel. SM & SP-SM Dense to very dense yellowish brown silty very fine sand, occasional pea gravel. SM 1935 END OF BORING.	35 55 1 4 55 1 55 5 4 55 5 4 4 55 4 4 29 5 4 29 38 55 5 4 4 29 38 55 5 4 7 10 12 55 5 4 7 10 12 55 5 4 7 10 12 55 5 4 7 55 10 7 55 11 7 55 12 7 55 13 7 55 14 7 55 16 7 55 16 7 55 16 7 55 16 7 55 16 7 55 16 7 55 16 8	

GROUND WATER OBSERVATION.	W.A.C. BORING LOG BORING NO. 2 SURFACE ELV.																			
24 HR AFTER BORING 1.75 M	LOCATION. CHAINGRAI PRACHANUROK DATE START CHAINGRAI DATE FINISH 31-OCT-	-43																		
SOILS DESCRIPTION	U STANDARD- PENETRATION - LIQUID LIMIT. Compressive Strength. Compressive Strength. INSITU VANE SHEAR. U U V - LIQUID LIMIT. - U U V - - - U U V - - - U U V - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - - - U U - - <td>DTAL- NSITY. d.?/w ●</td>	DTAL- NSITY. d.?/w ●																		
Soft to stiff light grey and brown silty clay, trace of very fine sand. ML-OL Loose dark grey silty very fine sand. SM & SP Medium to dense light grey very fine to fine sand, correctional correct sand,	SS 1 4 FS5 2 F3 FS5 3 F3 FS5 4 F3 FS5 5 F3 FS5 5 F3 FS5 7 F3 FS5 7 F3 FS5 8 F3 FS5 9 1000 FS5 10 1000 FS5 11 F3 FS5 12 F3																			
1995 END OF BORING.	1 1 1 1 1 1 1 1 <td>•0 •0 •0</td>	•0 •0 •0																		
													-						•	-
---	---	------------	--------------	--	----------	----------	--------------------------	---------------	--------------------	-----------	------------	------------	----------	------------------------------	---------------------------------	---------------------------------	-----------------	----------------------	--------------------------	----------------------------
ROJECT อาการศูนย์ ocation : อำเภอเมือ	RING LOG โทรศัพท์ค่ายเม็งราย ง จังหวัดเชียงราย					-	BORING DEPTH COORD	6 NO. (m.)	:; :;	8H 12.	- I 35			GROU WATE DATE DATE	IND E R LEV STAR FINIS	LEV () /EL (# ITED HED	m): 1): :	NOT 15/	12/ 12/	UNI 94
SOIL DESCR	IPTION	DEPTH (m.)	GRAPHIC' LOG	METHOD	SAMPLING	RECOVERY	(b 10 2	SPT lows	-N /ft) 0 40		PL v (*	vn 1 %)	L_ 80	(U O L X F 1	Su /sq.m JCT VT 2) ∆⊧ □1 3 4	*	T V (t/ 1.6	OTA UNIT EIGH T	IL 1 1 1.) 2.0
STIFF CLAY WITH FIN SAND, LI-BROWN, RED (CL) MEDIUM CLAY, LI-BRO (CL) LOOSE TO MEDIUM DEN TO MEDIUM SANDY SIL REDDISH BROWN (ML) DENSE SILTY FINE TO BROWN (SM) END OF BORING	E TO MEDIUM DISH BROWN 2.00 WN 3.00 SE SILT/FINE T, LI-BROWN, 10.50 MEDIUM SAND, 12.35	1 - 2 - 3		PA SSASSARSS PA SS ₩ SS ₩ SS ₩ SS ₩ SS ₩ SS ₩ SS ₩				17 17	3											
PA = POWER AUGERING	HA = HAND AUGERIN	G		w	0=	WA		JT			ST=	SHE	LBY	TUB	E	SS	= SP		SPO	ON

BORING LOC ROJECTอาคาร บก. ศูนย์การฝึก นศท. DCATION อ่า เภอ เมือง จังหวัด เชียงราย	ন . বা এ	nu. 1	ชียง	ราย	BC DE CC	RING PTH ORD.	NO (m.) _	8	3.15	1			GROU OBSE DATE DATE	IND E RVED STAI	UEV. (m WL (m RTED))4 24/ 25/	.50 7/9	93
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	RECUVERT	SF (blo	PT - N ws/ft D 30	40		PL 1	Wn (%) (%)	LL 	01 XF 1	Su(1/ JCT SVT 2 3	m ² } △ PP □ TV 3 4	(¥1 1/1	3
VERY LOOSE SILTY FINE TO MEDIUM SAND, DARK GREY (SM) 3.50 MEDIUM DENSE SILTY FINE TO MEDIUM SAND, BROWN (SM) 4.50 LOOSE SILTY FINE TO MEDIUM SAND, GREY (SM) 7.00 MEDIUM DENSE SILTY FINE TO MEDIUM SAND, GREY (SM)	1		PA SS PA SS PA SS PA SS PA SS VO SS VO SS VO SS VO SS VO SS VO SS VO SS SS SS SS SS SS PA SS SS PA SS SS PA SS SS PA SS SS SS SS SS SS SS SS SS SS SS SS SS	1 2 3 4 5 6 7 8 9	3 3 9													
13.00 VERY DENSE SILTY GRAVEL, GREY (GP-GM) 13.15 END OF BORING			wo ss wo ss				9 21 9 21 50	5 76										

:	PA	NYA	C	ONS	ULTANTS CO	., LTD.	93	
BORING LOO PROJECT คลังสินค้าปู่นชิเมนต์นครหลา LOCATION อำเภอเมือง จังหวัดเชียง	3 เง เราย				BORING NO. BH DEPTH (m.) 15 COORD.	1- 1 .08	GROUND ELEV.(m. OBSERVED WL (m. DATE STARTED DATE FINISHED)) -0.60 31/7/93 31/7/93
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC	METHOD	SAMPLING	SPT - N (blows/ft) 10 20 30 40	PL W _n LL → → → (%) 20 40 60 80	Su(t/m ²) ⊙UCT △ PP XFVT ⊡TV 1 2 3 4	ĩ₁ (t∕m ³) 1,6 1,8 2,0
MEDIUM CLAY, YELLOWISH BROWN (CL) 1.70	1.		PA SS PA		9 5	9		
STIFF CLAY, BROWN, REDDISH BROWN AND GREY (CL) 3.80	3 .		SS PA SS	2	011 Q14			
MEDIUM DENSE CLAYEY GRAVEL, GREY AND REDDISH BROWN (GC) 4.75	4 -		SS PA SS	4	©19 © 15			
VERY STIFF CLAY, REDDISH BROWN AND GREY	6 · 7 ·	•	PA SS PA	6	026			
8.50	8 · 9 ·		SS PA SS	7 8	929 50 C			
	10 - 11 -		PA	9	50 0			
HARD CLAY, DARK GREY	12 - 13 -		PA	10	5076" (
15.08	14	•	PA	11	50/5" 0			
END OF BORING	9					004		
				3				

K. ENGI	NEER	ING	С	ONSULTAN	ts co., lt	D.	
BORING LOO PROJECT โรงประกอบอาหารและชักรีค LOCATION SW. จทบ. ชร. อ.เมือง จ.	G เชียงร	าย		BORING NO. BH- DEPTH (m.) 19.9 COORD.	- 1 95	GROUND ELEV.(m. OBSERVED WL (m. DATE STARTED 2 DATE FINISHED 2) 24/10/94 4/10/94
SOIL DESCRIPTION	DEPTH (m.) GRAPHIC	LOG METHOD	RECOVERY	SPT - N (blows/ft) 10 20 30 40	PL Wn LL (%) 20 40 60 80	Su(t/m ²) ⊙UCT △ PP XFVT ⊡TV 1 2 3 4	¥t (t∕m ³) 1,61,82,0
STIFF TO VERY STIFF CLAY, WITH SAND, BROWN AND GREY (CL)	1	A 53 53 53 5 5	1 2 3 4	9 910 011 016			• •
4.50 HARD SANDY CLAY, BROWN	5	SS WO	5	0 36			
STIFF TO VERY STIFF CLAY, TRACE SAND, LI-GREY AND BROWN (CL)	6	SS WO SS WO	6 7 8 9 10	¢ 19 ¢ 19 ¢ 12 ¢ 12 ¢ 12 ¢ 18 ¢ 18 ¢ 19 ¢ 19			·
17.50 HARD CLAY, WITH SAND, LI- GREY AND BROWN	17	SS ₩0 SS	13	φ21 φ30			
(CL) 19.95 END OF BORING	19 - 11	SS	15	031			

K. ENGI	NEE	RI	NG	С	ONSU	JL	ΓA	NI	rs	C	0.,	L	۲D.								
BORING LOO PROJECT ตึกแถวขึ้นประทวน 8 ครอบค LOCATION <u>รพ. จทบ. ชร. อ.เมือง</u>	3 รัว จ.เชื	ยงร	าย		BORIN DEPTH COORI	G N(I (m.) .)!	BH-	-2				GR OB DA DA	OUN SER TE S TE F	D EL VED STAR	.EV. WL TED	(m.) (m.) 2	5/1	c /9	14 4	-
SOIL DESCRIPTION	DEPTH (m.)	GRAPHIC	METHOD	RECOVERY	S (Ы	PT-	- N (ft)	0	PI + 2	0 4	Wn (%) 0 60	11 	0	Su DUC KFV	(1/1 T 4 T 1 2 3	n ²) ∆ Pf ⊡ T\ 4		(1,6	¥, 1/1	³)	0
STIFF TO VERY STIFF SANDY CLAY, BROWN, GREY (CL,CH) (CL,CH) HARD CLAY, TRACE SAND, BROWN, GREY (CH) 10.50 VERY STIFF CLAY, WITH SAND, BROWN AND GREY (CL) 15.00 HARD CLAY, GREY AND BROWN (CL) 15.45 END OF BORING			A SSA SSA SS PA SS ₩ SS ₩ SS ₩ SS ₩ SS ₩	1 2 3 4 5 6 7 8 9 11 12			55 8	36													
9	-					-		0	_		-	+	+	-			_		+	-	

	~				BORING NO. B	1-1	GROUND ELEV. (m	.i
BORING LO	G				DEPTH (m.) 23	.45	OBSERVED WL (m	1-1.20
PROJECT ปรับปรุง รพ. จทบ. ซ.ร. ปี	36				COORD		DATE STARTED	11/11/93
LOCATION <u>อำเภอเมือง จังหวัดเชียงร</u> า	ย						DATE FINISHED	11/11/93
SOIL DESCRIPTION	DEPTH (m)	GRAPHIC	METHOD	SAMPLING RECOVERY	SPT - N (blows/ft) 10 20 30 40	PL W _n LL ⊢	Su(t/m ²) ⊙UCT △PP XFVT ⊡TV 1 2 3 4	τ ₁ (t/m ³) 1.6 1.8 2.0
LATERITIC COMPACTED SOIL (FILL) 1.00		10.0	PA					
VERY SOFT TO SOFT CLAY,	1'	∇	SS	1	20	4		
GREY (CL)	2 -	1	SS	2	Ø4		┠╍┼╍┝╌┼╍	+++
2.50	1_	ľπf	PA	Ī				
	13.		SS	4	0 15	8		9
	4 -		PA				┠┼┥┽	┨╌┤╶┤╢╌
			SS	5	021	6		6
	1.		PA					
	6 :		SS	6	913	1-9-3		0
•	7 -		wo				+ + + + + + + + + + + + + + + + + + +	┟┼┼┽┥
STIFF TO VERY STIFF CLAY.		1111	SS	7	012			
BROWN, YELLOWISH BROWN,			wo					
(CL,CH)	9-		SS	8			┢╌┼╌┼╌┼╌┼	┢┽┽┽
	10-		wo					
			55	9				
	11 -			-				++++
	12 -		wo					
	12		55	10	Q13	¢		
	13 -		wo				┝╌┼╌┼╌┼╶┼	
	14 -		SS	11	0 16	0		+ + + + + + + + + + + + + + + + + + +
			Y/O					
	15 -		SS	12	Q 18			
	16 -		wo					
5500			SS	13	Q23			
	17		wo					
	18 -		SS	14				++++
19.00	10		wo		23	I		
HARD CLAY, LI-GREY			55	15	50/5"			
(CL)	20-		WO	T				┨╌┧╌┧╌┧╴
			19230					

K. ENGI	NEI	ERI	NG	C	DNSU	IL.	ΓΑΝ	TS	C	:O.,	L	ГD.							
BORING LO PROJECT ปวับปรุง รพ. จทบ. ช.ร. ปี LOCATION อำเภอ เมือง จังหวัดเชียงรา	G зе				BORING DEPTH COORD	6 NC (m.) <u> </u>	H-1 3.4	5			GRI OB: DAT DAT	DUNC SERV TE S TE F	ED EL	.EV.(WL (TED HED	m.) _ m.) _ []	- 1.20	0 /9 /9	3
SOIL DESCRIPTION	DEPTH	GRAPHIC	METHOD	SAMPLING BECOVERY	SI (blo 10 2	PT -	- N (ft) (0 40		PL 	Wn (%) 0 60			Su () UC (FV1 1 2	1/m T 2 T 1 2 3	n ²) △ PP □ TV 4		(† 1,6	¥ ₁ / m ³ 1.8 2) 2.0
HARD CLAY, DARK GREY (CL)	21		wo ss	16			0/10'	·• •	2						+		+		
DENSE CLAYEY FINE TO MEDIUM SAND, GREY (SC/CL) 23.45	22		wo	17			47	0	H H			+					+		
END OF BORING												+							
												+				+			
												1							
															-	+	+		╞
	6													-		-			-
	No.		1					+									+	+	
											+	-			_	-	-		
								$\frac{1}{1}$						-				+-	
สถาเ	19										1	-				-		-	-
ลฬาลงก	5		ð	9									20.6				+	+-	Ţ
9	l d	b	10					1			-	+		_		+	+	+	+

NEE	ERI	NG	С	ONSUL	FAN	rs co., l7	D.	
G ชียง	ราย			BORING N DEPTH (m COORD.) BH-) 6.2	-1 5	GROUND ELEV.(m. OBSERVED WL (m. DATE STARTED 2 DATE FINISHED 2) NOT FOUND 5/10/94 5/10/94
DEPTH (m.)	GRAPHIC LOG	METHOD	SAMPLING	SPT (blows)	- N (11) 040	PL W _n LL 	Su(t/m ²) ⊙UCT △ PP X FVT ⊡ TV 1 2 3 4	ĩ, (t/m ³) 1,6 1,8 2,0
1 -		PA						
		SS PA	1		o 33	9		<u></u>
2 -		SS	3	d	3			
3 -		PA						
		SS	4		637	\$		
4 -		PA			\mathbb{N}			
5 -		SS	5		55	0		
		PA						
6		SS	6		50/10 0	0		
1	1							
-								
9-	q			24	2			
-		D G			0		2	
2	6				1			
	NEI 1 1 2 4 - - - - - - - - - - - - -	NEERI 2 2 2 3 4 5 6 1 1 1 1 1 1 1 1 1 1 1 1 1	NEERING DUISTU NEERING OUTON NEERING OUTON NEERING OUTON NEERING OUTON NEERING	NEERING Current 1 004130 1 004130 1 004130 2 004130 3 004130 3 004130 1 004130 1 004130 2 004130 3 004130 3 004130 1 004130 1 004130 2 004130 3 004130 3 004130 3 004130 3 004130 1 004130 2 004130 3 004130 4 004130 5 004130 6 0055 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 0 004130 <td>NEERING CONSULT BORING NG BORING NG DEPTH (m. COORD. 1 SPT 1 SS 1 1 SS 1 1 SS 3 2 SS 3 PA I 3 SS 4 4 PA 5 PA 6 SS 6</td> <td>NEERING CONSULTANT BORING NO. BH- DEPTH (m.) 6.2 1 0 9 1 1 0 9 1 0 2 1 0 9 1 0 2 0 3 2 9 9 1 0 20 30 40 1 5 1 0 20 30 40 1 5 1 0 20 30 40 1 5 1 0 33 3 3 3 2 9 3 3 3 3 3 3 3 3 5 1 0 3 <t< td=""><td>NEERING CONSULTANTS CO., LT BORING NO. BH-1 DEPTH (m.) 6.25 COORD. COORD. Example of the second</td><td>NEERING CONSULTANTS CO., LTD. BORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) BUSS 18 DORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) DORE STREE 2 DATE FINISHED 2 Table Stree</td></t<></td>	NEERING CONSULT BORING NG BORING NG DEPTH (m. COORD. 1 SPT 1 SS 1 1 SS 1 1 SS 3 2 SS 3 PA I 3 SS 4 4 PA 5 PA 6 SS 6	NEERING CONSULTANT BORING NO. BH- DEPTH (m.) 6.2 1 0 9 1 1 0 9 1 0 2 1 0 9 1 0 2 0 3 2 9 9 1 0 20 30 40 1 5 1 0 20 30 40 1 5 1 0 20 30 40 1 5 1 0 33 3 3 3 2 9 3 3 3 3 3 3 3 3 5 1 0 3 <t< td=""><td>NEERING CONSULTANTS CO., LT BORING NO. BH-1 DEPTH (m.) 6.25 COORD. COORD. Example of the second</td><td>NEERING CONSULTANTS CO., LTD. BORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) BUSS 18 DORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) DORE STREE 2 DATE FINISHED 2 Table Stree</td></t<>	NEERING CONSULTANTS CO., LT BORING NO. BH-1 DEPTH (m.) 6.25 COORD. COORD. Example of the second	NEERING CONSULTANTS CO., LTD. BORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) GROUND ELEV.(m.) GETTH (m.) BUSS 18 DORING NO BH-1 GETTH (m.) GROUND ELEV.(m.) GETTH (m.) DORE STREE 2 DATE FINISHED 2 Table Stree

GRO	OUND W	ATER OBSEF	VATIO	N.			V	NAC	B	DRI	NG	10	G			BO	RING	NO.	1		
DATE	TIME	EL. of HOLE	EL.of V	NATER			_						<u> </u>			SUF	RFACE	ELV.			
	L					CA	тю	N. DE	ENH4	A SC	QUA	REE	RAN	1CH		DAT	TE STA	ART			
24 HF	R.AFTER	BORING.	0.62	М.						CHA	AING	RAI				DAT	E FIN	VISH			_
					ш	E NO		STA		D-	•	LIQUI		т.			e half L pressi akD	Jncon ve Stre remole	fined- ingth. ded	TO	TAL-
so	DILS [DESCRIP	TION.		- PROFIL	IPLE TYP	ТН М.	PENE	TRATI	ON.	×	PLASI	NATU	RAL	CNIT	© One Per	e half hetrom	Pocke neter F	t - Idg.	۶d	,γw
					soll	SAN	DEP	BLOW		ET		wi0151	When the		ENT.	• SE	NSIII	SC		T	-0 M 3
							0.00	30	60		2	20 4	0 6	50 8	0		1 2	2 3	3	1	2
Sti	F ff light (ill soil grey silty мн-он)	clay.	_0.50 _1.50		55.01 55.02 55.03 55.04 55.05		9 17 				× 0 * *	•	•							•••
Stit and yu -sional ((ff to h ellowist very f CH, MH-	ard light g h brown sil ne sand. он,сс & м	rey, re ty clay Le OL 1	ddish ,, occo- 15,45		-55.06 -55.07 -55.09 -55.09 -55.10 -55.11 -55.12 -55.13	5.00- 10.00- 1500-	-25	37			•••••••••••••••••••••••••••••••••••••••									•••
	END	OF BORING									*										
		สเ		9			20.00	N	2	9					7						

GRO	UND W	ATER OBSE	RVATIO	N.			v	V.A.	С. В	ORI	NG	LOG	1			BOF	RING	NO.	2		
DATE	TIME	EL. of HOL	E EL.of	WATER	L_							-			-	SUF	FACE	ELV.			
			0.49		LC	CA	TIO	N. C	DEN	AL	SQUA	RE I	BRAN	NCH	ŀ	DAT	E STA				
24 HR	AFTER	BORING.	0.08	Μ.			-		_	CH	AINC	RAI				DAT	EFI	NISH		_	
						NO	$\left[\right]$	Γ	•		-• L	.IQUID	LIMIT			∆One Comp ∎ pea	half L pressi	Uncont ve Stre	fined- ngth. led	тот	TAL-
sc	ILS D	DESCRIF	TION	1	OFILE	TYPE	Υ	ST. PENI	ANDA ETRA	RD- TION.	0- F	PLASTI	CLIM	IT.		© One	half	NE SH Pocke	t -	JEN	5117.
					OIL PR	AMPLE	JEPTH				×ĸ	NOISTL	JRE CO		IT.	• SE	NSITI	VITY	ເບິ່ງ.	γd, ●	γw -0
					S	S		BLC	ws/	FT.	ļ	0	%	-	-+		K	SC.	-	T/N	M. ³
	E)	Il soil		0.55		\vdash	0.00	3		00	2	40	, 60	80	-		H	ΓÌ		+	1
Stit	ff brow	vnish dark	(grey	sility	1	SCOM															
clay.	1	(UL)		1.50	ten	-5502		6				dx 1				-0					-
		(10) 200 m	10000			-\$5.03		15				4	-	-+		-0	-			+	2
Stif yellowi	sh bro	ery stiff (wn siltv i	greyish tlay, orr	and asjonal	D	-55.05		24				XO	-	•	_		4			F	•
very fi	ne san	d.	11		1			1				1					1				
	(MH-)	տ,սнձ(1	-55.06	500	15		-		ox	•	_		ø				-+•	• •
					1		5.00	1				1									
Ct:4	f dock	OFPV cit		6.00_	A	-55.07		6-				0	>		_	-				-•	
of decu	ayed w	vood.	, ciuy,		1								1								
	(-7.50-	A	-55.08		15				q	•		_			+			0
					1							1									
						-55.09		150				0					-			4	••
	¢.						10.00		1			1					1				
	1	0.000	Inviet	and	1	-55.10			46			×.				<u> </u>		b		-+•	••
Stit reddish	n light	n silty clay	y, occas	ional	1	124	6	24	V								11				
very f	nine san	ND.				-55.11		-220		1-1		0					K-			+	•
	(CL	ant-UL)							1							1					
	4				10.10	-SS.12	2	-	51			0		-		\vdash	A	1		-+	•
						-55.13	15,00-		54			to				L		-		\square	•
														3							
						-55.14				64-		*		-		-		-	\vdash	\vdash	•
				1								$ \rangle$								11	
				18,45	K	-55.15				62-		oxe	-+				+	-b-	<u>+</u>	++	•
	END	OF BORIN	G																		
							20,00	1													
						4	4					1				1					
				19 *	9		n s	1/I	121	19	14	5	11	η4							
L		_	-	-			1		1		1	1						1	1	1	1

GRO	OUND W	ATER OBSEF	VATION	۹.				MAC	BOD	NG LOG	BOR	NG NO.	1
DATE	TIME	EL. of HOLE	EL.of V	VATER				W.A.C.	BUN	NG LUG	SURF	ACE ELV.	
							TIO	AMP	HOE MU	ANG	DATE	START	
24 HF	R.AFTER	BORING. 1	.26	Μ.				CHI	ANGRAI		DATE	FINISH	
so		DESCRIP	TION.		FILE	TYPE NO		STAN PENETI	DARD- RATION.	LIQUID LIMIT. O- PLASTIC LIMIT.	∆One h Compro- ■ peak INSITU	nalf Unconfined essive Strength remolded VANE SHEAR.	TOTAL- DENSITY.
					OIL PRC	AMPLE	EPTH. N			X NATURAL MOISTURE CONTEN	Pene IT. • SEN	trometer Rdg.	γd,γw ●—⊖
					S	ŝ	0	BLOW	S/FT.	%	0, \ /H	(sc ⊡,∎ / TSM	T/M. ³
 		sil and l					0.00	30	60	20 40 60 80	11	2 3	1.2
brown s gravel	Medium silty c and ve CL	to hard gn clay, occas ary fine s & ML-OL	cey and sional sand.	.0.35 d pea	No. Marina	-SS.1 -CS.2 -SS.3 -CS.4 -SS5 -SS5	- 5.00	4 12 20	9		80 80 80 80 80 80 80 80 80 80 80 80 80 8		• • • •
V and gre very fi rock.	Very st ey silt ine sar	iff to har y clay, t d and deco ML-OL	rd brow race o unposed	7.60- vn f 1 10.50-		-557 -558 -559 -5510	-10,00	19	125	• • •	∆¢ ∆¢		• • •
brownisi of very rock.	ard br sh yello y fine	own, light w silty c sand and ML-OL	lay, t decompo	and trace osed		-55.11 -55.12		- 40	62	*		8	••
E	END O	F BORING	1	5.45	The second second	-55.13	-15,00		60	ו		ø	•->
		ล์ถ	5			and a second	-20.00						

	GRC	UND W	ATER OBSER	VATIO	Ν.			V	V.A.	С. В	ORI	NG	LO	G			BOF	RING	NO	(1)	
	DATE	TIME	EL. of HOLE	EL.of V	NATER												SUF	FACE	ELV.	-0	.45 n	n.
			l .	I		LC	CA	TIOI	N.	EDU	ю. с	CCU	P. C	XOLLI XOLLI	EGE		DAT	E STA	ART			
	24 HF	RAFTER	BORING. 2.	10	М.							TAH	NGRA	1			DAT	E FIN	NISH		_	
							.			•		-	IQUI	DLIM	IT.		AOne	half	Incont	ined-		
					1		NON			-							 pea 	kD	remole	ded	TOT	AL-
	sc		ESCRIP	ION		FILE	ΥPE		ST PEN	ANDA ETRA	RD-	0-1	PLAST	LI DI	MIT.		INSIT	UVA	NE SH	EAR.	DENS	5011.
						BROI	LE T	≥ 						NATI	IRAL		Pen	etrom	eter F	ldg.	Yd.	Yw
						OIL	AMP	EPTI				×	NOIST	TURE	CONT	ENT.	• SE	NSITI	VITY		•	-
						S	Ś	۵	BLC	ows /	FT.			%/0				KS	SC.		T/M	1. ³
		Concret	e pavement		0.05			0.00	3	80 6	50	- 2	20 4	10	60 8	0	1				-1-	2
		To	p soil		.20																	
	Soft	dark q	rey silty (clay.			-SS.1		4				oj -	A. 1.			ø					
					-2.00	4			10				1				Ì					
							-55.2		~				i.	Ī								μ
							-SS.3		(96.00)	55			xo-						0			•
										/			1					1				1
							50 1		22	ł			1					1				
						1	-25.4	5.00	- 1			× 10 1000	1				1.000	1			•	-
	Stif	ftov	erv stiff	ight o	mev.	1							1					;				
	yellowi	sh bro	wn and brow	n sil	ty		-SS.S		15				xo	•							•	ю
	clay,	occasio	onal decom <u>r</u>	xosed :	rock.	1.11.1							1					Í,				
		CL.	CH & ML-	DL.			-SS.6			33			× -						0			
								1862		1			1						1			
					-9.00	11		125	577	1		1										
	Hard	grey	and yellowi	sh bro	Jwn		-SS.7		Rin	1	100	XI	-9						8	***		۲
	Sincy	citty.	ML-OL	_	10.20	hi		10.00	+	14												
						80	-SS. 8				150	-*				+ 11						
						PP						i										
	brown	and bro	grey, yei own cemente	d silt	y	60			23		150	i										
3	very f	ine sar	nd, occasio	nal de	ecom-	e 9	55.9				150	Ť										•
	posed	rock.				E P E						11										
			SM			99	-SS.10				120	*	-							-		•0
					15 45		55 11	15.00-			150			1								
		ENT			13.45		100.11				1.20											T
		END	OF BORL	NG.	2		9															
					9	9		n 9	Л	2			1		15							
			b				0															

GROUND WATER OBSERVATION.			W.A.	C. BORI	NG LOG	BORING NO.	2)
DATE TIME EL. of HOLE EL.of WATER				-	SURFACE ELV0.40 m.		
	LOC	CATIC	DN.	EDUC.	OCCUP. COLLEGE	DATE START	
24 HR.AFTER BORING. 2.20 M.	CHAINGRAI					DATE FINISH	
SOILS DESCRIPTION				•	- LIQUID LIMIT.	∆One half Unconfined- Compressive Strength.	TOTAL
	<u> </u>	DE N	ST	ANDARD-		peakgremolded INSITU VANE SHEAR.	DENSITY.
	ROFI	M E	PEN	ETRATION.		One half Pocket - Penetrometer Rdg.	
	OILF	EPTH			X MOISTURE CONTENT	• SENSITIVITY	•-○
	S	s o	BLC	WS / FT.	%	KSC.	T/M. ³
Concrete pavement 10.05		0.0	0 3	0 60	20 40 60 80		1 2
Top soil 0,25							
Soft grey and yellowish brown	-S	S.1	R4		o x •	2	••0
silty clay. ML-OL					1		
Very stiff light grey, yellowish	-S	S.2	190		0	(e	•0
brown and brown silty clay,	L.	5.3	-			4	•0
occasional decomposed rock.		5.5					
ML-OL & CH				32	í		
	S	5.4 5.0	of		Xo•	· · · · · · · · · · · · · · · · · · ·	•
6.00	s	S.5	19		× -•	8	••
Very stiff to hard brown white					1		
and brown silty clay.		-		57	j l		
CH & ML-OL	111	5.0		3/9			•0
	11-S	S.7		644		·	•••••
		10.0	20-				
	2	5.0		82	Į.		
		5.0			7		
Hard grey, white, yellowish		219	1.21				
brown and brown compacted silty clay, occasional decomposed rock.	115	S.9	1	97.	* •	···· · ··· •	
ML-OL	11						
	-S	S.10		> 150			
		15.0	π				
15.45	S.	S.11		>150 <	· · · · * · • • • • • • • • • • • • • • • • • •		•
END OF BORING.							
ี สภาข		T	P/I	019	いちわわち		
616		b c					
			-				

VITAE

Pichai Pattararattanakul was born in Trang, a southern province of Thailand, on April 3, 1974. He attended Prince of Songkla University, one of the most prestigious universities in southern Thailand, where he obtained a bachelor's degree in civil engineering in 1995. He received a master's degree from Chulalongkorn University with a major in structural engineering in 1998. His master's thesis title was "Effects of Different Shear Reinforcement Detailings on Behavior of Reinforced Concrete Shear Walls Subjected to Cyclic Loadings" and was published in the ACI Structural Journal in 2001. After graduation, he worked as a research assistant at the earthquake engineering and vibration research laboratory of Chulalongkorn University from 1998 to 1999, conducting the full-scale behavior of reinforced concrete columns subjected to cyclic loadings.

Pichai began his engineering career when he studied in the last year of master's degree with an engineering consulting firm. In 1999, he has started his Ph.D. in the area of geotechnical engineering at Chulalongkorn University, together with working as part time structural engineer with many consulting firms in Bangkok. He has been involved in design of various types of structure both concrete and steel structures, including low- and high-rise buildings, industrial buildings, water treatment tanks, machine foundations, and etc.

สถาบนวิทยบริการ จุฬาลงกรณ์มหาวิทยาลัย